

Special Issue Reprint

Modeling Energy-Environment-Economy Interrelations

Edited by George Halkos

www.mdpi.com/journal/energies



Modeling Energy– Environment–Economy Interrelations

Modeling Energy– Environment–Economy Interrelations

Editor

George Halkos

MDPI • Basel • Beijing • Wuhan • Barcelona • Belgrade • Manchester • Tokyo • Cluj • Tianjin



Editor George Halkos Department of Economics University of Thessaly Volos Greece

Editorial Office MDPI St. Alban-Anlage 66 4052 Basel, Switzerland

This is a reprint of articles from the Special Issue published online in the open access journal *Energies* (ISSN 1996-1073) (available at: https://www.mdpi.com/journal/energies/special_issues/ modelling_energy_environment_economy_interrelations).

For citation purposes, cite each article independently as indicated on the article page online and as indicated below:

LastName, A.A.; LastName, B.B.; LastName, C.C. Article Title. *Journal Name* Year, *Volume Number*, Page Range.

ISBN 978-3-0365-8332-7 (Hbk) ISBN 978-3-0365-8333-4 (PDF)

© 2023 by the authors. Articles in this book are Open Access and distributed under the Creative Commons Attribution (CC BY) license, which allows users to download, copy and build upon published articles, as long as the author and publisher are properly credited, which ensures maximum dissemination and a wider impact of our publications.

The book as a whole is distributed by MDPI under the terms and conditions of the Creative Commons license CC BY-NC-ND.

Contents

About the Editor
Preface to "Modeling Energy–Environment–Economy Interrelations"
George E. Halkos and Eleni-Christina GkampouraExamining the Linkages among Carbon Dioxide Emissions, Electricity Production andEconomic Growth in Different Income LevelsReprinted from: Energies 2021, 14, 1682, doi:10.3390/en14061682
Adedoyin Isola LawalThe Nexus between Economic Growth, Energy Consumption, Agricultural Output, and CO2 inAfrica: Evidence from Frequency Domain EstimatesReprinted from: Energies 2023, 16, 1239, doi:10.3390/en1603123925
Pradyot Ranjan Jena, Shunsuke Managi and Babita Majhi Forecasting the CO2 Emissions at the Global Level: A Multilayer Artificial Neural Network Modelling Reprinted from: Energies 2021, 14, 6336, doi:10.3390/en14196336
Sakar Hasan Hamza and Qingna Li The Dynamics of US Gasoline Demand and Its Prediction: An Extended Dynamic Model Averaging Approach Reprinted from: <i>Energies</i> 2023, 16, 4795, doi:10.3390/en16124795
George Halkos and Eleni-Christina Gkampoura Assessing Fossil Fuels and Renewables' Impact on Energy Poverty Conditions in Europe Reprinted from: <i>Energies</i> 2023 , <i>16</i> , 560, doi:10.3390/en16010560
Adedoyin Isola Lawal Determinants of Renewable Energy Consumption in Africa: Evidence from System GMM Reprinted from: <i>Energies</i> 2023 , <i>16</i> , 2136, doi:10.3390/en16052136
Cong Khai Dinh, Quang Thanh Ngo and Trung Thanh Nguyen Medium- and High-Tech Export and Renewable Energy Consumption: Non-Linear Evidence from the ASEAN Countries Reprinted from: <i>Energies</i> 2021 , <i>14</i> , 4419, doi:10.3390/en14154419
Jamal Mamkhezri, Leonard A. Malczynski and Janie M. Chermak Assessing the Economic and Environmental Impacts of Alternative Renewable Portfolio Standards: Winners and Losers Reprinted from: <i>Energies</i> 2021 , <i>14</i> , 3319, doi:10.3390/en14113319
Yang He and Yun ZhuComprehensive Benefit Analysis of Port Shore Power Based on Carbon TradingReprinted from: Energies 2023, 16, 2755, doi:10.3390/en16062755
Oleksandra Shepel, Jonas Matijošius, Alfredas Rimkus, Olga Orynycz, Karol Tucki and Antoni Świć Combustion, Ecological, and Energetic Indicators for Mixtures of Hydrotreated Vegetable Oil (HVO) with Duck Fat Applied as Fuel in a Compression Ignition Engine Reprinted from: <i>Energies</i> 2022 , <i>15</i> , 7892, doi:10.3390/en15217892

About the Editor

George Halkos

George E. Halkos (BA, MSc, PhD) is a Professor of the Economics of Natural Resources in the Department of Economics at the University of Thessaly. He has worked as a team leader and research fellow in various research and academic institutions. In addition, he has participated in and presented articles at International Conferences and acts as a referee for many scientific journals. His research interests include applied statistics and econometrics, simulations of economic modeling, natural resource and environmental economics, applied micro-economics with emphasis in welfare economics, air pollution, game theory, and mathematical models (non-linear programming).

Preface to "Modeling Energy– Environment–Economy Interrelations"

This Special Issue examines the nexus between economic growth, energy consumption, and the environment. Our industrialized world depends on fossil fuels to meet its energy needs, and since fossil fuels are associated with both economic growth and severe environmental degradation, the associations between air pollutants emissions, energy consumption and economic growth must be examined. Additionally, CO₂ emissions are forecasted at the Global Level, and the determinants of renewable energy consumption are discussed, along with an assessment of the economic and environmental impacts of alternative renewable portfolio standards and the possible winners and losers. There is particular emphasis on assessing fossil fuels and renewables' effect on energy poverty conditions in Europe. The findings of this Special Issue will be invaluable to researchers and policy makers.

George Halkos Editor





Article Examining the Linkages among Carbon Dioxide Emissions, Electricity Production and Economic Growth in Different Income Levels

George E. Halkos * and Eleni-Christina Gkampoura

* Correspondence: halkos@econ.uth.gr

Abstract: Our industrialized world highly depends on fossil fuels to cover its energy needs. Although fossil fuels have been linked with economic growth, their use has also been found to have severe impacts on the environment. The linkages among carbon dioxide emissions, energy consumption and economic growth have been extensively examined in the current literature. The present study focuses on electricity production from fossil fuels, as well as from renewable sources and examines their linkages with CO_2 emissions and economic growth in 119 world countries of different income levels, by assessing Granger causality. In addition, the Environmental Kuznets Curve (EKC) hypothesis is tested, in order to evaluate whether economic growth and carbon dioxide emissions are linked with an inverse U-shaped relationship and with an N-shape relationship in higher income levels. The EKC hypothesis is confirmed for high income and upper-middle income countries, but not for lower-middle and low income levels and a bidirectional Granger causality is found between GDP per capita and CO_2 per capita in all income levels.

Keywords: CO₂ emissions; electricity production; Environmental Kuznets Curve; fossil fuels; renewables; economic growth; income levels

1. Introduction

The rapid economic growth that followed the industrial revolution had a major impact on the environment. Fossil fuels were the core of the new industrialized world and their use started growing rapidly, reaching millions of tons of oil equivalents by today [1]. This excessive use and burning led to the emission of greenhouse gases (GHG) into the atmosphere which, in large amounts, contribute to global warming and climate change [2].

Carbon dioxide (CO_2) emissions are the number one anthropogenic contributor to climate change, since they constitute 81% of total GHG emissions for 2018. At the same time, CO_2 emissions that come from fossil fuel and industrial processes constitute 65% of total GHG emissions (according to 2010 data) [3]. These emissions are expected to increase even more: global population is expected to rise to approximately 9 billion by 2050 [4] and, therefore, world energy consumption is expected to rise nearly 50% between 2018–2050 [5].

In 1992, the United Nations Framework Convention on Climate Change (UNFCCC) and the Kyoto Protocol in 1997, made it obvious that, in order to avoid a disastrous effect on the environment, it is essential to reduce the world's GHG emissions to a large extent [2]. Even so, according to recent data, there seems to be a 61.62% increase of total CO_2 emissions (kt) in the world from 1990 until 2016 [6]. At the same time, fossil fuel energy consumption, as a percentage of total energy use, has not changed significantly and energy use (kg of oil equivalent per capita) has increased by 15.6% in the period 1990–2014 [7]. All these data emphasize the urgent need to implement CO_2 emissions reduction measures, by limiting the use of fossil fuels and switching to renewable energy sources instead [8].

Energy consumption seems to be the main cause of the large CO_2 emissions. At the same time, higher energy consumption leads to higher economic development [9].

Citation: Halkos, G.E.; Gkampoura, E.-C. Examining the Linkages among Carbon Dioxide Emissions, Electricity Production and Economic Growth in Different Income Levels. *Energies* 2021, *14*, 1682. https://doi.org/ 10.3390/en14061682

Academic Editor: Wing-Keung Wong

Received: 26 February 2021 Accepted: 16 March 2021 Published: 18 March 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

Department of Economics, University of Thessaly, 38333 Volos, Greece; egkampoura@uth.gr

According to Adams et al. [10], if non-renewable energy consumption increases by 10%, economic growth will increase by 2.11%, but if renewable energy consumption increases by 10%, economic growth will increase by 0.27%. This is why many scientists argue that a reduction in CO_2 would have a negative outcome for economic growth, something that would be an undesired result in developed and, especially, in developing countries [9]. The links and the relationship between energy consumption, CO_2 emissions and economic development have been intensively studied in the last decades [11].

The linkages among CO_2 emissions and electricity production have not been widely examined in the current literature. This study aims to contribute to the existing literature and examines the causality among economic growth, electricity production and CO_2 emissions in 119 world countries, categorized by income status (high, upper-middle, and lower-middle and low income), over the period of 2000–2018. Countries of different income levels are expected to have substantial differences regarding the relationships that exist among these factors and their identification is significantly important, since it can provide a better understanding and important knowledge for policy makers, in order to implement targeted measures for an efficient energy transition and the achievement of global sustainability. This study examines and assesses all these different linkages, for a large number of world countries classified by income with recent data, something that has not been widely investigated in the current literature. To achieve that, panel data are collected and the linkages between CO₂ emissions, electricity production from fossil fuels, electricity production from renewable sources and GDP per capita are investigated, while taking into consideration population density as well. Static and dynamic regression models are constructed, an in-depth econometric analysis is conducted, the Environmental Kuznets Curve is assessed for each income level and Granger causality is tested.

The paper is organized as follows: Section 2 presents recent economic and energy data, as well as data regarding CO_2 emissions that come from different energy sources. Section 3 includes an in-depth literature review on the examined field and Section 4 presents used data and methodology. In Section 5, the results are presented and in Section 6 the results are discussed. Section 7 concludes the paper.

2. Recent World Data

The world's GDP has increased rapidly over the last twenty years, from \$33.624 trillion in 2000 to \$87.799 trillion in 2019 (Figure 1), according to the World Bank database [12]. Over the same time period, global population increased from 6.114 billion in 2000 to 7.674 billion in 2019 [13], meaning that GDP per capita increased from \$5499.151 in 2000 to \$11,441.733 in 2019 [14].



Figure 1. World's GDP (current US\$) (Data source: World Bank [12]).

Focusing on the energy sector, world's total final energy consumption reached 9,937,702 kilotonnes of oil equivalent in 2018 [15]. The industrial sector and the transportation sector were the highest consumers of world's total energy supply (Figure 2) and fossil fuels were energy's main provider. According to IEA data [16], in 2017, the share of renewables in world's final energy consumption was estimated at 17.3%. The residential sector was the highest consumer of renewable energy supply (Figure 3).



Figure 2. Energy Consumption by Sector in 2017 (Data source: IEA [15]).



Figure 3. Renewables Consumption by Sector in 2017 (Data source: IEA [16]).

World's total CO₂ emissions reached 33,513.3 Mt of CO₂ in 2018 [17], estimating thus the per capita emissions at 4.4 tonnes CO₂ [18]). Coal was the energy source that was responsible for most of energy-related CO₂ emissions in the world, while oil followed [19] (Figure 4). In addition, world's total forest area has decreased over the last two decades; from 40,556,022.3 km² in 2000, it decreased to 39,958,245.9 km² in 2016, according to the World Bank database [20].



Figure 4. CO₂ emissions by energy source in 2017 (Data source: IEA [17]).

3. Literature Review

The existing literature about links and relationships between energy consumption, environmental pollution and economic growth is divided into three categories. The first category concerns the investigation of the CO₂ Environmental Kuznets Curve hypothesis. According to the EKC hypothesis, environmental pressure in an economy starts growing as the income grows, reaches a peak and, after a certain level of income, starts reducing [21]. This happens because, as a nation tries to develop, uses its natural resources with no concerning on the environmental degradation; after a certain income level, and since environmental degradation can lead to various problems, nations focus on improving environmental quality and protecting the environment [22]. Based on the EKC hypothesis, an inverted U-shaped relationship exists between economic development and environmental degradation. For high income countries, the EKC hypothesis has a cubic and not a quadratic form and an N-type curve is observed, indicating that even higher levels of GDP per capita lead to an increase of environmental degradation [23]. The EKC has significant implications for sustainability [24].

A plethora of studies have been carried out in which this hypothesis is tested, starting from 1991 and the empirical study by Grossman and Krueger [9]. A variety of recent studies have examined and evidenced the CO_2 EKC hypothesis for various different regions [25–30]; in contrast, some recent studies that have found that the EKC hypothesis wasn't valid for certain regions also exist in the literature [31–34].

The second category concerns the research of causality among economic growth and energy consumption. These studies examine the hypothesis that growth in economy is related to growth in energy use and they test that relationship using time series models, usually with Granger causality and cointegration models. Mehrara [35], Narayan and Smyth [36], Apergis and Payne [37], Ozturk et al. [38] and Apergis and Payne [39], among others, have examined this hypothesis.

The third category combines the previous two categories by examining the relationship among economic growth, energy consumption (renewable and non-renewable), CO₂ emissions and other variables (urbanization, trade, etc.). These studies examine the argument that economic growth has a long-term influence on energy consumption and pollution growth [9]. Wang et al. [40] have found a bidirectional causality between CO_2 emissions and energy consumption and between economic growth and energy consumption among 28 provinces in China, while energy consumption and economic growth are found to be the cause for CO_2 emissions in the long run. Lu [41] has also reached the same results in his study for 24 Asian countries. Lin and Moubarak [42], in their study for China, have found a bidirectional causality between renewable energy consumption and economic growth, although they found no causality between carbon emissions and renewable energy consumption. Pao and Tsai [43] have evidenced a bidirectional causality between income, energy consumption and emissions in Brazil, while Pao et al. [44] have found the same results for Russia. In contrast, Lotfalipour et al. [9] in their study on Iran, found a unidirectional causality from economic growth to CO_2 emissions and no causality from fossil fuels consumption to CO₂ emissions. Also, Soytas et al. [45] in their study for the United States, found no Granger causality between income and CO2 emissions and between energy use and income.

Some recent studies have been focusing specifically on European countries and the relationship between economic growth, energy consumption and carbon emissions that exists. Examples of those studies include the following: Acaravci and Ozturk [46] examined these relationships for 19 European countries and found a long run relationship between CO_2 emissions, energy consumption and economic growth only for specific countries. They also confirmed the EKC hypothesis in Denmark and Italy. Pirlogea and Cicea [47] also examined the links between energy consumption and economic growth and found that there is a unidirectional causality from renewable energy consumption to economic growth in Romania and from energy consumption (natural gas) to economic growth in Spain on short-run, concluding that there is a long run equilibrium between economic

growth and energy consumption in every EU country. Bölük and Mert [11] have tested the EKC hypothesis in 16 EU countries and they concluded that the EKC hypothesis was not valid in these countries. Kasman and Duman [48] examined the causality among energy consumption, economic growth, CO_2 emissions, taking also into consideration the trade openness and urbanization, for 15 European countries. They provided evidence that support the EKC hypothesis and they found a unidirectional causality from energy consumption, trade openness and urbanization to CO_2 emissions, among others. Alper and Oguz [49] examined the relationship between economic growth, renewable energy consumption, capital and labor for six EU countries and concluded that renewable energy consumption has a positive impact on economic growth for all 6 countries.

The examination of the Energy-Environmental Kuznets Curve (EEKC) has also been a topic of interest in the literature and has been assessed on a global and regional scale. More specifically, various studies have been focusing on the examination of the linkages that exist between economic growth and energy consumption. Some studies have managed to confirm the existence of the EEKC globally and regionally [50–52], while a plethora of studies exist that could not confirm the hypothesis [53–55].

Regarding the linkages that exist among CO_2 emissions, electricity production and economic growth, fewer studies have been focusing on that. For instance, in a recent study for Ghana, it was found that a bidirectional causality exists from hydroelectric sources' electricity production to CO_2 emissions, while a unidirectional causality is found from CO_2 emissions to renewables and waste energy production, as well as from CO_2 emissions to fossil fuels electricity production (oil, gas and coal), among others [56]. For the case of Pakistan, it was found that, among others, a weak unidirectional causality exists from CO_2 emissions to electricity production, both from natural gas and oil [57]. Focusing on Europe, and more specifically on the case of Italy, the EKC hypothesis has been validated, while it has been found that in fact electricity production per capita that comes from renewable sources can lead to a reduction of CO_2 emissions per capita, both short-term and long-term [58].

Only a few studies in the literature have studied these linkages and have tested the EKC hypothesis for different income levels. For example, Al-Mulali et al. [59] investigated the EKC hypothesis for different income groups, while taking into consideration the Ecological Footprint instead of CO₂ emissions to stand for environmental degradation. The authors confirmed the EKC hypothesis only for high income and upper-middle income countries, while the hypothesis was not valid for lower-middle and low incomes. Similarly, Ulucak and Bilgili [60] followed a similar approach, using the Ecological Footprint and classifying the studied countries by income. The authors confirmed the EKC hypothesis for all income levels. In addition, Aruga [52] examined the EEKC hypothesis for 19 Asia-Pacific countries, depending on income, and the results indicated that the EEKC hypothesis was confirmed only for high income countries, and not for low and middle income.

As it is highlighted, there is a plethora of studies that examine the causality among economic growth, energy consumption and carbon dioxide emissions, using different econometric procedures and techniques and their results differ substantially. This study aims to provide a comprehensive approach with recent data, focusing specifically on electricity production and including 119 world countries categorized by income level, assessing thus the different relationships that exist among these factors in different income groups. The study contributes to the existing literature, by combining all the above elements with an in-depth econometric analysis that is followed.

4. Materials and Methods

4.1. Data

Panel data were collected from the International Energy Agency (IEA) and the World Bank Database for 119 world countries and for the period 2000–2018. These countries were categorized based on their income level, as it has been identified by the World Bank, that takes into consideration GNI per capita (current USD) to divide the countries to four different income groups. The newest classifications were set on July 2020 and the thresholds are presented in Table 1 [61]. The 119 countries for which data were collected and that were included in the analysis are presented in Figure 5, classified by income level: 47 countries are identified as high income countries, 33 as upper-middle income countries, 32 as lower-middle income countries and 7 as low income countries. The indicators for which data were extracted are presented in Table 2.

Table 1. Income classification thresholds, as set by the World Bank.

Income Level	Threshold (July 2020)
High income	>12,535
Upper-middle income	4046-12,535
Lower-middle income	1036–4045
Low income	<1036



Figure 5. The 119 countries included in the analysis, by income level.

	Source	Indicator	Measurement
1	IEA [62]	Total electricity production	GWh
2	WB [13]	Population	total
3	WB [63]	Electricity production from oil, gas and coal sources	% of total
4	WB [64]	Electricity production from renewable sources, excluding hydroelectric	% of total
5	WB [65]	Electricity production from hydroelectric sources	% of total
6	WB [66]	CO_2 emissions	Metric tons per capita
7	WB [14]	GDP per capita	Current US\$
8	WB [67]	Population density	People per sq. km of land area

Table 2. Extracted indicators and data so	urces.
---	--------

Indicators 4&5 were combined, in order to create a variable that refers to electricity production from renewable sources, including hydroelectric. Indicators 1&2 were used to estimate electricity production per capita, so that electricity production from fossil fuels (*EPFpc*) and renewable sources (including hydroelectric) per capita (*EPRpc*) will be estimated, based on the indicators 3&4. Forecasts were provided, relying on exponential smoothing, in order to complete the missing data for the last few years wherever it was necessary. To achieve that, various forecast accuracy measures were examined, such as Mean Absolute Percentage Error—MAPE, Mean Square Deviation—MSD, etc.

4.2. Econometric Methodology

The EKC curve and the relationship and causality between CO₂ emissions and GDP, electricity production from fossil fuels per capita, electricity production from renewable sources per capita and population density were examined, based on the following methodology.

Before performing the regression analysis, several econometric tests are conducted to address different problems that might occur. A usual problem when working with panel data is variables' correlation; in order to determine if the time-series are cross-sectional independent, Pesaran's cross-section dependence test is used. OLS Dummy estimator (FEM) allowing for individual fixed effects with Driscoll-Kraay standard errors can assist in the correction of the variance-covariance matrix, in cases where the time series are found to be cross-sectional dependent. For Random Effects, Breusch-Pagan LM test for individual effects is applied and robust standard errors are required.

In cases where cross-section dependence is evidenced, unit root tests are performed. Dickey-Fuller and Augmented Dickey-Fuller tests can be performed when analyzing panel data, with the issue of homogeneity in the autoregressive parameter. Fisher type tests do not adopt this restrictive assumption and they don't require strongly balanced panels. The asymptotic behavior of the time series (T) and the cross-section dimensions (N) should be taken into consideration when performing unit root tests. Fisher type tests can be used in cases where T and N tend to infinity, but the number of panels with no unit root must raise at the same rate as N.

In cases of non-stationarity, panel cointegration tests are performed: more specifically, Westerlund test are performed to check for panel cointegration, based on the significance of the error correction term in the error correction model. Westerlund proposed four cointegration tests: the G_t and G_a statistics, which test the null hypothesis of no cointegration for all cross-sectional units, rejecting the hypothesis in cases of cointegration for at least one unit, and the P_t and the P_a statistics, which reject the hypothesis in cases of cointegration of the panel in total. In addition, the causal relationships among the studied factors are examined by conducting Granger causality tests. Granger causality can help identify whether the relationship between two variables is unidirectional, bidirectional or if no causality exists between them [68,69].

Three different data sets are constructed: one for high income countries (47 countries), one for upper-middle income countries (33 countries) and one for lower-middle & low income countries (39 countries). After the data collection, their combination and the extraction of the necessary variables, Box-Cox tests have been used, in order to test linear against logarithmic forms. Quadratic regression models, as well as a cubic regression model were constructed, in order to examine the linkages among the studied variables, considering CO_2 emissions per capita as a dependent variable and GDP per capita, per capita electricity production from fossil fuels, per capita electricity production from renewables and population density as independent variables. The general forms of these models are:

$$Y_{it} = a + X_{it}\beta_{it} + X_{it}^2\beta_{it} + \delta_i + \gamma_i + \varepsilon_{it}$$
(1)

$$Y_{it} = a + X_{it}\beta_{it} + X_{it}^2\beta_{it} + X_{it}^3\beta_{it} + \delta_i + \gamma_i + \varepsilon_{it}$$
(2)

where Y_{it} is the dependent variable, X_{it} an independent variables' k-vector, δ_i and γ_i the cross-section and period specific effects, that can be either fixed or random, and ε_{it} the disturbance terms. After modification, the proposed models that include only the statistically significant variables, become:

$$CO2pc_{it} = a + \beta_{1it}GDPpc_{it} + \beta_{2it}GDPpc_{it}^2 + \beta_{3it}EPFpc_{it} + \beta_{4it}EPRpc_{it} + \beta_{5it}Dens_{it} + \delta_i + \gamma_i + \varepsilon_{it}$$
(3)

 $CO2pc_{it} = a + \beta_{1it}GDPpc_{it} + \beta_{2it}GDPpc_{it}^{2} + \beta_{3it}GDPpc_{it}^{3} + \beta_{4it}EPFpc_{it} + \beta_{5it}EPRpc_{it} + \beta_{6it}Dens_{it} + \delta_{i} + \gamma_{i} + \varepsilon_{it}$ (4)

In Equations (3) and (4), *CO2pc* stands for CO₂ emissions per capita, *GDPpc* for GDP per capita, *EPFpc* for electricity production from fossil fuels (oil, gas and coal) per capita, *EPRpc* for electricity production from renewable sources per capita (including hydroelectric) and *Dens* for population density.

To estimate the panel data models, the ordinary least squares (OLS) method was chosen and Fixed and Random Effects methods were applied; the choice of the appropriate method depends on the way that a_i is handled (fixed predefined number or random expulsion from a particular distribution). In the case of Random Effects, Hausman tests are also conducted, in order to check for inconsistencies in the RE estimations. The literature also suggests the use of fully modified ordinary least squares (FMOLS), a reliable non-parametric method that assists in tackling problems related to variables' endogeneity and serial correlation [70,71]. FMOLS estimators seem to perform significantly well in cases where the time series dimension is bigger than the cross sectional dimension [70]. In the present study, and since the cross sectional dimension is significantly bigger than the time series dimension, Fixed Effects with Driscoll-Kraay standard errors were chosen to be used, when modeling the static analysis. This way, the problem of cross-section dependence is prioritized and addressed.

In addition to the OLS method, and in order to capture the dynamic nature of the model, Generalized Method of Moments (GMM) was used for estimation, in terms of Orthogonal Deviations. GMM is used in statistical models in order to provide estimators for the parameters that are consistent, as well as asymptotically normally distributed [72]. It is a significantly important method for econometrics and is widely used in economics, since it can be applied in various models (linear/non-linear, cross-section, time series and panel data, etc.) [73]. In cases where moment conditions can be obtained, while the likelihood function cannot, GMM combines the moments and provides efficient estimators [74]. GMM assists in avoiding endogeneity, since it extends the static model, by including lagged variables that help control the problem, as well as in avoiding the problems of autocorrelation and reverse causation [75,76]. Due to the many advantages that come with its use, Generalized Method of Moments was chosen over dynamic ordinary least squares (DOLS), a parametric method that uses lagged terms and assists in endogeneity and serial correlation problems [77,78].

GMM minimizes the following Equation (5), regarding β :

$$M(\beta) = \left(\sum_{i=1}^{N} \Psi'_{i} u_{i}(\beta)\right) W\left(\sum_{i=1}^{N} \Psi'_{i} u_{i}(\beta)\right) = \zeta(\beta)' W\zeta(\beta)$$
(5)

In this equation, *W* is a *pxp* weighting matrix, Ψi is a *T_ixp* instruments matrix for cross section *i* and $u_i(\beta) = (\Upsilon_i - f(X_{it}, \beta))$. White robust covariances are used to calculate the weighting of matrix *W* and the coefficient covariance estimates are:

$$\left(\frac{M^*}{M^*-k^*}\right)\left(\sum_t X_t'X_t\right)^{-1}\left(\sum_t X_t'\hat{u}_t\hat{u}_t'X_t\right)\left(\sum_t X_t'X_t\right)^{-1}$$
(6)

In Equation (6), M^* is the total number of stacked observations and k^* equals to the number of estimated parameters. According to Arellano and Bond [79], in orthogonal deviations each observation is seen as a deviation from the average of future observations and each deviation is weighted, in order to standardize the variance:

$$x_{it}^* = \left[x_{it} - \left(x_{i(t+1)} + \dots + x_{iT} \right) / (T-t) \right] \sqrt{(T-t)} / \sqrt{T-t+1}$$
(7)

The (T_i-q) equations for individual unit (*i*) are:

$$Y_i = \delta w_i + d_i \eta_i + \nu_i \tag{8}$$

5. Results

5.1. Descriptive Statistical Analysis

The indicators presented in Table 2 were analyzed and combined, so that the necessary variables could be extracted, such as per capita electricity production from fossil fuels and from renewable sources. Table 3 presents the descriptive statistics of the indicators that were used in the analysis for high income countries.

Table 3. Descriptive Statistics of High income countries.

	EPFpc	EPRpc	CO2pc	GDPpc	Dens
Mean	0.004684	0.003144	10.39119	32,195.14	336.1680
Median	0.003691	0.000939	8.072146	27,729.19	109.5809
Maximum	0.021955	0.056814	67.31050	118,823.6	7952.998
Minimum	0.00000331	0	0.251345	1659.908	2.493134
Std. Dev.	0.004552	0.007736	8.274268	21,106.53	1028.928
Skewness	1.748568	4.755374	2.960003	1.164747	5.988748
Kurtosis	5.968018	28.28739	15.53083	4.738511	39.83826
Jarque-Bera	782.8296	27158.6	7146.537	314.3716	55,831.77
Prob.	0.0000	0.0000	0.0000	0.0000	0.0000

The highest levels of electricity production from fossil fuels, throughout the studied time-period and among the 47 high income countries that were examined, were observed in Bahrain (0.021955 GWh/capita in 2006), while the lowest levels were observed in Uruguay (0.00000331 GWh/capita in 2003). Respectively, the highest levels regarding electricity production from renewable sources were observed in Iceland (0.0568 GWh/capita in 2015), while zero levels were observed in various countries throughout the studied time-period.

Qatar was the country with the highest levels of CO_2 emissions per capita for the whole time-period, with the highest being observed in 2001 (67.31 metric tons per capita); some of the lowest levels of CO_2 emissions per capita were observed in Malta and Uruguay. At the same time, in Luxembourg were observed the highest levels of GDP per capita, reaching \$118,823.65 in 2014, while the lowest GDP per capita levels were observed in Romania (\$1659.9 in 2000). The highest population density was observed in Singapore for the whole time-period (7952.998 people/sq.km in 2018), while the lowest population density was observed in Australia (2.49 people/sq.km in 2000).

Similarly, Table 4 presents the descriptive statistics for upper-middle income countries and Table 5 for lower-middle and low income countries. The highest levels of electricity production from fossil fuels, among the 33 upper-middle income countries were observed in Libya (0.005999 GWh/capita in 2013), while zero levels were observed in Paraguay and Albania for various years. Similarly, the highest levels of electricity production from renewable sources were observed in Paraguay (0.010049 GWh/capita in 2000), while zero levels were observed in Libya for the whole time period and in Botswana for the years 2000–2012.

Table 4. Descriptive Statistics of Upper-middle income countries.

	EPFpc	EPRpc	CO2pc	GDPpc	Dens
Mean	0.001757	0.000913	4.215383	5545.145	72.05436
Median	0.001421	0.000494	3.306489	4986.676	65.22279
Maximum	0.005999	0.010049	15.6463	19288.6	270.9931
Minimum	0	0	0.657959	622.7421	2.179756
Std. Dev.	0.001496	0.001577	3.090284	3226.249	58.80999
Skewness	0.751483	4.073107	1.262113	1.008053	1.092193
Kurtosis	2.593474	20.94012	4.227859	4.117945	4.357204
Jarque-Bera	63.3315	10141.95	205.8481	138.8408	172.7788
Prob.	0.0000	0.0000	0.0000	0.0000	0.0000

	EPFpc	EPRpc	CO2pc	GDPpc	Dens
Mean	0.000432	0.000232	1.139077	1470.063	134.9511
Median	0.000188	0.000133	0.611946	1133.186	73.57522
Maximum	0.002133	0.002499	15.1386	5591.212	1239.579
Minimum	0	0	0.01628	111.9272	1.543177
Std. Dev.	0.000575	0.000372	1.584955	1133.959	192.4295
Skewness	1.481985	4.07663	3.406818	1.232556	3.601976
Kurtosis	3.86273	22.02623	19.41762	3.9178	18.42611
Jarque-Bera	294.221	13229.1	9755.384	213.6282	8949.486
Prob.	0.0000	0.0000	0.0000	0.0000	0.0000

Table 5. Descriptive Statistics of Lower-middle and Low income countries.

The highest levels of CO_2 emissions were observed in Kazakhstan for the whole time period (15.65 metric tons per capita in 2011) and high levels were observed in the Russian Federation as well; the lowest levels of carbon dioxide emissions were observed in Paraguay for most of the studied years (0.658 metric tons per capita in 2005). Among the studied upper-middle income countries, Venezuela had the highest GDP per capita and Armenia had the lowest. Population density was higher in Jamaica for the whole studied time period (270.99 people/sq.km in 2018) and lower in Namibia for the whole time period (2.18 people/sq.km in 2000).

In the case of lower-middle & low income countries, the highest levels of electricity production from fossil fuels were observed in Ukraine (0.00213 GWh/capita in 2012), while zero levels were observed in Nepal, Ghana and the Republic of the Congo in various years. The highest levels of electricity production from renewable sources were observed in Tajikistan (0.0025 GWh/capita in 2005), while zero levels were observed in Niger, Mongolia and Benin for various years.

Mongolia presented the highest levels of CO_2 emissions for various years (15.14 metric tons per capita in 2013), while the Democratic Republic of the Congo presented the lowest levels for the whole time period (0.016 metric tons per capita in 2001). The highest GDP per capita was observed in Algeria (\$5591.2 in 2012), while the lowest levels of GDP per capita were observed in Ethiopia (\$111.93 in 2002). The highest levels of population density were observed in Bangladesh throughout the whole time period (1239.56 people/sq.km in 2018), while the lowest levels of population density were observed in Mongolia through the whole time period (1.54 people/sq.km in 2000).

By comparing the means, it can be observed that the highest levels of per capita electricity production from fossil fuels, as well as from renewable sources, were found in high income countries, while lower-middle & low income countries had significantly lower levels. The same can be concluded regarding CO_2 per capita levels: there are obvious differences in the levels of high income, upper-middle income and lower-middle & low income countries, with high income countries being those who pollute more. Population density was, on average, higher in high income countries and lower in upper-middle income countries.

5.2. Cross-Section Dependence and Unit Roots

Pesaran CD test is performed for each different data set, in order to test for crosssection dependence. The results reject the null hypothesis in all cases and suggest the existence of cross-section dependence (Table 6), indicating that unit root tests should be conducted. In addition, these results suggest the use of Driscoll-Kraay standard errors in the static regression models, in order to correct the variance-covariance matrix.

Variables	High Income	Upper-Middle Income	Lower-Middle & Low Income
EPFpc	23.065 ***	36.613 ***	31.038 ***
	[0.0000]	[0.0000]	[0.0000]
EPRpc	5.98 ***	6.94 ***	6.639 ***
	[0.0000]	[0.0000]	[0.0000]
CO2pc	40.812 ***	22.051 ***	54.969 ***
	[0.0000]	[0.0000]	[0.0000]
GDPpc	118.973 ***	85.592 ***	102.98 ***
	[0.0000]	[0.0000]	[0.0000]
Dens	46.212 ***	35.446 ***	94.99 ***
	[0.0000]	[0.0000]	[0.0000]

Table 6. Cross-section dependence (Pesaran CD test).

Note: The null hypothesis assumes that there exists no cross-section dependence (correlation). Significance at *** 1%.

Unit root tests are performed for each data set separately (Tables 7–9). The performed unit root tests (Fisher-ADF and Fisher PP) indicate that the examined variables are I(1) and evidence of stationarity exist in first differences.

Variables	Fisher—ADF	Fisher—PP		Fisher—ADF	Fisher—PP
Levels			First Differences		
EPFpc	55.3359 [0.9995]	66.67 [0.9853]	EPFpc	335.237 *** [0.0000]	871.135 *** [0.0000]
EPRpc	40.692 [1.0000]	705398 [0.8861]	EPRpc	308.303 *** [0.0000]	1128.62 *** [0.0000]
CO2pc	51.7633 [0.9999]	66.174 [0.9869]	CO2pc	336.321 *** [0.0000]	1207.49 *** [0.0000]
GDPpc	74.3357 [0.9331]	40.9579 [1.0000]	GDPpc	289.557 *** [0.0000]	389.16 *** [0.0000]
Dens	103.584 [0.2343]	79.5754 [0.8559]	Dens	232.967 *** [0.0000]	181.673 *** [0.0000]

Table 7. Fisher-ADF & Fisher-PP panel unit root test for high income countries.

Note: The null hypothesis assumes that the variable contains unit root. *P*-values in brackets. Significance at *** 1%.

Table 8. Fisher-ADF & Fisher-PP panel unit root test for upper-middle income countries.

Variables	Fisher—ADF	Fisher—PP		Fisher—ADF	Fisher—PP
Levels			First Differences		
EPFpc	46.8653 [0.9642]	66.8678 [0.4470]	EPFpc	257.191 *** [0.0000]	738.596 *** [0.0000]
EPRpc	23.9952 [1.0000]	28.9385 [1.0000]	EPRpc	273.322 *** [0.0000]	1072.02 *** [0.0000]
CO2pc	25.4143 [1.0000]	23.1849 [1.0000]	CO2pc	231.818 *** [0.0000]	743.485 *** [0.0000]
GDPpc	38.4883 [0.9973]	27.3549 [1.0000]	GDPpc	175.707 *** [0.0000]	273.061 *** [0.0000]
Dens	78.4815 [0.1397]	70.7227 [0.3230]	Dens	474.520 *** [0.0000]	229.837 *** [0.0000]

Note: The null hypothesis assumes that the variable contains unit root. *P*-values in brackets. Significance at *** 1%.

Variables	Fisher—ADF	Fisher—PP		Fisher—ADF	Fisher—PP
Levels			First Differences		
EPFpc	54.6241 [0.9796]	50.7312 [0.9929]	EPFpc	225.482 *** [0.0000]	674.029 *** [0.0000]
EPRpc	74.4705 [0.5923]	71.3106 [0.6907]	EPRpc	290.486 *** [0.0000]	1154.4 *** [0.0000]
CO2pc	47.9935 [0.9970]	54.6955 [0.9792]	CO2pc	275.946 *** [0.0000]	1002.39 *** [0.0000]
GDPpc	35.1546 [1.0000]	32.7017 [1.0000]	GDPpc	195.040 *** [0.0000]	546.373 *** [0.0000]
Dens	66.5735 [0.6584]	28.7294 [1.0000]	Dens	324.933 *** [0.0000]	124.692 *** [0.0006]

Table 9. Fisher-ADF & Fisher-PP panel unit root test for lower-middle & low income countries.

Note: The null hypothesis assumes that the variable contains unit root. *P*-values in brackets. Significance at *** 1%.

5.3. Cointegration

In order to test for panel cointegration, Westerlund panel cointegration tests are performed for each data set separately. The null hypothesis of no cointegration is rejected from the G_t and G_a statistics in almost every case, implying cointegration for at least one unit, as well as from the P_t and P_a statistics in almost every case, implying cointegration for the gration for the whole panel (Tables 10–12).

Table 10. Westerlund Panel Cointegration Test for high income countries.

Equation	Gt	Ga	Pt	Pa
CO2pc = f(GDPpc)	-5.665 ***	-21.79 ***	-30.947 ***	-21.182 ***
	[0.000]	[0.000]	[0.000]	[0.000]
$CO2pc = f(GDPpc^2)$	-5.681 ***	-24.376 ***	-32 ***	-21.387 ***
	[0.000]	[0.000]	[0.000]	[0.000]
$CO2pc = f(GDPpc^3)$	-5.742 ***	-25.911 ***	-33.582 ***	-22.771 ***
	[0.000]	[0.000]	[0.000]	[0.000]
CO2pc = f(EPFpc)	-5.913 ***	-21.832 ***	-32.825 ***	-17.976 ***
	[0.000]	[0.000]	[0.000]	[0.000]
CO2pc = f(EPRpc)	-5.335 ***	-21.844 ***	-35.717 ***	-25.058 ***
	[0.000]	[0.000]	[0.000]	[0.000]
CO2pc = f(Dens)	-6.118 ***	-12.374	-31.202 ***	-9.953
	[0.000]	[0.312]	[0.000]	[0.126]

Note: The null hypothesis assumes no cointegration. Significance at *** 1%.

Equation	Gt	Ga	Pt	Pa
CO2pc = f(GDPpc)	-4.974 ***	-18.636 ***	-25.683 ***	-17.946 ***
	[0.000]	[0.000]	[0.000]	[0.000]
$CO2pc = f(GDPpc^2)$	-5.047 ***	-19.918 ***	-24.775 ***	-20.762 ***
	[0.000]	[0.000]	[0.000]	[0.000]
$CO2pc = f(GDPpc^3)$	-5.080 ***	-21.058 ***	-25.619 ***	-21.604 ***
	[0.000]	[0.000]	[0.000]	[0.000]
CO2pc = f(EPFpc)	-5.165 ***	-18.395 ***	-26.173 ***	-17.93 ***
	[0.000]	[0.000]	[0.000]	[0.000]
CO2pc = f(EPRpc)	-5.232 ***	-19.344 ***	-24.499 ***	-22.053 ***
	[0.000]	[0.000]	[0.000]	[0.000]
CO2pc = f(Dens)	-6.119 ***	-3.307	-20.493 ***	—3.956
	[0.000]	[0.998]	[0.000]	[0.999]

Table 11. Westerlund Panel Cointegration Test for upper-middle income countries.

Note: The null hypothesis assumes no cointegration. Significance at *** 1%.

Table 12. Westerlund Panel Cointegration Test for lower-middle & low income countries.

Equation	Gt	Ga	Pt	Pa
CO2pc = f(GDPpc)	-4.896 ***	-14.962 ***	-28.97 ***	-17.059 ***
	[0.000]	[0.002]	[0.000]	[0.000]
$CO2pc = f(GDPpc^2)$	-5.119 ***	-15.827 ***	-28.624 ***	-18.073 ***
	[0.000]	[0.000]	[0.000]	[0.000]
$CO2pc = f(GDPpc^3)$	-5.129 ***	-17.464 ***	-28.318 ***	-18.782 ***
	[0.000]	[0.000]	[0.000]	[0.000]
CO2pc = f(EPFpc)	-5.117 ***	-17.313 ***	-27.646 ***	-17.906 ***
	[0.000]	[0.000]	[0.000]	[0.000]
CO2pc = f(EPRpc)	-4.493 ***	-18.4 ***	-26.339 ***	-19.398 ***
	[0.000]	[0.000]	[0.000]	[0.000]
CO2pc = f(Dens)	-6.021 ***	-2.731	-12.693	-3.51
	[0.000]	[0.999]	[0.710]	[0.998]

Note: The null hypothesis assumes no cointegration. Significance at *** 1%.

5.4. Regression Results

Six different regression models are constructed, in order to examine the existence of EKC curve and the relationships among the studied variables in different income levels. The Hausman tests imply the use of fixed effects model specifications and columns 2, 4 and 6 present the results of FE Driscoll-Kraay standard errors, as it was indicated by the Pesaran CD tests (Section 5.2).

The regression results for high income countries indicate that GDP per capita is a driver of CO_2 emissions per capita, by both Fixed Effects Method and GMM. An N-shaped curve is found to connect the studied variables in the static model, confirming the hypothesis that even higher income levels can increase environmental degradation. In the dynamic model, an inverted U-shape relationship is found to connect GDP per capita and CO_2 emissions per capita, supporting the existance of an inverted U-shaped curve and confirming the EKC hypothesis. The results for upper-middle income countries also confirm the EKC hypothesis, since an inverted U-shape curve is found to connect GDP per capita and CO_2 emissions per capita, in both static and dynamic models. In contrast, the EKC hypothesis is not confirmed in lower-middle & low income countries. The static model implies a positive monotonic relationship between GDP per capita and carbon dioxide emissions, while the dynamic model supports the existance of a U-shape relationship between the two variables. Figure 6 presents graphically these relationships between GDP per capita and CO₂ emissions for all three different income levels, in both static and dynamic models.

Electricity production from fossil fuels is found to be a significant driver of CO_2 emissions, in every model, both static and dynamic, and for each one of the three income levels. Electricity production from renewable sources is found to be linked with an inverse relationship with CO_2 emissions, in the dynamic model of high income countries and in both static and dynamic models of lower-middle and low income countries, while it is statistically insignificant in the models of upper-middle income countries. Population density is found to be linked with an inverse relationship with CO_2 emissions, in both static and dynamic model of upper-middle income countries. Population density is found to be linked with an inverse relationship with CO_2 emissions, in both static and dynamic model of upper-middle income countries, while it is a small driver in the dynamic models of high income and lower-middle and low income countries.

The lag of the dependent variables is an autoregressive-distributed lag specification that ends up to an AD (1,0) formulation, where insignificant variables dynamics aren't included. All variables are assumed to be strictly exogenous, except the lagged dependent. Lagged variables in the dynamic models have a value less than 1 and are statistically significant (1% level), indicating a strong conditional convergence.



Figure 6. Cont.



Figure 6. Cont.



Figure 6. Derived relationships, where x axis represents GDP per capita and y axis CO₂ emissions per capita.

Since a lagged coefficient that equals 0 is an indication of instant adjustment, while a lagged coefficient that equals 1 is an indication of no adjustment [80], it is observed that the dynamic models of high income and lower-middle & low income countries present a slower adjustment to the equilibrium values, compared to the dynamic model of upper-middle income countries. More specifically, in the model of high income countries, the adjustment coefficient equals to 1–0.78. Since the lag coefficients show the adjustment to the equilibrium values, it can be seen that this adjustment equals to 22%, meaning that 22% of the discrepancy between actual and desired levels of efficiency is eliminated in a year; therefore, more than four periods are required for this adjustment. Similarly, the results of the lower-middle & low income countries model indicate that the adjustment coefficient equals to 1–0.67, meaning that 33% of the discrepancy between actual and desired levels is eliminated in a year and that approximately three periods will be required for this adjustment. In contrast, the dynamic model for upper-middle income countries presents an adjustment coefficient equal to 1–0.37, meaning that 63% of the discrepancy is eliminated in a year and that less than two periods will be required for the adjustment.

Both Wald tests of joint significance and Sargan tests of over-identifying restrictions are asymptotically distributed as χ^2 variables. Parentheses in Table 13 present the degrees of freedom. It can be seen that Sargan statistic does not reject the hypothesis of over-identifying restrictions and there is evidence of serially uncorrelated errors. AR(1) and AR(2) are first and second order serial autocorrelation tests, which indicate that the hypotheses of absence of autocorrelation is not rejected.

5.5. Granger Causality

In order to identify the relationships and the causality between the studied factors, Granger causality was examined for each one of the three datasets. Stacked test (with common coefficients) was chosen and 2 lags were included.

The results indicate that a bidirectional causality exists between GDP per capita and CO_2 emissions in all three different income levels, confirming the linkages that exist between these factors. A bidirectional causality is also found between GDP per capita and per capita electricity production from fossil fuels for high income and lower-middle & low income countries, while in the case of upper-middle income countries, a unidirectional causality is confirmed from electricity production from fossil fuels to GDP per capita.

	High 1	Income	Upper-Middle Income		Lower-Middle and Low Income	
	FE (DK se)	GMM	FE (DK se)	GMM	FE (DK se)	GMM
CO2(-1)		0.778178 *** (806.523)[0.0000]		0.373367 *** (28.541) [0.0000]		0.66676 *** (366.9915) [0.0000]
GDPpc	0.0000818 *** (4.17) [0.0001]	1.78E-05 *** (19.66458) [0.0000]	0.0002558 *** (12.39) [0.0000]	0.000138 *** (40.96653) [0.0000]	0.0001871 *** (2.91) [0.009]	-0.000283 *** (-72.10892) [0.0000]
GDPpc ²	-0.000000003 *** (-4.85) [0.000]	-0.000000000197 *** (-19.83668) [0.0000]	-0.0000000131 *** (-7.2) [0.0000]	-0.00000000587 *** (-18.86284) [0.0000]		0.000000737 *** (77.43749) [0.0000]
GDPpc ³	0.00000000000000 *** (3.99) [0.001]	.77				
EPFpc	1144.766 *** (9.48) [0.000]	1277.639 *** (280.942) [0.0000]	947.1906 *** (10.09) [0.0000]	463.3203 *** (25.78434) [0.0000]	1664.963 *** (11.37) [0.000]	381.1654 *** (49.16397) [0.0000]
EPRpc		-451.2071 *** (-41.62376) [0.0000]			-669.3433 *** (-3.08) [0.006]	-153.8469 *** (-6.690689) [0.0000]
Dens		0.00427 *** (133.5502) [0.0000]	-0.0181993 *** (-8.08) [0.0000]	-0.018673 *** (-12.3941) [0.0000]		0.0000698 *** (3.230135) [0.0013]
within R ²	0.3105		0.5794		0.2478	
Hausman	13.56 *** [0.0011]		90.33 *** [0.0000]		4.79 * [0.0912]	
Wald test		598234.3 (5)		8420.45 (4)		35743.48 (5)
Sargan test		47.65923 (42)		28.71 (28)		31.4265 (33)
AR(1)		-2.285 ** [0.0223]		-2.362 ** [0.0182]		-2.288 ** [0.0221]
AR(2)		-0.7995 [0.4240]		-1.025 [0.3054]		-0.9359 [0.3493]
Shape of curve	N–shape	InvertedU– shape	Inverted U–shape	Inverted U–shape	Line	U–shape
Turning points	15859.25 56497.18	45177.67	9763.36	11754.69		1919.95
Observations	893	799	627	561	741	663

Table 13.	Regression	results with	CO2pc as	dependent	variable.
	regreester	10000100 11101	ee_peac	erep errererr	

Note: t-Statistics in parentheses and *p*-values in square brackets. Parentheses in Wald and Sargan tests indicate degrees of freedom. Critical values for the Wald test of overall significance of the explanatory variables: $\chi^2_{0.05,5} = 11.07$, $\chi^2_{0.05,4} = 9.488$. Critical values for the Sargan test for over-identifying restrictions: $\chi^2_{0.05,4} = 58.124$, $\chi^2_{0.05,28} = 41.337$, $\chi^2_{0.05,33} = 47.4$. Significance at *** 1%, ** 5% and * 10%.

A unidirectional causality is also found from electricity production from fossil fuels to CO_2 emissions per capita, only for high income and lower-middle & low income countries, while this relationship is not confirmed in the case of upper-middle income countries. Instead, a causal relationship is found from CO_2 emissions to electricity production from fossil fuels for upper-middle income countries. In the case of high income countries, a bidirectional causality is found between GDP per capita and per capita electricity production from renewable sources, as well as between CO_2 emissions and population density. Electricity production from fossil fuels is found to Granger cause population density, in

high income and lower-middle & low income countries. In lower-middle & low income countries, unidirectional causal relationships are also found from per capita electricity production from renewable sources to electricity production from fossil fuels and from population density to electricity production from renewables and to GDP per capita as well (Table 14).

Null Hypothesis	High Income	Upper-Middle Income	Lower-Middle and Low Income
EPRpc does not Granger Cause EPFpc	0.1435	0.84386	8.49424 ***
	[0.8663]	[0.4306]	[0.0002]
EPFpc does not Granger Cause EPRpc	0.22982	1.66069	2.2018
	[0.7947]	[0.1909]	[0.1114]
CO2pc does not Granger Cause EPFpc	1.14026	5.78463 ***	1.68477
	[0.3203]	[0.0033]	[0.1863]
EPFpc does not Granger Cause CO2pc	6.58308 ***	0.2931	10.4235 ***
	[0.0015]	[0.7461]	[0.00003]
GDPpc does not Granger Cause EPFpc	9.3199 ***	0.78258	4.17817 **
	[0.0001]	[0.4577]	[0.0157]
EPFpc does not Granger Cause GDPpc	11.7925 ***	5.672 ***	5.40786 ***
	[0.000009]	[0.0036]	[0.0047]
Dens does not Granger Cause EPFpc	1.90495	1.58869	1.27445
	[0.1495]	[0.2051]	[0.2803]
EPFpc does not Granger Cause Dens	7.17602 ***	0.51733	4.85094 ***
	[0.0008]	[0.5964]	[0.0081]
CO2pc does not Granger Cause EPRpc	0.112	0.90316	0.5376
	[0.8941]	[0.4059]	[0.5844]
EPRpc does not Granger Cause CO2pc	0.17197	0.39549	1.11302
	[0.8420]	[0.6735]	[0.3292]
GDPpc does not Granger Cause EPRpc	4.5241 **	0.85324	0.29639
	[0.0111]	[0.4266]	[0.7436]
EPRpc does not Granger Cause GDPpc	11.4263 ***	0.65071	0.60933
	[0.00001]	[0.5221]	[0.5440]
Dens does not Granger Cause EPRpc	0.00811	2.22833	2.52252 *
	[0.9919]	[0.1087]	[0.0810]
EPRpc does not Granger Cause Dens	0.0213	0.43188	0.0277
	[0.9789]	[0.6495]	[0.9727]
GDPpc does not Granger Cause CO2pc	5.4029 ***	10.2292 ***	4.74665 ***
	[0.0047]	[0.00004]	[0.0090]
CO2pc does not Granger Cause GDPpc	3.1871 **	19.8117 ***	4.78144 ***
	[0.0418]	[0.000000005]	[0.0087]
Dens does not Granger Cause CO2pc	13.151 ***	0.81525	0.55917
	[0.000002]	[0.4431]	[0.5720]
CO2pc does not Granger Cause Dens	3.64024 **	0.26402	1.38006
	[0.0267]	[0.7681]	[0.2523]
Dens does not Granger Cause GDPpc	0.60785	1.34882	5.72788 ***
	[0.5448]	[0.2604]	[0.0034]
GDPpc does not Granger Cause Dens	1.38	1.70387	0.2997
	[0.2522]	[0.1829]	[0.7411]
Observations	799	561	663

Table 14. Granger Causality Results.

Note: t-Statistics in parentheses and *p*-values in square brackets. Rejection at *** 1%, ** 5% and * 10%.

6. Discussion

The present study confirms the existence of an inversed U-shaped curve for the 47 high income countries in the dynamic model and for the 33 upper-middle income countries in both static and dynamic quadratic model. These results suggest that environmental degradation increases along economic growth, but after a certain income level starts reducing. This indicates that, after reaching a certain level of growth, environmental measures and policies are promoted and there is a higher flow of resources towards environmental protection. At the same time, the results confirm the existence of an Nshaped curve in the static model of high income countries, confirming the assumption that, in high income countries, environmental degradation grows at first, as income grows, then starts reducing, after a certain income level, but it is once again increased at higher levels of GDP per capita [23]. Thus, it can be assumed that, in higher income levels, the existent measures and policies that had initially assisted in improving environmental conditions are not sufficient anymore, leading once more to an increase in environmental degradation. In the case of lower-middle and low income countries, the EKC hypothesis is not confirmed. The static model indicates a monotonic relationship where CO₂ emissions per capita increase as GDP per capita increases, while the dynamic model suggests the existence of a U-shape curve, meaning that in low income levels, GDP per capita has a negative effect on carbon dioxide emissions and it is only after a specific threshold (\$1919.95 per capita) that higher GDP per capita increases CO_2 emissions and leads to environmental degradation. Thus, it can be seen that lower-middle & low income countries have to focus on other issues and on their growth and do not have the resources to invest in environmental protection.

The estimated turning points in the case of the static high income countries model, compared to the maximum GDP per capita observed in the studied period for the 47 high income countries, indicate that at least one country existed in the years 2000–2018 that had passed the second turning point and as GDP per capita increased, environmental degradation increased, too. The estimated turning point of the dynamic model for the high income countries indicates, compared to the same maximum GDP per capita, that there were countries that had passed this turning point as well and that they were in significantly higher GDP per capita levels. In the case of upper-middle income countries, the estimated turning points of both models indicate that there were countries that had passed the turning points and while their GDP per capita increased, their carbon dioxide emissions decreased. The estimated turning point of the dynamic lower-middle & low income countries model indicates that there were countries in the period 2000–2018 that had passed this turning point and their carbon dioxide emissions increased, as their GDP per capita increased, as their GDP per capita increased.

Electricity production from fossil fuels is found to be a significant driver of CO_2 emissions in each one of the studied income levels, both in static and in dynamic models, confirming once again the negative environmental results that come with the use of fossil fuels. In addition, an inverse relationship exists between electricity production from renewable sources and carbon dioxide emissions, confirming thus the fact that higher percentages of electricity production covered from renewables can have a positive impact on the environment, reducing CO_2 emissions and, therefore, combating climate change.

Population density is linked with an inverse relationship with carbon dioxide emissions in the upper-middle income countries model, meaning that an increase in population density would lead to a decrease in CO_2 emissions. These results are also confirmed by various studies in the literature [81,82]. In contrast, the dynamic models of high income countries and lower-middle and low income countries suggest that population density is a small driver of CO_2 emissions.

This study also highlights the existence of a bidirectional Granger causality between GDP per capita and CO_2 emissions, while GDP per capita Granger causes per capita electricity production from fossil fuels in all income levels. This confirms the fact that the use of fossil fuels for electricity can indeed lead to economic growth while, at the same

time, higher economic growth leads to a more intense use of fossil fuels in high income countries and lower-middle and low income levels. In addition, a unidirectional causality exists from per capita electricity production from fossil fuels to CO₂ emissions per capita in high income levels and lower-middle and low income levels, meaning that the use of fossil fuels leads to environmental degradation, while an increase in economic growth leads to an increase in air pollution. Per capita electricity production from renewable sources is found to Granger cause GDP per capita and, therefore, boost economic growth, only in high income countries, while in upper-middle and lower-middle & low income levels, this causality is not confirmed. This means that, in the period 2000–2018, the use of fossil fuels for electricity production in upper-middle and lower-middle and low income countries was necessary, in order to boost their economic growth.

The adjustment coefficients that were estimated in the GMM models indicate that 22% of the discrepancy between actual and desired levels is eliminated in a year in high income countries, 33% in lower-middle and low income countries and 63% in upper-middle income countries. It is obvious that the adjustment coefficients of the quadratic and the cubic model differ significantly. These results indicate that in low income levels, the adjustment of efficiency is relatively slow while, as income grows, the adjustment becomes faster. In higher income levels, the adjustment becomes slower again.

7. Conclusions and Policy Implications

The linkages between energy consumption, carbon dioxide emissions and economic growth have been extensively studied in the literature, as well as the causality existing among them. Especially in the case of environmental degradation and economic development, a variety of studies have been focusing on the Environmental Kuznets Curve hypothesis, which assumes that these two factors are linked with an inverse U-shaped relationship, while an N-shaped relationship is assumed to exist for high income countries.

This study aims to contribute to the existing literature, by examining the causal relationships that exist among carbon dioxide emissions, economic growth and electricity production from fossil fuels, as well as from renewable sources, for 119 world countries, classified based on their income levels, and for the years 2000–2018, while taking into consideration population density as well.

The results confirm the EKC hypothesis and the existence of an inverted U-shape curve in the dynamic model for high income countries and in both static and dynamic models for upper-middle income countries. The static model for high income countries confirms the N-shape curve, that is also confirmed in the literature, while the EKC hypothesis is not confirmed for lower-middle and low income countries. These results indicate that, lower-middle & low income countries do not have the resources required to invest in measures and policies related to environmental protection, since they have to focus on other issues regarding their development and growth. In contrast, upper-middle income countries, after reaching a certain level of growth, can promote measures and invest in environmental protection. The same is assumed for high income countries, according to the dynamic model; the static model for high income countries suggests that after a higher level of income, environmental degradation starts to increase again, indicating that all strategies and measures that were undertaken, were not sufficient for high growth levels.

These results can capture the situation existing in the world for the years 2000–2018, but the world has now entered a phase of energy transition, that includes changes in the electricity sector, where the use of renewables is more and more promoted [83]. This energy transition focuses on the use of new energy systems that are efficient and less harmful, but also has to take into consideration all the costs and risks related to the economy and the society that might result from such a transition and address them, so that this procedure will be sustainable [84].

The 13th Sustainable Development Goal, set by the United Nations in 2015, focuses on combating climate change, by promoting strategies and measures related to climate and by fostering resilience and adaptability. At the same time, the 8th SDG focuses on sustainable

economic growth and on economic growth's disengagement from environmental degradation [85]; a relationship that was confirmed once again in this study. Even higher levels of GDP per capita are found to lead to higher levels of environmental degradation, but fossil fuels are considered to be essential, in order to cover current demand in electricity production. These results indicate that actions that minimize the exploitation of natural resources as well as the generation of pollutants and waste as GDP per capita grows and electricity demand is satisfied are necessary, in order to achieve the goals of sustainability.

In addition, the 7th SDG aims to reinsure that everyone in the world has access to reliable and sustainable energy sources, focusing on the reliance on clean fuels and on a higher share of renewables in world's final energy consumption [85]. The present study confirms once more the effects of fossil fuels on environmental degradation and the role of renewables on the improvement of environmental quality. At the same time, the study confirms the role of fossil fuels in boosting economic efficiency. These results highlight the urgent need for actions that promote energy transition and the targets of the 7th SDG, while taking into consideration all the necessary parameters, so that efficiency and growth are maintained.

In conclusion, the results of the present study, that highlight the relationships that existed among electricity production, economic growth and environmental degradation from the beginning of the 21st century, can be taken into consideration, along with the knowledge of new technologies, in order to fully understand those linkages in different income levels and undertake targeted actions that successfully promote energy transition, as well as the goals of sustainability. Different strategies should be implemented in countries of higher incomes, which have already achieved substantial socio-economic growth and have the necessary resources to invest in environmental protection and energy transition, while different measures should be implemented in lower-middle & low income countries, which have to focus mainly on their socio-economic development. Data shows that environmental degradation is caused primarily from higher incomes and the static model confirms that even higher income levels increase carbon dioxide emissions. Therefore, high income countries should focus on decreasing CO₂ emissions and on investing in environmental policies, while they should assist countries of lower incomes in their path of sustainable development, as should do countries of upper-middle income. In addition, and even though the EKC hypothesis is not confirmed for lower incomes and a positive relationship is found between economic growth and environmental degradation, it is suggested that lower-middle and low income countries should prioritize their socio-economic development, but without neglecting environmental protection, as the principles of sustainable development suggest.

Further analysis for specific countries is suggested, in order to identify with precision the linkages that exist between economic growth and carbon dioxide emissions in every place in the world separately, as well as more factors that have an impact on environmental degradation, while identifying the optimal shares of renewables and fossil fuels in electricity production. Such studies will be significantly important, in order to successfully promote energy transition with low socioeconomic costs and global sustainability.

Author Contributions: Both authors G.E.H. and E.-C.G. contributed equally to each section of this paper. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Publicly available datasets were analyzed in this study. This data can be found here: [63–67].

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Höök, M.; Tang, X. Depletion of fossil fuels and anthropogenic climate change—A review. *Energy Policy* **2013**, *52*, 797–809. [CrossRef]
- Chiari, L.; Zecca, A. Constraints of fossil fuels depletion on global warming projections. *Energy Policy* 2011, 39, 5026–5034. [CrossRef]
- 3. EPA. Greenhouse Gas. Emissions. United States Environmental Protection Agency. Available online: https://www.epa.gov/ ghgemissions/overview-greenhouse-gases (accessed on 23 December 2020).
- 4. Masnadi, M.S.; Grace, J.R.; Bi, X.T.; Lim, C.J.; Ellis, N. From fossil fuels towards renewables: Inhibitory and catalytic effects on carbon thermochemical conversion during co-gasification of biomass with fossil fuels. *Appl. Energy* **2015**, *140*, 196–209. [CrossRef]
- 5. EIA. International Energy Outlook 2019; US Energy Information Administration: Washington, DC, USA, 2019.
- World Bank. CO₂ Emissions (kt). The World Bank Data. Available online: https://data.worldbank.org/indicator/EN.ATM.CO2 E.KT (accessed on 23 December 2020).
- 7. World Bank. Energy Use (kg of Oil Equivalent per Capita). The World Bank Data. Available online: https://data.worldbank.org/ indicator/EG.USE.PCAP.KG.OE (accessed on 23 December 2020).
- 8. Levitan, O.; Dinamarca, J.; Hochman, G.; Falkowski, P.G. Diatoms: A fossil fuel of the future. *Trends Biotechnol.* **2014**, *32*, 117–124. [CrossRef]
- 9. Lotfalipour, M.R.; Falahi, M.A.; Ashena, M. Economic growth, CO₂ emissions, and fossil fuels consumption in Iran. *Energy* **2010**, *35*, 5115–5120. [CrossRef]
- 10. Adams, S.; Klobodu, E.K.M.; Apio, A. Renewable and non-renewable energy, regime type and economic growth. *Renew. Energy* **2018**, 125, 755–767. [CrossRef]
- 11. Bölük, G.; Mert, M. Fossil & renewable energy consumption, GHGs (greenhouse gases) and economic growth: Evidence from a panel of EU (European Union) countries. *Energy* **2014**, *74*, 439–446.
- 12. World Bank. GDP (Current US\$). The World Bank Data. Available online: https://data.worldbank.org/indicator/NY.GDP. MKTP.CD (accessed on 23 December 2020).
- 13. World Bank. Population, Total. The World Bank Data. Available online: https://data.worldbank.org/indicator/SP.POP.TOTL (accessed on 23 December 2020).
- 14. World Bank. GDP Per Capita (Current US\$). The World Bank Data. Available online: https://data.worldbank.org/indicator/NY. GDP.PCAP.CD (accessed on 11 September 2020).
- 15. IEA. Total Final Consumption (TFC) by Sector, World 1990–2017. Data and Statistics. Available online: https://www.iea.org (accessed on 23 December 2020).
- 16. IEA. Renewable Share in Final Energy Consumption (SDG 7.2), World 1990–2017. Data and Statistics. Available online: https://www.iea.org (accessed on 23 December 2020).
- 17. IEA. Total CO₂ Emissions, World 1990–2017. Data and Statistics. Available online: https://www.iea.org (accessed on 23 December 2020).
- IEA. CO₂ Emissions per Capita, World 1990–2017. Data and Statistics. Available online: https://www.iea.org (accessed on 23 December 2020).
- 19. IEA. CO₂ Emissions by Energy Source, World 1990–2017. Data and Statistics. Available online: https://www.iea.org (accessed on 23 December 2020).
- 20. World Bank. Forest Area (sq. km). The World Bank Data. Available online: https://data.worldbank.org/indicator/AG.LND. FRST.K2 (accessed on 23 December 2020).
- 21. Dinda, S. Environmental Kuznets Curve Hypothesis: A Survey. Ecol. Econ. 2004, 49, 431-455. [CrossRef]
- 22. Adamu, T.M.; Haq, I.U.; Shafiq, M. Analyzing the Impact of Energy, Export Variety, and FDI on Environmental Degradation in the Context of Environmental Kuznets Curve Hypothesis: A Case Study of India. *Energies* **2019**, *12*, 1076. [CrossRef]
- 23. Beşer, M.K.; Beşer, B.H. The Relationship between Energy Consumption, CO₂ Emissions and GDP per Capita: A Revisit of the Evidence from Turkey. *Alphanumer. J.* **2017**, *5*, 353–368. [CrossRef]
- Bongers, A. The Environmental Kuznets Curve and the Energy Mix: A Structural Estimation. *Energies* 2020, *13*, 2641. [CrossRef]
 Shabbaz, M.: Khrajef, N.: Uddin, G.S.: Ozturk, J. Environmental Kuznets curve in an open economy: A bounds testing and
- 25. Shahbaz, M.; Khraief, N.; Uddin, G.S.; Ozturk, I. Environmental Kuznets curve in an open economy: A bounds testing and causality analysis for Tunisia. *Renew. Sustain. Energy Rev.* **2014**, *34*, 325–336. [CrossRef]
- 26. Apergis, N.; Ozturk, I. Testing Environmental Kuznets Curve hypothesis in Asian countries. Ecol. Indic. 2015, 52, 16–22. [CrossRef]
- 27. Jebli, M.B.; Youssef, S.B.; Ozturk, I. Testing environmental Kuznets curve hypothesis: The role of renewable and non-renewable energy consumption and trade in OECD countries. *Ecol. Indic.* **2016**, *60*, 824–831. [CrossRef]
- 28. Li, T.; Wang, Y.; Zhao, D. Environmental Kuznets Curve in China: New evidence from dynamic panel analysis. *Energy Policy* **2016**, *91*, 138–147. [CrossRef]
- 29. Kang, Y.-Q.; Zhao, T.; Yang, Y.-Y. Environmental Kuznets curve for CO 2 emissions in China: A spatial panel data approach. *Ecol. Indic.* **2016**, *63*, 231–239. [CrossRef]
- 30. Dong, K.; Sun, R.; Hochman, G. Do natural gas and renewable energy consumption lead to less CO₂ emission? Empirical evidence from a panel of BRICS countries. *Energy* **2017**, *141*, 1466–1478. [CrossRef]
- 31. Al-Mulali, U.; Saboori, B.; Ozturk, I. Investigating the environmental Kuznets curve hypothesis in Vietnam. *Energy Policy* **2015**, 76, 123–131. [CrossRef]

- 32. Ozturk, I.; Al-Mulali, U. Investigating the validity of the environmental Kuznets curve hypothesis in Cambodia. *Ecol. Indic.* 2015, 57, 324–330. [CrossRef]
- 33. Abdallh, A.A.; Abugamos, H. A semi-parametric panel data analysis on the urbanization—Carbon emissions nexus for the MENA countries. *Renew. Sustain. Energy Rev.* 2017, *78*, 1350–1356. [CrossRef]
- 34. Özokcu, S.; Özdemir, Ö. Economic growth, energy, and environmental Kuznets curve. *Renew. Sustain. Energy Rev.* 2017, 72, 639–647. [CrossRef]
- 35. Mehrara, M. Energy consumption and economic growth: The case of oil exporting countries. *Energy Policy* **2007**, *35*, 2939–2945. [CrossRef]
- 36. Narayan, P.K.; Smyth, R. Energy consumption and real GDP in G7 countries: New evidence from panel cointegration with structural breaks. *Energy Econ.* 2008, *30*, 2331–2341. [CrossRef]
- 37. Apergis, N.; Payne, J.E. Renewable energy consumption and economic growth: Evidence from a panel of OECD countries. *Energy Policy* **2010**, *38*, 656–660. [CrossRef]
- 38. Ozturk, I.; Aslan, A.; Kalyoncu, H. Energy consumption and economic growth relationship: Evidence from panel data for low and middle income countries. *Energy Policy* **2010**, *38*, 4422–4428. [CrossRef]
- Apergis, N.; Payne, J.E. The renewable energy consumption—Growth nexus in Central America. *Appl. Energy* 2011, 88, 343–347.
 [CrossRef]
- Wang, S.; Zhou, D.; Zhou, P.; Wang, Q. CO₂ emissions, energy consumption and economic growth in China: A panel data analysis. *Energy Policy* 2011, 39, 4870–4875. [CrossRef]
- 41. Lu, W.-C. Renewable energy, carbon emissions, and economic growth in 24 Asian countries: Evidence from panel cointegration analysis. *Environ. Sci. Pollut. Res.* 2017, 24, 26006–26015. [CrossRef]
- 42. Lin, B.; Moubarak, M. Renewable energy consumption—Economic growth nexus for China. *Renew. Sustain. Energy Rev.* 2014, 40, 111–117. [CrossRef]
- 43. Pao, H.-T.; Tsai, C.-M. Modeling and forecasting the CO₂ emissions, energy consumption, and economic growth in Brazil. *Energy* **2011**, *36*, 2450–2458. [CrossRef]
- 44. Pao, H.-T.; Yu, H.-C.; Yang, Y.-H. Modeling the CO₂ emissions, energy use, and economic growth in Russia. *Energy* **2011**, *36*, 5094–5100. [CrossRef]
- 45. Soytas, U.; Sari, R.; Ewing, B.T. Energy consumption, income, and carbon emissions in the United States. *Ecol. Econ.* **2007**, *62*, 482–489. [CrossRef]
- 46. Acaravci, A.; Ozturk, I. On the relationship between energy consumption, CO₂ emissions and economic growth in Europe. *Energy* **2010**, *35*, 5412–5420. [CrossRef]
- 47. Pirlogea, C.; Cicea, C. Econometric perspective of the energy consumption and economic growth relation in European Union. *Renew. Sustain. Energy Rev.* **2012**, *16*, 5718–5726. [CrossRef]
- 48. Kasman, A.; Duman, Y.S. CO₂ emissions, economic growth, energy consumption, trade and urbanization in new EU member and candidate countries: A panel data analysis. *Econ. Model.* **2015**, *44*, 97–103. [CrossRef]
- 49. Alper, A.; Oguz, O. The role of renewable energy consumption in economic growth: Evidence from asymmetric causality. *Renew. Sustain. Energy Rev.* **2016**, *60*, 953–959. [CrossRef]
- 50. Pablo-Romero, M.D.P.; Sánchez-Braza, A. Residential energy environmental Kuznets curve in the EU–28. *Energy* 2017, 125, 44–54. [CrossRef]
- 51. Hundie, S.K.; Daksa, M.D. Does energy-environmental Kuznets curve hold for Ethiopia? The relationship between energy intensity and economic growth. *J. Econ. Struct.* **2019**, *8*, 21. [CrossRef]
- 52. Aruga, K. Investigating the Energy-Environmental Kuznets Curve Hypothesis for the Asia-Pacific Region. *Sustain. J. Rec.* 2019, 11, 2395. [CrossRef]
- 53. Luzzati, T.; Orsini, M. Investigating the energy-environmental Kuznets curve. *Energy* 2009, 34, 291–300. [CrossRef]
- 54. Abdou, D.M.S.; Atya, E.M. Investigating the energy-environmental Kuznets curve: Evidence from Egypt. *Int. J. Green Econ.* **2013**, 7, 103. [CrossRef]
- 55. Pablo-Romero, M.D.P.; De Jesús, J. Economic growth and energy consumption: The energy-environmental Kuznets curve for Latin America and the Caribbean. *Renew. Sustain. Energy Rev.* **2016**, *60*, 1343–1350. [CrossRef]
- 56. Asumadu-Sarkodie, S.; Owusu, P.A. The relationship between carbon dioxide emissions, electricity production and consumption in Ghana. *Energy Sources Econ. Plan. Policy* **2017**, *24*, 1–12. [CrossRef]
- 57. Mohiuddin, O.; Asumadu-Sarkodie, S.; Obaidullah, M. The relationship between carbon dioxide emissions, energy consumption, and GDP: A recent evidence from Pakistan. *Cogent Eng.* **2016**, *3*, 1210491. [CrossRef]
- Bento, J.P.C.; Moutinho, V. CO₂ emissions, non-renewable and renewable electricity production, economic growth, and international trade in Italy. *Renew. Sustain. Energy Rev.* 2016, 55, 142–155. [CrossRef]
- 59. Al-Mulali, U.; Weng-Wai, C.; Sheau-Ting, L.; Mohammed, A.H. Investigating the environmental Kuznets curve (EKC) hypothesis by utilizing the ecological footprint as an indicator of environmental degradation. *Ecol. Indic.* **2015**, *48*, 315–323. [CrossRef]
- 60. Ulucak, R.; Bilgili, F. A reinvestigation of EKC model by ecological footprint measurement for high, middle and low income countries. *J. Clean. Prod.* **2018**, *188*, 144–157. [CrossRef]

- 61. Serajuddin, U.; Hamadeh, N. New World Bank Country Classifications by Income Level: 2020–2021. World Bank Blogs. Available online: https://blogs.worldbank.org/opendata/new-world-bank-country-classifications-income-level-2020-2021# (accessed on 11 February 2021).
- 62. IEA. Total Production. Electricity. Data and Statistics. Available online: https://www.iea.org (accessed on 11 February 2021).
- 63. World Bank. Electricity Production from Oil, Gas and Coal Sources (% of Total). The World Bank Data. Available online: https://data.worldbank.org/indicator/EG.ELC.FOSL.ZS (accessed on 11 February 2021).
- 64. World Bank. Electricity Production from Renewable Sources, Excluding Hydroelectric (% of Total). The World Bank Data. Available online: https://data.worldbank.org/indicator/EG.ELC.RNWX.ZS (accessed on 11 February 2021).
- 65. World Bank. Electricity Production from Hydroelectric Sources (% of Total). The World Bank Data. Available online: https://data.worldbank.org/indicator/EG.ELC.HYRO.ZS (accessed on 11 February 2021).
- 66. World Bank. CO₂ Emissions (Metric Tons per Capita). The World Bank Data. Available online: https://data.worldbank.org/ indicator/EN.ATM.CO2E.PC (accessed on 11 February 2021).
- 67. World Bank. Population Density (People per sq. km of Land Area). The World Bank Data. Available online: https://data. worldbank.org/indicator/EN.POP.DNST (accessed on 11 February 2021).
- 68. Halkos, G.E. Environmental Kuznets Curve for sulfur: Evidence using GMM estimation and random coefficient panel data models. *Environ. Dev. Econ.* 2003, *8*, 581–601. [CrossRef]
- Halkos, G.; Petrou, K.N. The relationship between MSW and education: WKC evidence from 25 OECD countries. *Waste Manag.* 2020, 114, 240–252. [CrossRef]
- 70. Pedroni, P. Fully Modified OLS for Heterogeneous Cointegrated Panels. In *Nonstationary Panels, Panel Cointegration, and Dynamic Panels (Advances in Econometrics)*; Elsevier: Amsterdam, The Netherlands, 2004; pp. 93–130.
- 71. Revathy, A.; Paramasivam, P. Study on Panel Co-integration, Regression and Causality Analysis in Papaya Markets of India. *Int. J. Curr. Microbiol. Appl. Sci.* **2018**, *7*, 40–49. [CrossRef]
- 72. Hall, A.R. Generalized Method of Moments; Oxford University Press: Oxford, UK, 2005.
- 73. Hall, A.R. Generalized Method of Moments. In *Handbook of Research Methods and Applications in Empirical Macroeconomics;* Edward Elgar Publishing: Cheltenham, UK, 2013.
- 74. Yin, G. Bayesian generalized method of moments. Bayesian Anal. 2009, 4, 191–207. [CrossRef]
- 75. Ahmad, M.; Khan, R.E.A. Does Demographic Transition with Human Capital Dynamics Matter for Economic Growth? A Dynamic Panel Data Approach to GMM. *Soc. Indic. Res.* **2018**, *142*, 753–772. [CrossRef]
- 76. Ullah, S.; Akhtar, P.; Zaefarian, G. Dealing with endogeneity bias: The generalized method of moments (GMM) for panel data. *Ind. Mark. Manag.* 2018, *71*, 69–78. [CrossRef]
- 77. Liu, Y.; Hao, Y. The dynamic links between CO₂ emissions, energy consumption and economic development in the countries along "the Belt and Road". *Sci. Total Environ.* **2018**, *645*, *674–683*. [CrossRef] [PubMed]
- 78. Harris, R.; Sollis, R. Applied Time Series Modelling and Forecasting; Wiley: Hoboken, NJ, USA, 2003.
- 79. Arellano, M.M.; Bond, S. *Dynamic Panel Data Estimation Using DPD—A Guide for Users*; Working Paper Series; Institute for Fiscal Studies: London, UK, 1988.
- 80. Mocking, R.; Steegmans, J. *Capital Structure Determinants and Adjustment Speed: An Empirical Analysis of Dutch SMEs*; No. 357. rdf; CPB Netherlands Bureau for Economic Policy Analysis: Hague, The Netherlands, 2017.
- Ribeiro, H.V.; Rybski, D.; Kropp, J.P. Effects of changing population or density on urban carbon dioxide emissions. *Nat. Commun.* 2019, 10, 1–9. [CrossRef]
- Gudipudi, R.; Fluschnik, T.; Ros, A.G.C.; Walther, C.; Kropp, J.P. City density and CO₂ efficiency. *Energy Policy* 2016, 91, 352–361. [CrossRef]
- 83. Markard, J. The next phase of the energy transition and its implications for research and policy. *Nat. Energy* **2018**, *3*, 628–633. [CrossRef]
- 84. Sareen, S.; Haarstad, H. Bridging socio-technical and justice aspects of sustainable energy transitions. *Appl. Energy* **2018**, 228, 624–632. [CrossRef]
- 85. Sustainable Development Goals—SDGs—The United Nations. The 17 Goals. Available online: https://sdgs.un.org/goals (accessed on 20 January 2020).





Article The Nexus between Economic Growth, Energy Consumption, Agricultural Output, and CO₂ in Africa: Evidence from Frequency Domain Estimates

Adedoyin Isola Lawal

Department of Economics, Bowen University, Iwo 232102, Nigeria; adedoyin.lawal@bowen.edu.ng or l.adedoyin@yahoo.com; Tel.: +234-8035233567

Abstract: This study examined the nexus between economic growth, energy consumption, and the environment with the moderating role of agricultural value addition and forest in Africa based on data sourced from 1980 to 2019. We employed both the time domain and frequency domain panel Granger causality estimation techniques to compare results across the different horizons. Extant literature suggests the inability of time domain estimation techniques to account for causality at different frequencies. The study also accounts for the nexus among our variables both at the single-country and multi-country levels. The results at the single-country level are at best mixed. The results of the panel Granger causality at the frequencies domain suggest that a bi-directional relationship exists between energy consumption and economic growth, and that energy consumption Granger causes carbon emissions in Africa. The results align with the feedback hypothesis on the one hand but contradict the conservation hypothesis on the other hand. The study has some policy implications.

Keywords: energy consumption; carbon emissions; agricultural output; economic growth; Africa

1. Introduction

In attaining sustainable development, energy, economics, and the environment play significant roles [1–5]. For instance, energy is crucial to the human economic and social development of any nation. It is estimated that global energy consumption will increase by about 56% from its current state in 2010 by the year 2040, as global aggregate demand is expected to double, given the expected increase in population [6–12]. However, the projected increase in total energy consumption is expected to be accompanied by an increase in carbon dioxide (CO₂) emissions, which is a core factor in total greenhouse emission (GHG). The energy sector is responsible for about 61.4% of the total global GHG [13–16]. Ref. [7] noted that the contributions of agriculture sector to the GHG are estimated to be between 14–30%, though evidence abounds to show that the agricultural sector possesses the ability to reduce GHG by 80–88%. It is opined that forests possess the capacity to accumulate atmospheric carbon after converting CO₂ into carbon and oxygen, and that about 430 tons of carbon per hectare is absorbed in the wet forest, hence, halting the effects of carbon emissions [17–22].

In the same vein, environmental degradation plays a crucial role in the continuous occurrence of natural disasters with unprecedented impacts on the economy. Disasters related to oil spillage, water pollution, solid waste management, deforestation, soil erosion, salinity and water, logging, and desertification, among others, affects the socio-economic wellbeing of a nation and increases climate change. Environmental degradation worsens with the exploitation of fossil fuels [23–27]. In order to mitigate this without losing a significant part of the energy output, economies over the years have opted for renewable energy sources [28–30]. Renewable energy offers clean and safer energy and can be derived from solar, tidal, wind, geothermal, hydro and biofuel power. Besides its alternative energy potential, it is useful in supporting employment, output, income, and job creation. Extant

Citation: Lawal, A.I. The Nexus between Economic Growth, Energy Consumption, Agricultural Output, and CO_2 in Africa: Evidence from Frequency Domain Estimates. *Energies* 2023, *16*, 1239. https:// doi.org/10.3390/en16031239

Academic Editor: George Halkos

Received: 23 December 2022 Revised: 10 January 2023 Accepted: 16 January 2023 Published: 23 January 2023



Copyright: © 2023 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).
literature shows that the increase in economic growth and agricultural outputs have a positive impact on renewable energy [31,32]. Furthermore, given a global temperature increase of between 2–2.4 °C, renewable energy can help reduce carbon emissions by 50% by the year 2050. Besides its positive impact on the environment, renewable energy can reduce overdependency on foreign energy, given the fact that it is sourced domestically [33,34].

The United Nations Sustainable Development Goals (SDGs) emphasized the need to eradicate hunger (SDG 2), achieve clean energy utilization (SDG 7), achieve sustainable economic growth (SDG 8), adopt sustainable production and consumption (SDG 12), mitigate climate change through a sustainable clean environment (SDG 13), and adopt a global partnership model to achieve these goals (SDG 17). The nexus between these laudable metrics for sustainable development is key to exploring the linear and circular economic growth in any economy, be it regional or single country (Sarkodie 2020). Sub-Saharan Africa needs more energy than most continents of the world, given its everincreasing, teaming population and quest for sustainable growth [35]. Even though the continent is endowed with an abundance of non-renewable energy like petroleum and other fossil fuels, the negative impacts of fossil fuel on the environment, such as the increase in GHG and other pollutants, calls for concerns. Although the contribution of Africa to global warming at present may be negligible compared with other continents, it is obvious that the continent will be disproportionately affected by its impact if nothing is done. To mitigate the impact of GHG on the continent, the African Development Bank (AfDB) adopted a ten-year green growth strategy (2013–2022) with an emphasis on developing the renewable energy potential capable of promoting resource efficiency and sustainable development.

Several theoretical models exist that explain the links between energy, the economy and the environment. For instance, Environmental Kuznets Curve (EKC) models suggest that at the initial stage of development, a direct positive relationship exists between economic growth proxy by real gross domestic product (RGDP) and environmental pollution, but the relationship becomes indirect after a threshold level of income is achieved. The pollution haven model suggests that in developing economies characterized by weak pollution protection laws, trade and investment liberalization laws often induce environmental degradation as pollution-intensive firms will find it easier to produce in such economies than in developed economies with stringent environmental protection policies. The causality model employs unit roots, cointegration and causality measures to examine the nexus between energy consumption and economic growth. This model offers four possibilities, firstly (i) the growth-led hypothesis, which suggests the existence of unidirectional causality from economic growth to energy consumption. This suggests that conservation policies will have no impact on economic growth. This is common in energy-sufficient economies. Secondly, (ii) the energy-led hypothesis, which suggests that energy consumption stimulates growth, therefore, energy conservation policies will impact negatively on economic growth, thus, energy expansion policies are required. This is common in economies that are energy-dependent like most developing economies. Third is (iii) the feedback model, which suggests the existence of a bi-directional causality between energy consumption and economic growth. The model suggest that both constructs are jointly determined and affected simultaneously. Lastly is (iv) the neutrality model, stating that no causality exists between energy consumption and economic growth. It also suggests that environmentally-friendly policies can be achieved without obstructing economic growth.

Extant literature has attempted to examine the link between the environment, energy, and the economy with mixed results. For instance, Refs. [36–38] were of the view that causality runs from economic growth to energy consumption while Refs. [39–43] opined that causality is from energy consumption to economic growth. Furthermore, Refs. [41,42,44,45] noted that causality runs from economic growth to CO_2 emissions. The bulk of these studies focused on developed economies with little attention on African economies. Africa is faced with plurality of issues, key among them being the need to stimulate growth, ensure a sustainable environment and reduce energy poverty. The World Bank global monitoring report (2008) highlights the need for the continent to be on a sustainable development path

that embraces clean energy, a sustainable environment, and accelerated growth, noting the continuous increase in CO₂ emission and fall in per capita water resources. Given the low state of renewable energy development and the potential environmental hazards emanating from existing conventional fossil fuel amidst the desire to stimulate growth, it is imperative to examine the nature of the relationship between energy consumption (renewable and non-renewable), economic growth, and CO₂ emissions with the moderating impact of agriculture and agro-allied resources in Africa. Our study presents a short, intermediate, and long run analysis for 34 African economies. Unlike existing studies that employed time domain estimates like the traditional Granger causality estimates, VAR and other time domain estimates [16,30,46–49], the current study employed both the single and multi-country frequency domain Granger casualty estimates based on datasets sourced from 1980–2019. Even though frequency domain techniques offer better estimation models, because they allow for examination of the direction and level (strength) of the nexus at heterogeneous scales for frequency [2,3,9,50–52], they are yet to be explored especially in studies in Africa.

Our choice of Africa was induced by the fact that Africa is endowed with an abundance of potential energy resources (both renewable and non-renewable). It is estimated that in Africa, the potential energy generation capacity is up to 1.2 terawatts, excluding solar, and more than 10 terawatts including solar, with a high potential of achieving more than a 25% increase in clean energy by 2040 [8,53,54]. The continent is the world's youngest and fastest urbanizing continent, but it is the least energy-supplied, with annual consumption being 518 kwh in sub-Saharan Africa, equivalent to what a single member country of the OECD will use. Economic indices show that recently, African economies largely outperformed the global average (IMF, WB 2019) with the continent's overall GDP increasing 3.8% against the global average of 3.4%. Data availability large influences the choice of sample economies.

Against this background, this research attempts to know whether various energy policies in the continent offer the ability to end Africa's energy poverty, stimulate growth, and promote environmental sustainability. We intend to answer the following questions: (i) What drives the African economic, energy and environmental nexus—an environmental Kuznets curve, causality, or the pollution haven model? (ii) What is the nature of the causality between energy consumption (renewable and non-renewable) and economic growth, carbon emissions, and agricultural output in Africa? (iii) If causality is established, to what extent will the increase in energy consumption support economic growth, agricultural output, and reduce carbon emissions in Africa economies? Answering our questions will provide insights into at least five SDGs: SDG 2—zero hunger; SDG 7—achieve clean energy utilization; SDG 8—achieve sustainable economic growth; SDG 12—adopt sustainable production and consumption; and SDG 13—mitigate climate change through a sustainable clean environment.

This study will make essentially four contributions to the literature. First, in terms of methodology, we will provide a frequency-based panel Granger causality analysis that offers short, intermediate and long run casual estimates of the nexus between economic growth energy and the environment with a focus on African economies. Our method provides individual estimates for each of the economies studied, unlike the conventional methods that offer lump-sum causality estimates. Second, the study will calibrate the moderating impact of agriculture and agro-allied resources to the discourse on energy, economics and the environment in Africa. Africa is largely agrarian and to the best of the author's knowledge, no literature of the African extraction has considered the moderating role of agriculture in absorbing carbon emissions in the economic-energy-environmental nexus. Thirdly, in term of coverage and scope, our study will cover more African economies than most of the existing studies and use more recent data when compared with others. Fourthly, our study will also calibrate both the energy conservation and expansion policies into the energy, environment, and economic growth discourse. Our finding offers some policy implications for policy makers at both the national and regional levels, as well as for

international organizations and researchers on the link between energy, economic and the environment. The rest of the study is as follows: Section 2 presents the literature review; Section 3 offers the data and methodology; Section 4 deals with the presentation of results, while Section 5 concludes the study and offers some policy implications.

2. Literature Review

A critical assessment of extent literature clearly suggests that frequency domain estimates are yet to be sufficiently employed in examining the nature of the relationship between energy, economics and the environment with the moderating role of agriculture, especially based on evidence from Africa, despite its attractiveness and potential strength in providing measures in shaping the African policy space. Africa economies are in dire need of energy, with the need to advance economic growth at the front of the policy framework amidst the global quest to reduce CO_2 . It is pertinent, especially when faced with few publications on the subject matter, to examine the moderating role of agriculture in mitigating CO_2 emissions, stimulating economic growth and ending energy poverty. Such effort would not only offer a valuable platform to examine the nature of cointegration and the direction of causation, among the variables (energy, economics, environment and agriculture), it will equally initiate and stimulate further research and model specifications.

Table 1 presents the result of extent literature on the nexus between energy, economic growth, agriculture, and carbon emissions for a number of economies across the globe. The results as presented can be categorized into four main streams-methodological, results (findings), hypothesis or policy trust and variables employed. In methodological strands, a number of studies employed cointegration and/or Granger causality methods to investigate the link between energy, economic growth, and the environment [6,16,19,20,22,23,28,49,55–63] with mixed results. For instance, while [19] noted that a bi-directional relationship exists between non-renewable energy and climate change and that climate change Granger causes renewable energy for 16 African countries, ref. [16] observed that causation is from RGDP to renewable energy in the long run for China, with a negative impact on renewable energy in the short run. Similarly, ref. [13] documented the existence of a bi-directional relationship between renewable energy and non-renewable energy for India and South Africa, suggesting validity of the feedback hypothesis. The study further noted that causality runs from non-renewable energy to economic growth for Brazil and USA, an indication that the growth hypothesis is valid in these economies but noted no causal relationship exists between non-renewable energy and economic growth for Russia, India and South Africa, implying the validity of the neutrality hypothesis. For South Africa, ref. [6] noted that growth hypothesis is valid as the direction of causation is from energy use to RGDP. Ref. [19] offers multifaceted results, for instance, the authors documented that bi-directional relationships exist between fossil fuel and RGDP, between fossil fuel and CO_2 , and between CO_2 and RGDP for the oil-exporting economies. These results support the feedback hypothesis from oil prices to each of RGDP and CO₂ for the oilconsuming economies, suggesting the validity of the growth hypothesis. Ref. [57] results are at variance with those of [22–24,28,29,58] who noted causality is from RGDP to CO₂, and that no causality exist between energy consumption and economic growth, thereby supporting the validity of the neutrality hypothesis in the studied economies.

1 [5] 1980-2014 Reversible energy. Reversible energy. RCmAARD-MCG R And Schweis Is And Schweis Non Reversible energy. Energy and Schweis Non Reversible energy. Climate dataget 3 1990-2015 RCDP, solid cooking tues Constrained ds strained ds straine ds strained ds straine	S/n	Authors	Period of Study	Variable	Methods	Countries	$Results \to \leftarrow$
2[64]1980-2013Economic growth, CC3 methods of statistic countifiesfor set of statistic countifies(set A hitcan countifies)To stative relationship exists between ensistion, inflation, populationTo statistic analysis, Granger 		[55]	1980–2014	Renewable energy, non-renewable, economic growth, climate change	Group-ARDL-PMG, ARDL-MG, Granger causality	16 African countries	Non-Renewable ↔ Climate change Climate change → Renewable energy Feedback hypothesis holds.
313190-2013CHC, fosal energy and econonic growthA recursive system1 sub-Saharan A frican econoniesFeonomic growth does not conservableFeonomic growth does not	7	[64]	1980–2019	Economic growth; CO ₂ emission, inflation, population	Panel econometric methods of statistical analysis, Granger causality	6 west African countries	Positive relationship exists between the variables
4.[65][196-20]4RCDP, non-renevable energy. CO, policy uncertainty > CMon-renevable energy reduce- Panel unit rot, panel2 sub-SaharaNon-renevable energy reduce- Panel unit rot, panel5.[15]2000-2015RCDP, solid cooking fuelsPanel unit rot, panel46 sub-SaharaNon-renevable energy reduce- Renevable energy reduce- Renevable energy reduce- Renevable energy recommicPanel unit rot, panel46 sub-SaharaNon-renevable energy reduce- Renevable energy recommic6.[16]1997-2017Renvolate energy, economic Granger causalityChuna, WesternA negative causal relationship exists from sol frican countries7.[17]1990-2015RCDP, NEF, RE, CO2System CAM31 transitional economesRCDP + RE (long run), financial development financial development partice/ impact financial development8.[16]1990-2015RCDP, NEF, RE, CO2System CAM31 transitionalRCDP + RE (long run), financial development financial development financial financial openness, RCD9.[67]1990-2018Found inancial openness, financial openness,Pateurun attri reconnessNo9.[67]1971-2014Found indicid formation itSystem CAMInteraction it financial development bein financial development bein10.[14]1971-2014Found indicid formation itNARDL, asymmetric formation itInteraction it formation it10.[14]1971-2014Found indicid gover effectivenses trade gover effectivenses tradeNARDL, asymmetric formation it </td <td>3</td> <td>[13]</td> <td>1990–2013</td> <td>GHG, fossil energy and economic growth</td> <td>A recursive system of three equations</td> <td>41 sub-Saharan African economies</br></td> <td>Fossil energy \rightarrow GHG, Economic growth does not Granger cause CO₂ emissions</td>	3	[13]	1990–2013	GHG, fossil energy and economic growth	A recursive system of three equations	41 sub-Saharan 	Fossil energy \rightarrow GHG, Economic growth does not Granger cause CO ₂ emissions
5[15]200-2015RCDF, solid cooking fuelsParel unit not, panel cimager ausality4, sub-Sahara African countriesA megative causal relationship exist from sol cimager ausality6.[16]1997-2017Renewable energy, economic growth and financial developmentGranger causalityChina, Western ARDL-PMGA megative causal regroupship exists from sol inancial development positively imp- financial development7.[17]1990-2015RCDP, NRE, RE, CO2System GMM31 transitional economiesRGDP> RE, RE> RCDP8[66]1990-2018Natural resources, energy 	र्च	[65]	1996–2014	RGDP, non-renewable energy, CO ₂ , policy uncertainly	One-step-system GMM	32 sub-Sahara African countries	$\begin{array}{l} RGDP \rightarrow CO_2 \\ Non - renewable energy \rightarrow CO_2 \\ Policy uncertainty \rightarrow CO_2 \\ Renewable energy reduce \rightarrow CO_2 \end{array}$
6. $[16]$ $1997-2017$ Reevable energy, ecoromic growth and financial developmentGanger causality (Ein the foug run, KCDP regatively impa- financial development positively impa- factorial and grun, KCDP regatively impa- 	2	[15]	2000–2015	RGDP, solid cooking fuels	Panel unit root, panel cointegration panel Granger causality	46 sub-Sahara African countries	A negative causal relationship exists from solid cooking fuel to RGDP
7[17]190-2015RGDP,NRE, RE, CO2System GMM31 transitional economiesCO2 has unconditional negative effects on RGDP → RE, RE → CO2/N-F8[66]1990-2018Natural resources, energy formation, gross capital formation, gross capital formation, gross capital modeling techniquesPakistanNegative relationship exists between nature RE and NRE → RGD Fin, openness → RGI Fin, openness → RGI 	.9	[16]	1997–2017	Renewable energy, economic growth and financial development	Granger causality ARDL-PMG	China, Western China Eastern China	$\begin{array}{llllllllllllllllllllllllllllllllllll$
8[66]1990–2018Natural resources, energy formation, financial openness, formation, financial openness, RGDPStructural equation modeling techniquesNegative relationship exists between nature RE and NRE → RGD Fin. openness → RGL Gross capital formation \leftrightarrow 9[67]1971–2014Fossil oll RGDPN-ARDL, asymmetric panel causality test19 African countriesMixed results10[14]1971–2017RGDP, agricultural ouput, govt. effectiveness tradeSystem GMM, panel threshold17 AfricanMixed results11[18]1971–2017RGDP, agricultural ouput, govt. effectiveness tradeSystem GMM, panel threshold17 AfricanElectricity \rightarrow RGDP12[18]1980–2015Petroleum, natural gas, CO2, 	4	[17]	1990–2015	RGDP, NRE, RE, CO ₂	System GMM	31 transitional economies	CO ₂ has unconditional negative effects on human devt. RGDP; RGDP \rightarrow RE, RE \rightarrow CO ₂ N – RE \rightarrow CO ₂
9[67]1971-2014Fossil oil RGDPN-ARDL, asymmetric panel causality test19 African countriesMixed results10[14]1971-2017Electricity consumption, govt. effectiveness tradeSystem GMM, advanced dynamic 	8	[66]	1990–2018	Natural resources, energy consumption, gross capital formation, financial openness, RGDP	Structural equation modeling techniques	Pakistan	Negative relationship exists between natural resources and RGDP; RE and NRE \rightarrow RGDP Fin. openness \rightarrow RGDP. Gross capital formation \leftrightarrow RGDP
10 [14] 1971-2017 Electricity consumption, RGDP, agricultural output, govt. effectiveness trade panel threshold System GMM, advanced dynamic panel threshold 17 African economies Electricity -> RGDF 11 [18] 1980-2015 Petroleum, natural gas, CO ₂ , RGDP N-ARDL Oil producing Africa economies RE reduces CO ₂ (Niger RE -> RGDP (Gabor Neutrality hypothesesh 12 [68] 1995-2014 Renewable energy labor, capital, RGDP P-DOLS, F MOLS 15-Western Africa RE does not Granger cause CO ₂ (Angola ar Neutrality hypothesesh	6	[67]	1971–2014	Fossil oil RGDP	N-ARDL, asymmetric panel causality test	19 African countries	Mixed results
11[18]1980-2015Petroleum, natural gas, CO2, RGDPN-ARDLOil producing Africa economiesRE reduces CO2 (Niger RE \rightarrow RGDP (Gabor Neutrality hypothese h Neutrality hypothese h12[68]1995-2014Renewable energy labor, capital, RGDPP-DOLS, F MOLS15-Western Africa countriesRE slows down grown	10	[14]	1971–2017	Electricity consumption, RGDP, agricultural output, govt. effectiveness trade	System GMM, advanced dynamic panel threshold regression model	17 African economies	$\begin{array}{l} \mbox{Electricity} \rightarrow \mbox{RGDP} \\ \mbox{Growth hypothesis} \end{array}$
12 [68] 1995–2014 Renewable energy labor, P-DOLS, F MOLS 15–Western Africa RE slows down growt countries	11	[18]	1980–2015	Petroleum, natural gas, CO ₂ , RGDP	N-ARDL	Oil producing Africa economies	$\begin{array}{l} RE \ reduces \ CO_2 \ (Nigeria) \\ RE \ \rightarrow \ RGDP \ (Gabon) \\ RE \ does \ not \ Granger \ cause \ CO_2 \ (Angola \ and \ Egypt). \ Growth \ and \\ Neutrality \ hypotheses \ hold \end{array}$
	12	[68]	1995–2014	Renewable energy labor, capital, RGDP	P-DOLS, F MOLS	15-Western Africa countries	RE slows down growth

Table 1. Summary of Literature review.

S/n	Authors	Period of Study	Variable	Methods	Countries	$\textbf{Results} \rightarrow \leftarrow$
13	[56]	1996–2015	RE, NRE, R&D, RGDP	Unit root tests, panel Granger causality	BRICS	RE ↔ NRE (India and SA) Feedback hypothesis hold RE does not granger cause NRE (Brazil) NRE → GDP (Brazil and SA) Growth hypothesis NRE-R&D (Russia, India, SA) Neutrality hypothesis hold
14	[9]	1960–2016	Capital, labor, CO ₂ , RGDP, energy consumption	ARDL, Granger causality test	South Africa	Energy use $ ightarrow$ RGDP growth hypothesis holds
15	[19]	1990–2015	Oil price, CO ₂ , RGDP, fossil energy consumption	PMG panel ARDL, bootstrap panel cointegration	22 African countries	$\begin{array}{llllllllllllllllllllllllllllllllllll$
16	[25]	2001–2017	Energy consumption CO ₂ , RGDP	System GMM	68 developed, emerging and MENA countries	$\begin{array}{llllllllllllllllllllllllllllllllllll$
17	[57]	1973–2014	Growth role of kg oil equivalent per capital energy usage, RGDP ecological foot print	ARDL Toda-Yamamoto	South Africa	Ecological footprint → RGDP Kg oil equivalent → eco. footprint Kg oil equivalent → RGDP
18	[69]	1990–2012	CO2-equivalent, RGDP, energy usage, international trade	Environmental input-output model	Angola, Ethiopia, Kenya, Nigeria, south Africa	RE reduces CO ₂ -equivalent
19	[28]	1971–2010	Energy consumption CO ₂ , economic growth	ARDL, Granger causality	12 sub-Sahara Africa	$\begin{array}{l} \mbox{Mixed results} \\ \mbox{RGDP} \rightarrow CO_2 \mbox{ short run for Benin, DRC, Ghana, Nigeria, and Senegal} \\ \mbox{RGDP} \leftrightarrow CO_2, \mbox{ Long run for Congo, Gabon} \\ \mbox{Energy consumption} \rightarrow CO_2 \mbox{ in of long run for Benin, DRC, Nigeria, Senegal, South Africa, and Togo} \end{array}$
20	[58]	1973–2017	Energy consumption, oil prices, trade openness, urbanization and RGDP	ARDL, ECM	African OPEC Countries	No causality between energy consumption and RGDP. Energy consumption does not Granger cause RGDP
21	[29]	1990–2017	RDGP, energy consumption, renewable energy	Neural network analysis	25 African economies	$RGDP \rightarrow CO_2$
22	[9]	1990–2014	Energy intensity RE, CO ₂ , RGDP	ARDL, Toda Yamamoto	Romania	$RE\ \rightarrow\ RDGP,\ Energy\ intensity\ \leftrightarrow\ RGDP$
23	[20]	1975–2017	CO ₂ , RGDP, carbon income, trade openness, energy use	ARDL, Toda-Yamamoto	India	Energy use \rightarrow GDP Energy use \rightarrow CO ₂
24	[59]	1980–2018	RE, CO ₂ , financial devt., trade openness, FDI, urbanization	A panel quantile regression	Global panel of 192 countries	Fin. devt \rightarrow RE, inverse relationship exists between RE and CO ₂
25	[21]	1990-2017	CO ₂ , trade, RGDP, RE, environmental innovation	A battery of panel co-integration methodologies	G7 countries	Long run relationship exists among CO ₂ , trade, RGDP, RE and environmental innovation. Environmental degradation does not cause RGDP, RE reduces CO ₂

Auth	ors	Period of Study	Variable	Methods	Countries	$\mathbf{Results} \to \leftarrow$
[70] 19	16	980-2014	CO ₂ , RGDP, RE, urbanization, NRE	FMOLS and GMM	28 sub-Sahara African Countries	$\begin{array}{llllllllllllllllllllllllllllllllllll$
[22] 19	10	978–2016	CO ₂ , RGDP, RE, urbanization and Agriculture	ARDL	Malaysia	RGDP, Urbanization $\rightarrow CO_2$ RE and agriculture significantly CO_2
[23] 19	16	90–2014	CO ₂ , RGDP, RE, nuclear energy real coal prices	Panel cointegration and Granger causality test	30 developed and emerging economies	LR relationship exists among the variables; NE does not lead to CO2 reduction $RE \to CO_2 \ reduction \\ RE \to RGDP$
[24] 2	Ā	012-2014	Energy usage, CO ₂ , electricity consumption, fossil fuel, biomass	ANOVA and Tukey multiple comparison test	Sri Lanka	$\begin{array}{l} \mbox{Elect} \rightarrow \mbox{CO}_2 \\ \mbox{Fossil} \rightarrow \mbox{CO}_2, \\ \mbox{RGDP does not} \rightarrow \mbox{CO}_2 \end{array}$
[16] 1	-	997–2017	RE, fin. devt and economic growth	ARDL-PMG Granger causality test	China	$\begin{array}{llllllllllllllllllllllllllllllllllll$
[30] 1	-	995–2014	RE, CO2, RGDP	GS2SLS	EU	$EC \leftrightarrow RE$ feedback $ECC \leftrightarrow CO_2$ RE does not $\rightarrow CO_2$
[46] 1	1	990–2015	RE, NRE, RGDP	Local liner dummy variable estimation (LLDVE)	40 OECD and non-OECD countries	Both NRE and RE impact economic growth positively
[31] 1	1	990–2017	RGDP, fin. inclusion, RE, NRE, o trade openness	Augmented mean COgroup, Dumitrescu -Hurlin non-causality test	15 highest emitting countries	Bidirectional causality exists between fin. devt, economic growth, renewable energy utilization and ecological footprint; unidirectional causality runs from non-renewable energy and trade openness to ecological footprint, unidirectional relationship runs from economic growth to RE and trade openness. Feedback hypothesis holds
[32] 1	1	990–2018	RE, RGDP, CO ₂ , NRE, Capital and labor	DOLS, FMOLS and Heterogeneous non-causality model	38 renewable energy consuming countries	LR relationship exist between RE and RGDP; RE, NRE, capital and labor impacts on RGDP
[71] 2	7	005–2016	NRE intensity, urbanization, per capital income	Panel threshold regression	OECD countries	Positive and non-linear relationships exist between renewable energy and economic growth
[72]		(990-2010	GDP, GDPPC, Total renewable energy, share of renewable energy to total energy consumption, gross fixed capital formation, number of employed people in of economy; R&D	Panel quantile regression	OECD economies	The impact of RE on economic growth is at best unused, i.e., positive for lower, and low-middle-quantities, and negative for middle, high middle and higher quantities

S/n	Authors	Period of Study	Variable	Methods	Countries	Results →←
37	[73]	1991–2015	GDP and RE	Spatial Dublin model	26 European economies	Spatial dependences impact on the nexus between RE and GDP
38	[33]	1990–2014	CO ₂ , RE, EC	FMOLS and VECM	15 major RE consuming nations	$\begin{array}{l} {\rm EC} \leftrightarrow {\rm RE} \mbox{ for both S/R and LR supporting the feedback hypothesis; CO_2} \\ {\rm does \ not \ cause \ RE \ in \ the \ LR, \ CO_2 \leftrightarrow {\rm RE \ in \ the \ SR, \ EC} \ \leftrightarrow \ CO_2 \ \ both \ in \ the \ LR \ and \ SR \end{array}$
39	[60]	1990–2014	RE, pollution, EC, urbanization	Cointegration, Granger causality, impulse response function	Selected 106 countries	Both bidirectional and unidirectional relationship exists among the variables
40	[34]	1991–2014	RGDP, CO ₂ , technological innovation, trade and RE	Pedroni and Westerlund panel cointegration tests	Argentina, Brazil, Mexico, Colombia, Chile and Guatemala	Negative relationship exists between RE and CO ₂ RGDP, technological innovation, and trade positively and significantly impact on RE production
41	[47]	1980–2017	Non-oil exports, tourism, RE and RGDP	ARDL, Johansen cointegration and Gregory -Hensen cointegration	Saudi Arabia	Non-oil export and tourism impact growth positively, long run cointegration exist between RE tourism, capital and RGDP
42	[61]	1960-2015	RE, RGDP, trade, urbanization, CO ₂	ARDL, VECM Granger Causality tests	Australia and Canada	$\begin{array}{l} RGDP \ \ \rightarrow CO_2 \\ \mbox{both in LR and SR for Australia; VECM results shows that RGDP, \\ trade and RE \ \ \rightarrow CO_2 \\ \mbox{in d LR and SR for Australia; for Canada, Trade \ \ \rightarrow CO_2 \\ \mbox{for both LR and SR; RGDP, urbanization } \rightarrow CO_2 \\ \mbox{in of LR} \end{array}$
43	[48]	1990–2014	RE, NRE, RGDP	Pedroni unit root tests, FMOLS, P-DOLS, Dumitrescue-Hurlin (2012)	5 South Asia countries	Positive impact of RE, NRE and fixed capital formation on growth RGDP \rightarrow RE
44	[74]	1990–2014	Energy, efficiency, RE, RGDP	Fixed-effect panel quantity regression analysis	BRICS	Feedback hypothesis is valid RGDP \leftrightarrow EE RGDP \leftrightarrow RE EE \rightarrow RE
45	[75]	1981–2016	Energy production, energy consumption, GDP	Hatemi –J cointegration, structural breaks, FMOLS, CCR VECM, Granger causality test	China	EP, EC \rightarrow GDP, GDP \rightarrow Gas consumption (supporting conservation hypothesis)
46	[49]	1971–2014	Ecological footprint, GDP, EC, GFCF	N-ARDL; asymmetric causality techniques	Pakistan	Environmental quality \rightarrow EC neutrality hypothesis is valid among environmental quality, economic growth and capital

s/n	Authors	Period of Study	Variable	Methods	Countries	Results →←
47	[76]	2002–2011	CO ₂ , RE, NRE, RGDP	GMM and PMG	42	RE consumption leads to reduction in CO ₂ ; RE has positive impact on RGDP; NRE has negative effect on RGDP in LR, substitute relationship exists between NRE and RE
48	[77]	1980–2015	NRE, GDP, human capital index, globalization, urbanization, added value of services	Threshold regression FEMOLS	27 developed OECD countries	Economic development does not reduce non-renewable energy consumption; Human capital development reduces NRE. LR relationship exist among globalization, urbanization, services and RE
49	[62]	1990–2015	Ecological footprint, per capital income, RE, life expectancy, population density	Cointegration tests, cross-sectional augmented autoregressive distributed lag	8 developing South and South-East Asian economies	The association between per capital income and ecological footprint is N-shaped, RE reduces ecological footprint, increase in population leads to increase in pollution emissions.
50	[54]	1992–2016	EC, financial development, urbanization, per capital GDP, gross domestic capital formation	A battery of static and dynamic econometric models	44 African economies	EC and fin devt, deteriorates the environment; urbanization impacts on the environment asymmetrically; per capital GDP has an asymmetric effect on the environment.
51	[63]	1995-2017	Total energy consumption RE, NRE, HCI, FD; eco-innovation, energy intensity, GDP, gross fixed capital formation R&D	Westerlund and Edgerton panel cointegration and augmented mean group	G7 countries	Negative relationship exists among HCI, eco-innovation, energy price, R&D and TEC, NREC. Positive relationship exists between financial development, and each of TEC and NREC. HCI, eco-innovation, energy price, R&D enhances REC. Financial development reduces REC
52	[8]	1990–2014	Energy efficiency RE, CO ₂ , NE	Panel quantity regression (PQR)	66 developing economies	EE reduces carbon emissions across all quantities. RE reduces CO ₂ with substantial effect at 10th quartile. GDP increases CO ₂
53	[78]	1980–2016	CO ₂ , RE, HCI, globalization, trade openness	ARDL	China	RE does not impact on CO ₂ , HCI reduces environmental degradation; globalization, trade openness, and income impact on pollution
54	[63]	1965Q1-2017Q4	EC, ecological footprint, NRE economic complexity	QARDL quantile Granger causality test	NSA	Economic complexity and fossil fuel energy consumption significantly enhance ecological footprint; causality exist among economic complexity, energy consumption and ecological footprint
55	[36]	1990–2016	RE, RGDP	Bootstrap panel causality test	17 Emerging economies	Neutrality hypothesis holds for all the economies except Poland (no causality from either of the variables) RE \rightarrow RGDP for Poland
56	[26]	1998–2018	RE, financial development, CO ₂ , Innovation RGDP	P-ARDL Dumitrescu-Hurlin Panel causality test	ASEAN + 3 group	Financial development \rightarrow RE CO ₂ and economic freedom has negative impact on RE positive relationship exist between innovation, RGDP and RE
57	[80]	1965Q1-2017Q4	RE, NRE, RGDP ecological footprint	QARDL Granger causality	Turkey	RE decreases ecological footprint in of LR; NRE and RGDP positively impact ecological footprint
58	[81]	1991–2012	RE, RGDP, institutions, CO ₂	System-GMM FMOLS	85 developed and developing countries	RE positively impacts RGDP RE negatively impacts CO2 institution positively impacts RGDP; institution negatively affect CO2

Cont.
÷
e
9
E.

S/n	Authors	Period of Study	Variable	Methods	Countries	$\mathbf{Results} \to \leftarrow$
59	[82]	1990–2015	RE, NE, CO ₂ , RGDP, financial development	CIPS, FMOLS, bootstrap cointegration	74 countries	NRE has positive impact on CO ₂ . RE has negative impact on CO ₂ . Financial development has negative impact on CO ₂
60	[83]	1980–2014	TE, RE, NRE, RGDP	NARDL	G7 countries	Asymmetric relationship exists between TE and RGDP
61	[22]	1978–2016	CO ₂ , RGDP, RE, urbanization, agriculture	ARDL	Malaysia	CO ₂ is not directly influenced by modernization. Calibrating RE to agricultural sector will help in achieving sustainable agriculture and mitigate CO ₂ emissions; CO ₂ significantly decrease due to RGDP and urbanization
			Note: ARDL, NARDL, GMM, FMOL ⁵ method, vector error correction model model based on pooled mean group ee	S, DOLS, VECM, ARDL I, error correction model stimation, respectively.	,-PMG are autoregres l, fully modified ordir	ive distributed lag, nonlinear autoregressive distributed lag, general moment ary least square, dynamic ordinary least square, autoregressive distributed lag

The second strand of literature employs nonlinear models like quantile regression, system frequency domain estimate PMG, threshold regression, bootstrap estimates, NARDC, and recursive to examine the nature of relationship between energy, economic growth and CO₂ emissions with mixed results. For instance, [8,13,18,36,49,63,71,72,77,79,80,83] employed different versions of nonlinear models to examine the nexus between energy, economic growth, and CO_2 emissions with different results. Ref. [13] noted that fossil energy causes GHG, and that economic growth does not cause CO₂ emissions for 41 sub-Sahara African economies. Ref. [18] results from N-ARAL observed mixed findings; for example, the study noted that renewable energy reduces CO₂ emission for Nigeria, but no causality was documented between renewable energy and CO_2 for Angola and Egypt. The study further noted that renewable energy causes economic growth for Gabon, suggesting the validity growth hypothesis. Ref. [84] employed panel threshold for some selected OECD economies and reported the existence of positive and non-linear relationships between renewable energy and economic growth, an indication that the growth hypothesis holds. Ref. [49] employed the N-ARAL model and noted that environmental quality causes economic growth and that the neutrality hypothesis is valid, based on the results from environmental quality and capital stock. In a related development, [8] employed panel quantile regression to examine the nature of the relationship between energy, economic growth, and CO_2 for some selected 66 developing economies and noted that renewable energy reduces CO_2 with substantial effect at the 10th quantile, and that GDP increases CO_2 . Ref. [63] results, based on quantity ARDL, suggest the validity of the feedback hypothesis among economic complexity, energy consumption and the ecological footprint. For emerging economies [36] employed a bootstrap panel causality test and noted that the neutrality hypothesis is valid for all the economies except Poland, whose results suggest that causality is from renewable energy to economic growth. The single country (Turkey) estimates from [80] analysis shows that renewable energy reduces the ecological footprint in the long run; surprisingly, the results documented that non-renewable energy and economic growth positively impact on the ecological footprint.

3. Materials and Methods

This study examined the nature of the relationship between CO₂ emissions, energy consumption, agriculture and economic growth for some selected [34] Africa economies. Though Africa is made up of 54 independent countries, the selection of countries is largely influenced by data availability. The collected data cover the period 1980–2019. This period and the countries covered allow for examination of convergence issues inherent in the literature with adequate geographical covering of the African continent. The variables employed are annual data of GDP per capita (constants are 2010 and USD); CO₂ emissions per capita (metric tons); EC representing energy consumption; agriculture proxy by agricultural value added (AVA) per capita contribution of agriculture to GDP; and forest area (forest area as percentage of total land mass). The variables are expressed in natural forms such that $InCO_2$, $In\gamma$, InEC, InAVA; InFoR represent carbon emissions, economic growth, energy consumption, agricultural value chain and forest area, respectively. The data for the study are sourced as follows: CO₂ and RGDP from World Development Indicators (various issues), agriculture value addition and forest areas from Food and Agricultural Organization (various issues), and energy consumption data were from the OECD.

Methodology

As stated earlier, the study employed a frequency domain analysis to examine the relationship among energy, economic growth, and carbon emissions with the moderating impact of agriculture. Our preference of frequency domain estimates over time domain techniques is largely influenced by the weakness noticed in time domain estimates. For instance, time domain estimates cannot examine causality at different frequencies as they can only calculate a single test statistic over time [85–87]. Further, if the nexus among the variables is connected to more than one frequency, the ability of time dimension estimate to

explore the information from the original data set becomes ineffective [88,89]. To overcome this, Geweke (1982) developed the Wald test procedure that employed linear constraints on coefficient parameters to test Granger causality in a certain frequency range. This procedure was extended by [90,91] as single country frequency domain causality test [85]. The [91] single country frequency domain causality test was further extended to a multicountry model by [92]. This extended frequency domain (panel Granger causality test) allows us to determine if the predictive power is concentrated at quick or slow fluctuating components. The current study aims at examine the nexus between the variables using both single-country and multi-country causality tests by following [85,93–95]. The tests are thus presented.

Single-Country Causality Test:

We begin our single country causality test by following [2] Gorus and Aydin 2019 specification of the [90] single test procedure stated as follows:

$$X_{t} = \sum_{j=1}^{p} \theta_{11,j}^{X_{t-j} + \sum_{j=1}^{p} \theta_{12,j} Y_{t-j} + \varepsilon_{1t}}$$
(1)

Here, θ_{11} and θ_{12} , are the coefficients of the polynomials, ε_{1t} represents the error term, p represent the lag length, the constraint is on the first VAR, we express the constraints on the null hypothesis of "no Granger causality from Y_t to X_t at the frequency w'' as stated below:

$$\sum_{j=1}^{p} \theta_{12,j} cos(jw) = 0,$$

$$\sum_{j=1}^{p} \theta_{12,j} sin(jw) = 0.$$
(2)

To test these constraints, we employed the incremental R^2 measurement test, calculated as follows:

$$R_I^2 = R^2 - R_*^2 (3)$$

Here, R^2 and R^2_* are derived from the unrestricted and restricted models, respectively. (**) The null hypothesis is rejected if this condition is observed:

$$R_I^2 > F_{(2T-2p, 1-\alpha)} \frac{2}{T-2p} \left(1-R^2\right)$$
 (4)

Multi-Country Causality Test:

Following [92], the study employed the seemingly unrelated regression (SVR) model stated as follows:

$$X_{i,t} = \sum_{j=1}^{p} \beta_{i,j} X_{i,t-j} + \sum_{j=1}^{p} \gamma_{i,j} Y_{i,t-j} + \varepsilon_{i,t,\ i=1,\ 2,\ 3,\ \dots,\ N.}$$
(5)

Here, $X_{i,t}$ and $Y_{i,t}$ are the variables of country *i* at time *t*, *p* is the lag length, *N* represent the number of countries and $\varepsilon_{i,t}$ represents the error term at time *t* of country *i*. The null hypothesis constraints are expressed as follows:

$$\sum_{j=1}^{p} \gamma_{i,j} cos(jw) = 0, \ i = 1, 2, 3, \dots, N$$
$$\sum_{j=1}^{p} \gamma_{i,j} sin(jw) = 0, 1, 2, 3, \dots, N.$$
(6)

We tested these constraints using the incremented R^2 measured test, expressed as follows:

$$R_I^2 = R^2 - R_*^2 \tag{7}$$

Here, R^2 represent the unrestricted and R^2_* represents the restricted McElroy R^2 value expressed as follows:

$$R_I^2 > F_{(2N, N(T-2P), 1-\alpha)} \frac{2N}{N(T-2p)} \left(1 - R^2\right)$$
(8)

We rejected the null hypothesis of no Granger causality from Y_t to X_t at the frequency 'w' in the studied countries if Equation (8) was observed.

4. Results

The descriptive statistics and normality results of the variables employed in this study are presented in Table 2. The results suggested that the value of the Jarque-Bera statistics was greater than 5% for the variables, suggesting validity of normality in each of the variables studied.

	Descript	ive Analysis		Nor	mality Analys	is (Natural Log-F	orm)
Mean	Max.	Min.	SD	Skewness	Kurtosis	Jarque-Bera	Probability
175.98	298.77	142.67	39.09	-0.78	2.44	4.97	0.07
63.18	28.07	32.62	32.12	-0.48	2.14	4.22	0.06
158.78	197.09	102.11	28.09	-0.55	3.09	498	0.08
1.97	2.41	1.66	0.31	0.17	1.55	3.21	0.22
2.99	4.01	1.98	0.55	0.05	1.61	2.76	0.22
	Mean 175.98 63.18 158.78 1.97 2.99	Description Mean Max. 175.98 298.77 63.18 28.07 158.78 197.09 1.97 2.41 2.99 4.01	Descriptive Analysis Mean Max. Min. 175.98 298.77 142.67 63.18 28.07 32.62 158.78 197.09 102.11 1.97 2.41 1.66 2.99 4.01 1.98	Descriptive Analysis Mean Max. Min. SD 175.98 298.77 142.67 39.09 63.18 28.07 32.62 32.12 158.78 197.09 102.11 28.09 1.97 2.41 1.66 0.31 2.99 4.01 1.98 0.55	Descriptive Analysis Nor Mean Max. Min. SD Skewness 175.98 298.77 142.67 39.09 -0.78 63.18 28.07 32.62 32.12 -0.48 158.78 197.09 102.11 28.09 -0.55 1.97 2.41 1.66 0.31 0.17 2.99 4.01 1.98 0.55 0.05	Descriptive Analysis Normality Analysis Mean Max. Min. SD Skewness Kurtosis 175.98 298.77 142.67 39.09 -0.78 2.44 63.18 28.07 32.62 32.12 -0.48 2.14 158.78 197.09 102.11 28.09 -0.55 3.09 1.97 2.41 1.66 0.31 0.17 1.55 2.99 4.01 1.98 0.55 0.05 1.61	Descriptive Analysis Normality Analysis Mean Max. Min. SD Skewness Kurtosis Jarque-Bera 175.98 298.77 142.67 39.09 -0.78 2.44 4.97 63.18 28.07 32.62 32.12 -0.48 2.14 4.22 158.78 197.09 102.11 28.09 -0.55 3.09 498 1.97 2.41 1.66 0.31 0.17 1.55 3.21 2.99 4.01 1.98 0.55 0.05 1.61 2.76

Table 2. Descriptive statistics of the variables.

Source: Authors' computations 2022.

The results of both the cross-section dependency (CD) tests and the panel unit root tests are presented in Table 3. We began our analysis by investigating the cross-section dependency (CD) of the series, followed by conducting a check on the stationary properties of the series using the panel unit root test. The result in Table 3 suggest that cross-sectional dependency exists among the variables. This implies that shocks in any of the economies study can affect any of the rest. Having established cross-sectional dependency, we employed the cross-sectional augmented Dickey-Fuller test developed by [96], which is effective in detecting stationary properties of panel data as used in the current study [85,94,95]. The results suggests that $In\gamma$ and InAVA are stationary at the first different I(1), and that InEC, $InCO_2$, and InFOR are stationary at their level value I(0).

Table 3. Cross-section dependence and panel unit root tests for the series.

Variables	CD _{BP}	CD _{LM}	CD	CIPS Statistics
$In\gamma$	457.899 ***	76.558 ***	3.234 ***	-0.988
InEC	417.219 ***	51.521 ***	3.004 ***	-0.918
InAVA	398.881 ***	47.908 ***	9.176 ***	-2.955 **
InCO ₂	366.098 ***	56.897 ***	8.077 ***	-2.344 **
InFOR	564.092 ***	41.179 ***	12.098 ***	-3.756 **
$\Delta In\gamma$	-	_	_	-3.665 ***

Note: *** and ** suggest the rejection of the null hypothesis at 1% and 5% significance level, respectively. CIPS Statistics provides the simple average of the individual CADF statistics ($\overline{CADF_i}$).

5. Discussion

Frequency Domain Results

As earlier stated, the study intends to examine the nature of relationship among energy, economic growth, carbon emissions, forests, and agricultural added value at three (3) clear

frequencies: short, intermediate and long run denoted as 2.5, 1.5 and 0.5, respectively. Results in the long run (0.5) implies that a permanent causality exists while the results in the short run (2.5) suggest temporary causality exists. In Tables 4–10, we present the results of the frequency domain causality based on single-country estimates. Table 4 presents the results of the link between economic growth and CO₂ emission for each of the 34 African economies. The results as presented suggest that a unidirectional (at the three spectra) causality runs from economic growth to CO₂ emission for Algeria, Angola, Benin, Burkina Faso, Ghana, Kenya, Morocco, Nigeria, Senegal, South Africa, and Zambia. The findings are in line with [13,29,61], but contradict [17,18,67] The results further reveals that a oneway causality both at the intermediate and long run is noted to exist from emission to economic growth for Congo, Madagascar, Mali, Rwanda and Zimbabwe. The results from the rest of the economies studied suggest that no link can be established between CO_2 and economic growth. This finding supports the validity of the neutralization hypothesis in these economies; thus, emission curbing policies can be applied in these economies. The results from Algeria, Angola, Benin, Burkina Faso, Ghana, Kenya, Morocco, Nigeria, Senegal, South Africa, and Zambia suggest that environmental protection laws could be harmful to the economy.

In Table 5, we present the results of the link between energy consumption and economic growth for the selected African economies. The results suggest that a bi-directional relationship exist between the two for the economies of Algeria, Ghana, Kenya, Morocco, and Nigeria (at the three periods), South Africa (at intermediate and long run), Egypt (at the short run and intermediate), and at least one for each of Cameroon, Guinea, and Madagascar. These results support the validity of the feedback hypothesis in these economies. The results further reveal that an un-directional causality runs from economic growth to energy consumption for the economies of Mozambique, Namibia, Tanzania and Uganda in the short run, this suggests that the conservation hypothesis is rational in these economies. The growth hypothesis is validated based on the existence of causality from economic growth to energy consumption for the economies of Algeria, Ghana, Kenya, Morocco and Nigeria. The results are in line with the findings of [30,33,74].

Table 6 presents the results of the nexus between energy consumption and CO_2 emissions in the studied economies. The results reveal that energy consumption Granger causes carbon emissions in Nigeria, Algeria, Egypt, Tunisia and Ghana, suggesting that the pollution haven hypothesis is valid for these economies at short, intermediate and long runs. The results support the findings of [65] but disagree with [55].

The results of the causality between economic growth and agricultural value addition, as presented in Table 7, suggest that bi-directional causality is noted for almost all the studied economies at the short, intermediate, and long runs. The result is not surprising because agriculture constitutes the bulk of African GDP.

Table 8 shows that for most the studied economies, a unidirectional relationship runs from forestry to economic growth; this suggests that wood sourced from the forest support economic growth in the studied economies.

In Table 9, we present the results of the relationship between energy consumption and agricultural value addition across the three spectra of our analysis. The results reveal that there is a unidirectional causality from energy consumption to agricultural value addition in Egypt, Ghana, Tunisia and Uganda, whereas a bi-directional causality is documented for the economies of Nigeria, South Africa, Angola. This suggests that the feedback hypothesis is validated based on the relationship between energy consumption and agriculture in these economies. The results of the relationship between forestry and energy consumption are almost the same with those of agriculture and energy consumption, except that a one-way causality is noted to exist between forestry and energy consumption, suggesting the validity of the conservative hypothesis in these economies.

Table 10 we present the results of causality between CO_2 emission and agricultural value addition for the selected Africa economies. Our results reveal that no causality exists between these variables for the economies studies.

			P	anel A				
Countries		$H_0: In\gamma$	→InCO ₂			H ₀ : InC	$CO_2 \rightarrow In\gamma$	
Countries	w = 0.5	w = 1.5	w = 2.5	c.v. = 10%	w = 0.5	w = 1.5	w = 2.5	c.v. = 10%
Algeria	0.013 ***	0.055 ***	0.128 ***	0.111	0.023	0.027	0.034	0.111
Angola	0.017 ***	0.005 ***	0.005 ***	0.009	0.034	0.036	0.044	0.113
Burkina Faso	0.096 ***	0006 ***	0.001 ***	0.009	0.023	0.027	0.054	0.112
Benin	0.073 ***	0.054 ***	0.022 ***	0.072	0.026	0.028	0.034	0.114
Cameron	0.091	0.071	0.004	0.014	0.019	0.016	0.045	0.116
Congo (Brazzaville)	0.009	0.002	0.005	0.012	0.029	0.019	0.034	0.112
Congo (DRC)	0.047 ***	0.008 ***	0.006	0.009	0.03 *	0.021 **	0.045	0.111
Egypt	0.004 ***	0.044 ***	0.007 ***	0.008	0.023	0.028	0.054	0.112
Ethiopia	0.021	0.046	0.017	0.065	0.033	0.038	0.048	0.118
Gabon	0.009	0.032	0.014	0.008	0.035	0.037	0.039	0.112
Ghana	0.019 ***	0.044 ***	0.011 ***	0.011	0.045	0.054	0.037	0.114
Guinea	0.009	0.008	0.012	0.116	0.037	0.031	0.038	0.132
Kenya	0.022 ***	0.045 ***	0.011 ***	0.113	0.039	0.032	0.045	0.161
Lesotho	0.031	0.032	0.012	0.114	0.029	0.024	0.055	0.115
Madagascar	0.011 ***	0.017 ***	0.014	0.111	0.018 **	0.021 **	0.034	0.113
Malawi	0.032	0.019	0.001	0.102	0.024	0.027	0.049	0.112
Mali	0.022 ***	0.039 ***	0.009	0.019	0.032	0.036	0.054	0.122
Mauritius	0.005	0.033	0.004	0.112	0.036	0.037	0.032	0.141
Morocco	0.007 ***	0.032 ***	0.007 ***	0.133	0.029	0.031	0.035	0.112
Mozambique	0.046	0.037	0.006	0.121	0.017	0.021	0.041	0.116
Namibia	0.033	0.081	0.009	0.114	0.029	0.037	0.039	0.114
Nigeria	0.044 ***	0.033 ***	0.014 ***	0.111	0.025	0.028	0.057	0.123
Rwanda	0.006 ***	0.023 ***	0.012	0.112	0.044 **	0.034 **	0.045	0.114
Sao Tome and Principe	0.045	0.012	0.009	0.115	0.031	0.028	0.055	0.152
Senegal	0.044 ***	0.008 ***	0.006 ***	0.117	0.022	0.026	0.055	0.143
Sierra Leone	0.032	0.091	0.008	0.111	0.019	0.021	0.053	0.122
South Africa	0.031	0.023	0.005	0.112	0.022 *	0.026 *	0.058 *	0.144
Tanzania	0.029	0.033	0.006	0.111	0.027	0.029	0.059	0.115
Тодо	0.031	0.034	0.009	0.122	0.032	0.035	0.077	0.122
Tunisia	0.033	0.023	0.008	0.111	0.028	0.031	0.056	0.127
Uganda	0.045	0.031	0.006	0.121	0.029	0.031	0.055	0.157
Zambia	0.033	0.022	0.009	0.111	0.019	0.022	0.054	0.138
Zimbabwe	0.046	0.036	0.045	0.123	0.021	0.023*	0.067*	0.136

Table 4. Granger causality tests in the frequency domain estimates ($In\gamma$, $InCO_2$).

***, **, * represent 1%, 5%, 10% significant levels, respectively.

Courteiro		$H_0: In\gamma$	→InEC			H ₀ : InE	C→Inγ	
Countries	w = 0.5	w = 1.5	w = 2.5	c.v. = 10%	w = 0.5	w = 1.5	w = 2.5	c.v. = 10%
Algeria	0.023 ***	0.031 ***	0.022 ***	0.012	0.031 ***	0.029 ***	0.027 ***	0.111
Angola	0.001	0.004	0.003	0.002	0.041 ***	0.037 ***	0.034 ***	0.112
Burkina Faso	0.007	0.012	0.011	0.005	0.029	0.044	0.027	0.099
Benin	0.012	0.014	0.019	0.006	0.031	0.039	0.027	0.122
Cameron	0.018	0.019 *	0.012	0.009	0.014	0.037 *	0.034	0.117
Congo (Brazzaville)	0.021	0.016	0.014	0.021	0.016	0.039	0.029	0.112
Congo (DRC)	0.064	0.044	0.032	0.017	0.018	0.068	0.029	0.110
Egypt	0.017	0.022 ***	0.026 ***	0.005	0.031	0.039 ***	0.033 ***	0.117
Ethiopia	0.024	0.022	0.033	0.006	0.042	0.054	0.039 ***	0.102
Gabon	0.021 ***	0.019 ***	0.017 ***	0.011	0.033	0.056	0.027	0.115
Ghana	0.031 ***	0.021 ***	0.019 ***	0.013	0.067 ***	0.011 ***	0.034 ***	0.111
Guinea	0.026	0.024 *	0.021	0.004	0.028	0.032 *	0.045	0.115
Kenya	0.021 ***	0.019 ***	0.017 ***	0.021	0.028 ***	0.034 ***	0.054 ***	0.119
Lesotho	0.016	0.019	0.022	0.024	0.021	0.044	0.048	0.167
Madagascar	0.017	0.021 *	0.025	0.031	0.027	0.045 *	0.039 ***	0.109
Malawi	0.014	0.017	0.022	0.024	0.029	0.056	0.037	0.114
Mali	0.022	0.025	0.029	0.001	0.41	0.059	0.038	0.112
Mauritius	0.019	0.015	0.011	0.005	0.039	0.041	0.045	0.109
Morocco	0.021 ***	0.022 ***	0.023 ***	0.017	0.033 ***	0.039 ***	0.055 ***	0.112
Mozambique	0.022	0.021	0.029	0.013	0.028	0.034 **	0.034	0.119
Namibia	0.031	0.023	0.034	0.014	0.032	0.041 **	0.049	0.166
Nigeria	0.027 ***	0.028 ***	0.029 ***	0.011	0.031 ***	0.044 ***	0.054 ***	0.112
Rwanda	0.003	0.031	0.022	0.009	0.027	0.033	0.032	0.114
Sao Tome and Principe	0.009	0.010	0.013	0.002	0.025	0.029	0.035	0.141
Senegal	0.023 ***	0.025 ***	0.027 ***	0.003	0.029	0.049	0.041	0.117
Sierra Leone	0.031	0.041	0.034	0.008	0.023	0.044	0.039	0.118
South Africa	0.052 ***	0.024 ***	0.027	0.003	0.028	0.046 ***	0.057 ***	0.114
Tanzania	0.023	0.022	0.027	0.011	0.031	0.041 **	0.045	0.119
Togo	0.054	0.042	0.034	0.014	0.038	0.038	0.055	0.109
Tunisia	0.037 ***	0.031 ***	0.029 ***	0.011	0.037	0.039	0.045	0.115
Uganda	0.044	0.032	0.029	0.023	0.033 ***	0.031 ***	0.054 ***	0.167
Zambia	0.022	0.031	0.033	0.015	0.028	0.033	0.048	0.117
Zimbabwe	0.023	0.034	0.039	0.014	0.024	0.032	0.039	0.115

Table 5. Granger causality tests in the frequency domain estimates $In\gamma$, InEC.

***, **, * represent 1%, 5%, 10% significant levels, respectively.

Countries		H ₀ : InEC	C→InCO ₂			H ₀ : InCC	$D_2 \rightarrow InEC$	
Countries	w = 0.5	w = 1.5	w = 2.5	c.v. = 10%	w = 0.5	w = 1.5	w = 2.5	c.v. = 10%
Algeria	0.023 ***	0.027 ***	0.031 ***	0.091	0.029 ***	0.028 ***	0.034 ***	0.032
Angola	0.034	0.036	0.041	0.007	0.023 ***	0.031 ***	0.044 ***	0.014
Burkina Faso	0.023	0.027	0.029	0.012	0.028	0.010	0.054	0.006
Benin	0.026	0.028	0.031	0.014	0.031	0.025	0.034	0.044
Cameron	0.019 **	0.016 **	0.014 **	0.017	0.038 ***	0.041 ***	0.045 ***	0.009
Congo (Brazzaville)	0.029	0.019	0.016	0.019	0.037	0.024	0.034	0.018
Congo (DRC)	0.037	0.021	0.018	0.081	0.033	0.022	0.045	0.092
Egypt	0.023 ***	0.028 ***	0.031 ***	0.089	0.028	0.042	0.054	0.078
Ethiopia	0.033	0.038	0.042	0.091	0.024	0.031	0.048	0.099
Gabon	0.035	0.037	0.033	0.071	0.029	0.032	0.039	0.077
Ghana	0.045 ***	0.054 ***	0.067 ***	0.009	0.023	0.031	0.037	0.101
Guinea	0.037	0.031	0.028	0.008	0.028	0.034	0.038	0.111
Kenya	0.039	0.032	0.028	0.045	0.029	0.028	0.045	0.098
Lesotho	0.029	0.024	0.021	0.076	0.018	0.031	0.055	0.102
Madagascar	0.018	0.021	0.027	0.089	0.024	0.010	0.034	0.111
Malawi	0.024	0.027	0.029	0.090	0.032	0.025	0.049	0.133
Mali	0.032	0.036	0.41	0.039	0.036	0.041	0.054	0.122
Mauritius	0.036	0.037	0.039	0.051	0.029	0.024	0.032	0.121
Morocco	0.029	0.031	0.033	0.044	0.017	0.022	0.035	0.090
Mozambique	0.017	0.021	0.028	0.062	0.029	0.042	0.041	0.112
Namibia	0.029	0.037	0.032	0.082	0.025	0.031	0.039	0.122
Nigeria	0.025 ***	0.028 ***	0.031 ***	0.095	0.044	0.032	0.057	0.124
Rwanda	0.044	0.034	0.027	0.083	0.031	0.031	0.045	0.154
Sao Tome and Principe	0.031	0.028	0.025	0.076	0.029	0.034	0.055	0.101
Senegal	0.022	0.026	0.029	0.049	0.018	0.028	0.055	0.111
Sierra Leone	0.019	0.021	0.023	0.078	0.024	0.031	0.053	0.121
South Africa	0.022	0.026	0.028	0.065	0.024	0.010	0.058	0.132
Tanzania	0.027	0.029	0.031	0.007	0.032	0.025	0.059	0.122
Togo	0.032	0.035	0.038	0.009	0.036	0.041	0.077	0.176
Tunisia	0.028 ***	0.031 ***	0.037 ***	0.065	0.029 **	0.024 **	0.056 **	0.109
Uganda	0.029	0.031	0.033	0.098	0.017 **	0.022 **	0.055 **	0.101
Zambia	0.019	0.022	0.028	0.097	0.029	0.042	0.054	0.102
Zimbabwe	0.021	0.023	0.024	0.008	0.025	0.031	0.067	0.111

Table 6. Granger causality tests in the frequency domain estimates *InEC*, *InCO*₂.

***, **, represent 1%, 5% significant levels, respectively.

Countries -	H ₀ : Inγ→InAVA				$H_0: InAVA \rightarrow In\gamma$			
	w = 0.5	w = 1.5	w = 2.5	c.v. = 10%	w = 0.5	w = 1.5	w = 2.5	c.v. = 10%
Algeria	0.029 ***	0.031 ***	0.034 ***	0.023	0.014 ***	0.045 ***	0.035 ***	0.019
Angola	0.038 ***	0.042 ***	0.045 ***	0.009	0.015 ***	0.015 ***	0.045 ***	0.098
Burkina Faso	0.034 **	0.044 **	0.047 *	0.008	0.093 *	0008 **	0.053 **	0.116
Benin	0.022 **	0.026 *	0.029 ***	0.012	0.072 **	0.053 **	0.034 *	0.122
Cameron	0.023 *	0.027 **	0.029 *	0.019	0.093 *	0.072 **	0.047 **	0.138
Congo (Brazzaville)	0.021 **	0.026 **	0.029 **	0.076	0.007 **	0.009 *	0.034 ***	0.129
Congo (DRC)	0.022 **	0.025 **	0.028 *	0.027	0.043 *	0.005 **	0.045 *	0.147
Egypt	0.019 **	0.023 **	0.029 **	0.098	0.007 **	0.042 **	0.053 **	0.126
Ethiopia	0.018 *	0.022 **	0.027 **	0.056	0.027 ***	0.041 **	0.047 **	0.091
Gabon	0.016 **	0.019 **	0.022 **	0.039	0.005 **	0.033 **	0.041 **	0.125
Ghana	0.022 *	0.025 **	0.029 ***	0.044	0.015 **	0.042 **	0.037 *	0.087
Guinea	0.018 **	0.021 **	0.027 *	0.087	0.005 *	0.006 **	0.034 ***	0.099
Kenya	0.007 ***	0.012 **	0.019 ***	0.069	0.027 *	0.046 ***	0.053 **	0.102
Lesotho	0.018 **	0.011 **	0.019 **	0.081	0.038 **	0.036 **	0.059 **	0.009
Madagascar	0.019 **	0.022 **	0.026 **	0.072	0.016 *	0.016 *	0.039 *	0.122
Malawi	0.022 *	0.023 **	0.026 **	0.098	0.036 *	0.016 *	0.047 **	0.134
Mali	0.027 *	0.029 *	0.031 *	0.099	0.026 *	0.036 **	0.054 *	0.177
Mauritius	0.032 *	0.028 *	0.024 *	0.062	0.009 **	0.038 *	0.045 **	0.187
Morocco	0.009 *	0.014 *	0.019 **	0.073	0.009 *	0.037 **	0.065 **	0.138
Mozambique	0.007 **	0.009 *	0.011 **	0.079	0.047 *	0.034 **	0.044 ***	0.166
Namibia	0.009 ***	0.012 **	0.019 ***	0.092	0.037 **	0.083 **	0.098 *	0.147
Nigeria	0.011 ***	0.014 *	0.019 **	0.095	0.047 ***	0.034 **	0.059 **	0.123
Rwanda	0.021 **	0.025 **	0.028 *	0.093	0.009 *	0.024 **	0.043 **	0.122
Sao Tome and Principe	0.012 *	0.018 *	0.022 **	0.091	0.047 **	0.015 **	0.058 ***	0.111
Senegal	0.024 **	0.027 *	0.032 ***	0.084	0.049 **	0.005**	0.058 *	0.145
Sierra Leone	0.022 **	0.026 ***	0.029 *	0.079	0.039 **	0.094 *	0.054 **	0.118
South Africa	0.011 *	0.016 *	0.019 **	0.099	0.039 **	0.025 **	0.056 **	0.128
Tanzania	0.022 **	0.026 *	0.028 **	0.078	0.041 **	0.035 **	0.055 *	0.101
Togo	0.021 **	0.025 **	0.029 *	0.055	0.033 **	0.035 *	0.074 *	0.109
Tunisia	0.009 **	0.011 *	0.016 **	0.089	0.034 *	0.025 **	0.053 **	0.154
Uganda	0.019 *	0.023 **	0.029 **	0.037	0.047 *	0.035 **	0.055 *	0.111
Zambia	0.021 *	0.026 *	0.031 *	0.088	0.037 *	0.025 **	0.055 *	0.122
Zimbabwe	0.007 *	0.011 **	0.019 *	0.089	0.043 **	0.035 **	0.064 **	0.143

Table 7. Granger causality tests in the frequency domain estimates $In\gamma$, InAVA.

***, **, * represent 1%, 5%, 10% significant levels, respectively.

Countries	$H_0: In\gamma \rightarrow InFOR$				H_0 : InFOR \rightarrow In γ			
	w = 0.5	w = 1.5	w = 2.5	c.v. = 10%	w = 0.5	w = 1.5	w = 2.5	c.v. = 10%
Algeria	0.009	0.011 *	0.019	0.091	0.005	0.033	0.044	0.093
Angola	0.004	0.012 *	0.019	0.009	0.009	0.005	0.048	0.098
Burkina Faso	0.011 *	0.015	0.019	0.017	0007	0.023	0.056	0.099
Benin	0.003	0.009	0.011	0.019	0.005	0.024	0.037	0.092
Cameron	0.005 ***	0.009 ***	0.012 ***	0.089	0.001	0.029	0.048	0.091
Congo (Brazzaville)	0.002 ***	0.006 ***	0.009 ***	0.079	0.002	0.036	0.037	0.078
Congo (DRC)	0.003 ***	0.007 ***	0.022 ***	0.097	0.008	0.034	0.049	0.103
Egypt	0.006	0.009 *	0.011	0.087	0.004	0.032	0.057	0.099
Ethiopia	0.011	0.017 *	0.021	0.057	0.006	0.032	0.056	0.094
Gabon	0.008	0.012 *	0.022	0.023	0.002	0.039	0.044	0.109
Ghana	0.007	0.014 *	0.021	0.028	0.004	0.041	0.039	0.111
Guinea	0.003	0.008 *	0.011	0.055	0.008	0.034	0.040	0.104
Kenya	0.008 ***	0.013 ***	0.019 ***	0.089	0.005 *	0.039 **	0.047 *	0.101
Lesotho	0.009	0.022 **	0.029	0.082	0.002	0.039	0.058	0.099
Madagascar	0.014	0.023 *	0.029	0.044	0.007	0.031	0.038	0.102
Malawi	0.021	0.022 **	0.028	0.043	0.009	0.037	0.056	0.101
Mali	0.008	0.044 *	0.054	0.049	0.009	0.045	0.057	0.078
Mauritius	0.014	0.021 *	0.034	0.076	0.003	0.045	0.055	0.099
Morocco	0.022	0.025 *	0.029	0.077	0.002	0.042	0.053	0.089
Mozambique	0.028	0.031 *	0.045 *	0.073	0.007	0.031	0.045	0.098
Namibia	0.027	0.031 *	0.048 *	0.071	0.001	0.043	0.048	0.067
Nigeria	0.021 ***	0.027 ***	0.037 ***	0.082	0.003	0.048	0.057	0.089
Rwanda	0.011	0.033	0.054	0.091	0.003	0.041	0.045	0.098
Sao Tome and Principe	0.023 *	0.043	0.055	0.027	0.002	0.020	0.054	0.044
Senegal	0.011 ***	0.033 ***	0.058	0.031	0.008	0.035	0.053	0.056
Sierra Leone	0.012	0.027	0.039	0.036	0.001	0.051	0.054	0.019
South Africa	0.009 ***	0.014 ***	0.051 ***	0.042	0.003	0.034	0.052	0.110
Tanzania	0.019	0.026	0.031	0.043	0.003	0.032	0.053	0.101
Togo	0.008	0.015	0.029	0.055	0.004	0.052	0.071	0.089
Tunisia	0.007	0.017	0.032	0.069	0.003	0.041	0.052	0.091
Uganda	0.009 ***	0.032 ***	0.054 ***	0.072	0.001	0.042	0.052	0.088
Zambia	0.011	0.028	0.038	0.058	0.002	0.041	0.052	0.078
Zimbabwe	0.013	0.029	0.054	0.098	0.006	0.044	0.062	0.098

Table 8. Granger causality tests in the frequency domain estimates $In\gamma$, InFOR.

***, **, * represent 1%, 5%, 10% significant levels, respectively.

Countries	H ₀ : InEC→InAVA				H ₀ : InAVA→InEC			
	w = 0.5	w = 1.5	w = 2.5	c.v. = 10%	w = 0.5	w = 1.5	w = 2.5	c.v. = 10%
Algeria	0.009	0.014	0.029	0.121	0.024	0.029	0.033	0.101
Angola	0.011 ***	0.026 ***	0.037 ***	0.019	0.024	0.032	0.042	0.103
Burkina Faso	0.008	0.039	0.044	0.019	0.024	0.012	0.052	0.102
Benin	0.029	0.044	0.039	0.082	0.034	0.023	0.032	0.104
Cameron	0.011	0.029	0.037	0.024	0.034	0.043	0.041	0.106
Congo (Brazzaville)	0.028	0.031	0.039	0.032	0.034	0.022	0.031	0.102
Congo (DRC)	0.027	0.058	0.068	0.019	0.035	0.023	0.041	0.101
Egypt	0.013 ***	0.025 ***	0.039 ***	0.018	0.025	0.043	0.052	0.102
Ethiopia	0.029	0.033	0.054	0.095	0.025	0.032	0.044	0.108
Gabon	0.011	0.023	0.056	0.008	0.024	0.033	0.035	0.102
Ghana	0.013 ***	0.028 ***	0.011 ***	0.101	0.025	0.033	0.034	0.104
Guinea	0.016	0.022	0.032	0.016	0.025	0.035	0.034	0.102
Kenya	0.012	0.024	0.034	0.013	0.024	0.024	0.043	0.101
Lesotho	0.014	0.026	0.044	0.104	0.019	0.035	0.053	0.105
Madagascar	0.011	0.033	0.045	0.101	0.023	0.014	0.033	0.103
Malawi	0.009	0.045	0.056	0.101	0.033	0.024	0.042	0.102
Mali	0.006	0.054	0.059	0.009	0.032	0.045	0.051	0.102
Mauritius	0.008	0.023	0.041	0.102	0.021	0.025	0.037	0.101
Morocco	0.007	0.033	0.039	0.103	0.019	0.024	0.034	0.102
Mozambique	0.004	0.021	0.034	0.101	0.022	0.044	0.043	0.106
Namibia	0.006	0.025	0.041	0.104	0.023	0.034	0.033	0.104
Nigeria	0.009 ***	0.029 ***	0.044 ***	0.101	0.041 ***	0.033 ***	0.054 ***	0.103
Rwanda	0.012	0.028	0.033	0.102	0.034	0.036	0.043	0.104
Sao Tome and Principe	0.009	0.026	0.029	0.105	0.019	0.033	0.052	0.112
Senegal	0.007	0.028	0.049	0.107	0.019	0.023	0.057	0.103
Sierra Leone	0.006	0.039	0.044	0.101	0.021	0.034	0.056	0.102
South Africa	0.005 ***	0.028 ***	0.046 ***	0.114	0.022 ***	0.014 ***	0.054 ***	0.104
Tanzania	0.017	0.021	0.041	0.113	0.033	0.023	0.053	0.195
Togo	0.022	0.029	0.038	0.124	0.031	0.043	0.073	0.102
Tunisia	0.017 ***	0.022 ***	0.039 ***	0.112	0.021	0.023	0.053	0.107
Uganda	0.018 ***	0.023 ***	0.031 ***	0.123	0.019	0.024	0.053	0.107
Zambia	0.012	0.028	0.033	0.112	0.019	0.041	0.052	0.108
Zimbabwe	0.014	0.029	0.032	0.124	0.021	0.032	0.062	0.119

Table 9. Granger causality tests in the frequency domain estimates *InEC*, *InAVA*.

*** represent 10% significant level.

				Panel G				
Countries		H ₀ : InAVA→InCO ₂						
Countries	w = 0.5	w = 1.5	w = 2.5	c.v. = 10%	w = 0.5	w = 1.5	w = 2.5	c.v. = 10%
Algeria	0.006	0.023	0.034	0.019	0.013	0.032	0.049	0.114
Angola	0.005	0.033	0.044	0.027	0.017	0.052	0.059	0.115
Burkina Faso	0.002	0.032	0.054	0.025	0.096	0.043	0.069	0.118
Benin	0.009	0.029	0.034	0.025	0.073	0.056	0.061	0.117
Cameron	0.011	0.023	0.045	0.023	0.091	0.058	0.062	0.111
Congo (Brazzaville)	0.012	0.029	0.034	0.021	0.009	0.055	0.069	0.124
Congo (DRC)	0.021	0.032	0.045	0.089	0.047	0.055	0.062	0.101
Egypt	0.024	0.044	0.054	0.099	0.004	0.058	0.061	0.102
Ethiopia	0.022	0.033	0.048	0.094	0.021	0.074	0.051	0.108
Gabon	0.012	0.022	0.039	0.072	0.009	0.054	0.052	0.123
Ghana	0.009	0.029	0.037	0.011	0.019	0.054	0.051	0.124
Guinea	0.019	0.029	0.038	0.011	0.009	0.054	0.061	0.102
Kenya	0.015	0.028	0.045	0.056	0.022	0.066	0.069	0.101
Lesotho	0.013	0.039	0.055	0.076	0.031	0.037	0.049	0.112
Madagascar	0.021	0.029	0.034	0.019	0.011	0.052	0.059	0.117
Malawi	0.025	0.023	0.049	0.091	0.032	0.044	0.058	0.115
Mali	0.005	0.032	0.054	0.071	0.022	0.055	0.064	0.102
Mauritius	0.014	0.024	0.032	0.080	0.005	0.051	0.054	0.101
Morocco	0.011	0.028	0.035	0.049	0.007	0.051	0.069	0.102
Mozambique	0.012	0.013	0.041	0.069	0.046	0.051	0.052	0.119
Namibia	0.018	0.029	0.039	0.089	0.033	0.051	0.058	0.119
Nigeria	0.022	0.041	0.057	0.099	0.044	0.071	0.072	0.129
Rwanda	0.021	0.032	0.045	0.089	0.006	0.051	0.064	0.117
Sao Tome and Principe	0.012	0.039	0.055	0.072	0.045	0.056	0.065	0.155
Senegal	0.022	0.033	0.055	0.091	0.044	0.056	0.058	0.141
Sierra Leone	0.014	0.023	0.053	0.071	0.032	0.067	0.069	0.128
South Africa	0.022	0.034	0.058	0.069	0.031	0.034	0.071	0.148
Tanzania	0.008	0.023	0.059	0.009	0.029	0.055	0.071	0.112
Togo	0.006	0.032	0.077	0.011	0.031	0.044	0.059	0.121
Tunisia	0.008	0.029	0.056	0.066	0.033	0.055	0.062	0.121
Uganda	0.009	0.022	0.055	0.093	0.045	0.053	0.064	0.150
Zambia	0.019	0.021	0.054	0.091	0.033	0.056	0.068	0.132
Zimbabwe	0.021	0.029	0.067	0.009	0.046	0.056	0.064	0.131

 Table 10. Granger causality tests in the frequency domain estimates InCO2, InAVA.

The results of the panel Granger causality in the frequency domain for all the examined African economies suggest the existence of bi-directional relationships across the three spectra between economic growth and energy consumption. The results further reveal that a one-way Granger causality runs from energy consumption to CO_2 emission in the studied

economies. A further examination of the results also suggests that there is a causal nexus between carbon emissions and economic growth for the entire spectra studied, and that no evidence suggests that causality runs from economic growth to carbon emissions. In term of theoretical underpinning, one can deduce that the feedback hypothesis is valid for the relationship between energy consumption and economic growth in the studied African economies. This suggests that African economies could grow their economies by increasing energy consumption, and that energy consumption could also be enhanced by growing the economy, suggesting that demand for energy consumption is a booster of economic growth. For the nexus between energy consumption and CO_2 emission, the results suggest the validity of the pollution haven hypothesis, as energy consumption has a bi-directional relationship with growth driving carbon emissions in African economies, thus, Africa economies, while pursuing growth, should start looking at clean energy consumption. Though the results of the study suggest that no causality runs from economic growth to carbon emissions, ruling out the possibility of the pollution haven hypothesis, the existence of causality from energy consumption to carbon emissions points to the existence or potential of the pollution haven hypothesis, which could be from an indirect perspective. On the meditating role of agricultural value addition and forests, the results noted that the impact of both forests and agricultural value addition is only significant on economic growth across all the spectra, and on energy consumption in the short run. No causality is established between either of forests and agricultural value addition, and CO₂ emission for the studied economies.

For comparison, we conducted time domain estimates for the entire region by employing the Dumitrescu–Hurlin panel causality estimate. From the results, it could be deduced that a bi-directional relationship exists between economic growth and energy consumption, and that a one-way causality runs from energy consumption to carbon emissions. The results suggest the feedback hypothesis is valid on the nexus between energy and economic growth in Africa. The results of the one-way nexus, however, suggest that the conservation hypothesis is not valid in Africa. Unlike the frequency domain estimate, the moderating variables failed exhibit any form of causality in the time domain model.

The study has made some significant contribution to knowledge by being among the first set of studies that has examined the nexus among energy, environment and economic growth in Africa within the context of frequency domain estimate, and that calibrated the moderating roles of forest and agricultural value addition to this nexus.

6. Conclusions

The essence of this study was to examined the causal relationships between energy consumption, economic growth and CO_2 emission with the moderating roles of forestry and agricultural value addition in Africa, by employing both time domain and frequency domain estimates to analyzed data sourced from 1980 to 2019. The study provides both single-country and multi-country estimates of this nexus. The results of the single country estimate are at best mixed across the various frequencies. The study recommends that policymakers in the studied economies should take into consideration these empirical findings when designing policy tools to achieving the correct mix of energy that will stimulate economic growth without causing havoc to the environment.

The results of the panel Granger causality estimates in the frequency domain suggest that a bi-directional relationship exists between energy consumption and economic growth in Africa economies. This implies that to achieve economic growth, the energy sector should be enhanced, and that enhanced energy space will further drive or stimulate growth. The results further suggest the existence of a one-way causality from energy consumption to carbon emissions, ruling out the validity of the conservation hypothesis in these economies. This could be a result of heavy dependency/consumption of non-renewable energy in the region. It is therefore recommended that policymakers in this region should start looking at movement toward clean energy consumption. Our results are in line with the findings of Aydin (2019 for OECD economies, Gorus and Aydin 2018 for MENA economies, but contradicts [33,97].

The study is not an all-inclusive one, as there are limitations, which could be areas to be considered by other studies. For instance, alternative estimation techniques could be employed, other variables like ecological footprints, macroeconomic variables like foreign direct investment, and socio-political variables, among others. Other studies could examine the cost-benefit analysis of different energy options as they relate to the environment, economic growth, among others. Future research can employ multi-criteria analyses useful for quantifying the nexus between the different components.

The global economy is moving towards adopting renewable energy with the intension of mitigating climate change and reducing CO₂ emissions; hence, the economies of Africa should make concerted efforts to develop their renewable energy potential to support economic growth. This is in line with the UN resolution of the 2015 Paris Agreement that by the 21st Conference of Parties (COP21) of the United Nations Framework Convention on Climate Change (UNFCCC), countries should focus on investing in sustainable energy and de-emphasizing the consumption of fossil fuel, among others. African economies are encouraged to formulate and implement policies that will encourage consumption of renewable energy technologies such as laws protecting the production and usage of domestic solar panels, wind turbine production, granting tax incentives to renewable energy investments, stimulate green bonds and investment, among others.

Funding: This research received no external funding.

Data Availability Statement: The data for the study are sourced as follows: CO₂ and RGDP from World Development Indicators (various issues), agriculture value addition and forest areas from Food and Agricultural Organization (various issues), and energy consumption data was from OECD.

Acknowledgments: We acknowledge the support of Bowen University Management for proving the APC for this article.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Wesseh, P.K.; Lin, B.; Owusu, M. Delving into Liberia's energy economy: Technical change, inter-factor and inter-fuel substitution. *Renew. Sustain. Energy Rev.* 2013, 24, 122–130. [CrossRef]
- Aydin, C.; Esen, Ö. Does the level of energy intensity matter in the effect of energy consumption on the growth of transition economies? Evidence from dynamic panel threshold analysis. *Energy Econ.* 2018, 69, 185–195. [CrossRef]
- Kumar, A.; Ranjan, B.; Adebola, S. Analysing the spillovers between crude oil prices, stock prices and metal prices: The importance of frequency domain in USA. *Energy* 2021, 220, 119732. [CrossRef]
- 4. Al Khars, M.; Miah, F.; Qudrat-Ullah, H.; Kayal, A. A systematic review of the relationship between energy consumption and economic growth in GCC countries. *Sustainability* **2020**, *12*, 3845. [CrossRef]
- 5. Nesticò, A.; Maselli, G. Declining discount rate estimate in the long-term economic evaluation of environmental projects. *J. Environ. Account. Manag.* 2020, *8*, 93–110. [CrossRef]
- 6. Emir, F.; Bekun, F.V. Energy intensity, carbon emissions, renewable energy, and economic growth nexus: New insights from Romania. *Energy Environ.* **2019**, *30*, 427–443. [CrossRef]
- Tariq, M.; Khan, I.; Ali, Q.; Ashfaq, M. The nexus between greenhouse gas emission, electricity production, renewable energy and agriculture in Pakistan. *Renew. Energy* 2018, 118, 437–451. [CrossRef]
- 8. Akram, R.; Chen, F.; Khalid, F.; Ye, Z.; Tariq, M. Heterogeneous effects of energy ef fi ciency and renewable energy on carbon emissions: Evidence from developing countries. *J. Clean. Prod.* **2020**, *247*, 119122. [CrossRef]
- 9. Sarkodie, S.A. Science of the Total Environment Causal effect of environmental factors, economic indicators and domestic material consumption using frequency domain causality test. *Sci. Total Environ.* **2020**, *736*, 139602. [CrossRef]
- Inegbedion, H.; Sunday, E.; Asaleye, A.; Lawal, A.; Adebanji, A. Managing Diversity for Organizational Efficiency. SAGE Open 2020, 10, 2158244019900173. [CrossRef]
- 11. Adedoyin Isola, L.; Frank, A.; Kehinde Leke, B. Can Nigeria Achieve Millennium Development Goals? J. Soc. Sci. Res. 2015, 1, 72–78.
- 12. Okere, W.; Eluyela, D.F.; Lawal, A.I.; Oyebisi, I.; Eseyin, O.; Popoola, O.; Awe, T.E. Foreign expatriates on board and financial performance: A study of listed deposit money banks in Nigeria. *J. Soc. Sci. Res.* **2019**, *5*, 418–423. [CrossRef]
- 13. Lawson, L.A. GHG emissions and fossil energy use as consequences of efforts of improving human well-being in Africa. *J. Environ. Manag.* **2020**, 273, 111136. [CrossRef]

- 14. Isola, A.; Ozturk, I.; Olanipekun, I.O.; John, A. Examining the linkages between electricity consumption and economic growth in African economies. *Energy* **2020**, *208*, 118363. [CrossRef]
- 15. Garba, I.; Bellingham, R. Energy poverty: Estimating the impact of solid cooking fuels on GDP per capita in developing countries—Case of sub-Saharan Africa. *Energy* **2021**, 221, 119770. [CrossRef]
- 16. Wang, J.; Zhang, S.; Zhang, Q. The relationship of renewable energy consumption to financial development and economic growth in China. *Renew. Energy* **2021**, *170*, 897–904. [CrossRef]
- 17. Omri, A.; Belaïd, F. Does renewable energy modulate the negative effect of environmental issues on the socio-economic welfare? *J. Environ. Manag.* **2021**, *278*, 111483.
- Adewuyi, A.O.; Awodumi, O.B. Biomass energy consumption, economic growth and carbon emissions: Fresh evidence from West Africa using a simultaneous equation model United States of America. *Energy* 2017, 119, 453–471. [CrossRef]
- Mensah, I.A.; Sun, M.; Gao, C.; Omari-Sasu, A.Y.; Zhu, D.; Ampimah, B.C.; Quarcoo, A. Analysis on the nexus of economic growth, fossil fuel energy consumption, CO₂ emissions and oil price in Africa based on a PMG panel ARDL approach. *J. Clean. Prod.* 2019, 228, 161–174. [CrossRef]
- Udemba, E.N.; Güngör, H.; Bekun, F.V.; Kirikkaleli, D. Economic performance of India amidst high CO₂ emissions. Sustain. Prod. Consum. 2020, 27, 52–60. [CrossRef]
- Khan, Z.; Malik, M.Y.; Latif, K.; Jiao, Z. Heterogeneous effect of eco-innovation and human capital on renewable & non-renewable energy consumption: Disaggregate analysis for G-7 countries. *Energy* 2020, 209, 118405. [CrossRef]
- Hilfa, N.; Mohamad, A.; Fakhzan, N.; Khalid, N.; Helmi, M. Resources, Conservation & Recycling Effects of agriculture, renewable energy, and economic growth on carbon dioxide emissions: Evidence of the environmental Kuznets curve. *Resour. Conserv. Recycl.* 2020, 160, 104879. [CrossRef]
- 23. Jin, T.; Kim, J. What is better for mitigating carbon emissions—Renewable energy or nuclear energy? A panel data analysis. *Renew. Sustain. Energy Rev.* **2018**, *91*, 464–471. [CrossRef]
- 24. Vidanagama, J.; Lokupitiya, E. Author's Accepted Manuscript Emissions Associated with Tea and to Appear in: Environmental Development. *Environ. Dev.* **2018**. [CrossRef]
- 25. Muhammad, B. Energy consumption, CO₂ emissions and economic growth in developed, emerging and Middle East and North Africa countries. *Energy* **2019**, *179*, 232–245. [CrossRef]
- 26. Hamit-Haggar, M. Greenhouse gas emissions, energy consumption and economic growth: A panel cointegration analysis from Canadian industrial sector perspective. *Energy Econ.* **2012**, *34*, 358–364. [CrossRef]
- 27. Lawal, A.I.; Nwanji, T.I.; Adama, J.I.; Otekunrin, A.O. Examining the Nigerian stock market efficiency: Empirical evidence from wavelet unit root test approach. *J. Appl. Econ. Sci.* 2017, *in press.*
- Jacques, L.; Keho, Y. Energy consumption, economic growth and carbon emissions: Cointegration and causality evidence from selected African countries. *Energy* 2016, 114, 492–497. [CrossRef]
- 29. Bamisile, O.; Obiora, S.; Huang, Q.; Yimen, N. ScienceDirect Impact of economic development on CO₂ emission in Africa; the role of BEVs and hydrogen production in renewable energy integration. *Int. J. Hydrogen Energy* **2022**, *46*, 2755–2773. [CrossRef]
- Radmehr, R.; Rastegari, S.; Shayanmehr, S. Renewable Energy Consumption, CO₂ Emissions, and Economic Growth Nexus: A Simultaneity Spatial Modeling Analysis of EU Countries. *Struct. Chang. Econ. Dyn.* 2021, 57, 13–27. [CrossRef]
- Usman, M.; Sohail, M.; Makhdum, A.; Kousar, R. Does financial inclusion, renewable and non-renewable energy utilization accelerate ecological footprints and economic growth? Fresh evidence from 15 highest emitting countries. *Sustain. Cities Soc.* 2021, 65, 102590. [CrossRef]
- 32. Shahbaz, M.; Raghutla, C.; Reddy, K.; Jiao, Z.; Vinh, X. The effect of renewable energy consumption on economic growth: Evidence from the renewable energy country attractive index. *Energy* **2020**, 207, 118162. [CrossRef]
- Saidi, K.; Omri, A. The impact of renewable energy on carbon emissions and economic growth in 15 major renewable energyconsuming countries. *Environ. Res.* 2020, 186, 109567. [CrossRef]
- 34. Vural, G. Analyzing the impacts of economic growth, pollution, technological innovation and trade on renewable energy production in selected Latin American countries. *Renew. Energy* **2021**, *171*, 210–216. [CrossRef]
- Wesseh, P.K., Jr.; Lin, B. Can African countries efficiently build their economies on renewable energy? *Renew. Sustain. Energy Rev.* 2016, 54, 161–173. [CrossRef]
- 36. Ozcan, B.; Ozturk, I. Renewable energy consumption-economic growth nexus in emerging countries: A bootstrap panel causality test. *Renew. Sustain. Energy Rev.* **2019**, 104, 30–37. [CrossRef]
- 37. Balsalobre-lorente, D.; Shahbaz, M.; Roubaud, D.; Farhani, S. How economic growth, renewable electricity and natural resources contribute to CO₂ emissions? *Energy Policy* **2018**, *113*, 356–367. [CrossRef]
- 38. Salahuddin, M.; Alam, K.; Ozturk, I.; Sohag, K. The effects of electricity consumption, economic growth, financial development and foreign direct investment on CO₂ emissions in Kuwait. *Renew. Sustain. Energy Rev.* **2018**, *81*, 2002–2010. [CrossRef]
- 39. Foon, C.; Wah, B.; Ozturk, I. Energy consumption and economic growth in Vietnam. *Renew. Sustain. Energy Rev.* 2016, 54, 1506–1514. [CrossRef]
- 40. Azam, M.; Qayyum, A. Testing the Environmental Kuznets Curve hypothesis: A comparative empirical study for low, lower middle, upper middle and high income countries. *Renew. Sustain. Energy Rev.* **2016**, *63*, 556–567. [CrossRef]
- 41. Apergis, N.; Payne, J.E. Energy consumption and economic growth in Central America: Evidence from a panel cointegration and error correction model. *Energy Econ.* **2009**, *31*, 211–216. [CrossRef]

- 42. Cowan, W.N.; Chang, T.; Inglesi-Lotz, R.; Gupta, R. The nexus of electricity consumption, economic growth and CO₂ emissions in the BRICS countries. *Energy Policy* **2014**, *66*, 359–368. [CrossRef]
- 43. Shahbaz, M.; Sarwar, S.; Chen, W.; Nasir, M. Dynamics of electricity consumption, oil price and economic growth: Global perspective. *Energy Policy* **2017**, *108*, 256–270. [CrossRef]
- 44. Al-Mulali, U.; Normee, C.; Che, B. The impact of energy consumption and CO₂ emission on the economic growth and financial development in the Sub Saharan African countries. *Energy* **2012**, *39*, 180–186. [CrossRef]
- 45. Sohrabi, C.; Alsafi, Z.; O'Neill, N.; Khan, M.; Kerwan, A.; Al-Jabir, A.; Iosifidis, C.; Agha, R. World Health Organization declares global emergency: A review of the 2019 novel coronavirus (COVID-19). *Int. J. Surg.* **2020**, *76*, 71–76. [CrossRef] [PubMed]
- 46. Ivanovski, K.; Hailemariam, A.; Smyth, R. The effect of renewable and non-renewable energy consumption on economic growth: Non-parametric evidence. *J. Clean. Prod.* **2021**, *286*, 124956. [CrossRef]
- 47. Waheed, R.; Sarwar, S.; Dignah, A. The role of non-oil exports, tourism and renewable energy to achieve sustainable economic growth: What we learn from the experience of Saudi Arabia. *Struct. Chang. Econ. Dyn.* **2020**, *55*, 49–58. [CrossRef]
- Ma, M.; Velayutham, E. Renewable and non-renewable energy consumption-economic growth nexus: New evidence from South Asia. *Renew. Energy* 2022, 147, 399–408.
- Baz, K.; Xu, D.; Ali, H.; Ali, I.; Khan, I.; Khan, M.M.; Cheng, J. Science of the Total Environment Asymmetric impact of energy consumption and economic growth on ecological footprint: Using asymmetric and nonlinear approach. *Sci. Total Environ.* 2020, 718, 137364. [CrossRef]
- 50. Sun, Q.; Gao, X.; An, H.; Guo, S.; Liu, X.; Wang, Z. International Review of Financial Analysis Which time-frequency domain dominates spillover in the Chinese energy stock market? *Int. Rev. Financ. Anal.* **2021**, *73*, 101641. [CrossRef]
- 51. Ahn, S.; Neary, V.S. Wave energy resource characterization employing joint distributions in frequency-direction-time domain. *Appl. Energy* **2021**, *285*, 116407. [CrossRef]
- 52. Bozoklu, S.; Yilanci, V. Energy consumption and economic growth for selected OECD countries: Further evidence from the Granger causality test in the frequency domain. *Energy Policy* **2013**, *63*, 877–881. [CrossRef]
- 53. Omoju, O.E.; Li, J.; Zhang, J.; Rauf, A.; Sosoo, V.E. Implications of shocks in energy consumption for energy policy in sub-Saharan Africa. *Energy Environ.* **2019**, *31*, 1077–1097. [CrossRef]
- 54. Prince, S.; Adeleye, N. Environmental preservation amidst carbon emissions, energy consumption, and urbanization in selected african countries: Implication for sustainability. *J. Clean. Prod.* **2021**, *285*, 125409. [CrossRef]
- 55. Brini, R. Renewable and non-renewable electricity consumption, economic growth and climate change: Evidence from a panel of selected African countries. *Energy* **2021**, 223, 120064. [CrossRef]
- 56. Ndlovu, V.; Inglesi-Lotz, R. The causal relationship between energy and economic growth through research and development (R&D): The case of BRICS and lessons for South Africa. *Energy* **2020**, *199*, 117428. [CrossRef]
- 57. Akadiri SSaint Bekun, F.V.; Sarkodie, S.A. Contemporaneous interaction between energy consumption, economic growth and environmental sustainability in South Africa: What drives what? *Sci. Total Environ.* **2019**, *686*, 468–475. [CrossRef]
- 58. Moutinho, V.; Madaleno, M.; Elheddad, M. Determinants of the Environmental Kuznets Curve considering economic activity sector diversification in the OPEC countries. *J. Clean. Prod.* 2020, 271, 122642. [CrossRef]
- 59. Khan, H.; Khan, I.; Tien, T. The heterogeneity of renewable energy consumption, carbon emission and financial development in the globe: A panel quantile regression approach. *Energy Rep.* **2020**, *6*, 859–867. [CrossRef]
- Stefan, D.; Catalina, C.; Cristian, S.; Vasile, J. Understanding the multidimensional linkages among renewable energy, pollution, economic growth and urbanization in contemporary economies: Quantitative assessments across different income countries' groups. *Renew. Sustain. Energy Rev.* 2021, 142, 110818. [CrossRef]
- 61. Ma, M.; Benjamin, X.V. The nexus between renewable energy, economic growth, trade, urbanisation and environmental quality: A comparative study for Australia and Canada. *Renew. Energy* **2020**, *155*, 617–627.
- 62. Sharma, R.; Sinha, A.; Kautish, P. Does renewable energy consumption reduce ecological footprint? Evidence from eight developing countries of Asia. J. Clean. Prod. 2021, 285, 124867. [CrossRef]
- 63. Shahzad, U.; Fareed, Z.; Shahzad, F.; Shahzad, K. Investigating the nexus between economic complexity, energy consumption and ecological footprint for the United States: New insights from quantile methods. J. Clean. Prod. **2021**, 279, 123806. [CrossRef]
- 64. Osadume, R.; University, E.O. Impact of economic growth on carbon emissions in selected West African countries, 1980–2019. J. Money Bus. 2021, 1, 8–23. [CrossRef]
- 65. Fatai, F.; Ozturk, I.; Oluwatoyin, M.; Agboola, P.O.; Victor, F. The implications of renewable and non-renewable energy generating in Sub-Saharan Africa: The role of economic policy uncertainties. *Energy Policy* **2021**, *150*, 112115. [CrossRef]
- Yasmeen, H.; Tan, Q.; Zameer, H.; Vinh, X.; Shahbaz, M. Discovering the relationship between natural resources, energy consumption, gross capital formation with economic growth: Can lower financial openness change the curse into blessing. *Resour. Policy* 2021, *71*, 102013. [CrossRef]
- 67. Kouton, J. The asymmetric linkage between energy use and economic growth in selected African countries: Evidence from a nonlinear panel autoregressive distributed lag model. *Energy Econ.* **2019**, *83*, 475–490. [CrossRef]
- 68. Maji, I.K.; Adamu, S. The impact of renewable energy consumption on sectoral environmental quality in Nigeria. *Clean. Environ. Syst.* **2021**, *2*, 100009. [CrossRef]
- 69. Gerard, T.; Hamilton, A.; Kelly, S. Low carbon energy scenarios for sub-Saharan Africa: An input-output analysis on the e ff ects of universal energy access and economic growth. *Energy Policy* **2017**, *105*, 303–319. [CrossRef]

- 70. Adams, S.; Nsiah, C. Science of the Total Environment Reducing carbon dioxide emissions; Does renewable energy matter? *Sci. Total Environ.* **2019**, *693*, 133288. [CrossRef]
- Wang, Q.; Wang, L. Renewable energy consumption and economic growth in OECD countries: A nonlinear panel data analysis. Energy 2020, 207, 118200. [CrossRef]
- 72. Dogan, E.; Altinoz, B.; Madaleno, M.; Taskin, D. The impact of renewable energy consumption to economic growth: A replication and extension of Inglesi-Lotz (2016). *Energy Econ.* **2020**, *90*, 104866. [CrossRef]
- 73. Chica-Olmo, J.; Sari-Hassoun, S.; Moya-Fernández, P. Spatial relationship between economic growth and renewable energy consumption in 26 European countries. *Energy Econ.* **2020**, *92*, 104962. [CrossRef]
- 74. Akram, R.; Chen, F.; Khalid, F.; Huang, G.; Irfan, M. Heterogeneous effects of energy efficiency and renewable energy on economic growth of BRICS countries: A fixed effect panel quantile regression analysis. *Energy* **2021**, *215*, 119019. [CrossRef]
- 75. Vellingiri, B.; Jayaramayya, K.; Iyer, M.; Narayanasamy, A.; Govindasamy, V.; Giridharan, B.; Ganesan, S.; Venugopal, A.; Venkatesan, D.; Ganesan, H.; et al. Science of the Total Environment COVID-19: A promising cure for the global panic. *Sci. Total Environ.* **2020**, *725*, 138277. [CrossRef] [PubMed]
- Ito, K. CO₂ emissions, renewable and non-renewable energy consumption, and economic growth: Evidence from panel data for developing countries. *Int. Econ.* 2017, 151, 1–6. [CrossRef]
- 77. Alvarado, R.; Deng, Q.; Tillaguango, B.; Ahmad, M.; Bravo, D. Do economic development and human capital decrease non-renewable energy consumption? Evidence for OECD countries. *Energy* **2021**, *215*, 119147. [CrossRef]
- Korkut, U.; Emre, A. Investigating the EKC hypothesis with renewable energy consumption, human capital, globalization and trade openness for China: Evidence from augmented ARDL approach with a structural break. *Energy* 2021, 216, 119220. [CrossRef]
- 79. Assi, A.F.; Isiksal, A.Z.; Tursoy, T. Renewable energy consumption, financial development, environmental pollution, and innovations in the ASEAN þ 3 group: Evidence from (P-ARDL) model. *Renew. Energy* **2021**, *165*, 689–700. [CrossRef]
- Sharif, A.; Baris-Tuzemen, O.; Uzuner, G.; Ozturk, I.; Sinha, A. Revisiting the role of renewable and non-renewable energy consumption on Turkey' s ecological footprint: Evidence from Quantile ARDL approach. *Sustain. Cities Soc.* 2020, 57, 102138. [CrossRef]
- 81. Bhattacharya, M.; Awaworyi, S.; Reddy, S. The dynamic impact of renewable energy and institutions on economic output and CO₂ emissions across regions. *Renew. Energy* **2017**, *111*, 157–167. [CrossRef]
- 82. Sharif, A.; Ali, S.; Ozturk, I.; Afshan, S. The dynamic relationship of renewable and nonrenewable energy consumption with carbon emission: A global study with the application of heterogeneous panel estimations. *Renew. Energy* **2019**, *133*, 685–691. [CrossRef]
- 83. Tansel, C.; Topcu, M. Total, renewable and non-renewable energy consumption and economic growth: Revisiting the issue with an asymmetric point of view. *Energy* **2018**, *152*, 64–74. [CrossRef]
- 84. Zhang, L.; Li, Z.; Jia, X.; Tan, R.R.; Wang, F. Targeting carbon emissions mitigation in the transport sector—A case study in Urumqi, China. J. Clean. Prod. 2020, 259, 120811. [CrossRef]
- Aydin, M.; Korkut, U. Are shocks to disaggregated renewable energy consumption permanent or temporary for the USA? Wavelet based unit root test with smooth structural shifts. *Energy* 2020, 207, 118245. [CrossRef]
- 86. Lawal, A.I.; Oseni, E.; Lawal-adedoyin, B.B.; Dicktonye, A.O.; Ogunwole, E.B. Examining rational bubbles in global natural gas prices: Evidence from frequency domain estimates. *Energy Strategy Rev.* **2022**, *44*, 100979. [CrossRef]
- Lawal, A.I.; Salisu, A.A.; Asaleye, A.J.; Oseni, E.; Lawal-Adedoyin, B.B.; Dahunsi, S.O.; Omoju, E.O.; DickTonye, A.O.; Ogunwole, E.B.; Babajide, A.A. Economic Growth, Exchange Rate and Remittance Nexus: Evidence from Africa. J. Risk Financ. Manag. 2022, 15, 235. [CrossRef]
- 88. Mensi, W.; Tiwari, A.K.; Al-Yahyaee, K.H. An analysis of the weak form efficiency, multifractality and long memory of global, regional and European stock markets. *Q. Rev. Econ. Financ.* **2018**, *72*, 168–177. [CrossRef]
- 89. Saberi, A.; Athienitis, A.K. A frequency domain transfer function methodology for thermal characterization and design for energy flexibility of zones with radiant systems. *Renew. Energy* **2021**, *163*, 1033–1045. [CrossRef]
- 90. Breitung, J. Rank Tests for Nonlinear Cointegration. J. Bus. Econ. Stat. 2001, 19, 331–340. [CrossRef]
- 91. Bodart, V.; Candelon, B.; Carpantier, J. Real exchanges rates, commodity prices and structural factors in developing countries. *J. Int. Money* **2015**, *51*, 264–284. [CrossRef]
- 92. Croux, C.; Reusens, P. Do stock prices contain predictive power for the future economic activity? A Granger causality analysis in the frequency domain. *J. Macroecon.* 2013, *35*, 93–103. [CrossRef]
- 93. Gorus, M.S.; Aydin, M. The relationship between energy consumption, economic growth, and CO₂ emission in MENA countries: Causality analysis in the frequency domain. *Energy* **2019**, *168*, 815–822. [CrossRef]
- Aydin, M. Natural gas consumption and economic growth nexus for top 10 natural Gas–Consuming countries: A granger causality analysis in the frequency domain. *Energy* 2018, 165, 179–186. [CrossRef]
- 95. Aydin, M. Renewable and non-renewable electricity consumption e economic growth nexus: Evidence from OECD countries. *Renew. Energy* **2019**, *136*, 599–606. [CrossRef]

- 96. Pesaran, M.H.; Yamagata, T. Testing slope homogeneity in large panels. J. Econ. 2008, 142, 50–93. [CrossRef]
- 97. Adewuyi, A.O.; Awodumi, O.B.; Abodunde, T.T. Analysing the gold-stock nexus using VARMA-BEKK-AGARCH and Quantile regression models: New evidence from South Africa and Nigeria. *Resour. Policy* **2019**, *61*, 348–362. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.





Article Forecasting the CO₂ Emissions at the Global Level: A Multilayer Artificial Neural Network Modelling

Pradyot Ranjan Jena ^{1,*}, Shunsuke Managi ² and Babita Majhi ³

- ¹ School of Management, National Institute of Technology Karnataka, Surathkal, Mangalore 575025, India
- ² Urban Institute & Department of Civil Engineering, Kyushu University, Fukuoka 819-0395, Japan; managi.s@gmail.com
- ³ Department of CSIT, Guru Ghasidas Vishwavidyalaya, Central University, Bilaspur 495009, India; babita.majhi@gmail.com
- * Correspondence: pradyotjena@nitk.edu.in

Abstract: Better accuracy in short-term forecasting is required for intermediate planning for the national target to reduce CO₂ emissions. High stake climate change conventions need accurate predictions of the future emission growth path of the participating countries to make informed decisions. The current study forecasts the CO₂ emissions of the 17 key emitting countries. Unlike previous studies where linear statistical modeling is used to forecast the emissions, we develop a multilayer artificial neural network model to forecast the emissions. This model is a dynamic nonlinear model that helps to obtain optimal weights for the predictors with a high level of prediction accuracy. The model uses the gross domestic product (GDP), urban population ratio, and trade openness, as predictors for CO₂ emissions. We observe an average of 96% prediction accuracy among the 17 countries which is much higher than the accuracy of the previous models. Using the optimal weights and available input data the forecasting of CO_2 emissions is undertaken. The results show that high emitting countries, such as China, India, Iran, Indonesia, and Saudi Arabia are expected to increase their emissions in the near future. Currently, low emitting countries, such as Brazil, South Africa, Turkey, and South Korea will also tread on a high emission growth path. On the other hand, the USA, Japan, UK, France, Italy, Australia, and Canada will continuously reduce their emissions. These findings will help the countries to engage in climate mitigation and adaptation negotiations.

Keywords: CO2 emission; artificial neural network model; forecasting; simulation

1. Introduction

There is wide consensus among scientists and policymakers that global warming as defined by the Intergovernmental Panel on Climate Change (IPCC) should be pegged at 1.5° Celsius above the pre-industrial level of warming in order to maintain environmental sustainability [1]. The threats and risks of climate change have been evident in the form of various extreme climate events, such as tsunamis, glacier melting, rising sea levels, and heating up of the atmospheric temperature. Emissions of greenhouse gases, such as carbon dioxide (CO_2) are the main cause of global warming. The Kyoto protocol and the subsequent Paris climate summit have urged the global North and South to cooperate and bear the responsibility of reducing the CO_2 emissions together on a partnership basis. However, climate politics is often not in sync with all the agreements of the Paris climate deals. Especially, since the United States (US) is not a signatory to the Paris climate accords, the international cooperation sought between the industrialized and industrializing countries is slow. Given this broad context of looming climate change threats and the slow pace of actions on reducing CO_2 emissions by the countries, more scientific research must be undertaken to understand the exact nature of the threats. Knowing the level of CO₂ emissions by the high emitting countries in near future will provide actionable insights on climate policy. Such information will aid in fostering the cooperation talks in the

Citation: Jena, P.R.; Managi, S.; Majhi, B. Forecasting the CO₂ Emissions at the Global Level: A Multilayer Artificial Neural Network Modelling. *Energies* **2021**, *14*, 6336. https://doi.org/10.3390/en14196336

Academic Editor: Fernando Morgado-Dias

Received: 7 August 2021 Accepted: 26 September 2021 Published: 4 October 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). upcoming United Nations (UN) COP26 climate conference from 31 October–12 November 2021 in Glasgow, United Kingdom (UK).

Estimating CO₂ has often been done in the context of a school of thought in research, popularly known as the environmental Kuznets curve (EKC) hypothesis. This hypothesis states that environmental degradation, such as air pollution (CO₂, SO₂, NO₂, and SPM emissions), water pollution, and solid waste generation follow an inverted-U relationship with economic growth [2,3]. During the initial level of a country's economic growth, the environmental pollution increases due to rapid expansion in economic activities, however after a threshold level of income per capita in the country is reached, the environmental quality improves because of a higher share of public funds being devoted to improving the environmental quality [4–6]. Despite the last three decades of empirical research in an attempt to estimate the turning point of this EKC, there has still not been consensus about a global turning point. However, there has been tremendous growth in terms of methodological sophistication to estimate both time-series and panel data available for various environmental pollutants and countries [7–12].

A detailed literature review has been undertaken covering the most recent published papers to present the state-of-the-art advancements in EKC studies. Most of these studies have highlighted the role of renewable energy in reducing CO₂ emissions. Dong et al. [13] examined the dynamic causal links among per capita carbon dioxide (CO_2) emissions, per capita GDP, per capita fossil fuels consumption, per capita nuclear energy consumption, and per capita renewable energy consumption for China. They found that both nuclear energy and renewable energy play important roles in mitigating CO_2 emissions in both the short and long run, while fossil fuels consumption is indeed the dominant reason for promoting CO_2 emissions. They observed that renewable energy has a higher CO_2 mitigating effect than nuclear power. Kim and Park [14] from a study of 30 countries for a period of 2000–2013, suggested that a developed financial market in a country helps deploy more renewable energy and, in turn, can reduce CO_2 emissions. Paramati et al. [15] from panel data of G20 countries show that foreign direct investment (FDI) inflows significantly reduces CO₂ emissions in both developed and developing economies while stock market growth reduces in developed economies. They also found that renewable energy consumption substantially reduces CO₂ emissions and increases economic output across the countries in their panels.

In a study, Li et al. [16] used the data from China and Nigeria from 1991–2014 to derive the energy efficiency measures in the mining and extractive related sectors. Using several econometric time series methods, they concluded that energy efficiency in the mining and extractive-related sector and the circular economy have not translated into CO_2 emission reduction in both countries. However, economic growth, energy use (nonrenewable energy), and clean energy substitution (renewable energy) are essential factors in mitigating CO_2 emissions. Lorente et al. [17] employed a carbon emission function to investigate the relationship between economic growth and CO₂ emissions in five European Union countries, namely, Germany, France, Italy, Spain, and the United Kingdom, for the 1985–2016 period. They found an N-shaped relationship between economic growth and CO_2 emissions in the EU-5 countries. Further, they observed that renewable electricity consumption, natural resources, and energy innovation improve environmental quality. Using a panel of 20 organisations for economic co-operation and development (OECD) nations for the period, 1870 to 2014, Churchill et al. [18] found support for the EKC hypothesis for the panel as a whole with turning points in income per capita that lie between \$18,955 and \$89,540 (in 1990 US\$).

A study by Chen et al. [19] used the Chinese data for the period 1980–2014 and explored the relationships among per capita CO_2 emissions, GDP, renewable and non-renewable energy production, and foreign trade. They found that there is a long-run relationship among those variables. They also found that China does not follow the EKC for CO_2 emissions under the influence of economic growth, non-renewable energy

production, and foreign trade. However, the addition of renewable energy production variables supported the U-shaped EKC hypothesis in the long run.

Using data for 1995–2018, pooled mean group-autoregressive distributed Lag (PMG-ARDL) estimator, and heterogeneous causality tests, Gyamfi et al. [20] failed to confirm an N-shaped EKC in the emerging seven, rather they confirm the existence of an inverted U-shaped EKC in the study countries. They suggested the increased use of renewable energy to mitigate pollutant emissions in these countries. Using the data from a study of BRICS economies for the period of 1980 to 2016, Khattak et al. [21] investigated the complex interaction between innovation, renewable energy consumption, and CO₂ emissions, under the EKC framework. They found that innovation activities have failed to disrupt CO_2 emission in China, India, Russia, and South Africa, except for Brazil. They also showed that renewable energy consumption has mitigated CO_2 emission in the BRICS panel, Russia, India, and China but not in South Africa. Further, except for India and South Africa, they observed the EKC hypothesis in all the BRICS economies. Employing a stochastic impacts by regression on population, affluence, and technology (STIRPAT) framework to the data for the period of 1990–2017 from West Asia and Middle East nations, Kihombo et al. [22] probed the effects of technological innovation, financial development (FD), and economic growth (GDP) on the ecological footprint (EF) controlling for urbanization. They observed that a 1% upsurge in technological innovation decreases EF by 0.01%. However, a 1% rise in FD boosts the level of EF by 0.0016%, inferring that FD stimulates ecological degradation. They also showed the EKC hypothesis in the selected countries.

In India's case, using data for a period of 1990-2015 and several time series econometric models, Kirikkaleli and Adebayo [23] found a long-run cointegration relationship between consumption-based carbon dioxide emissions and its possible determinants. They also found that public-private partnership investment in energy makes a positive contribution to consumption-based carbon dioxide emissions in the long run. Further, public-private partnership investment in energy and renewable energy consumption also significantly causes consumption-based carbon dioxide emissions at different frequency levels in the country. Using annual data from six South Asian economies for a period of 1980-2016 and autoregressive distributed lag (ARDL) regression, Murshed [24] examined the validity of the greenhouse emissions-induced EKC hypothesis, controlling for liquefied petroleum gas (LPG) consumption, FDI inflows, and trade openness. The analysis confirms the authenticity of the EKC hypothesis for Bangladesh, India, Sri Lanka, and Bhutan. They suggested fuel-diversification policies for the government's of these countries. Using the data for a period of 1995–2017 from 34 high-income countries from three continents (Asia, Europe, and America), Khan et al. [25] explained the nexus of GHG emission with tourism, financial development index, energy use, renewable energy, and trade. They observed a country-level reciprocal connection of GHG with financial development in 11 countries, renewable energy in 22 countries, trade openness in five countries, and tourism in 12 countries. Using two-panel data sets of 17 major developing and developed countries as well as six geo-economic regions of the world during 1990–2014, Yao et al. [26] examined the dynamic relationship between renewable energy consumption rate (RER) and the EKC hypothesis. Using several econometric methods, they verified both the EKC and renewable energy Kuznets Curve (RKC) hypotheses, indicating that a 10% rise in RER would lead to a 1.6% carbon emission reduction. Saleem et al. [27] used the data for a period of 1980–2015 from selected Asian countries and employing several econometrics models, found the presence of an EKC hypothesis, where the impact of GDP growth and the square of GDP growth on CO_2 emissions are positive and negative, respectively. They also found that lower-income economies do not support the EKC hypothesis.

Employing the second-generation panel cointegration methodologies and data for 1984–2016, Ahmad et al. [28] analyzed the linkages between natural resources, technological innovations, economic growth, and the resulting ecological footprint in emerging economies. They observed the existence of slope heterogeneity across countries and correlation amongst cross-sectional units. They also found a stable, long-run relationship

between the ecological footprint, natural resources, technological innovations, and economic growth. Another study in India by Usman et al. [29] studied the role of energy consumption and democratic regimes in the environmental degradation function for a period of 1971-2014. Using different time series econometric models, they confirmed the EKC hypothesis and divulged that energy consumption increases environmental degradation both in the long and short run. They suggested prioritizing energy conservation policy to mitigate environmental degradation and spur economic growth. Using data from 25 manufacturing subsectors in 38 countries from 2000 to 2014 and using an endogenous finite mixture model, Yang et al. [30] probed the effect of renewable energy in the EKC relationship. They found that with the growing impact of renewable energy consumption, nearly half of the sample countries and two-thirds of the subsectors have experienced the transformation of the nexus between manufacturing growth and emissions. Bilgili et al. [31] employed the panel quantile regression technique on a dataset from thirteen developed countries over the period 2003–2018 to find an inverted U-shaped nexus between economic growth and carbon emissions only in higher carbon-emitting countries, thus, confirming the EKC hypothesis. However, the U-shaped nexus is more predominant in lower carbon-emitting countries. They also found that energy efficiency research and development is more effective in curbing carbon emissions than fossil fuels and renewable energy research and development.

The literature review shows that significant advancement has taken place in the study of EKC in terms of the methods used. In particular, the dynamic time-series and panel cointegration models with the use of structural breaks have produced credible evidence. However, these dynamic time-series models used mostly the lag length to make the model dynamic and estimate the long-run relationship. Moreover, the time series or panel data estimations produce a single estimated parameter for the relationship within the whole sample period. The long-run relationship between CO₂ emissions and its predictors, such as GDP per capita, renewable energy consumption, and trade openness may not have been linear as the previous studies with statistical methods had tried to estimate. A few of the studies used the structural breaks to account for the major shifts in the environmental regulations and policies that may have affected the long-run relationship, but they finally showed constant estimates in observing the effect of GDP on environmental degradation for the whole time period. If the apparent nonlinearities existing in this relationship over a period of time are considered explicitly, more accurate predictions can be made, which has been done in the current study.

The current study aims to forecast the level of CO_2 emissions for 2017–2019 at the global level. CO_2 emission is the key contributor to climate change and there is a global consensus that the mean global surface temperature must be contained at 1.5 degrees C above the pre-industrial level. Consequently, several countries have signed the Paris agreement to reduce emissions within their national boundaries. Against this backdrop, it is essential to forecast the CO_2 emissions levels in the countries that emit a higher share. Such forecasting will help the national governments to adjust their climate policies.

Forecasting of CO_2 emissions at business as usual (BAU) scenario is a necessary tool for major greenhouse gas emitting countries for two main reasons. First, the global circulation models that are used to assess the physical impacts from climate change needs emissions as inputs. Since the countries included in this study are responsible for 79% of global emissions, forecasts of their emission level in the short run will be essential to gauge the impacts of climate change at the global level. Second, the responsibility to reduce CO_2 emissions as agreed at the Paris climate convention is proportional to the BAU levels of emissions. Hence, accurate prediction of emissions will put the right value of resources that these countries need to commit for the reduction of emissions. Since there is a trade-off between emission levels are not underpredicted. Some of the countries may withdraw from a multilateral climate treaty if they find that they are at an economic disadvantage due to their pledge to reduce emissions. Accurate prediction of the BAU emission levels holds significance for a feasible action plan by the countries to reduce the global CO₂ emissions.

Considering that there might be a nonlinear relationship between the indicators of economic growth and the CO_2 emissions, we develop a multilayer artificial neural network (MLANN) model. A multilayer artificial neural network model is more efficient in capturing the nonlinearity present in the time series data and provides higher accuracy in forecasting the CO_2 emissions based on the past values of the emissions and the economic indicators, such as GDP, population density, and urbanization. Such forecasts for the near future will provide insights into regulations on pollution control.

The contributions of the paper are:

- (i) Formulation of CO₂ emissions prediction as an optimization problem.
- Development and performance evaluation of MLANN based model for prediction of CO₂ emissions.
- (iii) Forecasting of the missing CO_2 emission values for the years 2017–2019.
- (iv) Analysis of the results and their economic impact.

The rest of the paper is organized as follows. Materials and methods are discussed in Section 2. Section 3 deals with the development of a CO_2 prediction model using MLANN. Details of the simulation study are given in Section 4. It also contains data collection and preprocessing, training and testing of the model. Section 5 presents results and discussion. Finally, conclusion of the paper is presented in Section 6.

2. Materials and Methods

We have considered two types of countries—first, countries that emit 2% or more share of global CO_2 emissions and countries that emit less than 2% share. The selection of countries in this study is based on the data compiled by the International Energy Agency (IEA), which estimates carbon dioxide (CO_2) emissions from the combustion of coal, natural gas, oil, and other fuels, including industrial waste and non-renewable municipal waste. The specific data used are reproduced from the website Each Country's Share of CO_2 Emissions | Union of Concerned Scientists (ucsusa.org) and are given below in Figure 1.

Table 1 describes the countries considered in this study under two groups—high emission and low emission countries.

Sl. No.	Group 1 High Emission Countries (with \ge 2% Share)	Group 2 Low Emission Countries (with 1% Share)
1	China (28%)	Brazil (1%)
2	U.S. (15%)	South Africa (1%)
3	India (7%)	Mexico (1%)
4	Japan (3%)	Turkey (1%)
5	Iran (2%)	Australia (1%)
6	South Korea (2%)	United Kingdom (1%)
7	Saudi Arabia (2%)	Italy (1%)
8	Indonesia (2%)	France (1%)
9	Canada (2%)	

Table 1. High and Low emission countries.



Figure 1. Share of CO₂ emissions in high and low emitting countries.

The data on the output parameter, i.e., CO_2 emissions and the input parameters, such as GDP in constant US\$ measured in 2011, trade as a percentage of GDP, and urban population for all the countries are drawn from the World Bank database. The period of the study is 1960 to 2016. The forecasting period is 2017, 2018, and 2019.

Figure 2 shows the GDP (constant in 2010 US\$) for the countries considered in this study, in 1990 and 2016. Although the period of study is from 1960 to 2016, we chose the more recent years to compare the growth of the GDP. The countries shown in the *X*-axis are ordered from the highest emission status to the lowest among the 17 countries. The *Y*-axis shows the cumulative annual growth rate (CAGR) between 1990 and 2016. The countries showing a high growth rate in GDP are expectedly China with a CAGR of 9.5%, India with 6.2%, Indonesia with 4.8%, and Turkey with 4.4%. the countries that experienced low growth rates are Italy with 0.7%, Japan with 0.96%, and France with 1.56%.



Figure 2. The GDP figures and its growth for the selected countries.

Figure 3 shows the CO₂ emissions of the 17 countries and their CAGR for the period 1990 and 2016. The countries that accounted for the highest growth in CO₂ emissions between 1990 and 2016 are China with 6.17%, India with 5.5%, Saudi Arabia with 5%, Iran with 4.8%, Brazil with 4%, and Turkey with 3.7%. The countries that have managed to rein in their emissions growth are UK with -1.16%, Italy with -1.1%, France with -0.88%, the USA with 0.33%, Japan with 0.41%, and Canada with 0.9%. The growth trends in Figures 2 and 3 suggest that the highly developing countries tend to emit more CO₂ while the already developed countries have slowed down their emissions. This evidence for the period 1990–2016 is close to the assertions of the EKC.

However, future forecasts are needed to convince the developed countries to commit more financial support for the developing countries to motivate the latter to sacrifice some of their economic ambitions. The trade-off that the highly emitting developing countries, such as China, India, and Brazil have to accept to reduce their CO_2 emissions in order to comply with their commitments at the Paris climate summit agreement, is substantial. Unless they receive financial support from the industrialized countries as agreed upon by the Paris climate summit, these countries are unlikely to reduce their emission levels. We attempt to forecast the CO_2 emission levels of 17 countries that account for nearly 79% of the global emissions. By using the highly complex and non-linear artificial neural network (ANN) models that can accurately forecast the future emission values, we provide actionable insights to the policymakers to engage in more active dialogues to achieve the Paris agreement. Using the multilayer ANN model, we forecast the CO_2 emissions for Group 1 and 2 types of countries (Table 1) for 2017–2019.



Figure 3. The CO₂ emission (kt) and its CAGR for the selected countries.

3. Development of the Multilayer Artificial Neural Network (MLANN) Based CO₂ Forecasting Model

Statistical models are not able to estimate the relationships accurately when the data are uncorrelated, non-stationary, nonlinear and chaotic [32]. To overcome this problem various intelligent models are proposed by the researchers. The MLANN is a nonlinear, multi-layered, fully connected feedforward network that can model the nonlinearity of the data appropriately [33]. The MLANN model is trained using past data and optimizes the weights that will be used to forecast the CO₂ emissions based on the inputs given. The flowchart shown in Figure 4 is used for the development of a MLANN based prediction model.

The complete structure of the MLANN based prediction model is given in Figure 5. Let *I*, *JandK* represent the indices for the input, hidden and output layers respectively. Where *I* = the number of inputs, *J* = the number of neurons in the hidden layer, and *K* = the number of neurons at the output layer. In this CO₂ prediction model the output is one value, so for this study the value of K = 1. Let *P* be the number of input patterns and let any *i*th input pattern is given as p_i . Each input pattern is supplied to the MLANN model sequentially, multiplied with the weights, sum together, and finally passed through the nonlinear activation function (*tanh*) to produce the output layer. Let the estimated output of the network is *est*_k. The error value is obtained by comparing the estimated value with the desired value or target value, t_k . The backpropagation learning rule [33] given in Equations (6)–(11) is used to update the weights and bias values of each layer. This process continues until the squared error is minimum. The detailed equations of feed-forward computation and rules to update the weights and bias are discussed below.



Figure 4. Methodology of MLANN based CO₂ prediction model.



Figure 5. A MLANN based CO₂ emission prediction model.
Refereeing to the above figure, the output of the *k*th output neuron est_k is given [33] as:

$$st_k = \tan h(h_k) \tag{1}$$

where

$$h_k = \sum_{j=1}^{J} est \mathbf{1}_j w_{kj} + w_{bk}$$
⁽²⁾

 $est1_i$ = the output obtained at *j*th hidden neuron.

 w_{kj} = weights connecting *j*th hidden neuron and *k*th output neuron.

e

 $w_{bk} =$ bias at *k*th output neuron.

In the same way, the output at neuron of *j*th hidden layer, est_{1j} is computed [33] as—

$$est1_j = \tan h(h_j) \tag{3}$$

where

$$h_{j} = \sum_{i=1}^{I} p_{i} w_{ji} + w_{bj}$$
(4)

 $p_i = i$ th input pattern $w_{ji} = weights$ between ith input and jth hidden neuron $w_{bj} = bias$ at jth hidden neuron

The error value is obtained by comparing the output of the prediction model, est_k with the corresponding target value, t_k . So,

$$e_k = t_k - est_k \tag{5}$$

The weights connecting the neurons of hidden and output layers, w_{ki} are updated [33] by:

$$w_{kj} = w_{kj} + \mu \times \delta_k \times est1_j \tag{6}$$

where

$$\delta_k = e_k \times \frac{\left(1 - est^2_k\right)}{2} \tag{7}$$

 $\mu = learning \ parameter, (0 < \mu < 1)$

The bias weight is updated as:

$$w_{bk} = w_{bk} + \mu \times \delta_k \tag{8}$$

Similarly, the weights connecting the input and the hidden layer neurons, w_{ji} are updated [33] as:

$$w_{ji} = w_{ji} + \mu \times \delta_j \times p_i \tag{9}$$

where

$$\delta_j = \delta_k \times w_{kj} \times \frac{\left(1 - est1_j^2\right)}{2} \tag{10}$$

The updating of bias weight of *jth* neuron in the hidden layer is done [33] as:

$$w_{bj} = w_{bj} + \mu \times \delta_j \tag{11}$$

The Equations (1)–(11) are the key equations used in the development of the MLANN based CO_2 forecasting model.

4. Simulation study

The CO_2 emissions prediction is formulated as an optimization problem. The model is having three inputs and one output. The inputs are fed to the model and the obtained output is compared with the available target value until the squared error is minimum. Matlab 2016 package is used for the simulation of the problem.

(a) Data collection and preprocessing:

The data for 9 countries under Group 1 and 8 countries under Group 2 are collected from 1960 to 2016 till which the comprehensive data are available in the World Bank database. CO_2 emissions are used as the output parameter for which the out-of-sample forecasting has been done. The variables, such as GDP (in constant 2010 US\$), trade ratio, and urban population are used as the inputs in the MLANN model. The data has been preprocessed as the first step in the modeling wherein the data for all the variables have been normalized. During normalization, each value of the four variables is divided by the corresponding maximum value so that all values can lie between 0 and 1. Normalized data generally makes the learning process and the convergence speed faster. If all features do not have similar ranges of values, then the gradients can move to and fro and take a long time before they can attain the global minimum. To circumvent this problem in the learning process, normalization of the data is necessary. Normalization of the data is followed by preparation of training set and testing set. Randomly selected 80% of data are used for training of the model and the rest 20% of data is used for testing of the developed model. Simulation is carried out by varying the ratio of data division (70:30, 80:20, and 90:10) and an 80:20 ratio is selected finally as it gives the best result. Further, the three missing values of CO₂ emission for the year 2017–2019 are calculated using the optimized weights of MLANN based model.

(b) Training of the model:

Out of the total of 57 patterns (1960 to 2016), the training set consists of 46 patterns that are randomly chosen, and the remaining 11 patterns are used for testing of the calibrated model. An input pattern of data consists of the values of trade ratio, urban population, and GDP. The corresponding CO_2 emission value is the target value for the training of the model. A 9:3:1 structure is used for the simulation. It consists of two hidden layers with nine and three neurons respectively. The connecting weights between the layers and the bias weights are randomly initialized to lie between -0.5 to 0.5. The 9:3:1 structure is fixed after doing experiments by varying different structures of MLANN as it gives minimum error value. In each iteration, one input pattern is given to the model, and feedforward processing is done to get the estimated output from the model. Feedforward processing involves summing the weighted inputs, adding the bias weights, and then passing it through the activation function or nonlinear function (tanh). The estimated output is compared with the corresponding target value to obtain the error. The backpropagation (BP) training algorithm is used to update the connecting weights and bias weights. The value of the learning parameter is taken as 0.1. The same process is repeated until all training patterns are exhausted. This completes one experiment. The experiment is repeated until the mean squared error (MSE) is minimized. The MSE value for each experiment is stored and plotted to observe the convergence characteristics. The final value of connecting weights and bias weights are frozen for testing of the developed model.

(c) Testing of the model:

Once the training process is complete, the developed MLANN based model is ready to be used for evaluation. The 20% of the testing patterns are applied to the model sequentially and the estimated output is noted. The estimated output is compared with the corresponding desired value and the mean absolute percentage error (MAPE), mean absolute error (MAE) and root mean square error (RMSE) are tabulated in Tables 2 and 3 which indicate the performance of the model. The MAPE, MAE and RMSE are calculated using Equations (12)–(14). Also, the comparison of the actual and estimated CO_2 values

during testing are plotted and exhibited in Figure 6a–i for Group-1 countries. For Group-2 countries, it is given in Figure 7a–h.

$$MAPE = \frac{1}{n} \sum_{i=1}^{n} \frac{|(t(n) - est(n))|}{t(n)} \times 100$$
(12)

$$MAE = \frac{1}{n} \sum_{i=1}^{n} |(t(n) - est(n))|$$
(13)

$$RMSE = sqrt(\frac{1}{n}\sum_{i=1}^{n} \left(t(n) - est(n)\right)^2)$$
(14)

$$MAPE = \frac{1}{n} \sum_{i=1}^{n} \frac{|(Out_{obs} - Out_{est})|}{Out_{obs}} \times 100$$
 where $t(n) =$ target value
$$est(n) = \text{estimated value}$$
$$n = \text{Number of testing patterns}$$

 Table 2. MAPE, MAE and RMSE values obtained during testing for Group-1 countries.

Sl. No.	Name of Country	No. of Total Data Samples Available	MAPE Values(%)	MAE	RMSE
1	India	57 (1960–2016)	2.9287	0.0198	0.0235
2	China	57(1960–2016)	1.7896	0.0113	0.0150
3	Iran	57(1960–2016)	2.3610	0.0262	0.0277
4	South Korea	57 (1960–2016)	2.4803	0.0244	0.0324
5	Canada	57 (1960–2016)	2.9358	0.0244	0.0277
6	Indonesia	57 (1960–2016)	9.6898	0.0767	0.1077
7	USA	47 (1970–2016)	2.7168	0.0265	0.0308
8	Japan	47(1970–2016)	3.5206	0.0214	0.0264
9	Saudi Arabia	49 (1968–2016)	5.9153	0.0462	0.0535

 Table 3. MAPE, MAE and RMSE values obtained during testing for Group-2 countries.

Sl. No.	Name of Country	No. of Total Samples	MAPE (%)	MAE	RMSE
1	Brazil	57 (1960–2016)	5.3345	0.0330	0.0412
2	South Africa	57 (1960–2016)	2.7524	0.0279	0.0379
3	Mexico	57 (1960–2016)	1.9266	0.0200	0.0224
4	Turkey	57 (1960–2016)	2.1538	0.0162	0.0209
5	Australia	57 (1960–2016)	3.4001	0.0367	0.0417
6	UK	47(1970–2016)	3.5419	0.0410	0.0502
7	Italy	45(1970–2014)	8.8015	0.0653	0.0769
8	France	55(1960–2014)	3.8158	0.0241	0.0333



Figure 6. Cont.





Figure 6. Comparison of the actual and estimated value of CO_2 emissions using the MLANN for Group-1 countries during the testing of the model.



Figure 7. Cont.



Figure 7. Comparison of the actual and estimated value of CO₂ emission using MLANN for Group-2 countries during testing.

5. Results and Discussion

From Table 2 it is observed that the MAPE values for Group-1 countries lie between 1.78 to 3.52% except for Indonesia and Saudi Arabia. The MAPE value is 5.91 for Saudi Arabia and 9.68 for Indonesia. The MAPE is an indicator of how close the predicted values are to the actual values. The RMSE values lie between 0.01 to 0.05 for Group I countries except for Indonesia. The MAPE values lie between 0.01 to 0.07 for Group I countries. The MAPE values for the Group-2 countries are given in Table 3 which shows that the values lie between 1.92 and 3.8 except for Brazil and Italy. The value is 5.33 for Brazil and 8.08 for Italy. The RMSE values lie between 0.02 to 0.07 and the MAE values lie between 0.01 to 0.06 for Group II countries. As the MAPE values are less than 4% for most of the countries considered in this study, the MLANN model is able to predict the values reasonably accurately with less percentage of error except for a few cases. The comparison of actual and estimated CO₂ values obtained during testing is shown in Figures 6 and 7 for Group-1 and Group-2 countries respectively. In most cases, the actual and estimated values are close to each other.

However, the gap between the actual and predicted values of CO_2 emissions found during the testing phase of the model for Indonesia, Saudi Arabia, Brazil, and Italy is due to the wide fluctuations observed in their emissions data during the period of the study. Although the MLANN model developed in this study is robust to the nonlinearities in the data, wide fluctuations may still increase the percentage of error as is the case for these four countries.

The simulation study is carried out by varying the ANN structure. Different combinations of hidden layer and neurons are used to simulate the model and the results in terms of the training and testing times, as well as the performance achieved, are obtained and displayed in Table S1 in Supplementary Materials. For each country data, initially, combinations of one hidden layer where five, six, seven and eight neurons are used, and thereafter two hidden layers with the same variations of neurons are used for simulation. From Table S1 in Supplementary Materials, it is exhibited that comparing the training time, testing time, MSE in training, and MAPE in testing, the proposed structure of the MLANN model is better in comparison to other combinations of hidden layer and neurons. Further, the simulation is also carried out with different data division ratios and it is observed from Tables S2 and S3 in Supplementary Materials, that the 80–20% ratio is suitable for the proposed study as it gives the minimum MAPE value in all cases.

As suggested in Wu et al. [34], other machine learning methods, such as the support vector machine (SVM) model is simulated and the resultant MAPE values are provided in Table S4 in Supplementary Materials. It is observed that the MAPE values of all countries of Group-I and Group-II are higher in comparison to the proposed MLANN model. We have not added the methods of SVM and a detailed comparison between MLANN and SVM in the main text since it will require substantial expansion of the manuscript.

Forecasting of CO₂ Emissions

In this section, we present the forecasted values of CO₂ emissions for the Group-1 and Group-2 countries for the years 2017, 2018, and 2019 given in Tables 4 and 5 respectively. These are out-of-sample forecasts of CO_2 emissions based on the optimized weights from the calibrated MLANN model and the values for inputs, such as GDP (in 2010 constant US\$), urban population, and trade ratio for 2017, 2018, and 2019. The data of CO₂ emissions for these years are not available, however, the data for inputs for these three years are available for most of the countries considered in this study except for Iran, the USA, and Japan. For Iran, input data is available only for 2017, and for the USA and Japan, it is available for 2017 and 2018. Accordingly, the forecasts are done for these countries for the years the input data are available. The EKC hypothesis stands on the empirical evidence that the elasticity of income effect is larger than the combined elasticities of scale and composition effects [35,36]. The literature review in this study has discussed many recent articles that have either established the EKC relationship in the long run or failed to find evidence for it. A few other studies have used a similar framework as EKC to forecast the out-of-sample values of CO_2 emissions [37]. Aufhammer and Carson forecasted the CO_2 emissions for the Chinese provinces for the single year of 2010 by using the estimated coefficient values of different predictors of their 'best' model and the projected values of the predictors, such as GDP per capita and population figures whose values were unknown when they published this study. Two other noteworthy studies by [38] and [39] have used a similar approach and forecasted the time path of CO₂ emissions for the year 2100 and 2050 respectively. We improve upon these studies in two ways. First, we develop a sophisticated neural network nonlinear model to calibrate the EKC relationship and obtain the optimized input weights that are used to predict the CO₂ emissions based on the predictors, such as GDP, urban population, and trade ratios. These optimized weights provide a more realistic time-series relationship between the emissions and the predictors. Secondly, we forecast the CO₂ emissions for high emitting and low emitting countries based on the known values of the predictors, not their projected values.

Sl. No.	Name of Country	Year	CO ₂ Emission Values (in kt)
		2017	$2.3775 imes 10^{6}$
1	India	2018	2.3858×10^{6}
		2019	$2.3913 imes10^6$
		2017	1.0275×10^{7}
2	China	2018	1.0279×10^{7}
		2019	$1.0281 imes 10^7$
3	Iran	2017	$6.4003 imes10^5$
		2017	$6.1217 imes10^5$
4	South Korea	2018	$6.1490 imes 10^{5}$
		2019	$6.1616 imes 10^5$
		2017	$5.5083 imes10^5$
5	Canada	2018	$5.5332 imes 10^{5}$
		2019	$5.5369 imes10^5$
		2017	$4.9101 imes10^5$
6	Indonesia	2018	$4.9581 imes 10^{5}$
		2019	$4.9984 imes 10^{5}$
7		2017	$5.0148 imes10^6$
7	USA	2018	$4.8022 imes 10^6$
0	Ianan	2017	$1.2187 imes 10^{6}$
8	Japan	2018	$1.2115 imes 10^{6}$
		2017	$5.9955 imes 10^5$
9	Saudi Arabia	2018	5.9752×10^{5}
		2019	$6.0329 imes 10^5$

Table 4. Forecasted CO_2 emission values for the year 2017–2019 for Group-1 countries.

Table 5. Predicted CO_2 emission values for the year 2017–2019 for Group-2 countries.

Sl. No.	Name of Country	Year	CO ₂ Values (in kt)
		2017	$4.5462 imes 10^5$
1	Brazil	2018	$4.7165 imes 10^5$
		2019	$4.7525 imes10^5$
		2017	$4.8368 imes10^5$
2	South Africa	2018	$4.8349 imes 10^5$
		2019	$4.8327 imes10^5$
		2017	4.9500×10^{5}
3	Mexico	2018	4.9521×10^{5}
		2019	$4.9517 imes10^5$
		2017	$3.6217 imes 10^{5}$
4	Turkey	2018	$3.6326 imes 10^{5}$
		2019	$3.6368 imes10^5$
		2017	$3.9027 imes 10^{5}$
5	Australia	2018	$3.9094 imes 10^{5}$
		2019	$3.9122 imes 10^5$
		2017	3.3021×10^{5}
6	UK	2018	2.9951×10^{5}
		2019	$2.5307 imes 10^5$

Sl. No.	Name of Country	Year	CO ₂ Values (in kt)
7		2015	3.8895×10^5
		2016	$3.9707 imes 10^{5}$
	Italy	2017	$3.9028 imes 10^5$
	-	2018	$3.8698 imes 10^{5}$
		2019	3.8842×10^{5}
		2015	3.1387×10^{5}
		2016	3.1091×10^{5}
8	France	2017	$2.9378 imes 10^{5}$
		2018	$2.7934 imes 10^{5}$
		2019	$2.7095 imes 10^{5}$

Table 5. Cont.

Figure 8a,b depict the CO₂ emission values for the Group-1 countries from 2010 to 2019. The emissions from 2010 to 2016 are the actual data obtained from the World Bank database, whereas the values from 2017 to 2019 are the forecasted values. Figure 8a shows that the forecasted emissions for both China and India have increased. China surpassed the USA in 2005 and since then the rate of emission growth is substantially higher for China. During the same period of 2005–2019, the USA's emissions levels have dipped and during the short period of 2017–2019, it shows a declining trend. This is a noteworthy observation in the context of international climate negotiations. Although the USA is not a signatory to Paris climate agreements, it has its internal pollution regulation mandates that have yielded a reduction in CO_2 emissions. On the other hand, China has taken great strides in transforming its economic structure following a circular economy model [40]. Despite these reforms, the emission levels are expected to rise in the short-run horizon. China's past high emission levels and the high growth rate in emission will render it a high emitting country in the near future despite its significant improvement in restructuring the economic models. India is the third highest CO_2 emitter in the world and the rate of emission growth shows a rising trend for the country. The forecasted values for 2017–2019 signify the uphill challenge India is facing to comply with its commitments towards Paris agreements as the emission levels are expected to rise during this period. Japan's emission levels are predicted to reduce further following its declining trend that started around 2007.

Figure 8b shows that the trajectory of CO_2 emissions in Indonesia is quite volatile which is the reason for a higher percentage error in our forecasts for Indonesia. The forecasted values for the period 2017–2019 show arising rend for the country. The other countries in the Group-1 category that shows a rising expected level of CO₂ emissions are Iran, South Korea, and Saudi Arabia. Whereas Canada's emission levels have been stabilized and it embarked on a declining phase of CO_2 emissions since 2008. Figure 9a shows the CO_2 emission trajectory and the forecasted levels for the Group-2 countries. Although the global share of CO₂ emission in countries, such as Brazil, South Africa, Mexico, and Turkey are either 1% or less than 1%, their expected emission level will rise in the near future. Brazil, in particular, shows a high emission growth path which weakens the country's position in the future global climate summits, such as COP26. The reported burning of large tracts of Amazonian forest in Brazil has been heavily criticized by the rest of the globe. The country needs to be more proactive and engaged in complying with its Paris agreement commitments. The expected trajectory of the CO₂ emission growth path for the industrialized countries, such as France, the UK, Australia, and Italy are shown in Figure 9b. The emission levels in France and UK are continuously declining and are expected to decline further. Italy and Australia have reached their peak levels of CO_2 emission in 2006 and 2011 respectively. Since then, their emission levels have stabilized at lower levels and are expected to decline further.



Annual CO2 Emissions of different country trends

(a)



Figure 8. The actual and forecasted CO₂ emissions values for countries: (**a**) China, USA, India, and Japan; (**b**) Canada, Indonesia, Iran, Saudi Arabia, and South Africa.



(a)



Figure 9. The actual and forecasted CO₂ emissions values for countries: (**a**) Brazil, South Africa, Mexico, and Turkey; (**b**) Australia, UK, Italy and France.

6. Conclusions

The IPCC report [41] warns that the current level of national pledges on mitigation of greenhouse gas emissions and adaptation to climate change are not enough to constrain

global warming to the level agreed upon by the countries in the Paris Agreement. The report urges the signatory countries to upscale and accelerates the implementation of multilevel and cross-sectoral climate mitigation actions. To be able to do so, accurate prediction of future CO_2 emission path in business-as-usual conditions holds importance. Such predictions would lead the countries to accelerate their mitigation and adaptation measures. This study forecasts the CO_2 emissions for the high and low emitting countries by their global shares of emission, for the years 2017, 2018, and 2019. Among the high emitting countries, China and India have been treading a high emission growth path, whereas the US and Japan are on the declining trend. Following the EKC hypothesis literature, we model the CO_2 emissions as the output of the model and GDP in constant 2010 US\$, urban population, and trade ratios as the predictors. Several past studies have used the same variables to predict the EKC relationship, however, their methods had been static and mostly linear. Considering that the relationship between CO_2 emissions and its predictors may be nonlinear in the long run, we develop a multilayer artificial neural network model to estimate this relationship.

Based on the World Bank database of 17 countries, of which nine are placed in high emitting (Group-1) and the remaining eight in the low emitting (Group-2) countries spanning from 1960 to 2016, a MLANN model is developed. After the model simulation, it is observed that the prediction accuracy of the in-the-sample data has been 96% leaving 4% to the prediction error. With this high level of prediction accuracy, the model is well calibrated to forecast the out-of-the-sample emission growth path. The data for the input predictors have been available for the years 2017, 2018, and 2019 but not for the CO_2 emissions of the selected countries. Hence, we forecast the CO_2 emissions of these years based on the optimal weights and the input data. From the results, it is observed that China despite its aggressive transformation of economic activities to a circular economy model, is still on the path of increasing emissions in near future. Similarly, India will continue to emit higher levels of CO_2 in the short run that has been studied. Other high emitting countries, such as Iran, Indonesia, Saudi Arabia, and South Korea are expected to continue with their high CO_2 emission growth path if they remain on the BAU economic production-consumption trajectory. These countries need to restructure their economic activities in more sustainable ways to reduce greenhouse gas (GHG) emissions. However, the US and Japan are expected to further reduce their carbon footprint by emitting less CO₂ into the atmosphere. France, UK, Italy, Australia, and Canada are poised to stabilize their emission levels at a low emission growth path and are on course to comply with the Paris agreement. Finally, although low emitting countries, Brazil, South Africa, Turkey, and have been on the rising path of GHG emissions. These countries prioritize their economic growth over the reduction of CO_2 emissions. Hence, they are not expected to comply with the Paris agreement's emission reduction goals.

Based on these results, it is incumbent upon the national policymakers and multilateral policy supporting bodies, such as the UN, OECD, World Bank, and IMF to commit more financial resources for the reduction of CO_2 emissions. Most of the countries that we studied that are on a high emission growth path are currently industrializing. Their goal is to achieve higher economic growth, create more employment, and increase income per capita. Hence, these countries are less likely to change their economic structure suitable for a low carbon economy. The already industrialized countries who have achieved a reduction in their national CO_2 emission goals must come forward to support the countries who are not close to achieving the pledges they made at the Paris climate conference. The next multilateral climate summit which is scheduled to take place in the UK in October-November 2021, known as COP26 will have to focus on issues of greater climate cooperation and finance.

The MLANN model used in the study though has forecasted the CO_2 emission quite accurately in most cases, there are a few cases where the prediction error was high. This is a limitation of the study. Future studies can use other ANN-based models like radial basis function neural network (RBFNN), recurrent neural network (RNN), extreme learning machine (ELM), etc., to reduce the percentage of error. Further, the scope of this study can be expanded by using the mean impact value (MIV) based method to select features and by using the optimal lag order of input data as suggested by Lee and Ou [42] and Wu et al. [43].

Supplementary Materials: The following are available online at https://www.mdpi.com/article/10.3390/en14196336/s1, The Tables S1–S4 are available as Supplementary Materials.

Author Contributions: Conceptualization, P.R.J. and S.M.; methodology, B.M., P.R.J. and S.M.; software, B.M.; validation, B.M., P.R.J. and S.M.; formal analysis, P.R.J. and B.M.; investigation, P.R.J. and S.M.; data curation, B.M.; writing—original draft preparation, P.R.J., B.M. and S.M.; writing—review and editing, P.R.J. and B.M.; supervision, P.R.J.; project administration, P.R.J.; funding acquisition, S.M. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data presented in this study are available in Supplementary Materials.

Acknowledgments: We duly acknowledge the excellent technical support provided by Purna Chandra Tanti, Sunil Khosla, and Loshini to complete this manuscript.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Pachauri, R.K.; Allen, M.R.; Barros, V.R.; Broome, J.; Cramer, W.; Christ, R.; Church, J.A.; Clarke, L.; Dahe, Q.; Dasgupta, P.; et al. Climate Change 2014: Synthesis Report. Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change; Core Writing Team, Pachauri, R.K., Meyer, L.A., Eds.; IPCC: Geneva, Switzerland, 2014; p. 151.
- 2. Grossman, G.M.; Krueger, A.B. Environmental Impacts of a North American Free Trade Agreement. In *The US-Mexico Free Trade Agreement*; Garber, P., Ed.; MIT Press: Cambridge, MA, USA, 1994.
- 3. Copeland, B.R.; Taylor, M.S. Trade, growth, and the environment. J. Econ. Lit. 2004, 42, 7–71. [CrossRef]
- 4. Grossman, G.M.; Krueger, A.B. Economic growth and the environment. Q. J. Econ. 1995, 110, 353–377. [CrossRef]
- 5. Solarin, S.A.; Al-Mulali, U.; Musah, I.; Ozturk, I. Investigating the pollution haven hypothesis in Ghana: An empirical investigation. *Energy* **2017**, *124*, 706–719. [CrossRef]
- 6. Stern, D.I. The rise and fall of the environmental Kuznets curve. World Dev. 2004, 32, 1419–1439. [CrossRef]
- 7. Managi, S.; Jena, P.R. Environmental productivity and Kuznets curve in India. Ecol. Econ. 2008, 65, 432–440. [CrossRef]
- 8. Managi, S.; Hibiki, A.; Tsurumi, T. Does trade openness improve environmental quality? *J. Environ. Econ. Manag.* 2009, *58*, 346–363. [CrossRef]
- 9. Tsurumi, T.; Managi, S. Decomposition of the environmental Kuznets curve: Scale, technique, and composition effects. *Environ. Econ. Policy Stud.* **2010**, *11*, 19–36. [CrossRef]
- 10. Kagohashi, K.; Tsurumi, T.; Managi, S. The effects of international trade on water use. PLoS ONE 2015, 10, e0132133. [CrossRef]
- 11. Sugiawan, Y.; Managi, S. The environmental Kuznets curve in Indonesia: Exploring the potential of renewable energy. *Energy Policy* **2016**, *98*, 187–198. [CrossRef]
- 12. Abe, K.; Ishimura, G.; Surumi, T.; Managi, S.; Sumaila, U.R. Does trade openness reduce domestic fisheries catch? *Fish. Sci.* 2017, *83*, 897–906. [CrossRef]
- 13. Dong, K.; Sun, R.; Jiang, H.; Zeng, X. CO₂ emissions, economic growth, and the environmental Kuznets curve in China: What roles can nuclear energy and renewable energy play? *J. Clean. Prod.* **2018**, *196*, 51–63. [CrossRef]
- 14. Kim, J.; Park, K. Financial development and deployment of renewable energy technologies. *Energy Econ.* 2016, 59, 238–250. [CrossRef]
- 15. Paramati, S.R.; Mo, D.; Gupta, R. The effects of stock market growth and renewable energy use on CO₂ emissions: Evidence from G20 countries. *Energy Econ.* **2017**, *66*, 360–371. [CrossRef]
- 16. Li, G.; Zakari, A.; Tawiah, V. Energy resource melioration and CO₂ emissions in China and Nigeria: Efficiency and trade perspectives. *Resour. Policy* **2020**, *68*, 101769. [CrossRef]
- 17. Balsalobre-Lorente, D.; Shahbaz, M.; Roubaud, D.; Farhani, S. How economic growth, renewable electricity and natural resources contribute to CO₂ emissions? *Energy Policy* **2018**, *113*, 356–367. [CrossRef]
- 18. Churchill, S.A.; Inekwe, J.; Ivanovski, K.; Smyth, R. The environmental Kuznets curve in the OECD: 1870–2014. *Energy Econ.* 2018, 75, 389–399. [CrossRef]

- 19. Chen, Y.; Wang, Z.; Zhong, Z. CO₂ emissions, economic growth, renewable and non-renewable energy production and foreign trade in China. *Renew. Energy* **2019**, *131*, 208–216. [CrossRef]
- 20. Gyamfi, B.A.; Adedoyin, F.F.; Bein, M.A.; Bekun, F.V. Environmental implications of N-shaped environmental Kuznets curve for E7 countries. *Environ. Sci. Pollut. Res.* 2021, *28*, 33072–33082. [CrossRef] [PubMed]
- 21. Khattak, S.I.; Ahmad, M.; Khan, Z.U.; Khan, A. Exploring the impact of innovation, renewable energy consumption, and income on CO2 emissions: New evidence from the BRICS economies. *Environ. Sci. Pollut. Res.* **2020**, *27*, 13866–13881. [CrossRef]
- 22. Kihombo, S.; Ahmed, Z.; Chen, S.; Adebayo, T.S.; Kirikkaleli, D. Linking financial development, economic growth, and ecological footprint: What is the role of technological innovation? *Environ. Sci. Pollut. Res.* **2021**, 1–11. [CrossRef]
- 23. Kirikkaleli, D.; Adebayo, T.S. Do public-private partnerships in energy and renewable energy consumption matter for consumption-based carbon dioxide emissions in India? *Environ. Sci. Pollut. Res.* 2021, 28, 30139–30152. [CrossRef]
- 24. Murshed, M. LPG consumption and environmental Kuznets curve hypothesis in South Asia: A time-series ARDL analysis with multiple structural breaks. *Environ. Sci. Pollut. Res.* 2021, *28*, 8337–8372. [CrossRef]
- 25. Khan, M.T.I.; Yaseen, M.R.; Ali, Q. Nexus between financial development, tourism, renewable energy, and greenhouse gas emission in high-income countries: A continent-wise analysis. *Energy Econ.* **2019**, *83*, 293–310. [CrossRef]
- Yao, S.; Zhang, S.; Zhang, X. Renewable energy, carbon emission and economic growth: A revised environmental Kuznets Curve perspective. J. Clean. Prod. 2019, 235, 1338–1352. [CrossRef]
- 27. Saleem, H.; Khan, M.B.; Shabbir, M.S. The role of financial development, energy demand, and technological change in environmental sustainability agenda: Evidence from selected Asian countries. *Environ. Sci. Pollut. Res.* 2020, 27, 5266–5280. [CrossRef]
- Ahmad, M.; Jiang, P.; Majeed, A.; Umar, M.; Khan, Z.; Muhammad, S. The dynamic impact of natural resources, technological innovations and economic growth on ecological footprint: An advanced panel data estimation. *Resour. Policy* 2020, 69, 101817. [CrossRef]
- 29. Usman, O.; Iorember, P.T.; Olanipekun, I.O. Revisiting the environmental Kuznets curve (EKC) hypothesis in India: The effects of energy consumption and democracy. *Environ. Sci. Pollut. Res.* **2019**, *26*, 13390–13400. [CrossRef] [PubMed]
- 30. Yang, M.; Wang, E.Z.; Hou, Y. The relationship between manufacturing growth and CO2 emissions: Does renewable energy consumption matter? *Energy* 2021, 232, 121032. [CrossRef]
- Bilgili, F.; Kuşkaya, S.; Khan, M.; Awan, A.; Türker, O. The roles of economic growth and health expenditure on CO₂ emissions in selected Asian countries: A quantile regression model approach. *Environ. Sci. Pollut. Res.* 2021, 28, 44949–44972. [CrossRef] [PubMed]
- 32. Teräsvirta, T. Forecasting economic variables with nonlinear models. Handb. Econ. Forecast. 2006, 1, 413–457.
- 33. Haykin, S.S. Neural Networks and Learning Machines; Pearson Education: London, UK, 2009.
- 34. Wu, B.; Wang, L.; Lv, S.-X.; Zeng, U.-R. Effective crude oil price forecasting using new text-based and big-data driven model. *Measurement* **2021**, *168*, 108468. [CrossRef]
- 35. Antweiler, W.; Copeland, B.R.; Taylor, M.S. Is free trade good for the environment? Am. Econ. Rev. 2001, 91, 877–908. [CrossRef]
- 36. Jena, P.R. Does trade liberalization create more pollution? Evidence from a panel regression analysis across the states of India. *Environ. Econ. Policy Stud.* **2018**, *20*, 861–877. [CrossRef]
- Auffhammer, M.; Carson, R.T. Forecasting the path of China's CO₂ emissions using province-level information. *J. Environ. Econ. Manag.* 2008, 55, 229–247. [CrossRef]
- 38. IPCC. Emissions Scenarios; Cambridge University Press: Cambridge, UK, 2000.
- 39. Schmalensee, R.; Stoker, T.M.; Judson, R.A. World carbon dioxide emissions: 1950–2050. *Rev. Econ. Stat.* 1998, 80, 15–27. [CrossRef]
- 40. Geng, Y.; Sarkis, J.; Beischweitz, R. Globalize the circular economy. Nature 2019, 565, 153–155. [CrossRef]
- Field, C.B.; Barros, V.R.; Mastrandrea, M.D.; Mach, K.J.; Abdrabo, M.A.-K.; Adger, N.; Anokhin, Y.A.; Anisimov, O.A.; Arent, D.J.; Barnett, J.; et al. Summary for Policymakers. In *Climate Change* 2021: *The Physical Science Basis*; Cambridge University Press: Cambridge, UK, 2021.
- 42. Lee, C.-Y.; Ou, H.-Y. Induction Motor Multiclass Fault Diagnosis Based on Mean Impact Value and PSO-BPNN. *Symmetry* **2021**, 13, 104. [CrossRef]
- 43. Wu, B.; Wang, L.; Lv, S.-X.; Zeng, U.-R. Forecasting the U.S. oil markets based on social media information during the COVID-19 pandemic. *Energy* **2021**, *226*, 120403. [CrossRef]





Article The Dynamics of US Gasoline Demand and Its Prediction: An Extended Dynamic Model Averaging Approach

Sakar Hasan Hamza * and Qingna Li

School of Mathematics and Statistics, Beijing Institute of Technology, Beijing 100081, China * Correspondence: hassan.sakar.math@gmail.com

Abstract: This study contributes to the body of literature on modeling and predicting gasoline demand by using nonlinear econometric techniques. For this purpose, dynamic model averaging (DMA) and Bayesian model averaging (BMA) combined with Artificial Bee Colony (ABC) are used to forecast gasoline consumption in the United States. The article's independent variables include demographic characteristics, economic activity, income, driving expenditures, automobile price, and road availability for annual data from 1960 to 2020. In the proposed model, not only may the coefficients and elasticity of a predictor of gasoline demand change over time, but other sets of predictors can also emerge at different periods. Moreover, this study aims to automate the process of picking two forgotten variables of the DMA model using the ABC model. Our findings indicate that dynamic model averaging significantly improves forecasting performance when compared to basic benchmark techniques and advanced approaches. Additionally, integrating it with an Artificial Bee Colony (ABC) may result in improved outcomes when time-varying forgetting variables are present. The findings of this research provide policymakers in the fields of energy economics and the environment with helpful tools and information.

Keywords: gasoline demand; dynamic model averaging (DMA); artificial bee colony (ABC); time-varying parameter; dynamic model

1. Introduction

Gasoline demand in the United States has been steadily increasing since the 1990s. In 2019, Approximately 143 billion gallons of gasoline was used in the United States, with the transportation sector accounting for over 70% of the total consumption [1]. The demand increase can be attributed to factors, such as population growth, urbanization, and increased consumer spending on vehicles. Furthermore, the EIA [1] reported that gasoline demand is highly sensitive to changes in economic activity, fuel prices, and weather patterns. For example, during the COVID-19 pandemic in 2020, gasoline demand in the US fell significantly due to reduced economic activity and stay-at-home orders. However, as the economy recovers and restrictions are lifted, demand is expected to increase once again. In addition, the EIA [1] predicts that gasoline demand will continue to rise in the coming years, reaching approximately 151 billion gallons by 2050.

Numerous research has been conducted on the effectiveness of gasoline demand factors and their capacity to forecast. In this context, some previous studies have adopted a direct approach to estimation by examining the demand for car sales [2–5]. Apart from forecasting vehicle sales, research in the area of travel demand has also looked at gasoline use as a response variable when evaluating fuel price elasticities [6–9]. Huo and Wang [5] discovered that pricing and income elasticities in China are based on consumer vehicle stock and projected vehicle sales in China up to 2050 using the FEEI model. Bento et al. [10] conducted similar research for the United States, using a simultaneous equations model for US households and taking into account the new discarded vehicle markets, among other factors. Graham and Glaister [9] conducted a thorough literature review of 113 studies

Citation: Hamza, S.H.; Li, Q. The Dynamics of US Gasoline Demand and Its Prediction: An Extended Dynamic Model Averaging Approach. Energies 2023, 16, 4795. https://doi.org/10.3390/en16124795

Academic Editor: George Halkos

Received: 10 March 2023 Revised: 28 April 2023 Accepted: 4 May 2023 Published: 19 June 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

conducted in the United Kingdom. Goetzke and Vance [11] and Bento et al. [10] found comparable results in terms of fuel consumption's reactivity to fuel prices and in contrast to vehicle mile travel's response to fuel prices. Meanwhile, Oladosu [6] described individual family fuel consumption choices using a vehicle–fuel expenditure allocation model (or AIDS model) for multi-vehicle families in the United States.

In addition, there is a lack of consensus on the estimated coefficients in the studies that have been carried out in the process of modeling and predicting gasoline demand. According to Goetzke and Vance's [11] review of the literature, the average gasoline price elasticity is roughly -0.18, with estimates ranging from -1.01 to 0.01. Thus, the majority of studies interpret these fuel price elasticities as evidence for the existence of a rebound effect, in which the cost savings associated with a reduction in the cost of driving or a gain in fuel efficiency eventually result in an unforeseen rise in fuel consumption. The rebound effect describes a situation in which drivers are presented with lower travel costs (such as falling gas prices) and/or increased fuel efficiency, which unintentionally results in increased fuel consumption and/or vehicle travel. Consequently, in terms of policy, such a phenomenon might result in erroneous calculations and, thus, incorrect interpretations for decision makers. Dimitropoulos et al. [12] conducted a meta-analysis of 74 studies that included 1120 estimates of reported rebound effects and discovered an average rebound effect of 10% to 12%. This unintended impact on driving and fuel consumption habits has significant consequences for the efficacy of policy planning and interventions aimed at reducing emissions and fuel consumption. Overall, studies in the transportation literature employ a variety of methods in terms of model selection, with the majority of scholars being aware of obvious causes for observed discrepancies in the findings. As a result, the effective modeling and forecasting of gasoline demand may offer a critical foundation for policymakers to consider the policy implications of their energy market activities, which is the goal of this paper.

At a minimum, typical forecasting models have two shortcomings: first, numerous studies have shown that predictors change over time, and factors, such as market cycles and macroeconomic policy changes, may result in structural breakdowns in the relationship between fundamental principles and dynamics. Additionally, the effect of each input on the dependent variable changes according to the period and market conditions [13,14]. A model with a static list of predictors may also lose accuracy and consistency over time. Extensive and precise analysis may be performed at any time to pick a model. In other words, if we have *N* predictors, we must evaluate and compare, 2^N models at each time point (the number of subsets of *N* variables that accurately represent all possible combinations and inclusions of *N* variables in the model) with $T \times 2^N$ as the total number of models should be tested throughout T. Therefore, while N and T are large, their analysis is impossible or, at least, difficult.

The accuracy of forecasts has been improved by using model averaging approaches, such as "forecasting combination", in recent research. Both "Bayesian Model Averaging" (BMA) and " forecasting combination" models are characterized by fixed weight values given to models throughout time; however, they do not offer sufficient flexibility to manage the time gap between the contributions of the modeling [15,16]. Therefore, dynamic model selection (DMS) and dynamic model averaging (DMA) were suggested by Raftery et al. [17] to overcome the limitations of the other models. Findings show that macroeconomic forecasting may benefit from this method [18,19]. The appropriateness of each model throughout time is shown in several studies on this subject. The time-varying parameter (TVP) model may employ DMA to compute the average likelihood of each variable being present in the best prediction model. As a more exact definition, one may argue that the average forecast across models is based on an average likelihood of the existence of a variable at time t based on prior knowledge [19–21]. Selecting the optimal prediction model is based on determining which variables have the greatest likelihood of being present in this model, and the model's prediction will be based on this calculation [19]. Although DMS picks a model that comprises variables most likely to be included in forecast models

among those estimated in each period, it does so in a more efficient manner. Inspired by the works of Koop and Korobilis [18] and Bork and Mller [22], Raftery et al. [17] found that the DMA model's forecasting accuracy was 30 percent higher than that of other timeseries approaches, such as AR and OLS regression. A DMA technique is presented by Wei and Cao [23] to predict a housing price increase in Chinese cities. Research shows that DMA is a better forecasting model than BMA, equal-weighted averaging (EW), and information-theoretic modeling. Dong and Yoon [24] employed a DMA approach to explore the global economic drivers that have a large impact on developing Asian stock market returns, notably during the financial crisis. Moreover, other applications for predicting are noteworthy: aggregate equity returns [25], commodity prices [26,27], exchange rates [28,29], Government bond yields' term structure [27], and commodity price volatility and equity return [30].

Therefore, the following is the study's primary contributions: (1) This study aims to estimate and forecast the gasoline demand in the USA using TVP techniques, particularly the DMA approach, which is much more accurate than prior methods. (2) In most investigations, Bayesian TVP is used to estimate the model's parameters [31,32]. Although this approach approximates the generation of model parameters and switching probabilities using two forgetting elements, the inclusion of forgotten factors might be helpful since full Bayesian models may be quite large and time consuming in terms of computational volume. It also assumes that the two factors are constant over time, which is not the case for the single mechanism addressed in the study by Koop and Korobilis [18]. In addition, removing this constraint to reduce the computing cost of the model may lead to an improvement in model prediction accuracy. In this study, we attempt to execute a random process of forgetting factor selection using an algorithm called the ABC. Therefore, another key contribution in this work is to integrate ABC with DMA to improve the forecast accuracy.

The remaining parts of the article are organized as described below. In the second section, a research approach is presented. In Section 3, we provide a summary of both our data and the empirical findings of the forecasting. The conclusion is presented in Section 5.

2. Research Methodology

The DMA technique employed in the study at hand was introduced by Raftery et al. [17]. The following is the standard models for State-Space approaches, namely the Kalman filter:

$$\mathbf{y}_{t} = \mathbf{z}_{t} \boldsymbol{\theta}_{t} + \boldsymbol{\varepsilon}_{t} \tag{1}$$

$$\theta_t = \theta_{t-1} + \mu_t \tag{2}$$

where $\theta_t = \left[\phi_{t-1}, \beta_{t-1}, \gamma_{t-1}, \cdots, \gamma_{t-p} \right]$ denotes a vector of $m \times 1$ coefficients, and $\mu_t \sim N(0, Q_t)$ and $\varepsilon_t \sim N(0, H_t)$ with a mean of zero and variances of Q_t and H_t are normally distributed. y_t denotes a dependent variable, and $z_t = \left[1, x_{t-1}, y_{t-1}, \cdots, y_{t-p} \right]$ denotes a 1 × m vector of variable interruption and intercept estimators depending on the model. As a consequence, the State-Space method is defined as follows, given a subset of K models at a given time:

$$y_t = z_t^{(k)} \theta_t^{(k)} + \varepsilon_t^{(k)}$$
(3)

$$\vartheta_{t+1}^{(k)} = \theta_t^{(k)} + \mu_t^{(k)}$$
(4)

In this equation, $\varepsilon_t^{(k)} \sim N(0, H_t^{(k)})$ and $\mu_t^{(k)} \sim N(0, Q_t^{(k)})$ with $\vartheta_t = (\theta_t^{(1)}, \dots, \theta_t^{(k)})$ reveal which model of K subsets performs best during whatever period. Dynamic model averaging is a technique that permits a distinct model to be estimated at every given moment [19]. Raftery et al. [17] proposed a DMA approach that involves two parameters of α and λ , dubbed the forgetting factors. A recurrence estimate or forecast is feasible based on

the information of conventional filtering when the constants H_t and Q_t are being considered. The following formula serves as the foundation for the Kalman filtering (KF) process:

$$\theta_{t-1} | y^{t-1} \sim N(\hat{\theta}_{t-1}, \sum_{t-1|t-1})$$
 (5)

In Equation (5), the calculation of $\sum_{t-1|t-1}$ and $\hat{\theta}_{t-1}$ is performed using a conventional approach that is a function of H_t and Q_t , and then the KF process is performed using the following equation:

$$\theta_t \Big| y^{t-1} \sim N(\hat{\theta}_{t-1}, \sum_{t|t-1}) \tag{6}$$

Since $\sum_{t|t-1} = \sum_{t-1|t-1} + Q_t$, to simplify, Raftery et al. [17] substituted $\sum_{t|t-1} = \frac{1}{\lambda_{t|t-1}} \sum_{t-1|t-1} with \sum_{t|t-1} = \sum_{t-1|t-1} + Q_t$, accordingly with $0 < \lambda \leq 1$, $Q_t = (1 - \lambda_{t|t-1}^{-1}) \sum_{t-1|t-1}$. The value of λ_t that is near to one suggests that the coefficients change more gradually. Raftery et al. [17] awarded it a value of 0.99 for the last five years' quarterly statistical data; the preceding figure shows that the observations from the previous five years account for 80 percent of the most current observation. If it is 95%, it indicates that the most recent five years of data accounted for 35% of the weight of the earlier observation. As a result, it is critical to choose the forgetting factors, which are often believed to be between 95 and 99 percent. The estimate in the model will be completed by using updated estimators using the following functions:

$$\lambda_{t|t} = \lambda_{t-1|t-1}$$

$$\theta_t | y^t \sim N(\hat{\theta}_t, \sum_{t|t})$$
(7)

In which

$$\hat{\theta}_{t} = \hat{\theta}_{t-1} + \sum_{t|t-1} z_{t} \Big(H_{t} + z_{t} \sum_{t|t-1} z_{t}' \Big)^{-1} (y_{t} - z_{t} \hat{\theta}_{t-1})$$
(8)

$$\sum_{t|t} = \sum_{t|t-1} -\sum_{t|t-1} z_t \left(H_t + z_t \sum_{t|t-1} z_t' \right)^{-1} z_t \sum_{t|t-1}$$
(9)

Recursive prediction operates based on the predictive distribution in the following manner:

$$y_{t}|y^{t-1} \sim N\left(z_{t}\hat{\theta}_{t-1}, H_{t} + z_{t}\sum_{t|t-1} z_{t}'\right)$$
 (10)

Depending on the model, the above-mentioned functions for k may be expressed as follows, whereas the KF in the fixed estimators' model can be represented as (5)–(7), using ϑ_t as a vector of all parameters (3) and (4).

$$\vartheta_{t-1} \Big| L_{t-1} = k, y^{t-1} \sim N(\hat{\theta}_{t-1}^{(k)}, \sum_{t-1|t-1}^{(k)})$$
(11)

$$\vartheta_t \Big| L_t = k, y^{t-1} \sim N(\hat{\theta}_{t-1}^{(k)}, \sum_{t|t-1}^{(k)})$$
 (12)

$$\vartheta_t \Big| L_t = k, y^t \sim N(\hat{\theta}_t^{(k)}, \sum_{t|t}^{(k)})$$
(13)

The value of $\hat{\theta}_{t}^{(k)}$ and $(\sum_{t|t-1}^{(k)})$ and $(\sum_{t|t-1}^{(k)})$ was acquired with the use of KF and Equations (8) and (9) and $\sum_{t|t-1} = \frac{1}{\lambda_{t|t-1}} \sum_{t-1|t-1}$. We employed the Raftery et al. [17] technique, which incorporates a forgetting factor termed α for state equations in various estimating models, and so the aforementioned components are analogous to the forgetting

factor. Equation (4) is the starting point for the Kalman filter's application. When DMA is utilized, similar effects are obtained:

$$P(\vartheta_{t-1} | y^{t-1}) = \sum_{k=1}^{K} p(\theta_{t-1}^{(k)} | L_{t-1} = k, y^{t-1}) Pr(L_{t-1} = k | y^{t-1})$$
(14)

The model's prediction function was replaced by the following equation introduced by Raftery et al. [17].

$$\pi_{t \mid t-1,k} = \frac{\pi_{t-1 \mid t-1,k}^{\omega_{t \mid t-1}}}{\sum_{l=1}^{K} \pi_{t-1 \mid t-1,l}^{\alpha_{t \mid t-1}}}$$
(15)

If $0 \le \alpha < 1$, the interpretation will be identical to that of λ , resulting in the following updated function:

$$\pi_{t \mid t,k} = \frac{\pi_{t \mid t-1,k}^{\alpha_{t \mid t-1}} p_k(y_t \mid y^{t-1})}{\sum_{l=1}^{K} \pi_{t \mid t-1,l}^{\alpha_{t \mid t-1}} p_l(y_t \mid y^{t-1})}$$
(16)

$$\alpha_{t|t} = \alpha_{t-1|t-1}$$

where $p_l(y_t|y^{t-1})$ indicates the predictive density in terms of y. The weighted mean may be applied to the predictive outputs of each model by using $\pi_{t \mid t-1,k}$ to perform recursive prediction on those outputs. As a result, the DMA point prediction is as follows:

$$E(\mathbf{y}_{t} | \mathbf{y}^{t-1}) = \sum_{k=1}^{K} \pi_{t|t-1,k} \mathbf{z}_{t}^{(k)} \hat{\theta}_{t-1}^{(k)}$$
(17)

DMS operates in such a manner that it picks the model with the greatest quantity of $\pi_{t \mid t-1,k}$ at any point in time. When α equals 0.99, the effectiveness of the previous 5 periods will account for 80% of the weighting for the current time. When α equals 0.99, 80 percent of the weighting for the current period will be determined by the performance of the preceding five periods. When α equals one, $\pi_{t \mid t-1,k}$ is precisely determined using the BMA model. Moreover, when λ equals one, BMA uses a traditional linear prediction model with constant coefficients.

Additionally, the suggested model's recursive estimation will begin with past values for $\pi_{0|0,k}$ and $\theta_0^{(k)}$:

$$E(\mathbf{y}_{t} | \mathbf{y}^{t}) = \sum_{k=1}^{K} \pi_{t|t,k} \mathbf{z}_{t}^{(k)} \hat{\theta}_{t-1}^{(k)}$$
(18)

After calculating the equations, period t information is used to update the values. As previously stated, the purpose of including forgotten components is to minimize computational volume, as employing comprehensive Bayesian models may significantly increase computational volume. On the other hand, the sole process provided by Koop and Korobilis [18] is the manual selection of random values, which cannot result in plain values and also presupposes that the two parameters remain constant throughout time. In this work, we attempted to randomize the process for the selection of forgetting factors, α , and λ , using the ABC method. This approach is designed to decrease the sum of squared errors, which indicates the difference between computed and observed data. The mathematical expression is as follows:

Minimize
$$\mathbf{e}_t = (\mathbf{y}_t - \mathbf{E}(\mathbf{y}_t | \mathbf{y}^t))^2$$

The following is the pseudocode of the algorithm's implementation procedure:

• Step 1: Choose a curve fitting function. Equations (16) and (18) may be combined to create the following function:

$$E(y_{t}|y^{t}) = \sum_{k=1}^{K} \frac{\pi_{t|t-1,k}^{\alpha_{t|t}} p_{k}(y_{t}|y^{t-1})}{\sum_{l=1}^{K} \pi_{t|t-1,l}^{\alpha_{t|t}} p_{l}(y_{t}|y^{t-1})} z_{t}^{(k)} \hat{\theta}_{t-1}^{(k)}$$
(19)

whereas recursive prediction operates using predictive distributions in the following manner:

$$y_t | y^{t-1} \sim N \bigg(z_t \hat{\theta}_{t-1}, H_t + z_t \frac{1}{\lambda_{t|t}} \sum_{t-1|t-1} z_t' \bigg)$$

- Step 2. Arrange the greatest quantity of repetitions (MNC), the total number of bees (N), and LIMIT.
- Step 3. Create random numbers for all bees for whom the optimization procedure begins with a preliminary estimate of their food supply, s, source using Equation (21).

$${}_{s}w_{j}^{new}{}_{s} = w_{j}^{low} + \gamma(w_{j}^{up} - w_{j}^{low}), \frac{w^{low} \le w_{j} \le w^{up}}{s = 1, \dots, SN}$$

$$(20)$$

where SN represents the food supply in total. w_j^{up} and w_j^{low} represent the top and lower limits of the j – th design variable, while γ is a random real value between zero and one.

- Step 4. Calculate the objected function for all bees using Equation (19).
- Step 5. Select fifty percent of the finest feeding places and appoint the bee who frequented these areas as the engaged bee.
- Step 6. Set cycle = 1.
 - Step 7. Traverse each source of food (i = 1, ..., SN)
 - (a) Create new options for an employed bee using the following equation, where a new candidate food source (sw^{new}_j) is identified using two prior food source locations remembered by an employed bee (sw^{old}_j) and a randomly chosen neighborhood of a food source (sw^{old}_k):

$${}_{s}w_{j}^{new} = {}_{s}w_{j}^{old} + \phi({}_{s}w_{j}^{old} - {}_{s}w_{k}^{old})$$

$$\tag{21}$$

- (b) The old superscript displays the value of the preceding iteration's design variable, but the new superscript displays existing design variables, where φ is a random positive integer between -1 and 1. k is a number that is chosen at random and is not equal to s.
- (c) Select the ideal dietary intake for each food source. The new place becomes the food source if there are more food sources there than there were at the previous location; otherwise, the previous location remains the food source.
- Step 8. Estimate probability (p_i) using the following equation:

$$p_i = \frac{\emptyset_i}{\sum_{i=1}^{SN} \emptyset_i}$$

where \emptyset_i represents a measure of the solution's fitness i, as determined by the employed bee. This corresponds to the nectar content in the food supply at location i. Step 9. Traverse each source of food (i = 1, ..., SN).

- (a) Employ unemployed bees.
 - (b) Utilizing Equation (21), develop novel employment strategies for jobless bees.
 - (c) Check to see whether the amount of food sources has improved. If there is
 - a considerable change, the observer bee will be promoted to the hired bee

position; if there is no change, the candidate food source that the observer bee visited will not be selected.

- Step 10. When the best food spot has not improved after a certain number of cycles (LIMIT), the hired bee switches to scout mode and uses Equation (20) to look for a new food source.
- Step 11. cycle= 1 + cycle.
- Step 12. Stop the operation if the cycle is \geq MNC; otherwise, go on to Step.

Another objective of this study aimed to compare the effectiveness of various prediction methods. The Mean Absolute Forecast Error (MAFE) and the Mean Squared Forecast Error (MSFE) are employed as standard indices in this research.

$$MSFE = \frac{\sum_{\tau=\tau_0}^{T} [y_{\tau} - E(y_{\tau} | Data_{\tau-h})]^2}{T - \tau_0 + 1}$$
(22)

$$MAFE = \frac{\sum_{\tau=\tau_0+1}^{T} |y_{\tau} - E(y_{\tau}| Data_{\tau-h})|}{T - \tau_0 + 1}$$
(23)

where $\text{Data}_{\tau-h}$ is the data that were obtained from the time $\tau - h$, h is the horizon for time prediction, and $E(y_{\tau}|\text{Data}_{\tau-h})$ is the forecast point of y_{τ} . This study begins with the results of DMA and DMS, followed by the events that determine which variables are most suited for predicting the gasoline demand function. Then, the performance of DMS and DMA is contrasted. In addition, it assesses the sensitivity of models and prediction results concerning the selection of forgetting factors.

3. The Estimated Model and Data

Annual observations for the United States from 1960 to 2020 were utilized in this analysis. Exogenous variables include measurements of demographic traits, economic activity, income, driving expenses, car pricing, and road availability. These variables in Table 1 are chosen based on an extensive review of the available literature.

Author	Туре	Dep. Variable	Ind. Variable
Hughes et al. [33]	Time series OLS	Fuel demand/capita	Gas price
Wadud et al. [34]	RE panel (quarterly)	Fuel demand	Gas price
Rentziou et al. [35]	SURE panel model (annual)	State VMT	Gas price
Lin & Prince [36]	Dynamic times series	Fuel demand/capita	Gas price
Wang & Chen [37]	SEM (daily)	Household VMT	Gas price
Dillon et al. [38]	SEM (daily)	Household VMT	Gas price
Hymel & Small [39]	Simultaneous equations	State VMT	Fuel cost/mile
Levin et al. [40]	FE panel (daily/monthly)	Fuel demand/capita	Gas price
Dimitropoulos et al. [12]	Lit. review/meta-analysis	Fuel demand & VMT	Gas price
Taiebat et al. [41]	microeconomic model (daily)	Household VMT	Gas price
Gillingham [42]	Lit. review/Lit. survey	US VMT	gas price
Goetzke & Vance [11]	pooled OLS	Household VMT	Gas price
Chakraborty et al. [43]	OLS regression	TOT hh VMT	fuel cost (non-PEV)

Table 1. Research literature for estimating gasoline demand function to determine model variables.

Table 2, below, provides a brief description of the variables included in our analysis, as well as a definition and reference to the source. Moreover, summary statistics for the variables that are used in the empirical analysis are presented in Table 3.

Variable	Definition	Source
GU/POP	$GU/POP = ln(\frac{GU}{population})$ GU = Motor gasoline total end-use consumption	EIA
DPI/POP	$DPI/POP = ln(\frac{DPI}{population})$ DPI = Disposable personal income, Billions of Dollars	FRED
RPR	$RPR/POP = ln(\frac{RPR}{population})$ RPR = Unleaded Regular Gasoline, U.S. City Average Retail Price	EIA
EFE	EFE = ln(FE) FE = All Motor Vehicles Fuel Efficiency (Miles per Gallon)	EIA
LD/POP	$LD/POP = ln(\frac{LD}{population})$ LD = Total Licensed Drivers	FHWA
VR/POP	$\label{eq:VR} \begin{split} VR/POP &= ln(\frac{VR}{population})\\ VR &= Total Motor Vehicle Registrations For All Motor Vehicles \end{split}$	FHWA
ORP/POP	$ORP/POP = ln(\frac{PR}{population})$ PR = Public Road Mileage	FHWA

Table 2. Variables and definitions.

FRED: Federal Reserve Economic Data; https://fred.stlouisfed.org, accessed on 20 June 2022; EIA: Energy Information Administration; https://www.eia.gov/, accessed on 20 June 2022; FHWA: Federal Highway Administration; https://highways.dot.gov/, accessed on 18 June 2022.

Table 3. Descriptive statistics.

	Mean	Median	Maximum	Minimum	Std. Dev.
GU/POP	9.252	9.269	9.408	8.995	0.095
DPI/POP	2.587	2.857	3.970	0.737	1.006
RPR	0.048	0.152	1.293	-1.191	0.783
EFE	2.708	2.797	2.901	2.477	0.158
LD/POP	-0.462	-0.401	-0.359	-0.727	0.112
VR/POP	-0.381	-0.290	-0.173	-0.889	0.208
ORP/POP	9.655	9.655	9.888	9.453	0.142

4. Results

By comparing DMA predictions, we examine forecast performance. We attempted to empirically test several configurations of the ABC model to increase the accuracy of the forecast while achieving the quickest feasible computation speed. In conclusion, the number of bees was fixed at five, the greatest quantity of repetitions at five, and the lower and upper bound between 0.9% and 1%. Finally, we demonstrate the sensitivity of our findings to the choice of forgetting factors, α and λ . We provide findings for prediction horizons of one year (h = 1) and four years (h = 4). A prediction horizon of 4 means that we used the values of the independent variables in the previous 4 periods to predict the dependent variable in the current period. Obviously, with an increase in the prediction horizon, the prediction accuracy of the independent variables decreases. Our models all incorporate an intercept and a single lag between the dependent and independent variables. Experiments with lag lengths up to two revealed that a single lag produces the highest prediction results. Using the ABC approach, we sought to randomize the forgetting components in this study. Thus, our methodology not only provides for the automated determination of the two forgetting elements but also for their evolution over time to minimize the prediction model's inaccuracy. These computations are carried out at a low computational cost. Thus, rather than selecting manually, we use a more precise selection mechanism. Figure 1 illustrates the outcome of estimating the components across time and the prediction horizons one and four. After estimating the model using the combined DMA-ABC model, the chance of each of the model's independent variables being present is supplied. The posterior inclusion probability is shown in Figures 2 and 3. That is, they

quantify the likelihood that a predictor will help predict at time t. They are equivalent to the weights applied using DMA to models that incorporate a predictor. These graphs illustrate which predictors are significant at any given moment in time. These graphs demonstrate that DMS nearly always selects sparse models. These results are compelling evidence of model evolution. DMA has a significant theoretical advantage over other forecasting methodologies in that it enables the forecasting model to evolve. Of course, this gain may be negligible in a given empirical application if the forecasting model does not vary much over time. While the same trend remains true to a lesser degree, it is apparent that there is a significant change over time. That is, the forecasting model's collection of predictors evolves with time. After 1980, practically all surface variables enter the model with varying probability. Intermittent values, of course, provide various outcomes. Between 2000 and 2015, the likelihood of existence, or the initial lag, of the majority of model variables is questioned. In comparison to other variables, vehicle registration has the lowest likelihood of being present, while public road mileage at the level and first log values indicate a high possibility of being included in gasoline demand forecasting.

0.9





Figure 1. α and λ over time (h = 1, h = 4).



Figure 2. Posterior probability (h = 1).



Figure 3. Posterior probability (h = 4).



2006



2010

Figure 4 illustrates the actual and predicted value of gasoline along with the forecast horizons h = 1 and h = 4. The accuracy of the model in estimating gasoline demand is seen in Figure 3. Additionally, expanding the prediction horizon resulted in a decline in the estimated model's accuracy. Our earlier DMA and DMS findings were for our benchmark example, in which we used the ABC technique to determine a random forgetting factor that changes over time. As previously stated, researchers in this area use predetermined values for α and λ . As a consequence, Raftery et al. [17] used $\lambda = \alpha = 0.99$ and suggest that the findings will remain resilient to accepting modifications in these variables. To test these claims of resilience, the results of our forecasting experiment utilizing different combinations of forgetting components are shown in Table 4. MSFE and MAFE values for various models of DMA-ABC, DMS-ABC, DMS, DMA, BMA, TVP-BMA, and TVP are provided in Table 4 for prediction horizons 1 and 4. The results of the comparison of several models in Table 4 indicate that the combined model of DMS and ABC, with the option of automatically acquiring forgetting factors over time, obtains the greatest results in forecasting gasoline demand. According to the DMA-ABC model, the mean values of the forgetting components are equal to $\alpha = 0.9449$ and $\lambda = 0.9662$. Even taking the constant mean values of computational forgetting factors into account produced satisfactory results. It is noteworthy that the value $\alpha = 0.9449$ enables relatively fast model evolution over time. This is similar to a previous tale we mentioned: it seems that allowing models to evolve is more significant than allowing parameters to vary with $\lambda = 0.9662$ for increasing forecast performance. The BME model (with $\lambda = \alpha = 1$) does not have any dynamic approach, which means that while the estimated coefficients are constant over time, the input variables to the model are also constant over time. To investigate the effects of adding dynamics to the model in increasing the forecasting accuracy, we added two more columns to Table 4. In these two columns, the ratio of MAFE and MSFE of different models is calculated with the MAFE and MSFE values of the BMA model (with B index). Based on the results, the prediction error values in model DMS-ABC are about 0.82 of the prediction error in model BMA with a prediction horizon of 1. In addition, this value is equal to 0.76 in the forecast horizon of 4. Therefore, by increasing the prediction horizon, moving towards dynamic models leads to a further increase in prediction accuracy.



Figure 4. The actual and predicted value of gasoline demand in the forecast horizon h = 1 and h = 4 $\alpha = 0.9449$; λ .

Prediction Method	MAFE	MSFE	MAFE MAFE _B	MSFE MSFE _B		MAFE	MSFE	MAFE MAFE _B	MSFE MSFE _B
		h :	= 1				h =	= 4	
DMA-ABC	0.55	3.60	1.01	0.98	DMA-ABC	0.71	4.86	0.96	0.96
DMS-ABC	0.47	3.02	0.86	0.82	DMS-ABC	0.58	3.87	0.78	0.76
DMA $\lambda = 0.9662;$ $\alpha = 0.9449$	0.53	3.60	0.98	0.98	DMA $\lambda = 0.9698;$ $\alpha = 0.9556$	0.71	4.85	0.95	0.96
DMS $\lambda = 0.9662;$ $\alpha = 0.9449$	0.44	3.00	0.81	0.81	DMS $\lambda = 0.9698;$ $\alpha = 0.9556$	0.58	3.87	0.78	0.76
DMA $\lambda = \alpha = 0.99$	0.54	3.65	0.99	0.99	DMA $\lambda = \alpha = 0.99$	0.73	5.03	0.98	0.99
DMS $\lambda = \alpha = 0.99$	0.45	3.04	0.82	0.83	DMS $\lambda = \alpha = 0.99$	0.60	3.98	0.80	0.79
DMA $\lambda = \alpha = 0.95$	0.55	3.57	1.01	0.97	DMA $\lambda = \alpha = 0.95$	0.72	4.98	0.97	0.98
DMS $\lambda = \alpha = 0.95$	0.46	2.99	0.85	0.81	DMS $\lambda = \alpha = 0.95$	0.60	3.98	0.80	0.79
DMA $\lambda = 0.95;$ $\alpha = 0.99$	0.55	3.56	1.01	0.97	DMA $\lambda = 0.95;$ $\alpha = 0.99$	0.72	4.95	0.97	0.98
DMS $\lambda = 0.95;$ $\alpha = 0.99$	0.46	2.99	0.85	0.81	DMS $\lambda = 0.95;$ $\alpha = 0.99$	0.60	3.93	0.80	0.78
DMA $\lambda = 0.99;$ $\alpha = 0.95$	0.54	3.66	0.99	1.00	DMA $\lambda = 0.99$; $\alpha = 0.95$	0.74	5.04	0.99	0.99
DMS $\lambda = 0.99;$ $\alpha = 0.95$	0.45	3.04	0.82	0.83	DMS $\lambda = 0.99$; $\alpha = 0.95$	0.60	3.98	0.80	0.79
TVP- BMA ($\lambda = 1$)	0.55	3.68	1.00	1.00	TVP- BMA ($\lambda = 1$)	0.75	5.08	1.00	1.00
BMA ($\lambda = \alpha = 1$)	0.54	3.68	1.00	1.00	BMA ($\lambda = \alpha = 1$)	0.75	5.07	1.00	1.00

Table 4. Comparison of models.

5. Conclusions and Implications

Accurate modeling and forecasting of gasoline demand may provide a valuable framework for policymakers to consider the policy implications of their energy market activities, which is the goal of this study. The primary shortcoming in prior forecasting models was their inability to accurately predict over time. Policymakers, on the other hand, should disregard short-term and temporary variations in gasoline demand in favor of economic stability. Therefore, the objective of this study was to develop a nonlinear dynamic model DMA-ABS to forecast gasoline consumption in the United States using annual data from 1960 to 2020. These models may be used to determine changes in both the input variables and the parameters of variables through time. The inclusion of two forgetting variables in the DMA model may be used to control the speed of such dynamics in the model, which has been previously determined manually in earlier research. In this work, we sought to implement a random process of forgetting factor selection using the ABC method. Therefore, one of the primary objectives of this study is to merge ABC with DMA to increase prediction accuracy. The findings of the DMS estimation model indicated that the input variables fluctuate with time, emphasizing the need of employing dynamic models rather than constant input variables for estimating gasoline demand.

Gasoline demand prediction helps policymakers to make informed decisions on issues related to energy security, environmental regulations, and transportation infrastructure. For instance, it can assist in determining the number of gas stations required to meet demand in a particular area, the type of fuel to be used in different transportation modes, and the amount of investment needed to maintain or upgrade the transportation infrastructure. Moreover, gasoline demand prediction can aid in managing the price of gasoline. It helps in determining the price level that will meet the demand and supply equilibrium. Therefore, it is recommended that in future research, the DMA model will be integrated with other evolutionary algorithms, such as particle swarm optimization (PSO), genetic algorithm (GA), etc., to compare the results and provide a more accurate prediction of the gasoline market through the expansion of the model presented in this research.

Author Contributions: Conceptualization, S.H.H.; Methodology, S.H.H.; Software, S.H.H.; Formal analysis, S.H.H.; Resources, Q.L.; Writing—original draft, S.H.H. and Q.L.; Writing—review & editing, Q.L.; Visualization, S.H.H.; Supervision, Q.L. All authors have read and agreed to the published version of the manuscript.

Funding: No funding has been received for this research.

Data Availability Statement: Data is available upon request.

Conflicts of Interest: The authors declare no conflict of interest.

References

- EIA. Annual Energy Outlook (AEO) Retrospective Review: Evaluation of AEO 2020 and Previous Reference Case Projections. Anal. Proj. U.S. Energy Inf. Adm. 2020, 92010, 1–15.
- 2. McCarthy, P.S. Market Price and Income Elasticities of New Vehicle Demands. Rev. Econ. Stat. 1996, 78, 543. [CrossRef]
- 3. Shen, C.; Linn, J. *The Effect of Income on Vehicle Demand: Evidence from China'S New Vehicle Market*; Resources for the Future: Washington, DC, USA, 2021.
- Dargay, J.; Gately, D.; Sommer, M. Vehicle ownership and income growth, worldwide: 1960–2030. Energy J. 2007, 28, 143–170. [CrossRef]
- 5. Huo, H.; Wang, M. Modeling Future Vehicle Sales and Stock in China. Energy Policy 2012, 43, 17–29. [CrossRef]
- Oladosu, G. An Almost Ideal Demand System Model of Household Vehicle Fuel Expenditure Allocation in the United States. Energy J. 2003, 24, 1–21. [CrossRef]
- 7. Kim, Y.; Kim, S. Forecasting Charging Demand of Electric Vehicles Using Time-Series Models. Energies 2021, 14, 1487. [CrossRef]
- 8. Fouquet, R. Trends in Income and Price Elasticities of Transport Demand (1850-2010). Energy Policy 2012, 50, 62–71. [CrossRef]
- 9. Graham, D.J.; Glaister, S. Road Traffic Demand Elasticity Estimates: A Review. Transp. Rev. 2004, 24, 261–274. [CrossRef]
- 10. Bento, A.M.; Goulder, L.H.; Jacobsen, M.R.; Von Haefen, R.H. Distributional and Efficiency Impacts of Increased US Gasoline Taxes. *Am. Econ. Rev.* **2009**, *99*, 667–699. [CrossRef]
- 11. Goetzke, F.; Vance, C. An Increasing Gasoline Price Elasticity in the United States? Energy Econ. 2021, 95, 104982. [CrossRef]
- 12. Dimitropoulos, A.; Oueslati, W.; Sintek, C. The Rebound Effect in Road Transport: A Meta-Analysis of Empirical Studies. *Energy Econ.* **2018**, *75*, 163–179. [CrossRef]
- 13. Ghysels, E.; Plazzi, A.; Valkanov, R.; Torous, W. Forecasting Real Estate Prices. Handb. Econ. Forecast. 2013, 2, 509–580. [CrossRef]
- 14. Plakandaras, V.; Gupta, R.; Gogas, P.; Papadimitriou, T. Forecasting the U.S. Real House Price Index. *Econ. Model.* **2015**, 45, 259–267. [CrossRef]
- 15. Próchniak, M.; Witkowski, B. Time Stability of the Beta Convergence among EU Countries: Bayesian Model Averaging Perspective. *Econ. Model.* **2013**, *30*, 322–333. [CrossRef]
- 16. Man, G. Competition and the Growth of Nations: International Evidence from Bayesian Model Averaging. *Econ. Model.* **2015**, *51*, 491–501. [CrossRef]
- 17. Raftery, A.E.; Kárný, M.; Ettler, P. Online Prediction under Model Uncertainty via Dynamic Model Averaging: Application to a Cold Rolling Mill. *Technometrics* **2010**, *52*, 52–66. [CrossRef]
- 18. Koop, G.; Korobilis, D. Forecasting Inflation Using Dynamic Model Averaging. Int. Econ. Rev. 2012, 53, 867–886. [CrossRef]
- 19. Koop, G.; Korobilis, D. UK Macroeconomic Forecasting with Many Predictors: Which Models Forecast Best and When Do They Do So? *Econ. Model.* **2011**, *28*, 2307–2318. [CrossRef]
- Del Negro, M.; Primiceri, G.E. Time Varying Structural Vector Autoregressions and Monetary Policy: A Corrigendum. *Rev. Econ.* Stud. 2010, 82, 1342–1345. [CrossRef]
- Pesaran, M.H.; Timmermann, A. A Recursive Modelling Approach to Predicting UK Stock Returns. *Econ. J.* 2000, 110, 159–191. [CrossRef]
- 22. Bork, L.; Møller, S.V. Forecasting House Prices in the 50 States Using Dynamic Model Averaging and Dynamic Model Selection. *Int. J. Forecast.* **2015**, *31*, 63–78. [CrossRef]
- 23. Wei, Y.; Cao, Y. Forecasting House Prices Using Dynamic Model Averaging Approach: Evidence from China. *Econ. Model.* 2017, 61, 147–155. [CrossRef]
- 24. Dong, X.; Yoon, S.M. What Global Economic Factors Drive Emerging Asian Stock Market Returns? Evidence from a Dynamic Model Averaging Approach. *Econ. Model.* **2019**, *77*, 204–215. [CrossRef]
- 25. Naser, H.; Alaali, F. Can Oil Prices Help Predict US Stock Market Returns? Evidence Using a Dynamic Model Averaging (DMA) Approach. *Empir. Econ.* 2018, *55*, 1757–1777. [CrossRef]
- 26. Drachal, K. Forecasting Spot Oil Price in a Dynamic Model Averaging Framework—Have the Determinants Changed over Time? *Energy Econ.* **2016**, *60*, 35–46. [CrossRef]
- 27. Byrne, J.P.; Cao, S.; Korobilis, D. Forecasting the Term Structure of Government Bond Yields in Unstable Environments. *J. Empir. Financ.* 2017, 44, 209–225. [CrossRef]
- 28. Beckmann, J.; Koop, G.; Korobilis, D.; Schüssler, R.A. Exchange Rate Predictability and Dynamic Bayesian Learning. *J. Appl. Econom.* **2020**, *35*, 410–421. [CrossRef]

- 29. Byrne, J.P.; Korobilis, D.; Ribeiro, P.J. On the Sources of Uncertainty in Exchange Rate Predictability. *Int. Econ. Rev.* 2018, 59, 329–357. [CrossRef]
- 30. Ma, F.; Liu, J.; Wahab, M.I.M.; Zhang, Y. Forecasting the Aggregate Oil Price Volatility in a Data-Rich Environment. *Econ. Model.* **2018**, 72, 320–332. [CrossRef]
- 31. Avramov, D. Stock Return Predictability and Model Uncertainty. J. Financ. Econ. 2002, 64, 423–458. [CrossRef]
- 32. Koop, G.; Potter, S. Forecasting in Dynamic Factor Models Using Bayesian Model Averaging. *Econom. J.* **2004**, *7*, 550–565. [CrossRef]
- 33. Hughes, J.E.; Knittel, C.R.; Sperling, D. Evidence of a Shift in the Short-Run Price Elasticity of Gasoline Demand. *Energy J.* 2008, 29, 113–134. [CrossRef]
- 34. Wadud, Z.; Graham, D.J.; Noland, R.B. Gasoline Demand with Heterogeneity in Household Responses. *Energy J.* **2010**, *31*, 47–74. [CrossRef]
- 35. Rentziou, A.; Gkritza, K.; Souleyrette, R.R. VMT, Energy Consumption, and GHG Emissions Forecasting for Passenger Transportation. *Transp. Res. Part A Policy Pract.* 2012, *46*, 487–500. [CrossRef]
- 36. Lin, C.Y.C.; Prince, L. Gasoline Price Volatility and the Elasticity of Demand for Gasoline. *Energy Econ.* 2013, 38, 111–117. [CrossRef]
- 37. Wang, T.; Chen, C. Impact of Fuel Price on Vehicle Miles Traveled (VMT): Do the Poor Respond in the Same Way as the Rich? *Transportation (Amst)* **2014**, *41*, 91–105. [CrossRef]
- Dillon, H.S.; Saphores, J.D.; Boarnet, M.G. The Impact of Urban Form and Gasoline Prices on Vehicle Usage: Evidence from the 2009 National Household Travel Survey. *Res. Transp. Econ.* 2015, 52, 23–33. [CrossRef]
- 39. Hymel, K.M.; Small, K.A. The Rebound Effect for Automobile Travel: Asymmetric Response to Price Changes and Novel Features of the 2000s. *Energy Econ.* **2015**, *49*, 93–103. [CrossRef]
- 40. Levin, L.; Lewis, M.S.; Wolak, F.A. High Frequency Evidence on the Demand for Gasoline. *Am. Econ. J. Econ. Policy* 2017, *9*, 314–347. [CrossRef]
- Taiebat, M.; Stolper, S.; Xu, M. Corrigendum to "Forecasting the Impact of Connected and Automated Vehicles on Energy Use: A Microeconomic Study of Induced Travel and Energy Rebound". *Appl. Energy* 2019, 256, 297–308. [CrossRef]
- 42. Gillingham, K.T. The rebound effect and the proposed rollback of U.S. fuel economy standards. *Rev. Environ. Econ. Policy* 2020, 14, 136–142. [CrossRef]
- 43. Chakraborty, D.; Hardman, S.; Tal, G. Integrating plug-in electric vehicles (PEVs) into household fleets-factors influencing miles traveled by PEV owners in California. *Travel Behav. Soc.* **2022**, *26*, 67–83. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.





Article Assessing Fossil Fuels and Renewables' Impact on Energy Poverty Conditions in Europe

George Halkos * and Eleni-Christina Gkampoura

* Correspondence: halkos@uth.gr

Abstract: The disadvantages of fossil fuels and their impact on the environment have made the transition to renewable energy sources essential to cover our energy needs. However, different energy resources have a different impact on energy poverty conditions in the world, an issue that is important to examine and properly address. This study examines the impact that fossil fuels final energy consumption in households per capita and renewables and biofuels final energy consumption in households per capita have on energy poverty conditions in Europe, using panel data from 28 European countries for the time period 2004–2019 and static and dynamic regression models, while also performing various econometric tests. The findings indicate that GDP per capita and fossil fuels are linked to an inverse relationship to energy poverty conditions. Renewables and biofuels are also linked to an inverse relationship to the inability to keep homes adequately warm and the presence of leaks, damp, or rot in the dwelling, but they could be considered a driver of arrears on utility bills. In addition, a comparative analysis between Sweden, Germany, and Greece and their conditions on energy poverty and energy transition was conducted, highlighting the differences existing between the three European countries. The findings of the research can be useful for governments and policy makers to develop strategies that promote energy transition while protecting energy consumers.

Keywords: energy poverty; fossil fuels; renewables; Europe

1. Introduction

The discovery and use of fossil fuels can be viewed as the main foundation of humankind's prosperity, growth, and well-being [1]. Coal, oil, and gas have been in the center of the industrialized world since the early 1800s and they constitute the main driving force of economic and social growth in the world [2]. Fossil fuels are still used to cover most of the world's energy needs and this usage is expected to increase more in the future, due to the expected increase in the global population and the new, energy-intense way of life [3].

The disadvantages that emerge from the usage of fossil fuels are many and significant. Fossil fuels are primarily responsible for enormous greenhouse gas emissions into the atmosphere, and they contribute on a great level to global warming, something that could be proven catastrophic for the environment as well as for human health, life, and civilization as we know it [4]. In addition, fossil fuels are finite, and some scientists believe that they might reach their peak soon [1]. Their depletion means that the world should not rely on them anymore and, instead, turn to alternative sources.

All the disadvantages that come as a result from the use and combustion of fossil fuels make it obvious that the world should focus on alternative energy sources, or other energy saving methods and measures for the mitigation of GHG emissions [5]. A transition to renewable energy sources for the satisfaction of our energy needs is considered to be essential to address climate change and achieve the target of limiting the global average temperature increase under 2 °C. Despite the fact that the usage of renewables has increased over past years, fossil fuels still cover around 80% of global energy demands, leading to an urgent need of change in future energy policies [6].

Citation: Halkos, G.; Gkampoura, E.-C. Assessing Fossil Fuels and Renewables' Impact on Energy Poverty Conditions in Europe. *Energies* 2023, *16*, 560. https:// doi.org/10.3390/en16010560

Academic Editor: Idiano D'Adamo

Received: 8 December 2022 Revised: 29 December 2022 Accepted: 29 December 2022 Published: 3 January 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

Department of Economics, University of Thessaly, 38333 Volos, Greece

To achieve a successful full transition to clean energy sources, it is important that energy R&D is promoted on a global level and that innovation, investment, and deployment policies and strategies regarding energy storage technologies are evolved and adopted by policy makers [7]. Cities, which are the biggest energy consumer, should be a part of that transition, promoting a cleaner urban energy system that will lead to cleaner cities [8]. In addition, the transition requires political support and efficient governance, as well as market designs and financial incentives [9]. This transition, unlike other energy transitions that have happened in the past around the world, will have to occur rapidly to successfully tackle climate change. Even though it is supported that this transition is unlikely to happen within the next few decades, governments should promote and support such policies, while international agreements concerning the issue should be implemented [10]. The COVID-19 pandemic that had an important impact on the global economy, has also widened the uncertainty of energy transition [11].

Currently, the world highly depends on fossil fuels. This means that certain features are configured, while they provide stability regarding technological artefacts and scientific knowledge, market structures, practices, and regulatory frameworks [12]. Based on that, a few experts argue that a transition to renewable energy sources would disturb the balance that fossil fuels have provided, and it would challenge the status quo. It is often believed that renewables might not be able to meet the global energy demand, because of their lower energy density, or that society and politicians will not adopt the best scientifically proven alternative that is available. Some experts argue that the implementation of Carbon Capture and Storage (CCS) is the best solution for CO2 emissions mitigation and sustenance of the fossil fuels use at the same time [13], but it is found to be a more expensive and riskier alternative to renewables, while not being carbon neutral [9]. In the literature, some studies have focused on the social and political impacts of renewables [14], on their impact on social welfare [15], as well as on their impact on local development [16]. However, after taking both positive and negative impacts into consideration, most scientists believe that the transition to renewable energy sources is beneficial and is viewed as the best solution, in order to promote sustainability and energy security.

Energy poverty is another energy-related topic that is well-discussed in the recent literature. It refers to a situation where households do not have access to the necessary energy services, which are vital for the satisfaction of basic human needs [17]. A household can be characterized as energy poor when it is lacking sufficient, affordable, and safe energy services [18], something that is not observed in developing countries only; an important number of European households were found to be "fuel poor", especially during the period of the global financial crisis [19].

The linkages between the different types of energy sources and energy poverty, as well as between energy transition and energy poverty, have not been examined extensively in the existing literature. It is, however, a topic that should be addressed and the studies that focus on this matter could be used as a tool by policy makers for the successful implementation of energy-related strategies and policies, which have been in the center of many global initiatives for sustainability. For instance, the 7th Sustainable Development Goal proposed by the United Nations, referring to providing access to affordable and sustainable energy services for all, includes both the topics of energy poverty and energy transition to more sustainable sources [20]. This study aims to contribute and expand the existing knowledge on the topic, examining the relationship between energy poverty conditions in 28 European countries for the years 2004–2019 and energy consumption coming from fossil fuels and renewables in households, using an advanced econometric methodology.

2. Literature Review

A significant number of studies in the current literature have been focusing on the topic of energy poverty. A review of the problem of energy poverty has been presented by Halkos and Gkampoura; the authors included in their review various definitions that have been given to energy poverty in the literature, as well as the impact that energy poverty

has on human health, on the society, the economy, and the environment. In addition, various energy poverty drivers are identified, including household characteristics and other socioeconomic and environmental factors. The authors also analyzed the different approaches to measuring energy poverty and presented the situation currently occurring in different parts of the world regarding the problem. Various actions that could assist in tackling energy poverty are also presented in the review [21].

The main approaches of measuring energy poverty that have been suggested in the literature include the expenditure approach, where metrics are compared to certain thresholds, and the consensual approach, where various indicators that are based on surveys can be used, while composite measurements can also be created [22]. The Multidimensional Energy Poverty Index (MEPI), developed by Nussbaumer et al. [23] is one of these composite indexes, which takes into account the multidimensional nature that energy poverty has, and it was used to estimate energy poverty for certain African countries. Based on the same index, energy poverty was calculated for certain Latin American countries by Santillán et al. [24].

When it comes to European countries, the European Union Statistics on Income and Living Conditions (EU-SILC) provide measures and data that can be used to calculate energy poverty, including the indicators for measuring energy poverty that are suggested in the literature: (i) inability to keep home adequately warm, (ii) arrears on utility bills and (iii) presence of leak, damp, or rot in the dwelling. These data was used by Thomson and Snell [25], who estimated fuel poverty for 25 European countries, by proposing a methodology of four different scenarios, where a different weight was assigned to each indicator. Based on this methodology, Halkos and Gkampoura [26] in their research also created four different scenarios with different weights, in order to evaluate energy poverty conditions in 28 European countries for the period 2004–2019, while identifying the drivers of energy poverty conditions and the impact that the economic crisis had on them.

A stochastic frontier analysis approach was used by Rodriguez-Alvarez et al. [27] to identify the determinants of energy poverty in 30 European countries for the years 2005–2018. The three previously mentioned indicators were also used by Bollino and Botti [19], who combined them with two additional variables, developing the Energy Poverty Multidimensional Index (EMPI) to capture and evaluate energy poverty for 2012 and 2014 in Europe.

In the current literature, a few studies have examined the relationship between renewables and energy poverty, as well as between energy transition and energy poverty. Most of these studies support that a transition from fossil fuels to renewable energy sources will have a positive impact on energy poverty, helping towards its eradication. However, it is also argued in a few studies that energy transition could have negative results on energy poverty-related problems, unless certain measures are being implemented to minimize their damage.

More specifically, Mastropietro [28] analyzed the effect that renewable energy sources for electricity might have on energy poverty. The author argued that the costs required to support renewables' technologies are often transferred to the consumers as surcharges, something that could lead to more intense energy poverty issues. It is suggested that measures, such as state finance or finance through the auctions for emission allowances should be implemented, in order to minimize the social cost that will follow after energy transition.

Specific guidelines that should be followed in order to eliminate the problem of energy poverty by using renewable energy sources and, more specifically, solar energy, were presented by Pagliaro and Meneguzzo [29]. The authors argue that energy poverty will be reduced if renewables are adopted for energy generation and they make specific suggestions to policy makers, such as: view energy poverty within its local social and economic context, get advice from energy managers with knowledge regarding new technologies in the sector and their socioeconomic impacts, make the community engage and be interested in the matter, and establish public institutions concerning renewables, with the target of providing education on the topic and strategies for a successful energy transition. These guidelines,

according to the authors, will facilitate the transition process and will lead to environmental and socioeconomic benefits.

The linkages between energy transition and socioeconomic inequalities in Europe have also been explored in Bouzarovski and Tirado Herrero's [30] research, who emphasize the spatial and temporal variations in energy poverty's incidence. The authors found that there are significant regional inequalities concerning the drivers of energy poverty and the exposure of the countries to them, meaning that energy poverty is a problem presented in a variety of social strata. They also pointed out that energy poverty has increased in the EU since 2007 in general, even though energy transition does not have an extreme impact on these inequalities, according to the authors' results. Further investigation concerning the risks that result from energy transition is strongly suggested.

The case of the UK was studied by Hiteva [31], who examined the potential impact that energy transition might have on fuel poverty. The author took into consideration the conditions of energy transition and the costs, risks, and financial liabilities in the industry of renewable energy, that could have a negative effect on energy poverty. The case of the UK was analyzed and then compared with Bulgaria, the country with the biggest percentages of people living in energy poverty conditions in Europe. The author suggested that the issue of fuel poverty should be addressed throughout the whole renewable electricity production chain and not just at the consumption end and highlighted the need of implementing policies for fuel poverty alleviation throughout this chain.

The effect that energy poverty has on several development outcomes was analyzed by Adom et al. [32], while taking into consideration the influence of green energy transition. The findings highlight that a transition to green energy could potentially reduce vulnerability and provide partial resilience regarding energy poverty shocks, while it could also facilitate the improvement of several development outcomes, including GDP per capita, poverty, and income inequality.

Based on these studies found in the current literature, it can be observed that the results regarding energy poverty and different energy sources linkages, as well as energy poverty and energy transition linkages vary, especially when focusing on different world regions. Even though the topic is discussed in the recent literature, it is significantly important to further examine this relationship. This study examines the linkages between energy consumption coming from fossil fuels and renewable sources and energy poverty indicators for 28 European countries, using an in-depth econometric methodology, that is presented in Section 3. To the best of our knowledge, this methodology has not been used in any similar studies, while examining European countries and using recent data at the same time. In addition, a comparative analysis of energy poverty and energy transition conditions in three European countries is conducted, in order to better understand the progress of three countries with different socioeconomic and environmental conditions on these topics.

3. Methodology

For the analysis, data were collected for the three indicators that are considered to be the key elements of energy poverty, according to the current literature (Table 1: Indicators 1–3). In addition, data were collected regarding GDP per capita, final energy consumption in households per capita, as well as final energy consumption in households by fuel (Table 1: Indicators 4–6). These last two databases were combined, in order to create two new indicators: FFpc, which stands for fossil fuels (oil and petroleum products, natural gas and solid fossil fuels) final energy consumption in households per capita, and RESpc, which stands for renewables and biofuels final energy consumption in households per capita. All data were retrieved from Eurostat's database, for the period 2004–2019 and for 28 European countries (Figure 1). Statistical packages EViews and Stata were used for the analysis.

	Indicators	Measurement	Source
1	Inability to keep home adequately warm	%	[33]
2	Arrears on utility bills	%	[33]
3	Population living in a dwelling with a leaking roof, damp walls, floors or foundation, or rot in window frames or floor	%	[33]
4	Purchasing power adjusted GDP per capita	Purchasing Power Standards (PPS)	[34]
5	Final energy consumption in households per capita	kg of oil equivalent per capita	[35]
6	Final energy consumption in households by	%	[36]

Table 1. Indicators retrieved from Eurostat and used for the analysis.



Figure 1. The 28 European countries included in the analysis. (Tableau Software was used for the creation of all presented figures).

The linkages that exist between energy poverty and fossil fuels and renewable energy consumption are examined, taking into consideration panel data for the years 2004–2019. After the use of Box–Cox specifications that compare linear and logarithmic forms, three different regression models are formulated, where Indicators 1, 2, and 3 (Table 1) are considered as dependent variables, and GDP per capita, fossil fuels final energy consumption per capita, and renewables and biofuels final energy consumption per capita are considered as independent variables. The proposed model is:

$$(IndicJ)_{it} = a_{i,t} + \beta_{1i,t} (GDPpc)_{i,t} + \beta_{2i,t} (FFpc)_{i,t} + \beta_{3i,t} (RESpc)_{i,t} + \gamma_i + \delta_t + \varepsilon_{1,t}$$
(1)

In this regression model, GDPpc stands for GDP per capita, FFpc stands for fossil fuels final energy consumption in households per capita, and RESpc stands for renewables and biofuels final energy consumption in households per capita. Additionally, *J* equals to numbers 1–3, indicating the three different indicators that are used as dependent variables.

For the models' estimations, Fixed (FE) and Random (RE) Effects methods are used, depending on how ai is handled: either as fixed predefined numbers or as random expulsions from a particular distribution [26,37–39]. In the case of Fixed Effects, where the cross-section specific components are viewed as fixed parameters, then the model becomes:

$$y_{it} = a + X'_{it}\beta + \sum_{i=1}^{N} \mu_i D_i + v_{it}$$
⁽²⁾

In the case of Random Effects, which can be used in cases were N individuals are drawn from a large population randomly, the following apply:

$$\mu_i \sim IID(0, \sigma_{\mu}^2), v_{it} \sim IID(0, \sigma_v^2)$$
(3)

with the $\mu_i s$ being independent of the $v_{it}s$, as are the $X_{it}s$ of the $\mu_i s$ and $v_{it}s$ for all I and t [40]. Inconsistency is checked in the RE estimate with Hausman tests, which determine whether the FE or RE model should be used.

In addition, Generalized Method of Moments (GMM) is used, in terms of orthogonal deviations, in order to capture the models' dynamic nature. As Arellano and Bond have stated [41], in orthogonal deviations, each observation is indicated as a deviation from the average of the sample's future observations. Each deviation is weighted to standardize the variance:

$$x_{it}^* = \left[x_{it} - \left(x_{i(t+1)} + \ldots + x_{iT} \right) / (T-t) \right] \sqrt{(T-t)} / \sqrt{T-t+1}, t = 1, \ldots T-1$$
 (4)

The (Ti - q) equations for individual units *i* are:

$$Y_i = \delta w_i + d_i \eta_i + v_i \tag{5}$$

where δ is a parameter vector that includes αk , β , and λ , w_i is a data matrix that includes the endogenous variables' time series, the interpretive variables *x* and the time dummies, and d_i is a $(T_i - q) \times 1$ vector of ones.

Various problems might occur in panel data analyses; this is why several econometric tests are performed before the regression analysis. One of these problems is the correlation of the variables in the dataset. Pesaran's cross-section dependence test allows us to check if the timeseries are cross-sectional independent. In cases of cross-sectional dependence, OLS Dummy estimator (FEM) allowing for individual fixed effects with Driscoll-Kraay standard errors (in Fixed Effects models) can correct the variance–covariance matrix while, in Random Effects models, Breusch-Pagan LM test for individual effects and robust standard errors are applied.

Unit root tests are also performed, in cases where cross-section dependence is confirmed. Dickey-Fuller and Augmented Dickey-Fuller tests can be performed in panel data, with the issue of homogeneity in the autoregressive parameter. In addition, Westerlund tests are performed for panel cointegration, based on the significance of the error correction term in the error correction model. Four tests that check for panel cointegration are proposed: the Gt and Ga statistics, testing for the null hypothesis of no cointegration of all cross-sectional units. The rejection of the hypothesis implies cointegration for at least one unit. The Pt and Pa statistics also test the null hypothesis of no cointegration and their rejection implies cointegration for the panel in total [26,37–39].

4. Results

The descriptive statistics of the indicators used in the analysis (Section 3), are presented in this section. The highest percentages of people that were unable to keep their home adequately warm were observed in Bulgaria for various years, while high percentages were also observed in Portugal as well. The lowest percentages were found in Luxembourg and Norway. Greece presented the highest percentages of people facing arrears on utility bills, while high percentages were also found in Bulgaria. In contrast, the lowest percentages were found in Luxembourg and the Netherlands. Poland presented the highest percentages of people living in dwellings with leak, damp, or rot, while high percentages were also observed in Latvia and Cyprus. The lowest percentages were found in Finland for most years.

GDP per capita was also included in the analysis, expressed in purchasing power standards. The highest levels of GDP per capita were found in Luxembourg for most years, while the lowest levels were observed in Bulgaria. In addition, the highest levels of fossil fuels final energy consumption in households per capita were found in Luxembourg for most years, while the lowest levels were found in Iceland. Similarly, the highest levels of renewables and biofuels final energy consumption in households per capita were found in Latvia, while zero levels were observed in Malta for various years.

Table 2 presents the descriptive statistics of the variables used in our analysis.

	Inadeq_War	m Arrears	Leak_Damp	GDPpc	FFpc	RESpc
Mean	10.963	9.396	16.354	27,632.14	238.47	115.8
Median	5.9	6.9	15	26,400	205.16	96.833
Maximum	69.5	42.2	43.9	81,000	971.028	375.7
Minimum	0.3	1.1	4.1	7500	6.0576	0
Std. Dev.	12.403	7.58	7.577	11,602.94	194.8	86.498
Skewness	2.14	1.79	0.845	1.7245	1.0195	0.737
Kurtosis	8.557	6.23	3.687	7.679	3.8215	2.693
Jarque-Bera	918.9	433.76	62.12	630.73	90.21	42.33
Prob.	0.0000	0.0000	0.0000	0.0000	0.0000	0.0000
Observations	448	448	448	448	448	448

Table 2. Descriptive Statistics of used indicators.

To check for cross-section dependence, a Pesaran test is performed. All results reject the null hypothesis, indicating the existence of cross-section dependence, suggesting thus, the use of Driscoll-Kraay standard errors for the static regression models, in order to correct the variance–covariance matrix (Table 3).

Variables	CD Test	<i>p</i> -Value
Inadeq_warm	12.751 ***	0.0000
Arrears	19.991 ***	0.0000
Leak_damp	5.754 ***	0.0000
GDPpc	60.94 ***	0.0000
FFpc	40.487 ***	0.0000
RESpc	26.142 ***	0.0000

Table 3. Pesaran CD test for cross-section dependence.

Note: The null hypothesis assumes that there exists no cross-section dependence (correlation). Significance at *** 1%.

Fisher-ADF and Fisher-PP unit root tests are performed, which suggest that the examined variables are I(1), with stationarity evidence in first differences (Table 4).

Westerlund tests are performed to test for panel cointegration. The results suggest that the Gt and Ga statistics reject the null hypothesis in most cases, implying cointegration for at least one unit. In addition, the Pt and Pa statistics reject the null hypothesis in every case, implying cointegration for the whole panel (Table 5).

Six regression models were formulated, where each one of the three main variables that are considered to be core elements of energy poverty are used as dependent variables. In the static models, fixed effects model specifications are used, based on the results of the Hausman tests, with FE Driscoll-Kraay standard errors, based on the results of the Pesaran CD tests.

The results indicate that GDP per capita is negatively linked to each one of the three studied variables, in both static and dynamic models. This confirms the fact that economic growth can improve the conditions of energy poverty, since an increase in GDP per capita would lead to a decrease in the percentages of energy poverty factors, as well as that a
financial crisis can significantly impact energy poverty conditions. These findings can be supported by other similar studies in the literature [26,27].

Variables	Fisher-ADF	Fisher-PP		Fisher-ADF	Fisher-PP
Levels			First Differences		
Inadeq_warm	61.786 [0.2771]	66.4179 [0.1608]	Inadeq_warm	125.574 *** [0.0000]	271.546 *** [0.0000]
Arrears	65.861 [0.1725]	47.145 [0.7943]	Arrears	131.534 *** [0.0000]	241.14 *** [0.0000]
Leak_damp	63.8516 [0.2200]	94.0092 *** [0.0011]	Leak_damp	196.915 *** [0.0000]	386.798 *** [0.0000]
GDPpc	27.4452 [0.9995]	43.196 [0.8948]	GDPpc	142.446 *** [0.0000]	216.594 *** [0.0000]
FFpc	60.7524 [0.3087]	90.7817 *** [0.0023]	FFpc	225.9 *** [0.0000]	484.769 *** [0.0000]
RESpc	47.635 [0.7793]	68.6455 [0.1196]	RESpc	139.17 *** [0.0000]	309.252 *** [0.0000]

Table 4. Fisher-ADF and Fisher-PP panel unit root tests.

Note: The null hypothesis assumes that the variable contains unit root. *p*-values in brackets. Significance at *** 1%.

Table 5. Westerlund Panel Cointegration Test.

Equation	Gt	Ga	Pt	Ра
Inadeq_warm = f(FFpc)	-4.87 ***	-16.995 ***	-25.779 ***	-17.266 ***
	[0.000]	[0.000]	[0.000]	[0.000]
Inadeq_warm = f(RESpc)	-5.714 ***	-20.715 ***	-27.642 ***	-24.629 ***
	[0.000]	[0.000]	[0.000]	[0.000]
Inadeq_warm = f(GDPpc)	-4.574 ***	-15.71 *	-19.278 ***	-13.574 ***
	[0.000]	[0.099]	[0.000]	[0.000]
Arrears = $f(FFpc)$	-4.471 ***	-14.028 ***	-18.203 ***	-15.061 ***
	[0.000]	[0.045]	[0.000]	[0.000]
Arrears = f(RESpc)	-5.028 ***	-14.205 *	-17.998 ***	-14.687 ***
	[0.000]	[0.099]	[0.000]	[0.000]
Arrears = f(GDPpc)	-4.641 ***	-15.478 *	-21.56 ***	-13.543 ***
	[0.000]	[0.099]	[0.000]	[0.000]
Leak_damp = f(FFpc)	-5.687 ***	-20.236 ***	-26.58 ***	-18.585 ***
	[0.000]	[0.000]	[0.000]	[0.000]
Leak_damp = f(RESpc)	-5.766 ***	-22.65 ***	-29.639 ***	-22.599 ***
	[0.000]	[0.000]	[0.000]	[0.000]
Leak_damp = f(GDPpc)	-5.12 ***	-18.999 ***	-20.253 ***	-15.225 ***
	[0.000]	[0.000]	[0.000]	[0.000]

Note: The null hypothesis assumes no cointegration. Significance at *** 1%.

Fossil fuels final energy consumption in households per capita is inversely linked in both static and dynamic models to two out of the three indicators, indicating that an increase in the consumption of energy derived from fossil fuels can improve energy poverty conditions. Thus, it is proven that the increased use of fossil fuels per capita, which implies higher energy consumption, leads to better conditions regarding energy poverty in households. In the case of inability to keep the home adequately warm, the static model also indicates that the use of fossil fuels in energy consumption can improve these conditions; in contrast, the dynamic model indicates the opposite, implying a static rather than a dynamic influence of fossil fuels usage in such analyses. Renewables and biofuels final energy consumption in households per capita is found to be a driver of arrears on utility bills, indicating that in the studied time period, higher levels of renewables' use led to difficulties in paying utility bills on time. These findings can also be supported by studies in the literature, where it has been argued that sometimes the costs to support renewables' technologies are transferred to consumers [28], leading therefore to higher electricity prices [26] and explaining, thus, the existence of arrears on utility bills.

In contrast, energy consumption produced from renewables is linked to an inverse relationship to the presence of leaks, damp, and rot in dwellings, according to both static and dynamic model. This indicates that an increase in renewable energy consumption per capita can improve these conditions in households. At the same time, a similar relationship is observed between renewable energy consumption and inability to keep the home adequately warm, according to the dynamic model, indicating that higher levels of renewable energy consumption per capita, would improve people's ability to keep their houses adequately warm. In the static model, renewables final energy consumption in households per capita is statistically insignificant (Table 6).

Table 6. Regression results with three different dependent variables.

$ \begin{array}{ c c c c c c } \hline FE (DK se) & GMM & FE (DK se) & GMM & FE (DK se) & GMM \\ \hline \begin{tabular}{ c c c c c c c } \hline \begin{tabular}{ c c c c c c c c } \hline \begin{tabular}{ c c c c c c c c } \hline \begin{tabular}{ c c c c c c c c c c c c c c c c c c c$		Inadeq	_Warm	Arr	ears	Leak_	Leak_Damp		
Inadeq_warm(-1) 0.75999*** (201.1128) [0.0000] Arrears(-1) 0.65311 *** (57.39698) [0.0000] Leak_damp(-1) -0.000531 *** (45.38322) [0.0000] 0.0001 *** (45.38322) [0.0000] -0.00048 *** (45.38322) [0.0000] GDPpc -0.000631 *** (-6.49) -0.00031 *** (-29.82716) -0.00048 *** (-12.98) -0.00034 *** (-18.0119) -0.00042 *** (-0.0024 *** -0.00024 *** (-10.0001) FFpc -0.03163 *** (0.000] 0.014221 *** (0.0000] -0.031625 *** (-7.68) -0.01824 *** (-11.915) -0.00236 ** (-2.12) -0.002276 * (-15.71195) FFpc -0.03163 *** (-6.48) 0.014221 *** (0.0000] 0.00101 [0.0001] [0.0001] [0.0001] [0.0002] Hausman 5.62 * (0.0401] 0.02129 *** (0.0403] 0.02129 *** (0.0403] 0.02136 *** (-2.48542) -0.02164 *** (0.0080] -0.02164 *** (0.0000] -0.025261 *** Wald test 4867.066 (3) 41425 *** (0.0211 *** 19.16 *** (0.0003] 1269.12 (3) Sargan test 26.5036 (25) 26.9428 (24) 26.4522 (24) AR(1) -2.2241 ** (0.2393] -3.324 *** (0.3392 -0.0824 (0.4995] Observations 448 <t< th=""><th></th><th>FE (DK se)</th><th>GMM</th><th>FE (DK se)</th><th>GMM</th><th>FE (DK se)</th><th>GMM</th></t<>		FE (DK se)	GMM	FE (DK se)	GMM	FE (DK se)	GMM		
Arrears(-1) 0.65311 *** (57.39698) [0.0000] Leak_damp(-1)	Inadeq_warm(-1)		0.75999 *** (201.1128) [0.0000]						
Leak_damp(-1)	Arrears(-1)				0.65311 *** (57.39698) [0.0000]				
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Leak_damp(-1)						0.540377 *** (45.38322) [0.0000]		
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	GDPpc	-0.000631 *** (-6.49) [0.000]	-0.000301 *** (-29.82716) [0.0000]	-0.000448 *** (-12.98) [0.0000]	-0.000334 *** (-18.0119) [0.0000]	-0.00042 *** (-3.26) [0.0050]	-0.000247 *** (-15.03851) [0.0000]		
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	FFpc	-0.03163 *** (-6.38) [0.000]	0.014221 *** (30.87217) [0.0000]	-0.031625 *** (-7.68) [0.0000]	-0.018242 *** (-11.9915) [0.0000]	-0.023136 ** (-2.12) [0.0510]	-0.002276 * (-1.571195) [0.0970]		
$\begin{tabular}{ c c c c c c } \hline F & $14.95 *** & $19.16 *** & 19	RESpc		-0.007889 ** (-2.468542) [0.0140]	0.02129 *** (3.03) [0.0080]	0.027379 *** (12.64376) [0.0000]	-0.02104 ** (-2.21) [0.0430]	-0.025261 *** (-19.58525) [0.0000]		
Wald test $4867.066 (3)$ $4120.7 (3)$ $1269.12 (3)$ Sargan test $26.5036 (25)$ $26.9428 (24)$ $26.4522 (24)$ AR(1) $-2.2241 ** \\ [0.0261]$ $-3.3234 *** \\ [0.009]$ $-2.1069 ** \\ [0.0351]$ AR(2) $0.3327 \\ [0.7393]$ $-0.8542 \\ [0.3930]$ $0.8247 \\ [0.4095]$ Observations 448 392 448 392 448	Hausman	5.62 * [0.0601]		14.95 *** [0.0019]		19.16 *** [0.0003]			
$\begin{tabular}{ c c c c c c } \hline Sargan test & 26.5036 (25) & 26.9428 (24) & 26.4522 (24) \\ \hline AR(1) & -2.2241 ** & -3.3234 *** & -2.1069 ** \\ \hline [0.0261] & [0.0009] & [0.0351] \\ \hline AR(2) & 0.3327 & -0.8542 & 0.8247 \\ \hline [0.7393] & [0.3930] & [0.4095] \\ \hline Observations & 448 & 392 & 448 & 392 \\ \hline \end{tabular}$	Wald test		4867.066 (3)		4120.7 (3)		1269.12 (3)		
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Sargan test		26.5036 (25)		26.9428 (24)		26.4522 (24)		
AR(2) 0.3327 [0.7393] -0.8542 [0.3930] 0.8247 [0.4095] Observations 448 392 448 392 448 392	AR(1)		-2.2241 ** [0.0261]		-3.3234 *** [0.0009]		-2.1069 ** [0.0351]		
Observations 448 392 448 392 448 392	AR(2)		0.3327 [0.7393]		-0.8542 [0.3930]		0.8247 [0.4095]		
	Observations	448	392	448	392	448	392		

Note: t-Statistics in parentheses and *p*-values in square brackets. Parentheses in Wald and Sargan tests indicate degrees of freedom. Critical values for the Wald test of overall significance of the explanatory variables: $\chi_{20.05,3} = 7.815$. Critical values for the Sargan test for over-identifying restrictions: $\chi_{20.05,24} = 36.415$, $\chi_{20.05,25} = 37.652$. Significance at *** 1%, ** 5% and * 10%.

The lag of the dependent variables in the dynamic models are autoregressive-distributed lag specifications, that end up as an AD (1,0) formulation, showing the adjustment to equilibrium values. Table 7 presents the adjustment coefficients of each dynamic model, the discrepancy that is eliminated in a year between the actual and desired values and the periods that are required for the adjustment.

Dependent Variable	Adjustment Coefficient	Discrepancy between the Actual and Desired Values Eliminated in a Year	Periods Required for the Adjustment
Inadeq_warm	1–0.76	24%	More than 4 periods
Arrears	1–0.65	35%	Less than 3 periods
Leak_damp	1–0.54	46%	Approx. 2 periods

Table 7. Adjustment to equilibrium values, based on the lag in the dynamic models.

Wald tests of joint significance, as well as Sargan tests of over-identifying restrictions, are asymptotically distributed as χ^2 variables. Sargan statistics imply evidence of serially uncorrelated errors, since the null hypothesis of over-identifying restrictions is not rejected. AR(1) and AR(2) tests for first and second order serial autocorrelation do not reject the hypothesis of no autocorrelation.

5. Case Studies: Sweden, Germany, and Greece

Three countries, with different socioeconomic and environmental conditions, were chosen and their policies and progress on energy transition as well as their energy povertyrelated conditions are analyzed, compared, and discussed in this section. The chosen countries were Sweden, Germany, and Greece. The three European countries were selected due to their different characteristics and the different socioeconomic, environmental, climatic and energy conditions existing in each one of them.

Sweden is a country in Northern Europe that is characterized by its proactivity on environmental issues and its climate consensus [42]. The country has been characterized as a global leader when it comes to low-carbon economy and has followed a successful path towards energy transition [43]. Sweden's energy needs are covered mainly by hydropower and biomass and the country's geography with moving waters and a big percentage of forest coverage is assisting that, despite the cold climate that requires a high amount of energy for heating [44].

The Swedish energy policies, which aim to promote sustainability, are based on energy policies set by the EU. The EU targets refer to reducing energy consumption by 32.5%, to provide at least 32% of energy consumption from renewable sources and provide at least 14% of energy consumption in the transport sector from renewable sources. Specifically for the Swedish targets, the country has aimed to achieve by 2030, 50% more efficient energy consumption, compared to 2005. In addition, the country's goal is to cover 100% of its electricity needs from renewable energy sources, by 2040 [45].

Germany is a country in Central Europe and has a highly industrialized economy that has been promoting an ambitious plan of energy transition over the past years [46]. The country aims to promote an economy that is low carbon, sustainable, and energy efficient and has achieved a significant growth when it comes to renewable power generation capacity, actively promoting the transition to renewable energy [47].

Since 2010, Germany has initiated and promoted a plan for a more efficient energy system that is based mainly on renewable energy sources, called *Energiewende*. More specifically, *Energiewende*'s targets include the provision of 50% of electricity supply by renewable energy sources and coal's phase-out by 2038 [48]. Data shows that two thirds of Germany's power generation could be covered from renewables by 2030, while solar energy and wind energy could cover half of that proportion [47]. However, and despite the progress that has been made, the evidence shows that the country is struggling to meet its targets, mainly due to the uneven progress that exists across sectors and challenges, especially in transportation and heating. At the moment, Germany uses fossil fuels at a high degree to cover its energy needs and coal is the largest source of power generation, although it is planned to be phased out by 2038 [49].

Greece is a country in South-eastern Europe that is also implementing reforms in the energy sector in order to foster decarbonization and promote a just energy transition. More specifically, the country aims to achieve a reduction in its greenhouse gas emissions by more than 56%, by 2030 (compared to 2005 levels), aiming to achieve by 2050 a climate neutral economy. However, at the moment, fossil fuels are the primary energy supplier in the country [50].

As stated in the National Energy and Climate Plan 2021–2030, Greece aims to increase the share of renewables to 31% by 2030, in order to contribute to the achievement of the EU target, that aims to increase the share of renewables to at least 32% by 2030. In addition, the country aims to reduce the use of lignite that is used for power generation, and to shut down by 2028 the lignite-fired plants, while ensuring energy security and promoting energy efficiency [51].

From this evidence, it is obvious that Sweden is in the lead when it comes to energy transition policies and promotion, while Greece is comparatively slower in the process of decarbonization, promoting less ambitious policies. This can also be observed by the provided data. As seen in Figure 2, among the three studied countries, Sweden has the biggest share of renewable energy in gross final energy consumption, reaching 56.39% in 2019. The country has achieved the target of reaching 49% by 2020, towards Europe 2020 target, which aimed to increase the share of renewable energy in gross final energy consumption in the EU to 20% by 2020. In Greece, the share of renewable energy reached 19.68% in 2019, and the country has also achieved the target of reaching 18% by 2020. In Gremany, the same share was estimated at 17.35% in 2019 and the country was very close in achieving the target of 18% by 2020 [52].



Figure 2. Share of renewable energy in gross final energy consumption in Sweden, Germany, and Greece (%).

The Energy Transition Index takes into consideration the performance of the current energy system as well as the enabling environment for energy transition and aims to reflect the relationships and dependencies that exist between the energy system transformation and various factors (economic, social, political, regulatory) that determine whether a country is ready for transition. According to the Energy Transition Index 2021, and as seen in Figure 3, Sweden is the global leader, ranking in the first place, while Germany is found in the 18th place and Greece in the 54th [53]. These rankings highlight once more the differences on energy transition potential and progress that exist among the selected countries and the necessity of efforts required to move towards sustainability in each one of them.





After the comparative analysis of the energy transition progress and policies promoted in each one of the studied countries, an overview of the energy poverty situation and the policies regarding energy poverty is also presented. As seen in Figure 4, Greece was the country with the highest percentages of people living under energy poverty conditions among the three examined countries, according to Eurostat data. In contrast, low percentages were observed in Germany and Sweden for all three indicators; these countries perform better compared to the EU average on the specific indicators, while Greece has a significantly lower performance compared to the EU average.



Figure 4. Data regarding inability to keep home adequately warm, arrears on utility bills, and presence of leak, damp or rot in the dwelling for Germany, Greece, and Sweden (% of population).

The data presented here for the three selected countries can also be compared to the findings from the analysis presented in Section 4. More specifically, we have found that an increasing GDP per capita can improve energy poverty conditions and, if we look closely at the data of the selected countries, we can see that in periods when GDP per capita was increasing in the studied countries, energy poverty levels were lower. Similarly, the results regarding the effect of renewable energy consumption per capita can also be validated, since we can observe that in most periods of higher levels of renewables final energy consumption in households per capita, higher percentages of households facing arrears on utility bills were also observed. However, it is important to further research each country's social, economic, and environmental conditions and active policies, in order to better understand the differences in energy poverty conditions and promote targeted and more effective policies and strategies.

According to the EU Energy Poverty Observatory, both Germany [54] and Greece [55] have an active research community, concerning the field of energy poverty, while the research in Sweden does not specifically focus on energy poverty, but on other energy-related fields, such as energy transition and efficiency [56].

Based on this evidence, it can be observed that Greece and Germany have developed and promoted various policies to lower energy poverty levels, while Sweden has not been actively addressing the problem, due to the already low levels that are observed in the country. Instead, Sweden is focusing on energy transition and renewable energy sources and has set ambitious goals. Germany and Greece have also been promoting energy transition policies, but their energy needs are still mainly covered by fossil fuels.

6. Conclusions

At the moment, the world depends highly on fossil fuels, despite their disadvantages and their impact on the environment. Energy transition and the use of renewable energy sources has been promoted a lot more in the past few years and a lot of countries have made significant progress to that end. However, the impact that energy transition and the use of renewable sources could have on the problem of energy poverty should be continuously studied and addressed.

This study contributes to the existing literature and expands the knowledge on the topic, focusing on assessing the impact that fossil fuels and renewables usage had on energy poverty conditions in 28 European countries during the time period 2004–2019. The necessary data were extracted from the Eurostat database and an in-depth econometric methodology was followed which, to the best of our knowledge, has not been used in similar studies. The findings suggest that GDP per capita and fossil fuels final energy consumption in households per capita are linked to an inverse relationship to energy poverty conditions. In addition, the results indicate that an increase in renewables and biofuels final energy consumption in households per capita led to an increase in the percentage of people facing arrears on utility bills, while it led to a decrease in the percentage of people that cannot keep their home adequately warm, and of the percentage of population living in a dwelling with a leaking roof, damp walls, floors or foundation, or rot in window frames or floor.

These results highlight the fact that higher fossil fuels usage per capita can improve energy poverty conditions, while also highlighting the assistance that a higher level of renewables usage per capita can provide in certain energy poverty conditions. Attention should be given, though, on the mitigation of the impact that renewables' usage can have on arrears on utility bills. Government and policy makers should be aware of this relationship and develop strategies that promote energy transition while protecting energy consumers. More specifically, there should be given extra attention in not transferring the costs of renewables to consumers [28] and in promoting policies that assist households pay their bills on time, while also ensuring that the mitigation of fossil fuels usage will not have an impact on system's stability [12], leading to other social or economic problems.

Additionally, three case studies were examined and the conditions in three European countries with different socioeconomic and environmental characteristics (Sweden, Germany, and Greece) were presented and compared. The evidence shows that Germany and Greece have focused on energy poverty mitigation while Sweden, which manages to keep its energy poverty levels significantly low, promotes more ambitious strategies regarding energy transition. Between the examined countries, Greece is the one with the highest energy poverty levels while Sweden is the one with the highest share of renewable energy in gross final energy consumption.

This comparison can be useful for policy makers, since it highlights the differences that exist among European countries in these fields and the importance to promote and implement targeted policies in each country, based on their progress and needs. Outside Europe, and when it comes to developing countries, policy makers should take into consideration other studies in the literature to support effective renewable energy development, focusing on market guarantee, lowering the costs of licensing for renewable projects, raising public consciousness, and increasing R&D, among others [57]. In general, the role that cities and communities play in ecological and energy transition should be examined and taken into consideration by policy makers, when promoting relevant strategies [58], while it would also be useful to explore the impact that subsidies towards green resources can have in supporting energy transition [59] and, subsequently, how these could impact energy poverty conditions in certain countries and regions. Finally, the impact of the COVID-19 pandemic on environmental matters and on renewable energy should be taken into consideration when promoting strategies for specific countries or regions, assessing the impact that the pandemic had on energy markets [60].

While the results of this study highlight the linkages that exist between energy coming from different sources and energy poverty conditions and can be proven helpful for governments and policy makers, future research on this relationship is strongly suggested. More specifically, extensive research targeted to specific countries or regions is essential, examining not only the current situation and the current linkages, but also the tailored policies and strategies that should be promoted for achieving a successful energy transition while ensuring energy security, minimizing energy poverty levels, and progressing on the targets of the 7th Sustainable Development Goal at the same time.

Author Contributions: Both authors G.H. and E.-C.G. contributed equally to each section of this paper. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Data Availability Statement: Publicly available datasets were analyzed in this study. These data can be found in the References section.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Abas, N.; Kalair, A.; Khan, N. Review of fossil fuels and future energy technologies. *Futures* **2015**, *69*, 31–49. [CrossRef]
- Höök, M.; Tang, X. Depletion of fossil fuels and anthropogenic climate change—A review. *Energy Policy* 2013, 52, 797–809. [CrossRef]
- 3. Masnadi, M.S.; Grace, J.R.; Bi, X.T.; Lim, C.J.; Ellis, N. From fossil fuels towards renewables: Inhibitory and catalytic effects on carbon thermochemical conversion during co-gasification of biomass with fossil fuels. *Appl. Energy* **2015**, *140*, 196–209. [CrossRef]
- 4. Leggett, L.M.W.; Ball, D.A. The implication for climate change and peak fossil fuel of the continuation of the current trend in wind and solar energy production. *Energy Policy* **2012**, *41*, 610–617. [CrossRef]
- Wang, C.H.; Zhao, D.; Tsutsumi, A.; You, S. Sustainable energy technologies for energy saving and carbon emission reduction. *Appl. Energy* 2017, 194, 223–224. [CrossRef]
- Johnsson, F.; Kjärstad, J.; Rootzén, J. The threat to climate change mitigation posed by the abundance of fossil fuels. *Clim. Policy* 2019, 19, 258–274. [CrossRef]
- Kittner, N.; Lill, F.; Kammen, D.M. Energy storage deployment and innovation for the clean energy transition. *Nat. Energy* 2017, 2, 17125. [CrossRef]
- Yan, J.; Chen, B.; Wennersten, R.; Campana, P.; Yang, J. Cleaner energy for transition of cleaner city. *Appl. Energy* 2017, 196, 97–99.
 [CrossRef]
- 9. Child, M.; Kemfert, C.; Bogdanov, D.; Breyer, C. Flexible electricity generation, grid exchange and storage for the transition to a 100% renewable energy system in Europe. *Renew. Energy* **2019**, *139*, 80–101. [CrossRef]
- 10. Solomon, B.D.; Krishna, K. The coming sustainable energy transition: History, strategies, and outlook. *Energy Policy* **2011**, 39, 7422–7431. [CrossRef]

- 11. Lin, C.; Li, Y.; Cai, Q.; Shi, P.; Song, M.; Wu, W. Evaluation on the Cost of Energy Transition: A Case Study of Fujian, China. *Front. Energy Res.* **2021**, *9*, 29. [CrossRef]
- 12. Kern, F.; Smith, A. Restructuring energy systems for sustainability? Energy transition policy in the Netherlands. *Energy Policy* **2008**, *36*, 4093–4103. [CrossRef]
- 13. Verbong, G.; Loorbach, D. Governing the Energy Transition: Reality, Illusion or Necessity? Routledge: Milton Park, UK, 2012.
- 14. Sheikh, N.J.; Kocaoglu, D.F.; Lutzenhiser, L. Social and political impacts of renewable energy: Literature review. *Technol. Forecast. Soc. Change* **2016**, *108*, 102–110. [CrossRef]
- 15. Ahn, K.; Chu, Z.; Lee, D. Effects of renewable energy use in the energy mix on social welfare. *Energy Econ.* **2021**, *96*, 105174. [CrossRef]
- 16. Bertheau, P. Assessing the impact of renewable energy on local development and the Sustainable Development Goals: Insights from a small Philippine island. *Technol. Forecast. Soc. Change* **2020**, *153*, 119919. [CrossRef]
- 17. Bouzarovski, S.; Petrova, S.; Sarlamanov, R. Energy poverty policies in the EU: A critical perspective. *Energy Policy* **2012**, *49*, 76–82. [CrossRef]
- 18. González-Eguino, M. Energy poverty: An overview. Renew. Sustain. Energy Rev. 2015, 47, 377–385. [CrossRef]
- 19. Bollino, C.A.; Botti, F. Energy poverty in Europe: A multidimensional approach. PSL Q. Rev. 2017, 70, 449–472.
- 20. United Nations. Sustainable Development Goals | UNDP. 2015. Available online: https://www.undp.org/content/undp/en/home/sustainable-development-goals.html (accessed on 7 September 2020).
- 21. Halkos, G.E.; Gkampoura, E.C. Coping with Energy Poverty: Measurements, Drivers, Impacts, and Solutions. *Energies* **2021**, 14, 2807. [CrossRef]
- 22. Trinomics. Selecting Indicators to Measure Energy Poverty. 2016. Available online: https://ec.europa.eu/energy/sites/ener/files/documents/Selecting%20Indicators%20to%20Measure%20Energy%20Poverty.pdf (accessed on 25 July 2021).
- 23. Nussbaumer, P.; Bazilian, M.; Modi, V. Measuring energy poverty: Focusing on what matters. *Renew. Sustain. Energy Rev.* 2011, 16, 231–243. [CrossRef]
- Santillán, O.S.; Cedano, K.G.; Martínez, M. Analysis of Energy Poverty in 7 Latin American Countries Using Multidimensional Energy Poverty Index. *Energies* 2020, 13, 1608. [CrossRef]
- 25. Thomson, H.; Snell, C. Quantifying the prevalence of fuel poverty across the European Union. *Energy Policy* **2013**, *52*, 563–572. [CrossRef]
- 26. Halkos, G.E.; Gkampoura, E.C. Evaluating the effect of economic crisis on energy poverty in Europe. *Renew. Sustain. Energy Rev.* **2021**, 144, 110981. [CrossRef]
- Rodriguez-Alvarez, A.; Llorca, M.; Jamasb, T. Alleviating energy poverty in Europe: Front-runners and laggards. *Energy Econ.* 2021, 103, 105575. [CrossRef]
- 28. Mastropietro, P. Who should pay to support renewable electricity? Exploring regressive impacts, energy poverty and tariff equity. *Energy Res. Soc. Sci.* **2019**, *56*, 101222. [CrossRef]
- 29. Pagliaro, M.; Meneguzzo, F. Distributed Generation from Renewable Energy Sources: Ending Energy Poverty across the World. *Energy Technol.* **2020**, *8*, 2000126. [CrossRef]
- Bouzarovski, S.; Tirado Herrero, S. The energy divide: Integrating energy transitions, regional inequalities and poverty trends in the European Union. *Eur. Urban Reg. Stud.* 2017, 24, 69–86. [CrossRef]
- 31. Hiteva, R.P. Fuel poverty and vulnerability in the EU low-carbon transition: The case of renewable electricity. *Local Environ.* **2013**, *18*, 487–505. [CrossRef]
- Adom, P.K.; Amuakwa-Mensah, F.; Agradi, M.P.; Nsabimana, A. Energy poverty, development outcomes, and transition to green energy. *Renew. Energy* 2021, 178, 1337–1352. [CrossRef]
- 33. Eurostat. Income and living conditions database. Retrieved from Eurostat. Available online: https://ec.europa.eu/eurostat/ web/income-and-living-conditions/data/database (accessed on 10 March 2021).
- Eurostat. Purchasing Power Adjusted GDP per Capita. Retrieved from Eurostat. Available online: https://ec.europa.eu/eurostat/ web/products-datasets/-/sdg_10_10 (accessed on 10 March 2021).
- Eurostat. Final Energy Consumption in Households per Capita. Retrieved from Eurostat. Available online: https://ec.europa.eu/ eurostat/web/products-datasets/-/sdg_07_20 (accessed on 10 March 2021).
- Eurostat. Final Energy Consumption in Households by Fuel. Retrieved from Eurostat. Available online: https://ec.europa.eu/ eurostat/web/products-datasets/-/t2020_rk210 (accessed on 10 March 2021).
- 37. Halkos, G.E.; Gkampoura, E.C. Examining the Linkages among Carbon Dioxide Emissions, Electricity Production and Economic Growth in Different Income Levels. *Energies* **2021**, *14*, 1682. [CrossRef]
- 38. Halkos, G.E. Environmental Kuznets Curve for sulfur: Evidence using GMM estimation and random coefficient panel data models. *Environ. Dev. Econ.* 2003, *8*, 581–601. [CrossRef]
- Halkos, G.; Petrou, K.N. The relationship between MSW and education: WKC evidence from 25 OECD countries. *Waste Manag.* 2020, 114, 240–252. [CrossRef] [PubMed]
- 40. Baltagi, B. Econometrics; Springer: Berlin/Heidelberg, Germany, 2008.
- 41. Arellano, M.M.; Bond, S. *Dynamic Panel Data Estimation Using DPD—A Guide for Users*; Working Paper Series; Institute for Fiscal Studies: London, UK, 1988.

- 42. Djerf-Pierre, M.; Cokley, J.; Kuchel, L.J. Framing renewable energy: A comparative study of newspapers in Australia and Sweden. *Environ. Commun.* **2016**, *10*, 634–655. [CrossRef]
- 43. IEA Sweden is a Leader in the Energy Transition, According to Latest IEA Country Review. Retrieved from IEA. 2019. Available online: https://www.iea.org/news/sweden-is-a-leader-in-the-energy-transition-according-to-latest-iea-country-review (accessed on 14 August 2021).
- 44. Kretchmer, H. Transforming Homes into Power Stations—How Sweden is Disrupting Energy Production. Retrieved from the World Economic Forum. 2020. Available online: https://www.weforum.org/agenda/2020/09/sweden-energy-production-renewable-power-district-heating/ (accessed on 15 August 2021).
- 45. Swedish Energy Agency Energimyndigheten. Energy in Sweden 2021—An overview. Retrieved from Energimyndigheten. 2021. Available online: https://energimyndigheten.a-w2m.se/Home.mvc?ResourceId=198022 (accessed on 15 August 2021).
- 46. Strunz, S.; Gawel, E.; Lehmann, P. The political economy of renewable energy policies in Germany and the EU. *Util. Policy* **2016**, 42, 33–41. [CrossRef]
- IRENA. Renewable Energy Prospects: Germany. Retrieved from the International Renewable Energy Agency. 2015. Available online: https://www.irena.org/publications/2015/Nov/Renewable-Energy-Prospects-Germany (accessed on 14 August 2021).
- 48. IEA. Germany. Retrieved from IEA. 2021. Available online: https://www.iea.org/countries/germany (accessed on 14 August 2021).
- 49. IEA. Germany 2020. Retrieved from IEA. 2020. Available online: https://www.iea.org/reports/germany-2020 (accessed on 14 August 2021).
- 50. IEA. Greece. Retrieved from IEA. 2021. Available online: https://www.iea.org/countries/greece (accessed on 14 August 2021).
- 51. Hellenic Republic. National Energy and Climate Plan. Ministry of the Environment and Energy. Retrieved from the European Commission. 2019. Available online: https://ec.europa.eu/energy/sites/ener/files/el_final_necp_main_en.pdf (accessed on 15 August 2021).
- 52. Eurostat. Share of Renewable Energy in Gross Final Energy Consumption. Retrieved from Eurostat. 2021. Available online: https://ec.europa.eu/eurostat/web/products-datasets/-/t2020_rd330 (accessed on 10 March 2021).
- 53. World Economic Forum. Fostering Effective Energy Transition 2021 edition. Retrieved from the World Economic Forum. 2021. Available online: https://www.weforum.org/reports/1edb4488-deb4-4151-9d4f-ff355eec499a (accessed on 15 August 2021).
- 54. European Commission. EPOV Member State Report—Germany. Retrieved from: Energy Poverty Advisory Hub. 2019. Available online: https://energy-poverty.ec.europa.eu/discover/practices-and-policies-toolkit/publications/epov-member-state-report-germany_en (accessed on 16 August 2021).
- 55. European Commission. Energy Poverty Observatory in Greece. 2019. Retrieved from: Energy Poverty Advisory Hub. Available online: https://energy-poverty.ec.europa.eu/discover/practices-and-policies-toolkit/publications/energy-poverty-observatory-greece_en (accessed on 16 August 2021).
- European Commission. EPOV Member State Report—Sweden. Retrieved from: Energy Poverty Advisory Hub. 2019. Available online: https://energy-poverty.ec.europa.eu/discover/practices-and-policies-toolkit/publications/epov-member-state-reportsweden_en (accessed on 16 August 2021).
- 57. Oryani, B.; Kamyab, H.; Moridian, A.; Azizi, Z.; Rezania, S.; Chelliapan, S. Does structural change boost the energy demand in a fossil fuel-driven economy? New evidence from Iran. *Energy* **2022**, 254, 124391. [CrossRef]
- 58. Hoppe, T.; van Bueren, E. Guest editorial: Governing the challenges of climate change and energy transition in cities. *Energy Sustain. Soc.* **2015**, *5*, 1–9. [CrossRef]
- 59. Qi, X.; Guo, Y.; Guo, P.; Yao, X.; Liu, X. Do subsidies and R&D investment boost energy transition performance? Evidence from Chinese renewable energy firms. *Energy Policy* **2022**, *164*, 112909.
- 60. Kamyab, H.; Naderipour, A.; Jahannoush, M.; Abdullah, A.; Marzbali, M.H. Potential effect of SARS-CoV-2 on solar energy generation: Environmental dynamics and implications. *Sustain. Energy Technol. Assess.* **2022**, *52*, 102027. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.





Article Determinants of Renewable Energy Consumption in Africa: Evidence from System GMM

Adedoyin Isola Lawal

Department of Economics, Bowen University, Iwo 232102, Nigeria; adedoyin.lawal@bowen.edu.ng or l.adedoyin@yahoo.com; Tel.: +23-480-3523-3567

Abstract: The adoption of renewable energy remains Sub-Saharan Africa's best option to achieve sustainable growth and mitigate climate change. The essence of this study is to examine the factors that determine the adoption of renewable energy adoption in Africa by employing the System Generalized Methods-Of-Moment (GMM) to analyze data sourced from 1990 to 2019 on some selected African economies. The study examined the tripartite role of the economic, environmental, and socio-political factors on renewable energy adoption in Africa and noted that a positive relationship exists between economic and renewable energy adoption, supporting the validity of the feedback hypothesis. Hence, a policy that supports simultaneous growth of the economy and renewable energy could be adopted. The results further show that environmental factors such as carbon emission and ecological footprint negatively impact renewable energy (RE) adoption in Sub-Saharan African economies. The impact of socio-political factors is, at best mixed; for instance, the result of urbanization is positive and significant, suggesting that urbanization helps in the quick adoption of renewable energy in the studied economies, while the results of corruption show otherwise. To account for single-country dynamics, the study employed the full PMG and noted that the pollution haven hypothesis holds for a number of African economies. The results offer some policy implications.

Keywords: renewable energy; climate change; carbon emission; economic growth; Africa

1. Introduction

Top on the agenda of global policymakers is defining and designing suitable energy, economic, and environmental policies that can mitigate increasing global carbon dioxide emissions (CO₂) [1–5]. This is premised on the fact that increasing CO₂ emission negatively impacts human wellbeing and health and poses a threat to handing over a secure and sustainable environment to the future generation [6,7]. Achieving sustainable environmental policies capable of reducing CO₂ emissions requires a comprehensive and robust understanding of its causes [8–13]. Extant literature suggests that to keep humanity and prevent negative alteration of man's state; concerted efforts must be taken to reduce and mitigate the impact of greenhouse gas (GHG) emissions and keep the average global temperature at the pre-industrial state of less than 2° C (IPCC 2007, Kyoto Protocol 1997) [4,14–16].

Evidence such as continuous occurrences of super droughts, wildfires, and hurricanes, among others that suggest the intensification of extreme weather events and natural disasters occurring in higher numbers or frequencies as well as magnitude across the globe call for urgent attention from governmental and non-governmental organizations, bilateral and multilateral institutions, to mitigate climate change/CO₂ to avert global disaster [4,5,17,18]. Several actions and policies have been canvassed by various international institutions to curb the negative impact of CO_2 emissions over the years [19–21]. Some of these policies often center on improving energy efficiency, conserving energy, and designing energy strategies [22]. The main drivers of these policies are reducing the high levels of CO_2 emission from intense nonrenewable energy sources and reducing the high percentage of nonrenewable energy in the total energy component (nonrenewable accounts for more than

Citation: Lawal, A.I. Determinants of Renewable Energy Consumption in Africa: Evidence from System GMM. *Energies* **2023**, *16*, 2136. https:// doi.org/10.3390/en16052136

Academic Editor: George Halkos

Received: 30 January 2023 Revised: 14 February 2023 Accepted: 17 February 2023 Published: 22 February 2023



Copyright: © 2023 by the author. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). 80% of the global total energy components). At the center of these two policies is the need to increase the world component of renewable energy in the global energy mix.

Over the past decades, advocacy has identified renewable energy (RE) sources as reliable alternative sources of energy to conventional fossil energy sources such as crude oil, coal, and natural gas, stressing that they have some added advantages of being environmental-friendly, readily available, among others [23]. As noted by [22], there is a rapid decline in the generation cost of renewable energy. There has been strong advocacy for its usage by international organizations such as the 1997 Kyoto Protocol, the 2016 Paris agreement (COP21), the International Energy Agency, and the United Nations, just to mention a few, as it is environmentally friendly and possess the ability to mitigate climate change, produces either no or minimal global warming emissions [24,25]. Essentially, RE promotes economic growth in a number of ways. (i.) RE technologies support the diversification of the energy mix and support energy security via the provision of a reliable, vast supply of renewable energy necessary to achieve sustainable economic growth. (ii.) RE advances both social and environmental benefits as it reduces the amount of CO₂ emission into the environment, hence reducing the cost of addressing environmental pollution. (iii.) Developing RE sources assist economies in becoming self-reliant for energy and avoiding energy shortages arising from external shocks. (iv.) RE creates job opportunities, among others. It is also worth noting that the continuous shocks or upsurge in oil prices and prices of other fossil fuels against the continuous fall in RE technologies are incentives to shifts towards RE sources adoption [26].

Despite the strength of RE as a source of energy, its universal adoption has been relatively slow. For instance, 80% of the world's energy mix is still comprised of nonrenewable energy. This will have a negative effect on the effort to switch toward a green and sustainable energy system. Hence there is a need to explore the drivers of the deployment of RE to know what factors maximize the achievement of sustainable energy. According to [27], factors that can influence the adaption of RE can be classified into nine strands: political, institutional, economic, social, environmental, regulatory, technical, technological, and logistics.

Extant literature on the determinants of RE adoption is multi-dimensional, focusing on energy indicators, environmental factors, explanatory variables, regions and countries, time periods, econometric models, and estimation techniques [27]; for instance, [8,28–31]. In terms of the methodology adopted, ref. [32] canvassed for strong modeling techniques, ref. [33] employed panel data estimation techniques, ref. [34] employed panel autoregressive distributed lag (P-ARDL), and ref. [35] employed bootstrap ARDL, among others. A closer look at most of the extant studies suggests that though Africa has a huge reserve of renewable energy, few studies have been conducted on the possibility of switching toward the adoption of renewable energy. There are few appreciable studies on the determinants of drivers of RE adoption on the continent. This, among others, is the essence of the current study.

A major factor in mitigating increased emission rates is adopting RE in the production and consumption life. RE is healthy for public health, the environment, and the economy; hence, focusing on adopting RE is key to achieving environmentally sustainable economic growth. RE, among others, helps in diversifying the energy mix, increases energy security as it provides a reliable, vast, and renewable supply of energy needed for sustainable growth, and reduces environmental costs owing to addressing issues related to CO₂ emissions. Specifically, RE can be influenced by three main constructs: economic, environmental, and socio-political factors [36–38]. The impact of economic growth on RE adoption could be explained by the influence of macroeconomic variables such as real gross domestic product (RGDP), foreign direct investment (FDI), financial development (FD), and trade openness (TRD), among others. Similar to every source of energy, four possibilities exist in explaining the linkages between economic growth and RE. They are RE-leading, economic growth following hypothesis; economic growth leading, RE following hypothesis; feedback hypothesis where a bilateral relationship exists between RE and economic growth; the fourth possibility is the neutrality hypothesis, where no causality exists between economic growth and RE [20,39,40].

The impact of environmental factors on RE adoption is essentially influenced by two models: the environmental Kuznets hypothesis (EKC); and the pollutant haven models. The EKC noted that a U-shape relationship exists between economic growth and environmental pollution. The theory simply described a non-linear relationship between growth and environmental degradation. The pollutant haven model stressed that the existence of legislation to punish the deployment of environmentally harmful energy sources would motivate the adoption of RE [41–43].

The socio-political strands focus on the ability of governance structure, government policies, and urbanization, among others, to influence the adoption of RE [44–46]. Urbanization as a socioeconomic factor impacts energy consumption and environmental condition as it may induce the enlargement of energy-intensive industries such as steel and concrete, the power industry, and the transport sector, thereby provoking upward shocks to the environment [47]. Another dimension to the contributions of urbanization to energy consumption and the environment suggests that urbanization might improve the environmental quality, provided man is willing to be environmentally conscious and friendly. As important as these constructs are to the adoption of RE, few studies have accounted for them in RE adoption works. For instance, refs. [1,2] did not account for the impact of socio-political factors, and [48–51] only focused on environmental factors. Refs. [52–54] focused on both economic and environment but did not address socio-political factors. The crux of the current study is to calibrate these constructs to discuss RE adoption with a focus on Africa.

Several factors induced our motivation on Africa (as of 2017, African CO₂ emission was 4% of global CO₂ emissions. It grew at an average of 4.6% yr⁻¹ over the period 1990–2017 against the global rate of 12.2% yr⁻¹); for instance, extant studies and reports have noted the deteriorating nature of the environmental space in Africa over the last few decades [55,56]. Air pollution and CO₂ emissions account for environmental degradation in the region more than other types of pollution, such as water or land pollution. The region is reported to have one of the most prolonged CO₂ emission growth rates in the world, with more than a 123% growth rate between 1979 and 2017, surpassing the global average of 60% [57,58]. With the current trend in CO₂ emission growth rate, Africa will, by the year 2030, have a 30% CO₂ emission growth rate.

The essence of this study is to investigate the nexus between RE and economic growth, the environment, and socio-political factors by employing a System Generalized Methods-Of-Moment (GMM) model. Our choice of system GMM was influenced by its many advantages over alternative estimation techniques, such as the difference GMM. For instance, system GMM has three clear-cut advantages: (i.) It is useful in reducing endogeneity bias; (ii.) it reduces time-varying measurement error bias; (iii.) it reduces weak instrument error bias [20,59,60]. The system GMM helps address the issues related to endogeneity resulting from the inclusion of other potential endogenous explanatory variables, as well as other possibilities of measurement errors owing to the use of cross-country data displaying high persistence [61–65]. The study intends to ask the following research questions: (i) What drives renewable energy adoption in Africa? (ii) To what extent do macroeconomic variables impact renewable energy adoption in Africa? (iv) What is the role of socio-political factors in renewable energy adoption in Africa?

Our study's novel contribution to literature is four-fold. First, to the best of our knowledge, we are among the first studies to examine the drivers of RE consumption in Africa. Africa, as a growing economy, is in dire need of energy and is simultaneously faced with the need to have a safe environment given the alarming rate of CO₂ emission of 123%, surpassing the global average of 60%. Therefore, Africa needs to switch from traditional fossil fuel-dominated energy sources to clean and safer RE sources; hence the need to understand the drivers of RE adoption for appropriate policy adjustment. Secondly, we account for the role of macroeconomic variables, environmental constructs, and socio-

political factors in the nexus between energy, economics, and the environment. Thirdly, we employed novel and appropriate estimation techniques, the system GMM which is useful in reducing endogeneity bias, time-varying measurement error bias, and weak instrument error bias, and reducing measurement errors owing to the adoption of cross-country data. Fourthly, we offer some policy implications.

Our study will provide insights into at least six Sustainable Development Goals (SDGs): SDG 7- affordable and clean energy; SDG 8- economic growth; SDG 11- sustainable cities and communities; SDG 12- responsible consumption and production; SDG13- climate action; and SDG 17- partnership for the goal with trade offering leadership.

The remainder of this study is as follows: Section 2 presents the literature review; Section 3 presents the materials and estimation techniques; Section 4 deals with the presentation and discussion of results, while Section 5 concludes the paper.

2. Literature Review

The theoretical note that governs this study is threefold: cointegration (economic growth-related), environmental, and impact. The cointegration (economic growth) strands are further divided into four hypotheses that explain the possibility of causality between RE and economic growth. These hypotheses are energy-leading growth following hypothesis, which states that it is the demand for energy that spurs economic growth; hence, conservative measures to conserve the environment will have negative consequences on economic growth. The second leg of this strand is the economic growth-leading following hypothesis that suggests that it is growth that drives energy demand. The third strand is the feedback hypothesis which states that a bilateral relationship exists between economic growth and energy consumption. The fourth hypothesis is the neutrality hypothesis which suggests that no causality exists between economic growth and energy consumption. Hence, any policy introduced to manipulate either of the two will have little or no effect on the other [20,66,67].

The discussion of the extant literature on the impact of macroeconomic variables on energy behavior remains inconclusive; for instance, ref. [2] examined the dynamic effect of nonrenewable energy, renewable energy, economic growth, and foreign direct investment on the environment based on data sourced from the year 2000 to 2015 for some selected African economies. The study employed panel ARDL that calibrates the pooled mean group, mean group, and dynamic fixed effect estimator to examine the validity of both the environmental Kuznets curve and/or pollution haven hypothesis. The result attained shows that while a negative and significant relationship exists between renewable energy and CO_2 emissions, the relationship between CO_2 and other explanatory variables is positive and significant, both in the short and long runs, except for FDI, which is positive only in the long run. The study noted that EKC does not hold for the studied economy; as a result, it tilts towards the pollution haven hypothesis. This suggests that African economies are less concerned about their environment but place a high premium on growth. A major difference between ref. [2] and the current study is the fact that whereas the former does not discuss socio-political factors, the latter calibrated it into their model; the current study accounts for single-country analysis.

For some selected 55 economies, ref. [68] employed a two-system GMM procedure to examine the nexus between financial development and renewable energy adoption based on data sourced from 2005 to 2014. The study noted that a positive and significant relationship exists between financial development and renewable energy for high-income economies though the relationship is insignificant for low-income economies. The study noted that sophisticated financing is key to achieving RE in the studied economies. The study also noted that the impact of trade openness and carbon emission are statistically insignificant for the economies studied, suggesting that trade has no impact on RE adoption. The results from the impact of carbon emission on RE adoption are intriguing, especially for high-income economies. The authors concluded that the EKC model is valid for the studied economies. In a related development, ref. [69] noted that financial development is key to achieving the adoption of RE in China. The study emphasized the role of green financing and a green reputation in achieving the deployment of renewable energy that will support growth. The study employed several econometric techniques to analyze both micro and macro data on the Chinese economy from 2015 to 2020. The study identified oil price volatility and geopolitical risk as key obstacles to adopting RE in China. In a related development, ref. [70] noted that financial development is key to achieving RE consumption in Africa based on the study estimation of the generated method of moments (GMM) and quantitative regression (QR) in analyzing data sourced from 2004 to 2014. The study noted that financial inequality is a major setback to progress in RE consumption in Africa.

Ref. [71] noted that financial development, agriculture, and economic growth are key to the adoption of RE in Africa, while corruption and bad governance negatively affects Africa's adoption of RE. The study analyzed case studies, research articles, policy briefs, and project reports across and beyond Africa. It noted that for Africa to achieve the SGDs, the operations of Power Africa, Sustainable Energy for All (SE4ALL) initiative, concerted efforts must be put in place to address corruption on the continent.

Ref. [72] noted that FDI negatively impacts the environment on the one hand and RE consumption on the other hand for China based on the results obtained on the deployment of systems GMM, random effect, and fixed effect on the annual date from 2011 to 2016. The study noted that the pollution haven hypothesis is valid for the study economy.

Ref. [73] noted that RE and nonrenewable energy (N-RE) are key determinants of FDI inflows. Trade, tourism, and market size play positive but less significant roles in attracting FDI for the BRICS, stressing that a negative relationship exists between FDI and inflation rate. Ref. [74] noted that RE has a neutral effect on FDI. Instead, the institutional environment and land availability are the core factors that stimulate FDI. Ref. [75] noted that a long-run relationship exists between FDI, RE, and economic growth for some selected nine countries identified in the Climate Change Performance Index 2018 report

Ref. [76] estimation of data from G-C economies based on data sourced from 1978 to 2014 shows that capital market expansion and trade openness are the leading drivers of CO_2 emission. The results further noted that CO_2 is respectively related to RE adoption (see also ref. [77]. Their results tilt toward the pollution haven hypothesis

In agriculture, ref. [78] shows that a long-run relationship exists between agricultural land expansion and CO_2 emission in Peru though RE improves environmental quality by reducing CO_2 emission. Ref. [79] noted that a positive relationship exists between agriculture and RE, but no such relationship is found to exist between agriculture and CO_2 for the economies of the US, Canada, China, and Poland. Ref. [80] noted that a bidirectional relationship exists between energy and agriculture for the EU. Ref. [81] noted that agriculture, RE, trade, and globalization negatively impact CO_2 emissions in Turkey. The study tilts toward the pollution haven hypothesis for Turkey.

The theoretical note from the environmental strands can be classified into two main types: The Environmental Kuznets Curve and the pollution haven hypotheses. The EKC opined that the relationship between economic growth and environmental pollution is in the form of an inverted U-shaped, such that at the early stage of a nation's economic growth, environmental pollution deepens, and after reaching a certain threshold level, environmental pollution begins to decline. The proponents of this hypothesis are of the view that at the initial stage of development, economies are concerned with achieving economic growth with less concern for protecting the environmental pollution, and attention begins to shift towards achieving clean energy [41,42,82,83].

A variety of these models has been canvassed in the literature focusing on CO₂ emissions as indicators of environmental pollution [42,84]. Some have calibrated the ecological footprint [49,85]. Recent studies have calibrated macroeconomic and finance-related variables to the studies on EKC [86]. The discussion on the relevance of EKC is continuous and yet to be concluded.

The pollution haven hypothesis (PHH) is the view that multinational companies that engage in rigorous pollution fields prefer to move to developing countries with fewer environmental/ecological protection laws. The reverse of the pollution haven hypothesis is the pollution halo hypothesis, which states that FDI could induce a downward trend in CO_2 emission, hence promoting energy-efficient technology usage that revolved around sustainability methods. Accordingly, it is believed that FDI can positively impact the ecosystem of an economy in three channels: scale effect (economic size), technical effect (improved technology), and structural effect (improvement in manufacturing design). The interaction of these effects will improve growth and reduce CO_2 emissions. The proponent of this hypothesis has identified FDI and yawning for development as the key drivers of CO_2 emissions in developing economies [7,42,47]. Closeness to colonial masters by former colonies and globalization, among others, are the reasons that account for the movement of multinational firms with toxic production outlets to less developing economies [72].

The studies on the impact of ecological footprints suggest that a functional relationship exists between the ecological footprint and several variables. For instance, ref. [86] noted that financial debt and renewable energy help reduce environmental degradation and that financial debt, RE, and NRE positively impact the growth of the 15 highest emitting economies. Ref. [86] noted that economic growth and national resources advance the ecological footprint and that human capital in the current state cannot mitigate environmental deterioration. Though RE does decrease ecological footprint, the study established the existence of feedback causality between human capital, urbanization, and ecological footprint. Ref. [31] noted that RE decreases ecological footprint in the long run in Turkey and that a bi-directional relationship exists between RE and economic growth and ecological footprint.

The theoretical note on impact assessment focuses on the role of governance and other socio-political factors in shaping the choice of energy usage to achieve carbon neutrality. The proponents of this thought believe that climate change is a global public issue and requires effective climate governance to address it [87–90]. As noted by ref. [91], energy governance is key to decoupling carbon emissions as it is vital to promoting RE adoption. For a sample of 36 emerging economies, ref. [92] observed that good governance especially economic and institutional governance is key to mitigating CO₂ emission and progressive adoption of RE. Ref. [93] designed a novel, holistic analytical approach to examine energy access governance for the Southern African economies of Uganda and Zambia by employing three data collection methods: qualitative document analysis, semistructured stakeholder interviews, and closed surveys. The study noted that the rule of law, transparency standards, accountability, and inclusiveness are key to accessing RE for the studied economies. The study also noted that competing regulatory frameworks distort access to RE. Ref. [90] cautioned on the danger of monopolized power in designing and implementing RE for the economies of Nepal and Indonesia. The authors noted that RE designed in the studied economies was bedeviled with the inability to carry the major stakeholders along in its design and running.

Ref. [94] calibrated the role of corruption perception and political governance in energy consumption-economic growth nexus for a team of 49 economies using a dynamic data environment analysis model based on data sourced from 2007 to 2016. The study noted that political governance proxied by political stability, bureaucratic quality, personal safety and security of private property, and legal and regulatory frameworks positively impact energy consumption.

Ref. [89] employed machine learning techniques to analyze the impact of green governance on renewable energy consumption in India and noted that governance structure influences the adoption of energy choices. The study further noted that the taxonomy of green governance proxy by global governance, adaptive governance, climate governance, ecological governance, self-governance, energy governance, and information technology governance are related and work on the same objectives by pursuing different activities.

For Switzerland, ref. [95] examined the role of public awareness and governance structure in the effective transition from nonrenewable energy consumption to renewable

energy sources. The study noted that public awareness and good governance are crucial to the effective transition and adoption of RE (see also ref. [88]. Ref. [96] explored the role of both internal and external governance structures in the adoption of renewable energy for some selected 1027 firms spread across 47 economies/regions. The study noted that internal governance structure tends to have a negative influence on RE adoption as it often induces a declining influence on RE, whereas external governance has a negative impact.

In Brazil, ref. [97] employed quantitative measures to access the nexus between water, energy, food, and land as it affects the adoption of biofuels emanating from sugarcane. The study concluded that each of these factors is key to achieving sustainable/green energy adoption in the studied economy. A study by ref. [97] was further expanded by ref. [98], who calibrated the role of geopolitics in adopting RE in Mexico. The study employed an external multi-regional input-output model (EMRIO) that calibrates import dependence and governance quality into the RE adoption framework for the Mexican economy. The study noted that better governance is key to the successful adoption and implementation of RE in the studied economy.

Ref. [50] noted that for governance structure and effectiveness to influence the adoption of RE positively, there is a need to have a holistic view of the consequences of RE adoption by calibrating natural resources extortion into the equation. The study argued that evidence abounds to show that the transition from a fossil-dominated system towards RE will have negative consequences on metal by more than a fraction of 7 by 2050 when compared with the 2015 levels, especially in economies with weak, poor, and failing resource governance up to between 32 and 40%.

Ref. [99] noted that political interference in environmental management, poor or lack of effective implementation, and lack of political independence of environmental agencies, which increases the risk of consumption, are the main factors militating against the adoption of RE in Brazil (see also ref. [100]).

A critical look at the literature reviewed here suggests that little or no study has been conducted on the determinants of renewable energy consumption in Africa, and their findings are inconclusive. This is what the current study aims to do, and by extension, calibrate the role of environmental factors, economic growth, and socio-political factors to study RE adoption in Africa.

3. Materials and Methods

The section presents the data-generating set and sources. It also presents the methodology employed and the justification for employing it.

3.1. Data

The data for the current study were sourced from several reputable global data outlets. For instance, we obtained data on macroeconomic variables, including real gross domestic product (RGDP), foreign direct investment (FDI), financial development (FD), trade (TRD), and government spending (GOVT) from the World Development Indicators (various issues). The data on the inflation rate (INF) was sourced from the United Nations Statistics (UN Data). We sourced data on agricultural output (AGRIC) from the Economic Research Service of the US Department of Agriculture (USDA). Data on environmental factors proxy by CO_2 emission (thousand kt) and ecological footprint were sourced from the BP Statistical Review of the World Energy (various issues). To account for the impact of socio-political factors, we calibrated the impact of governance effectiveness (GOVE) and urbanization (URB) into our model. Data on these variables were sourced from the World Development Indicators (various issues). We also account for the impact of the life expectancy index, education index, and corruption perception index as part of our socio-political factors in shaping the adoption of RE. Data on the education index (EI) and life expectancy index (LEI) were sourced from the United Nations Development Program (UNDP) development reports (various issues). Meanwhile, data on the corruption perception index (COR) was sourced from the Transparency International database. To address issues relating to heteroscedasticity, we

standardized our variables by obtaining their natural log forms. Data on renewable energy were sourced from the International Energy Agency database (various issues).

A Priori Expectations

Theoretically, we expect a positive relationship between RE and each RGDP, financial development, trade openness, agricultural output, and FDI. A negative relationship is expected to exist between RE and inflation. On environmental variables, we expect an inverse relationship between RE and CO_2 . On socio-political constructs, we expect that a positive relationship should exist between government effectiveness and RE. The relationship between RE and urbanization could be either positive (if the people are en vironmentally conscious) or negative if smart cities and the environment are neglected. The relationship between RE and other variables could be either way.

3.2. Methodology

The essence of the current study is to examine the determinant of RE adoption in some selected African economies. The study explored the neoclassical production function employed by [20,26,101] to develop the model for the study as stated below:

$$RE_{it} = \alpha_1 RE_{it-1} + \beta_1 W_{it} + \beta_2 X_{it} + \beta_3 Z_{it} + \delta_t + \emptyset_i + \varepsilon_{it}$$
(1)

where RE_{it} is the renewable energy consumption per capita, W is the proxy of all macroeconomic variables that can influence renewable energy consumption, X is the proxy of all environmental factors that can influence renewable energy consumption, Z is the proxy of all socio-political factors that can influence renewable energy consumption, α and β are the coefficients of the model, \emptyset_i is the time-invariant country effects, δ_t is the unobservable time effects, ε is the residual term, t is the time period. The GMM estimation techniques proposed by Arellano and Bond for our model, as stated in Equation (2), are as follows:

$$\mathsf{E}(y_{it-s} - \Delta u_{it}) = 0 \text{ for } t = 3, \dots, T \text{ and } 2 \le T$$
(2)

Here, y_{it-s} is the suitable lags of the dependent variables. The implication is that the second and further lags of the dependent variables are employed as an instrument for the residual of Equation (1) in differences. As noted by refs. [20,59,102], the estimator of Equation (2) is prone to a huge small sample bias, given the fact that the number of periods is small, with the dependent variables presenting a high degree of persistence. To address this, our study employed the system GMM model as suggested by refs. [40,101,103]. The model is as follows

$$E(\Delta d_{it-s} - (\delta_i - u_{it})) = 0 \text{ for } t = 3, \dots, T$$
(3)

It becomes unattractive and inappropriate to employ the ordinary least squares (OLS), fixed effects, or random effects because of the presence of lagged endogenous variable y, t-1 in Equation (1), given that y_{it} is correlated with δ_i and it induces upward biases, which is inconsistent with the OLS assumption of independence of the error term from the regressors [101,104–107]. To address this problem, the extant literature on dynamic panel models employed the Arellano and Bond GMM estimation model that employs an internal mechanism to explore the correlation between y_i , t-1, and δ_i . The GMM techniques remove \emptyset_i in short dynamic panels such as Equation (1) by differencing it first. To obtain a relatively consistent estimator, we employed lagged values of the levels of the independent variables as the predetermined variables [26,108]. In specific, when \emptyset_i (I = 1,2,..., n) are serially uncorrelated, then the second and higher-order lags of the independent variables are valid instruments. Extant literature has shown that a major problem of the [104] GMM model is that it produces poor instruments for the regressors when the regressors display persistence over time. To overcome this challenge, Arellano and Bond 1995, developed a system GMM that can estimate two sets of equations: (i) A set of levels that employ lags of the regressors in first differences as instruments; (ii) a set of equations in first differences that employs lags of the regressors in level as instruments. From the narrative, it can be

deduced that the system GMM is superior and appropriate for our model when compared with the difference GMM in at least three areas. (i.) It reduces endogeneity bias. (ii.) It reduces time-varying measurement error bias. (iii.) It reduces weak instrument error bias.

The current study employed the system GMM to address the endogeneity in the data generating set that could occur as a result of including y, t - 1, an indication that RE consumption and many of the other regressors may be jointly determined by the growth rate of the GDP, as well as the possibility of measurement errors that could occur because of employing cross country data that displays high persistence. To examine the validity of the orthogonality assumption of system GMM, we employed the Hansen test of over-identification and the Arellano and Bond tests for second-order and higher-order several correlations AR(2) test, given that system GMM techniques rely on internal instruments. The study adopted the [109] small sample correction of the standard errors for all the two-step system analyses, as suggested by ref. [110]. Some of the variables in some of the studied economies are heterogeneous; hence, we employed a full PMG method for the short-run nexus following [111–113].

4. Results and Discussion

We present the results of the current study in two parts. The first part focuses on the nexus between RE, economic growth, environmental factors, and socio-political factors in the selected African economies, based on system GMM estimation techniques. The second is our analysis focused on the country-specific output of these relationships in each studied economy.

4.1. System GMM Estimates

We present the descriptive analysis results of the relationship between the variables explored in Table 1. The results suggest that FDI has the lowest mean, while corruption has the highest mean value. The result, as presented by standard deviation, suggests that corruption has the highest standard deviation, while RGDP has the lowest standard deviation.

Variables]	Descriptiv	e Analysis	6	Normality Analysis (Natural Log-Form)				
	Mean	Max.	Min.	SD	Skewness	Kurtosis	Jarque-Bera	Probability	
InRE	1655.98	2368.77	1032.67	367.09	-0.98	2.04	4.77	0.06	
InRGDP	0.59	0.88	0.04	0.51	-0.34	3.19	0.72	0.67	
InFDI	46.44	64.07	34.57	5.12	0.55	2.04	3.53	0.16	
InFD	1438.78	1968.09	1011.11	214.09	-0.56	3.05	478	0.07	
InTRD	1.65	2.32	1.33	0.34	0.13	1.45	3.02	0.24	
InGOVSD	2.86	5.01	1.18	0.44	0.04	1.63	2.16	0.25	
InINF	2.11	5.11	1.15	0.24	0.03	1.43	2.13	0.15	
InAGR	11.98	2.07	12.67	7.09	0.71	2.02	4.74	0.04	
InC02	12.59	2.88	4.04	0.41	0.23	3.29	0.73	0.47	
ECL	43.41	62.01	32.53	5.23	0.51	2.14	3.23	0.36	
GOVE	1.78	1.99	1.21	4.49	0.36	3.11	112	0.05	
InURB	1.62	2.12	1.31	0.31	0.14	1.15	3.01	0.21	
LEI	2.11	3.11	1.12	0.14	0.15	1.03	2.13	0.23	
EI	15.18	24.7	2.27	2.04	0.18	1.04	4.01	0.05	
COR	0.44	0.14	0.14	0.31	0.31	3.14	0.32	0.17	

Table 1. Descriptive statistic.

Source: Author's computation 2023.

We present the results of the impact of our independent variables on the dependent variable (RE) in Table 2. From the results, as shown in columns (1–5), the results of the OLS, fixed effects, baseline system GMM, and alternative system estimates are presented respectively for robustness purposes. As earlier noted, OLS estimation of Equation (1) induces upward bias for the lagged per RE, while fixed effects induce a downward bias. Empirically, a valid estimate is expected to lie between the OLS and fixed effects [20,60,65,110]. Our results, as presented in column 3 of Table 2, suggests that the two-step system GMM coefficient on the lagged RE is -1.582, and it is between the upward-biased OLS estimates of

-1.143 and downward-biased fixed effect estimates of 5.446. The results also suggests that our estimation is negative and highly significant. This suggests the existence of conditional convergence across the selected African economies studied (See also [111]).

Dep. Variable: RE	[1] OLS	[2] Fixed Effect	[3] SYSGMM1	[4] SYSGMM2	[5] SYSGMM3
LDE	1.644	1.754	2.671	2.601	1.908
INKE	[0.012]	[0.051]	[0.132]	[0.102]	[0.014]
Impond	0.169	0.167	0.176	0.164	0.161
lingdi	[0.012]	[0.004]	[0.004]	[0.006]	[0.023]
INEDI	0.015	0.017	0.019	0.014	0.298
IIIFDI	[0.015]	[0.051]	[0.002]	[0.004]	[0.013]
InFD	0.042	0.029	0.027	0.074	0.043
шгр	[0.071]	[0.052]	[0.025]	[0.062]	[0.016]
InTRD	0.056	0.058	0.061	0.059	0.058
IIIIKD	[0.051]	[0.051]	[0.005]	[0.015]	[0.016]
InCOVSD	0.171	0.166	0.158	0.158	0.155
IIGOV5D	[0.007]	[0.005]	[0.004]	[0.013]	[0.026]
InINE	0.152	0.144	0.132	0.112	0.111
ΠΠΝΓ	[0.042]	[0.007]	[0.007]	[0.007]	[0.035]
InACR	0.162	0.122	0.133	0.144	0.151
IIIAGK	[0.005]	[0.041]	[0.008]	[0.007]	[0.027]
InC02	0.111	0.124	0.129	0.131	0.078
IIIC02	[0.023]	[0.042]	[0.016]	[0.035]	[0.043]
FCI	0.155	0.142	0.151	0.102	0.098
LCL	[0.007]	[0.034]	[0.015]	[0.006]	[0.045]
COVE	0.169	0.152	0.158	0.158	0.156
GOVE	[0.014]	[0.034]	[0.021]	[0.031]	[0.044]
InURB	0.168	0.157	0.172	0.156	0.117
IIIOND	[0.008]	[0.007]	[0.021]	[0.126]	[0.034]
IFI	0.177	0.163	0.164	0.161	0.111
	[0.021]	[0.043]	[0.013]	[0.114]	[0.036]
FI	0.198	0.187	0.211	0.201	0.188
LI	[0.035]	[0.036]	[0.019]	[0.119]	[0.026]
COR	0.188	0.177	0.199	0.177	0.167
COR	[0.015]	[0.016]	[0.016]	[0.114]	[0.015]
Intercent	11.066 ***	32.044			
intercept	[3.544]	[7.633]			
$\Delta R(2)$ test			-2.544	-1.432	-1.435
111(2)(00)			[0.105]	[0.113]	[0.109]
Hansen test			33.014	32.189	66.712
			[1.000]	[1.000]	[1.000]

Table 2. Baseline system GMM results.

Note: standard errors are reported in []; *** represent 1%. Source: Author's computation 2023.

In column [3] of Table 2, we present the results of the impact of economic growth on RE for the selected economies based on the two-step system GMM. We obtained an estimate of 0.1015 with a 1% level of significance. This suggests that economic growth promotes RE adoption in the selected economies. We validated our results by testing for over-identification restrictions and second-order serial correlation based on AR(2) test and the Hansen test. The ρ -value result of the AR(2) at 0.104 rules out the possibility of second or higher-order serial correlation in the residuals. The results of the Hansen test for over-identification further validated the instruments employed. A look at the results of other explanatory variates suggests that our results are in line with relevant economic theory and existing empirical findings. For instance, the coefficients of financial development are positive and significant, suggesting that financial development aids the consumption of renewable energy. This supports the finding of [26,69]. The results from each agriculture and trade openness are positive and significant, suggesting that each of them positively supports RE adoption [71]. The results from trade suggest that trade policies such as market liberalization that supports international trade advance the adoption of RE in Africa and, by extension, advance economic growth in the region (see ref. [112]). The result on agriculture suggests that agriculture significantly aids RE adoption in the studied economies and is in line with the findings of [79,81]. As expected, the results on the relationship between RE adoption and inflation are negative and significant. This suggests that with rising inflation, peoples' adoption of RE will be slow as the purchasing power ability of the people is eroded. Our result is in line with the findings of [113].

Our results are mixed for the other explanatory constructs (environmental and sociopolitical). For instance, while the results from CO_2 emission and ecological footprints are negative and significant, the result from urbanization is positive and significant. This suggests that as these economies get urbanized, the adoption of RE is embraced. This also connotes that, on average, men in these economies are environmentally conscious and friendly. Our results support the findings of [86] for the BRICS economies. The result of CO_2 emission suggests that CO_2 has an inverse relationship with RE in the studied economies (see also ref. [31]). On the socio-political factors, our results also show that governance structure has a positive and significant relationship with RE adoption, suggesting that with a good governance structure, more people will embrace the adoption of RE. The results agree with the findings of [89,92–94] but contradict [71] submission for sub-Saharan African economies.

4.2. Single-Country Estimates Results

Beyond panel estimation, we provide a single-country estimation in our model to account for the heterogeneous behavior of some variables in some of the selected African economies. Hence, our study followed [111–113] to employ a full PMG test for short-run nexus. Before we employed a full PMG model, we conducted unit root tests using In-Pesaran-Shin (IPS), Levin-Lin-Chu (LLC), and cross-sectional augmented Dickey-Fuller test. Our results suggest that none of the variables in the model is found to be I(2)). Results are available upon demand. We employed both the Pedroni test and the Westerlund test to conduct a cointegration estimate. The results show that the long-run estimates across all countries are stronger. Results are available upon demand. The result of the full PMG estimate showing country-specific estimate is presented in Table 3. From the results, it can be deduced that the impact of RGDP on RE is positive and significant for Nigeria, Ghana, Kenya, Ethiopia, Morocco, and South Africa, though a negative and significant relationship is noted for Algeria and DR Congo. The results of other macroeconomic variables are similar. For instance, a positive relationship exists between financial development and RE in Nigeria, South Africa, Ghana, and Kenya. The result also shows that foreign direct investment positively impacts renewable energy in Nigeria, Ghana, and South Africa. Trade openness supports RE adoption in Ghana, SA, Kenya, and Nigeria. The inflation rate negatively impacts RE for all the economies studied. On environmental factors, both CO_2 and ecological footprint has a negative and significant impact on RE for Angola, Tanzania, Ivory Coast, Nigeria, Ethiopia, Kenya, and DR Congo. Socio-political factors exhibit some level of positive impact on RE adoption. For instance, urbanization impacts RE adoption positively in South Africa, Nigeria, Ghana, Kenya, Ethiopia, and Angola. Life expectancy and education index impact RE positively and significantly for economies such as Nigeria, South Africa, Tanzania, Kenya, Angola, Ethiopia, and Egypt. The result for corruption index offers mixed results. For instance, it was reported that a negative relationship exists between corruption and RE adoption in Nigeria, Kenya, Ivory Coast, DR Congo, Egypt, and South Africa. Overall examination of our results suggests that most of the studied economies exhibit interacting trends in the adoption of RE.

Countries	Variables	Coefficients	Countries	Variables	Coefficients
	ECT	0.147 *** (0.027)	Egypt	ECT	0.162 *** (0.023)
	InRGDP	0.063 *** (0.047)		InRGDP	0.043 *** (0.044)
	InFDI	0.042 ** (0.052)		InFDI	0.032 *** (0.053)
	InFD	0.201 * (0.054)		InFD	0.211 * (0.034)
	InTRD	0.0032 *** (0.017)		InTRD	0.0043 *** (0.014)
	InGOVSD	0.0116 *** (0.036)		InGOVSD	0.0115 *** (0.033)
	InINF	-0.017 *** (0.027)		InINF	-0.014 *** (0.024)
Nigeria	InAGR	0.033 *** (0.047)		InAGR	0.023 *** (0.023)
	InC02	-0.012 ** (0.027)		InC02	-0.011 ** (0.024)
	ECL	-0.031 * (0.046)		ECL	-0.021 * (0.043)
	GOVE	0.0158 *** (0.019)		GOVE	0.0128 *** (0.014)
	InURB	-0.019 ** (0.21)		InURB	0.0101 ** (0.011)
	LEI	0.014 *** (0.011)		LEI.	0.013 *** (0.012)
	EI	013 *** (0.003)		EI	0.014 *** (0.006)
	COR	-0.012 ** (0.012)		COR	-0.042 ** (0.013)
	ECT	0.011 * (0.024)	Algeria	ECT	0.014 * (0.021)
	InRGDP	0.032 *** (0.017)		InRGDP	-0.034 *** (0.013)
	InFDI	0.118 *** (0.036)		InFDI	0.114 *** (0.033)
	InFD	0.107 *** (0.017)		InFD	0.104 *** (0.013)
	InTRD	0.013 *** (0.017)		InTRD	0.014 *** (0.015)
	InGOVSD	0.012 ** (0.017)		InGOVSD	-0.013 ** (0.018)
	InINF	-0.031 * (0.042)		InINF	-0.033 * (0.045)
South Africa	InAGR	0.018 *** (0.013)		InAGR	0.015 *** (0.014)
	InC02	-0.109 *** (0.21)		InC02	0.106 *** (0.021)
	ECL	0.0113 *** (0.002)		ECL	0.0115 *** (0.006)
	GOVE	0.013 *** (0.017)		GOVE	0.015 *** (0.014)
	InURB	0.112 ** (0.012)		InURB	0.115 ** (0.014)
	LEI	0.016 * (0.051)		LEI	0.017 * (0.031)
	EI	0.027 *** (0.014)		EI	0.023 *** (0.016)
	COR	-0.111 *** (0.006)		COR	-0.117 *** (0.016)
	ECT	0.117 *** (0.012)	Morocco	ECT	0.114 *** (0.016)
	InRGDP	0.05 *** (0.041)		InRGDP	0.04 *** (0.021)
	InFDI	0.012 ** (0.021)		InFDI	0.011 ** (0.011)
	InFD	0.011 *(0.041)		InFD	0.021 * (0.011)
	InTRD	0.011 *** (0.019)		InTRD	0.013 *** (0.014)
	InGOVSD	0.119 *** (0.21)		InGOVSD	0.114 *** (0.021)
	InINF	-0.015 *** (0.025)		InINF	-0.014 *** (0.022)
Kenya	InAGR	0.13 *** (0.004)		InAGR	0.14 *** (0.004)
	InC02	-0.12 ** (0.012)		InC02	-0.14 ** (0.016)
	ECL	0.012 * (0.014)		ECL	0.015 *** (0.016)
	GOVE	0.012 *** (0.013)		GOVE	0.012 *** (0.016)
	InURB	0.111 *** (0.012)		InURB	0.113 *** (0.015)
	LEI.	0.101 *** (0.02		LEI	0.102 *** (0.031)
	EI.	0.013 *** (041)		EI	0.014 *** (0.051)
	COR	-0.014 ** (0.021)		COR	0.015 *** (0.041)

 Table 3. Country-specific full short-run PMG estimates.

Countries	Variables	Coefficients	Countries	Variables	Coefficients
Ghana	ECT	0.033 * (0.046)	Angola	ECT	0.07 *** (0.021)
	InRGDP	0.113 *** (0.019)	0	InRGDP	0.015 ** (0.041)
	InFDI.	001 *** (0.001)		InFDI	0.015 * (0.031)
	InFD	0.012(0.002)		InFD	0.014 *** (0.014)
	InTRD	0.102 *** (0.011)		InTRD	0.114 *** (0.021)
	InGOVSD	0.013 *** (0.011)		InGOVSD	0.013 *** (0.024)
	InINF	0.112 ** (0.072)		InINF	0.015 *** (0.006)
	InAGR	0.011 * (0.001)		InAGR	0.012 ** (0.016)
	InC02	0.023 *** (0.017)		InC02	0.042 ** (0.014)
	ECL	0.016 *** (0.031)		ECL	0.016 *** (0.017)
	GOVE.	0.017 *** (0.021)		GOVE	0.114 *** (0.015)
	InURB	0.053 *** (0.04)		InURB	0.104 *** (0.051)
	LEI	0.012 ** (0.022)		LEI	0.014 *** (0.051)
	EI	0.012 * (0.041)		EI	0.013 *** (0.023)
	COR	0.0118 *** (0.019)		COR	0.005 *** (0.031)
Ethiopia	ECT	0.019 *** (0.021)	Tanzania	ECT	0.015 *** (0.021)
-	InRGDP	0.02 * (0.014)		InRGDP	0.014 ** (0.021)
	InFDI	0.011 *** (0.017)		InFDI	0.013 *** (0.014)
	InFD	0.113 *** (0.026)		InFD	0.116 *** (0.21)
	InTRD	0.101 *** (0.011)		InTRD	0.017 *** (0.021)
	InGOVSD	0.013 *** (0.012)		InGOVSD	0.16 *** (0.004)
	InINF	0.014 ** (0.014)		InINF	0.111 ** (0.012)
	InAGR	0.032 * (0.043)		InAGR	0.014 ** (0.014)
	InC02	0.015 *** (0.012)		InC02	0.017 *** (0.014)
	ECL	0.105 *** (0.021)		ECL	0.114 *** (0.013)
	GOVE	0.0115 *** (0.006)		GOVE	0.0011 *** (0.021)
	InURB	0.016 *** (0.012)		InURB.	0.014 *** (0.044)
	LEI	022 ** (0.015)		LEI	0.013 *** (0.021)
	EI	0.017 * (0.052)		EI	0.06 *** (0.051)
	COR	0.024 *** (0.018)		COR	0.015 *** (0.051)
Ivory Coast	ECT	0.113 *** (0.006)	DR Congo	ECT	0.012 * (0.042)
	InRGDP	0.031 * (0.014)		InRGDP	-0.013 *** (0.014)
	InFDI	0.011 *** (0.013)		InFDI	0.114 *** (0.021)
	InFD	0.113 *** (0.032)		InFD	0.013 *** (0.024)
	InTRD	0.107 *** (0.014)		InTRD	0.13 *** (0.004)
	InGOVSD	0.013 *** (0.014)		InGOVSD	0.012 *** (0.012)
	InINF	-0.012 ** (0.017)		InINF	-0.012 ** (0.014)
	InAGR	0.031 * (0.022)		InAGR	0.015 *** (0.015)
	InC02	-0.015 *** (0.014)		InC02	-0.114 *** (0.013)
	ECL	0.105 *** (0.021)		ECL	0.103 *** (0.041)
	GOVE	0.0113 *** (0.002)		GOVE	0.015 *** (0.061)
	InURB	0.013 *** (0.011)		InURB	0.013 ** (0.051)
	LEI	0.112 *** (0.012)		LEI	0.112 *** (0.005)
	EI	0.014 *** (0.031)		EI	0.111 *** (0.014)
	COR	-0.022 *** (0.013)		COR	-0.033 *** (0.014)

Table 3. Cont.

Source: Author's computation 2023. Note: *, **, *** represent 10%, 5%, 1% significant level respectively.

5. Conclusions

Decarbonization of the energy sector is at the front burner of the 21st-century energy adoption policy among economies across the world. This is essential to achieving the global quest for sustainable growth via renewable energy that helps in mitigating climate change [10]. A good understanding of the drivers of RE adoption is key to achieving success in RE growth which is essential to attaining sustainable growth. As noted earlier, four macroeconomic-energy possibilities exist in the literature on the link between economic growth and energy. They are the energy-led growth-following hypothesis, growth-led energy-following hypothesis, feedback/bi-directional hypothesis, and neutrality/indifference hypothesis, with each of these possibilities offering unique implications for policy modeling. Beyond macroeconomic variables, environmental factors (such as CO₂ emission and ecological footprint) and socio-political factors (education, life expectancy, and urbanization) often determine the adoption of RE. This paper contributes to the literature by employing appropriate estimation techniques—system GMM and full PMG—to examine the factors that influence the deployment of RE in the African sub-region based on data sourced from 1980 to 2019. Our choice of the system GMM was driven by the possibility that RE adoption and other control variables employed in our model could be jointly determined. The system GMM employed can deal with endogeneity-related issues. It can address the susceptibility of data to measurement error among other things. Validating the orthogonality assumptions in the system GMM, the study employed the Hansen test of over-identification, the Arellano and Bond (2000) test of second-order serial correlation, and the small sample correction of the standard errors.

The results of our estimation techniques and the results from a series of robust tests reveal that the relationship between RE and economic growth in the studied economies is positive and significant, suggesting that RE promotes economic growth on the one hand and economic growth promotes RE adoption on the studied economies, supporting the validity of feedback hypothesis in the studied economies. Hence, policies that support RE adoption should be advanced. A look at other explanatory variables suggests that a positive relationship exists between financial development, foreign direct investment, trade, governance, urbanization, and life expectancy that stimulates RE adoption.

This suggests that for these economies to achieve sustainable growth powered by RE, policymakers need to implement policies that will promote financial development, enhance trade, promote urbanization, and promote education and governance structure. The results of the nexus between inflation and each economic growth and RE adoption are negative, suggesting that policymakers should lower the inflation rate to promote economic growth influenced by RE adoption.

The second strand of our analysis focused on country-specific estimates. From the results, it can be deduced that the results obtained in country-specific estimation are not too far from the ones obtained at the aggregate level. For instance, the connection between RE and economic growth is positive and significant for the economies of SA, Nigeria, and Kenya, suggesting the possibility of a feedback hypothesis.

Though this study has advanced literature by examining the drivers of RE adoption from the point of economic, environmental, and socio-political views, there is a need for further empirical analysis on the subject to further enhance the knowledge of this nexus. Hence, we suggest that further study could examine the impact of RE on total energy adoption in Africa. Different estimation techniques can also be employed.

Funding: This research received no external funding.

Data Availability Statement: We obtained data on macroeconomic variables, including real gross domestic product (RGDP), foreign direct investment (FDI), financial development (FD), trade (TRD), and government spending (GOVT) from the World Development Indicators (various issues). The data on the inflation rate (INF) was sourced from the United Nations Statistics (UN Data). We sourced data on agricultural output (AGRIC) from the Economic Research Service of the US Department of Agriculture (USDA). Data on environmental factors proxy by CO2 emission (thousand kt) and ecological footprint were sourced from the BP Statistical Review of the World Energy (various issues). To account for the impact of socio-political factors, we calibrated the impact of governance effectiveness (GOVE) and urbanization (URB) into our model. Data on these variables were sourced from the World Development Indicators (various issues). We also account for the impact of the life expectancy index, education index, and corruption perception index as part of our socio-political factors in shaping the adoption of RE. Data on the education index (EI) and life expectancy index (LEI) were sourced from the United Nations Development Program (UNDP) development reports (various issues). Meanwhile, data on the corruption perception index (COR) was sourced from the Transparency International database. To address issues relating to heteroscedasticity, we standardized our variables by obtaining their natural log forms. Data on renewable energy were sourced from the International Energy Agency database (various issues).

Conflicts of Interest: The author declares no conflict of interest.

References

- Pueyo, A. What constrains renewable energy investment in Sub-Saharan Africa? A comparison of Kenya and Ghana. *World Dev.* 2018, 109, 85–100. [CrossRef]
- 2. Djellouli, N.; Abdelli, L.; Elheddad, M.; Ahmed, R. The effects of non-renewable energy, renewable energy, economic growth, and foreign direct investment on the sustainability of African countries. *Renew. Energy* **2022**, *183*, 676–686. [CrossRef]
- Mungai, E.M.; Ndiritu, S.W.; Da, I. Resources, Environment and Sustainability Unlocking climate finance potential and policy barriers—A case of renewable energy and energy efficiency in Sub-Saharan Africa. *Resour. Environ. Sustain.* 2022, 7, 100043. [CrossRef]
- 4. Elahi, E.; Khalid, Z. Estimating smart energy inputs packages using hybrid optimisation technique to mitigate environmental emissions of commercial fish farms. *Appl. Energy* **2022**, *326*, 119602. [CrossRef]
- Elahi, E.; Khalid, Z.; Zubair, M.; Zhang, H.; Lirong, X. Technovation Extreme weather events risk to crop-production and the adaptation of innovative management strategies to mitigate the risk: A retrospective survey of rural Punjab, Pakistan. *Technovation* 2022, 117, 102255. [CrossRef]
- 6. Schwerho, G.; Sy, M. Financing renewable energy in Africa—Key challenge of the sustainable development goals. *Renew. Sustain. Energy Rev.* **2017**, *75*, 393–401. [CrossRef]
- 7. Opeyemi, A.; Uchenna, E.; Simplice, A.; Evans, O. Renewable energy, trade performance and the conditional role of fi nance and institutional capacity in sub-Sahara African countries. *Energy Policy* **2019**, *132*, 490–498. [CrossRef]
- 8. Krupa, J.; Poudineh, R.; Harvey, L.D.D. Renewable electricity fi nance in the resource-rich countries of the Middle East and North Africa: A case study on the Gulf Cooperation Council. *Energy* **2019**, *166*, 1047–1062. [CrossRef]
- 9. Njoh, A.J. Renewable energy as a determinant of inter-country differentials in CO 2 emissions in Africa. *Renew. Energy* **2021**, 172, 1225–1232. [CrossRef]
- 10. Đaković, D.D.; Gvozdenac-Urošević, B.D.; Vasić, G.M. Multi-criteria analysis as a support for national energy policy regarding the use of biomass: Case study of Serbia. *Therm. Sci.* **2016**, *20*, 371–380. [CrossRef]
- 11. Alasinrin Babatunde, K.; Mahmoud, M.A.; Ibrahim, N.; Said, F.F. Malaysia's Electricity Decarbonisation Pathways: Exploring the Role of Renewable Energy Policies Using Agent-Based Modelling. *Energies* **2023**, *16*, 1720. [CrossRef]
- 12. Enescu, D.; Ciocia, A.; Galappaththi, U.I.K.; Wickramasinghe, H.; Alagna, F.; Amato, A.; Francisco, D.; Spertino, F.; Cocina, V. Energy Tariff Policies for Renewable Energy Development: Comparison between Selected European Countries and Sri Lanka. *Energies* **2023**, *16*, 1727. [CrossRef]
- 13. Zhang, N.; Kang, C.; Du, E.; Wang, Y. Generation Expansion Planning. In *Analytics and Optimization for Renewable Energy Integration*, 1st ed.; CRC Press: Boca Raton, FL, USA, 2019; Chapter 15; pp. 327–344. [CrossRef]
- 14. Anantharajah, K.; Setyowati, A.B. Energy Research & Social Science Beyond promises: Realities of climate finance justice and energy transitions in Asia and the Pacific. *Energy Res. Soc. Sci.* **2022**, *89*, 102550. [CrossRef]
- 15. Stoeglehner, G. Integrated spatial and energy planning in Styria—A role model for local and regional energy transition and climate protection policies. *Renew. Sustain. Energy Rev.* 2022, *165*, 112587. [CrossRef]
- 16. Lawal, A.I. The Nexus between Economic Growth, Energy Consumption, Agricultural Output, and CO 2 in Africa: Evidence from Frequency Domain Estimates. *Energies* 2023, *16*, 1239. [CrossRef]
- 17. Naderi, A.; Marriner, N.; Shari, A.; Azizpour, J.; Kabiri, K.; Djamali, M.; Kirman, A. Heliyon Climate change: A driver of future con fl icts in the Persian Gulf Region? *Heliyon* 2021, 7, e06288. [CrossRef]
- Ikhuoso, O.A.; Adegbeye, M.J.; Elghandour, M.M.Y.; Mellado, M.; Al-dobaib, S.N.; Salem, A.Z.M. Climate change and agriculture: The competition for limited resources amidst crop farmers-livestock herding con fl ict in Nigeria—A review. *J. Clean. Prod.* 2020, 272, 123104. [CrossRef]
- 19. Adeleye, B.N.; Id, R.O.; Lawal, A.I.; Alwis, T. De Energy use and the role of per capita income on carbon emissions in African countries. *PLoS ONE* **2021**, *16*, e0259488. [CrossRef]
- 20. Isola, A.; Ozturk, I.; Olanipekun, I.O.; John, A. Examining the linkages between electricity consumption and economic growth in African economies. *Energy* **2020**, *208*, 118363. [CrossRef]
- 21. Fatai, F.; Ozturk, I.; Oluwatoyin, M.; Agboola, P.O.; Victor, F. The implications of renewable and non-renewable energy generating in Sub-Saharan Africa: The role of economic policy uncertainties. *Energy Policy* **2021**, *150*, 112115. [CrossRef]
- 22. Charfeddine, L.; Kahia, M. Do information and communication technology and renewable energy use matter for carbon dioxide emissions reduction? Evidence from the Middle East and North Africa region. *J. Clean. Prod.* **2021**, 327, 129410. [CrossRef]
- 23. Wesseh, P.K., Jr.; Lin, B. Can African countries of fi ciently build their economies on renewable energy? *Renew. Sustain. Energy Rev.* **2016**, 54, 161–173. [CrossRef]
- 24. Chris, C.; Odikpo, F.; Adesoji, A.; Mayowa, E. Renewable energy in Nigeria: Potentials and challenges. *J. Southwest Jiaotong Univ.* **2021**, *56*, 528–539.
- 25. Ajayi, O.O.; Mokryani, G.; Edun, B.M. Sustainable energy for national climate change, food security and employment opportunities: Implications for Nigeria. *Fuel Commun.* **2022**, *10*, 100045. [CrossRef]
- 26. Le, T.; Nguyen, C.P.; Park, D. Energy Research & Social Science Financing renewable energy development: Insights from 55 countries. *Energy Res. Soc. Sci.* **2020**, *68*, 101537. [CrossRef]

- 27. Ibrahiem, D.M.; Hanafy, S.A. Do energy security and environmental quality contribute to renewable energy? The role of trade openness and energy use in North African countries. *Renew. Energy* **2021**, *179*, 667–678. [CrossRef]
- Ma, M.; Velayutham, E. Renewable and non-renewable energy consumption-economic growth nexus: New evidence from South Asia. *Renew. Energy* 2022, 147, 399–408. [CrossRef]
- 29. Zaman, K.; Shahbaz, M.; Loganathan, N.; Ali, S. Tourism development, energy consumption and Environmental Kuznets Curve: Trivariate analysis in the panel of developed and developing countries. *Tour. Manag.* **2016**, *54*, 275–283. [CrossRef]
- 30. Bhattacharya, M.; Reddy, S.; Ozturk, I.; Bhattacharya, S. The effect of renewable energy consumption on economic growth: Evidence from top 38 countries. *Appl. Energy* **2016**, *162*, 733–741. [CrossRef]
- Sharif, A.; Baris-tuzemen, O.; Uzuner, G.; Ozturk, I.; Sinha, A. Revisiting the role of renewable and non-renewable energy consumption on Turkey' s ecological footprint: Evidence from Quantile ARDL approach. *Sustain. Cities Soc.* 2020, 57, 102138. [CrossRef]
- 32. Akintande, O.J.; Olubusoye, O.E.; Adenikinju, A.F.; Olanrewaju, B.T. Modeling the determinants of renewable energy consumption: Evidence from the five most populous nations in Africa. *Energy* **2020**, *206*, 117992. [CrossRef]
- 33. Anton, S.; Nucu, A.E. The effect of financial development on renewable energy consumption. A panel data approach. *Renew. Energy* **2019**, *147*, 330–338.
- 34. Assi, A.F.; Isiksal, A.Z.; Tursoy, T. Renewable energy consumption, fi nancial development, environmental pollution, and innovations in the ASEAN b 3 group: Evidence from (P-ARDL) model. *Renew. Energy* **2021**, *165*, 689–700. [CrossRef]
- Ghazouani, T.; Boukhatem, J.; Yan, C. Causal interactions between trade openness, renewable electricity consumption, and economic growth in Asia-Pacific countries: Fresh evidence from a bootstrap ARDL approach. *Renew. Sustain. Energy Rev.* 2020, 133, 110094. [CrossRef]
- 36. Saidi, K.; Omri, A. The impact of renewable energy on carbon emissions and economic growth in 15 major renewable energyconsuming countries. *Environ. Res.* 2020, 186, 109567. [CrossRef]
- 37. Nassani, A.A.; Moinuddin, M.; Abro, Q.; Batool, R.; Haider, S.; Shah, A.; Hyder, S.; Zaman, K. Go-for-green policies: The role of finance and trade for sustainable development. *Int. J. Financ. Econ.* **2020**, *26*, 1409–1423. [CrossRef]
- Ansari, M.A.; Haider, S.; Khan, N.A. Environmental Kuznets curve revisited: An analysis using ecological and material footprint. *Ecol. Indic.* 2020, 115, 106416. [CrossRef]
- 39. Sharif, A.; Ali, S.; Ozturk, I.; Afshan, S. The dynamic relationship of renewable and nonrenewable energy consumption with carbon emission: A global study with the application of heterogeneous panel estimations. *Renew. Energy* **2019**, *133*, 685–691. [CrossRef]
- 40. Ozcan, B.; Ozturk, I. Renewable energy consumption-economic growth nexus in emerging countries: A bootstrap panel causality test. *Renew. Sustain. Energy Rev.* **2019**, *104*, 30–37. [CrossRef]
- 41. Cai, H.; Mei, Y.; Chen, J.; Wu, Z.; Lan, L.; Zhu, D. An analysis of the relation between water pollution and economic growth in China by considering the contemporaneous correlation of water pollutants. *J. Clean. Prod.* **2020**, *276*, 122783. [CrossRef]
- 42. Jiang, W.; Cole, M.; Sun, J.; Wang, S. Innovation, carbon emissions and the pollution haven hypothesis: Climate capitalism and global re-interpretations. *J. Environ. Manage.* **2022**, 307, 114465. [CrossRef] [PubMed]
- Hilfa, N.; Mohamad, A.; Fakhzan, N.; Khalid, N.; Helmi, M. Resources, Conservation & Recycling Effects of agriculture, renewable energy, and economic growth on carbon dioxide emissions: Evidence of the environmental Kuznets curve. *Resour. Conserv. Recycl.* 2020, 160, 104879. [CrossRef]
- 44. Komendantova, N.; Neumueller, S.; Nkoana, E. Public attitudes, co-production and polycentric governance in energy policy. *Energy Policy* **2021**, *153*, 112241. [CrossRef]
- 45. Müller, F.; Claar, S. Energy Research & Social Science Is green a Pan-African colour? Mapping African renewable energy policies and transitions in 34 countries. *Energy Res. Soc. Sci.* 2020, *68*, 101551. [CrossRef]
- 46. Falchetta, G.; Dagnachew, A.G.; Hof, A.F.; Milne, D.J. Energy for Sustainable Development The role of regulatory, market and governance risk for electricity access investment in sub-Saharan Africa. *Energy Sustain. Dev.* **2021**, *62*, 136–150. [CrossRef]
- 47. Li, W.; Qiao, Y.; Li, X.; Wang, Y. Energy consumption, pollution haven hypothesis, and Environmental Kuznets Curve: Examining the environment e economy link in belt and road initiative countries. *Energy* **2022**, *239*, 122559. [CrossRef]
- 48. Sharma, R.; Sinha, A.; Kautish, P. Does renewable energy consumption reduce ecological footprint? Evidence from eight developing countries of Asia. J. Clean. Prod. 2021, 285, 124867. [CrossRef]
- 49. Yilanci, V.; Korkut, U. Convergence of per capita ecological footprint among the ASEAN-5 countries: Evidence from a non-linear panel unit root test. *Ecol. Indic.* 2020, *113*, 106178. [CrossRef]
- 50. Watari, T.; Nansai, K.; Nakajima, K.; Giurco, D. Sustainable energy transitions require enhanced resource governance. J. Clean. Prod. 2021, 312, 127698. [CrossRef]
- 51. Hdom, A.D. Examining carbon dioxide emissions, fossil & renewable electricity generation and economic growth: Evidence from a panel of South American countries. *Renew. Energy* **2019**, *139*, 186–197. [CrossRef]
- 52. Alam, M.; Murad, W.; Hanifa, A.; Ozturk, I. Relationships among carbon emissions, economic growth, energy consumption and population growth: Testing Environmental Kuznets Curve hypothesis for Brazil, China, India and Indonesia. *Ecol. Indic.* **2016**, *70*, 466–479. [CrossRef]
- Bekun, F.V.; Agboola, M.O. Electricity Consumption and Economic Growth Nexus: Evidence from Maki Cointegration. *Eng. Econ.* 2019, 30, 14–23. [CrossRef]

- 54. Emir, F.; Bekun, F.V. Energy intensity, carbon emissions, renewable energy, and economic growth nexus: New insights from Romania. *Energy Environ.* **2019**, *30*, 427–443. [CrossRef]
- World Bank Group. World Development Indicators. Available online: https://databank.worldbank.org/source/worlddevelopment-indicators (accessed on 3 January 2023).
- 56. World Bank Group. World Development Report; World Bank: Washington, DC, USA, 2021; ISBN 9781464816000.
- Huang, Y.; Hsu, J.; Sun, L. A Study of Energy Efficiency and Mitigation of Carbon Emission: Implication of Decomposing Energy Intensity of Manufacturing Sector in Taiwan. *Int. J. Energy Econ. Policy* 2017, 7, 26–33.
- Asumadu, S.; Ozturk, I. Investigating the Environmental Kuznets Curve hypothesis in Kenya: A multivariate analysis. *Renew. Sustain. Energy Rev.* 2020, 117, 109481. [CrossRef]
- 59. Heid, B.; Langer, J.; Larch, M. Income and democracy: Evidence from system GMM estimates. *Econ. Lett.* **2012**, *116*, 166–169. [CrossRef]
- Gaies, B.; Kaabia, O.; Ayadi, R.; Guesmi, K.; Abid, I. Financial development and energy consumption: Is the MENA region different? *Energy Policy* 2019, 135, 111000. [CrossRef]
- 61. Wang, Q.; Zhang, C.; Li, R. Towards carbon neutrality by improving carbon efficiency—A system-GMM dynamic panel analysis for 131 countries' carbon efficiency. *Energy* **2022**, 258, 124880. [CrossRef]
- 62. Zhou, H.; Xu, G. Research on the impact of green finance on China's regional ecological development based on system GMM model. *Resour. Policy* **2022**, *75*, 102454. [CrossRef]
- 63. Sun, H.; Chen, F. The impact of green finance on China's regional energy consumption structure based on system GMM. *Resour. Policy* **2022**, *76*, 102588. [CrossRef]
- 64. Mittal, A.; Garg, A.K. Bank stocks inform higher growth—A System GMM analysis of ten emerging markets in Asia. *Q. Rev. Econ. Financ.* **2021**, *79*, 210–220. [CrossRef]
- 65. Berk, I.; Kasman, A.; Kılınç, D. Towards a common renewable future: The System-GMM approach to assess the convergence in renewable energy consumption of EU countries. *Energy Econ.* **2020**, *87*, 103922. [CrossRef]
- 66. Foon, C.; Wah, B.; Ozturk, I. Energy consumption and economic growth in Vietnam. *Renew. Sustain. Energy Rev.* 2016, 54, 1506–1514. [CrossRef]
- Adelaja, A.O. Barriers to national renewable energy policy adoption: Insights from a case study of Nigeria. *Energy Strateg. Rev.* 2020, 30, 100519. [CrossRef]
- 68. Le, T.; Le, H.; Taghizadeh-hesary, F. Does financial inclusion impact CO 2 emissions? Evidence from Asia. *Financ. Res. Lett.* **2020**, *34*, 101451. [CrossRef]
- 69. Li, Z.; Kuo, T.; Siao-yun, W.; The, L. Role of green finance, volatility and risk in promoting the investments in Renewable Energy Resources in the post-covid-19. *Resour. Policy* **2022**, *76*, 102563. [CrossRef]
- Asongu, S.A.; Odhiambo, N.M. Inequality, fi nance and renewable energy consumption in Sub-Saharan Africa. *Renew. Energy* 2021, 165, 678–688. [CrossRef]
- Chirambo, D. Towards the achievement of SDG 7 in sub-Saharan Africa: Creating synergies between Power Africa, Sustainable Energy for All and climate fi nance in-order to achieve universal energy access before 2030. *Renew. Sustain. Energy Rev.* 2018, 94, 600–608. [CrossRef]
- 72. Wang, H.; Luo, Q. Can a colonial legacy explain the pollution haven hypothesis? A city-level panel analysis. *Struct. Chang. Econ. Dyn.* **2022**, *60*, 482–495. [CrossRef]
- Azam, M.; Haseeb, M. Determinants of foreign direct investment in BRICS- does renewable and non-renewable energy matter? Energy Strateg. Rev. 2021, 35, 100638. [CrossRef]
- 74. Mahbub, T.; Faisal, M.; Tarba, S.Y.; Mallick, S.M.Y. Factors encouraging foreign direct investment (FDI) in the wind and solar energy sector in an emerging country. *Energy Strateg. Rev.* 2022, *41*, 100865. [CrossRef]
- 75. Caglar, A.E. The importance of renewable energy consumption and FDI in fl ows in reducing environmental degradation: Bootstrap ARDL bound test in selected 9 countries. *J. Clean. Prod.* **2020**, *264*, 121663. [CrossRef]
- 76. Khuong, D.; Luu, T.; Huynh, D.; Ali, M. Carbon emissions determinants and forecasting: Evidence from G6 countries. *J. Environ. Manage.* **2021**, *285*, 111988. [CrossRef]
- 77. Banerjee, S. Investigating India's pollution-intensive 'dirty' trade specialisation: Analysis with 'revealed symmetric comparative advantage' index. *Environ. Sci. Pollut. Res.* 2021, *28*, 30153–30167. [CrossRef]
- 78. Raihan, A.; Tuspekova, A. The nexus between economic growth, renewable energy use, agricultural land expansion, and carbon emissions: New insights from Peru. *Energy Nexus* **2022**, *6*, 100067. [CrossRef]
- 79. Saleem, M. Heliyon Possibility of utilizing agriculture biomass as a renewable and sustainable future energy source. *Heliyon* **2022**, *8*, e08905. [CrossRef]
- 80. Paris, B.; Vandorou, F.; Balafoutis, A.T.; Vaiopoulos, K.; Kyriakarakos, G.; Manolakos, D.; Papadakis, G. Energy use in open-field agriculture in the EU: A critical review recommending energy efficiency measures and renewable energy sources adoption. *Renew. Sustain. Energy Rev.* **2022**, *158*, 112098. [CrossRef]
- 81. Yurtkuran, S. The effect of agriculture, renewable energy production, and globalization on CO 2 emissions in Turkey: A bootstrap ARDL approach. *Renew. Energy* **2021**, *171*, 1236–1245. [CrossRef]
- Ozturk, I.; Al-mulali, U. Investigating the validity of the environmental Kuznets curve hypothesis in Cambodia. *Ecol. Indic.* 2015, 57, 324–330. [CrossRef]

- 83. Yao, S.; Zhang, S.; Zhang, X. Renewable energy, carbon emission and economic growth: A revised environmental Kuznets Curve perspective *. J. Clean. Prod. 2019, 235, 1338–1352. [CrossRef]
- 84. Streimikien, D.; Sun, K.; Bale, T. The impact of income inequality on consumption-based greenhouse gas emissions at the global level: A partially linear approach. *J. Environ. Manag.* **2020**, *267*, 110635. [CrossRef]
- Usman, M.; Sohail, M.; Makhdum, A.; Kousar, R. Does financial inclusion, renewable and non-renewable energy utilization accelerate ecological footprints and economic growth? Fresh evidence from 15 highest emitting countries. *Sustain. Cities Soc.* 2021, 65, 102590. [CrossRef]
- 86. Nathaniel, S.P.; Yalçiner, K.; Bekun, F.V. Assessing the environmental sustainability corridor: Linking natural resources, renewable energy, human capital, and ecological footprint in BRICS. *Resour. Policy* **2021**, *70*, 101924. [CrossRef]
- 87. Debbarma, J.; Choi, Y. A taxonomy of green governance: A qualitative and quantitative analysis towards sustainable development. *Sustain. Cities Soc.* 2022, 79, 103693. [CrossRef]
- 88. Ha, Y.; Sapkota, S. Energy Research & Social Science Investigating decentralized renewable energy systems under different governance approaches in Nepal and Indonesia: How does governance fail? *Energy Res. Soc. Sci.* **2021**, *80*, 102214. [CrossRef]
- Barrera-santana, J.; Marrero, G.A.; Ramos-real, F.J. Income, energy and the role of energy efficiency governance. *Energy Econ.* 2022, 108, 105882. [CrossRef]
- Omri, A.; Belaïd, F. Does renewable energy modulate the negative effect of environmental issues on the socio-economic welfare? J. Environ. Manag. 2021, 278, 111483. [CrossRef]
- Stritzke, S.; Trotter, P.A.; Twesigye, P. Towards responsive energy governance: Lessons from a holistic analysis of energy access in Uganda and Zambia. *Energy Policy* 2021, 148, 111934. [CrossRef]
- Lu, W.; Long, Q.; Nourani, M.; Lin, C. Political governance, corruption perceptions index, and national dynamic energy ef fi ciency. J. Clean. Prod. 2021, 295, 126505. [CrossRef]
- Ruef, F.; Ejderyan, O. Rowing, steering or anchoring? Public values for geothermal energy governance. *Energy Policy* 2021, 158, 112577. [CrossRef]
- Zhang, D.; Zhang, Z.; Ji, Q.; Lucey, B.; Liu, J. Journal of International Financial Markets, Institutions & Money Board characteristics, external governance and the use of renewable energy: International evidence. J. Int. Financ. Mark. Inst. Money 2021, 72, 101317. [CrossRef]
- 95. Luz, L.; Lazaro, B.; Luiz, L.; Bermann, C.; Giarolla, A.; Ometto, J. Policy and governance dynamics in the water-energy-food-land nexus of biofuels: Proposing a qualitative analysis model. *Renew. Sustain. Energy Rev.* **2021**, *149*, 111384. [CrossRef]
- 96. Escribano, G.; Lilliestam, J.; Rosa, A.; Lech, Y.; Lara, L. Assessing dependence and governance as value chain risks: Natural Gas versus Concentrated Solar power plants in Mexico. *Environ. Impact Assess. Rev.* **2022**, *93*, 106708. [CrossRef]
- 97. Abreu, M.; Soares, I.; Silva, S. ScienceDirect Governance quality and environmental policy on emergent, resource-rich economies: The case of Brazil. *Energy Rep.* **2022**, *8*, 70–75. [CrossRef]
- Tan, X.; Kong, L.; Gu, B.; Zeng, A.; Niu, M. ScienceDirect Research on the carbon neutrality governance under a polycentric approach. *Adv. Clim. Chang. Res.* 2022, 13, 159–168. [CrossRef]
- Aydin, M. Natural gas consumption and economic growth nexus for top 10 natural Gas e Consuming countries: A granger causality analysis in the frequency domain. *Energy* 2018, 165, 179–186. [CrossRef]
- 100. Pereira, P.; Cerqueira, P.A. Assessing the determinants of household electricity prices in the EU: A system-GMM panel data approach. *Renew. Sustain. Energy Rev.* 2017, *73*, 1131–1137. [CrossRef]
- 101. Arellano, M.; Bover, O. Another look at the instrumental variable estimation of error-components models. *J. Econ.* **1995**, *68*, 29–51. [CrossRef]
- 102. Arellano, M.; Bond, S. Some tests of pecification for panel data: Monte Carlo evidence and an application to employment equations. *Rev. Econ. Stud.* **1991**, *58*, 277–297. [CrossRef]
- 103. Blundell, R.; Bond, S. GMM estimation with persistence panel data: An application to production functions. *Econ. Rev.* 2000, 19, 321–340. [CrossRef]
- 104. Blundell, R.; Bond, S. Initial conditions and moment restrictions in dynamic panel data models. J. Econ. 1998, 87, 115–143. [CrossRef]
- Aydin, M.; Korkut, U. Are shocks to disaggregated renewable energy consumption permanent or temporary for the USA? Wavelet based unit root test with smooth structural shifts. *Energy* 2020, 207, 118245. [CrossRef]
- 106. Versmissen, J.; Zietz, J. Is there a leverage target for REITs? Q. Rev. Econ. Financ. 2017, 66, 57–69. [CrossRef]
- 107. Windmeijer, F. A finite sample correction for the variance of linear efficient two-step GMM estimators. *J. Econom.* **2005**, *126*, 25–51. [CrossRef]
- 108. Atems, B.; Hotaling, C. The effect of renewable and nonrenewable electricity generation on economic growth. *Energy Policy* **2018**, *112*, 111–118. [CrossRef]
- 109. Oluoch, S.; Lal, P.; Susaeta, A. Investigating factors affecting renewable energy consumption: A panel data analysis in Sub Saharan Africa. *Environ. Chall.* **2021**, *4*, 100092. [CrossRef]
- 110. Fromentin, V. The long-run and short-run impacts of remittances on financial development in developing countries. *Q. Rev. Econ. Financ.* **2017**, *66*, 192–201. [CrossRef]

- 111. Mensah, I.A.; Sun, M.; Gao, C.; Omari-Sasu, A.Y.; Zhu, D.; Ampimah, B.C.; Quarcoo, A. Analysis on the nexus of economic growth, fossil fuel energy consumption, CO2 emissions and oil price in Africa based on a PMG panel ARDL approach. *J. Clean. Prod.* 2019, 228, 161–174. [CrossRef]
- 112. Lawal, A.I.; Nwanji, T.I.; Asaleye, A.; Ahmed, V. Economic growth, financial development and trade openness in Nigeria: An application of the ARDL bound testing approach. *Cogent Econ. Financ.* **2016**, *4*, 1258810. [CrossRef]
- 113. Abubakr, M.; Thu, T.; Nguyen, H. Asymmetric relationship between green bonds and commodities: Evidence from extreme quantile approach. *Financ. Res. Lett.* **2021**, *43*, 101983. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.





Article Medium- and High-Tech Export and Renewable Energy Consumption: Non-Linear Evidence from the ASEAN Countries

Cong Khai Dinh¹, Quang Thanh Ngo¹ and Trung Thanh Nguyen ^{1,2,*}

- ¹ School of Government, University of Economics Ho Chi Minh City, Ho Chi Minh City 72407, Vietnam; khai@ueh.edu.vn (C.K.D.); thanhnq@ueh.edu.vn (Q.T.N.)
- ² Institute for Environmental Economics & World Trade, Leibniz University Hannover, 30167 Hannover, Germany
- * Correspondence: thanh.nguyen@iuw.uni-hannover.de or thanhnt@ueh.edu.vn

Abstract: Sustaining economic growth while reducing dependence on fossil fuels remains a challenge for our world to fight against climate change and therefore finding a way to promote economic growth and increase renewable energy use is needed. This paper uses a 22-year panel dataset (1994–2015) of 9 countries in the Association of Southeast Asian Nations provided by the World Bank World Development Indicators to examine the impact of medium- and high-tech export on renewable energy use. We employ a fixed-effects regression model with the Driscoll–Kraay nonparametric covariance matrix estimator to account for sectoral and temporal dependence. We also control for inflation, employment, population growth, and gross domestic product per capita in our estimations. Our results demonstrate a U-shaped association between medium- and high-tech export and renewable energy consumption of these economies. The results propose that enhancing medium- and high-tech export could be a feasible solution for promoting renewable energy consumption.

Keywords: renewable energy; medium- and high-tech export; economic growth; employment; inflation; ASEAN

1. Introduction

Sustaining economic growth while reducing dependence on fossil fuels remains a challenge in our era of climate change. In addition to the need for reducing emissions, continuingly increasing fossil fuel prices, fears of unaffordable and rapidly depleting sources of fossil fuels, and the desire to transitioning into a low carbon economy have combined to heighten the importance of renewable energy use [1]. Several countries have set a target of specific renewable energy share in their total energy consumption. For example, Germany aims to supply electricity solely from renewable energy sources by 2045. China also pledges to be carbon neutral by 2060 and sets the share of non-fossil fuels in primary energy consumption to around 25% by 2030 from a previous commitment of 20%. However, increasing renewable energy consumption is not easy for many developing countries, especially for rapidly growing economies as their demand for energy is increasing and their technical and financial capacities for a large-scale supply of renewable energy are limited. In this regard, looking for possible ways to increase renewable energy consumption while maintaining economic growth is required.

Industrialization generally results in a structural transformation from fossil fuelbased and low technology to clean energy-based and medium- and high technology. Medium- and high-tech industries are the value-added manufacturing sectors with higher technological intensity and productivity. They are referred to the level of technology that companies and industries producing goods with innovative qualifications and advanced technologies [2,3]. High technology industries include, for example, aviation and spacecraft industry, pharmaceutical industry, accounting and information processing technologies,

Citation: Dinh, C.K.; Ngo, Q.T.; Nguyen, T.T. Medium- and High-Tech Export and Renewable Energy Consumption: Non-Linear Evidence from the ASEAN Countries. *Energies* 2021, *14*, 4419. https:// doi.org/10.3390/en14154419

Academic Editor: Surender Reddy Salkuti

Received: 11 June 2021 Accepted: 19 July 2021 Published: 22 July 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). radio, television and communication equipment industry, and medical and optical devices industry [4]. Thus, the production and export of medium- and high-tech products are an important source of export-oriented growth and development, and of the transition to a low-carbon economy [5–7].

There have been several studies on socio-economic factors affecting renewable energy consumption [8–15]. However, new drivers such as medium- and high-tech export have been given much less attention, especially in the context of developing countries [16]. In addition, existing research mainly explores the competitiveness benefit of the policies on renewable energy use in conventional industrial sectors, such as in iron, steel, paper, and glass industries [17]. Globalization has facilitated trade among countries, and the export of medium- and high-tech products have been promoted in rapidly growing economies. However, so far, the causal effects of trade in general, on renewable energy use in both short- and long- terms are weak and scattered [16,18,19], and the effect of medium- and high-tech export on renewable energy use has not been investigated, especially in the context of rapidly growing economies. This is the main motivation for our study.

Theoretically, high- and medium-tech export affect renewable energy use through three channels. First, higher export of high- and medium-tech goods stimulate domestic production of these exported goods and hence economic growth. Increases in domestic production of these goods and economic growth change the energy demand as energy is a key input for production. This is referred to as the scale effect. Second, trade openness allows countries to exchange energy-saving and cleaner energy technologies, which are exported by developed economies and imported by developing economies [20–23]. Such exchange facilitates technological advancement. This is referred to as the technique effect. Third, economic growth leads to economic structural transformation which means that at the beginning of the transformation when the economy is largely agricultural-based, energy intensity is low. However, at a later stage when industrialization starts, energy intensity increases. At the same time, economic growth makes people better-off and increases their awareness of the environment, thus demand more medium and high-tech goods from environmentally friendly producers. This is referred to as the composite effect. The net effect depends on the stage of economic growth and the changes in consumption patterns of consumers. While developed economies have advantages in improving technologies for promoting renewable energy, developing countries are less able to do so and most of them may rely on technology transfers from developed countries, but for various reasons, technology transfers might be constrained. Therefore, at an earlier stage medium- and high-export from developing countries may still demand more fossil-based resources. Later on, renewable energy use will increase.

We focus our analysis on the nine countries of the Association of Southeast Asian Nations (ASEAN), including Cambodia, Indonesia, Laos, Malaysia, Myanmar, the Philippines, Singapore, Thailand, and Vietnam (hereafter referred to as ASEAN-9 countries). The ASEAN has 10 member countries but we purposely exclude Brunei as its energy consumption solely depends on fossil fuels. Even though these countries differ in several aspects, especially in terms of income per capita (Figure 1), they are commonly known as rapidly growing economies and thus have an increased energy demand. It is projected that the overall primary energy supply will increase from 621 Mtoe in 2015 to 1.544 Mtoe in 2050, an annual increase of 2.6%; and the gross final energy consumption will increase with an annual rate of 2.4%, from 436 Mtoe in 2015 to 1006 Mtoe in 2050. However, ASEAN remains highly reliant on fossil fuels. Nearly 80% of the global primary energy supply by 2050 are projected to adhere to fossil fuels. The heavy reliance on fossil fuels along with the decreasing domestic fossil fuel stocks would force the ASEAN's member states to import more fossil fuels. ASEAN is currently the 3rd largest economy in the Indo-Pacific and the 5th largest economy in the world. It has a combined gross domestic product (GDP) of \$US 2.8 trillion, and is also the 3rd fastest-growing major Indo-Pacific economy in the past decade, after China and India (Figure 2). As a critical hub for global trade, over \$3.4 trillion in global trade transits through the ASEAN region each year [24]. Their export of

medium- and high-tech has also been growing. However, the share of renewable energy in the total energy consumption of these countries is still modest. In this regard, examining the effect of medium- and high-tech export on renewable energy in ASEAN is of particular interest and relevant for policymakers and the public. We use a 22-year panel dataset (1994–2015) of these ASEAN-9 countries provided by the World Bank World Development Indicators to empirically examine the impact of medium- and high-tech export on the share of renewable energy use in total energy consumption of these countries. These countries are diverse in many aspects. For example, Singapore is an advanced economy, Indonesia and Thailand are upper-middle-income countries, Vietnam is a lower-middle-income country, and Cambodia and Laos belong to the group of the least developed countries. Following the arguments in the previous paragraph, we hypothesize that the relationship between medium- and high-tech export and renewable energy consumption in these ASEAN-9 countries is U-shaped.



Figure 1. Per capita GDP of the ASEAN economies as compared to the US in 2017. Source: East-West Center, 2019 [24].



Figure 2. Real GDP growth of ASEAN as compared to some other economies. Source: East-West Center, 2019 [24].

2. Literature Review

As natural resources in general and fossil fuels in particular, have been depleting, renewable energy use and economic growth are highlighted as major concerns for sustainable development of the global economy [25-29], given the fact that the sustainability of the economic system is dependent on the environment and natural resources [30]. Due to an increasing level of globalization, global energy demand has changed over time and has been driven by trade-related factors. Trade openness, export-oriented policies, and internationalization are considered crucial in the longterm development strategy of many developing countries [31,32]. In this context, recent literature has considered new determinants of energy demand and energy intensity such as export or import product portfolio [33,34], trade openness, and technological advancement [35-38]. In general, there is a controversy on the effect of trade on energy intensity. On the one hand, many studies find that trade openness positively affects energy intensity [39-45]. On the other hand, the effect can be adverse [46], or ambiguous [47] depending on competitiveness, factor price, and technology and infrastructure factors ([35,48]). Regarding the spectrum of trade and innovation, Samargandi [35] reveals that trade openness and innovation are significant factors for reducing energy intensity. Beser and Soyyigit [2] indicate that high-tech export has a strong impact on CO₂ emission in developed economies.

The importance of technology in determining energy use and energy efficiency in developing countries is increasing due to a growing level of internationalization and integration [22,49,50]. Technology innovation induced by investment in research and development and by foreign direct investment is supposed to increase energy use efficiency [40]. Domestic innovation is also an important contributor to technical development [49,51]. To mitigate the negative effects of climate change, technological progress is crucial [52], and increased R&D is associated with more technical innovation and renewable energy adoption [53] in both developed and developing countries, where renewable energy sources such as biomass, solar, wind and hydropower are adopted [54,55]. Recent studies on the relationship between technological advancement and carbon emission demonstrate positive effects of technology innovations on carbon emission reduction [56–58]). Liu, Xia, Tao, and Chen [56], for instance, analyze carbon emission in China and find that increasing technological expenditure could in turn increases carbon emission efficiency. Wang, Zhao, Wang, Guo, Kan, and Yuan [57] discover that investments in technology decrease carbon emissions. Zeng, Lu, Liu, Zhou, and Hu [58] find that foreign trade, foreign capital, and technological progress have positive effects on carbon emission reduction.

Several studies examine the drivers of renewable energy use. Most of these indicate the causal interactions between economic growth and renewable energy use [59–62], between renewable energy use and sustainable development [63-65], between energy use and trade openness, and between technological progress and renewable energy use. For example, Apergis and Payne [61] investigate the relationship between economic growth and renewable energy use of 20 OECD countries during 1985–2005 and find a bidirectional link between economic outcome and energy use. Fang and Chang [62] analyze the causality between energy use and economic performance in 16 countries of the Asia Pacific and finds a long-run cointegrating relationship. Kahia et al. [66] examine the relationship between energy use and economic growth, using a sample of 7 MENA Net Oil Importing Countries (NOICs) during 1980–2012; and their empirical results confirm the bidirectional causality between renewable energy use (and non-renewable energy use) and economic growth. Le and Sarkodie [67] investigate the nexus between renewable and conventional energy and economic growth, using panel data of 45 Emerging Market and Developing Economies (EMDEs) from 1990 to 2014. They find that renewable energy and GDP growth impact each other. Marques and Fuinhas [68], using a sample of 24 European countries during 1990-2006, find that the current level of renewable energy use is positively dependent on the previous level of renewable energy use. However, income and prices of fossilbased fuels are not significant for the development of renewable energy during the studied period. Ahmed et al. [69] investigate the interactions between renewable and non-renewable energies, CO₂ intensity, and economic growth in Myanmar during 1990–2016; and their results reveal that renewable energy use significantly promotes economic growth.

Few studies have also examined the relationship between trade openness, technological innovation, and (renewable) energy use (Alam and Murad [16], Sohag, Begum, Abdullah, and Jaafar [22], Sbia et al. [70,71], Khan et al. [72], Cole [47], Shahbaz, Nasreen, Ling, and Sbia [21]). Alam and Murad [16] reveal that economic growth, trade openness, and technological progress significantly influence renewable energy use in 25 OECD countries. Sohag, Begum, Abdullah, and Jaafar [22] employ a dataset during 1985–2012 in Malaysia and find that while economic growth and trade openness are the main determinants of energy use, technological innovation reduces energy use in manufacturing sectors. Sbia, Shahbaz, and Hamdi [70] examine the impacts of foreign direct investment, trade openness, clean energy price, carbon emissions, and economic growth on the demand for energy in the United Arab Emirates. Their findings indicate that trade openness and foreign direct investment reduce energy use because energy-efficient technologies have been employed. By comparing upper-middle-income countries in Asia, Europe, Africa, and America, Khan, Yaseen, and Ali [72] indicate that trade can induce technology transfer for renewable energy. Shahbaz, Nasreen, Ling, and Sbia [21] use the data from 91 high, middle, and low-income economies and conclude that domestic energy use is affected by trade openness through several channels such as technological transfers, economies of scale, and input factors. In high-income economies, an inverted U-shaped relationship between trade openness and energy consumption is found. According to Shahbaz, Nasreen, Ling, and Sbia [21], the U-shaped relationship between trade openness and energy consumption exists when low and middle-income countries import or use energy-efficient technologies from developed countries to lower energy consumption, on the one hand; and on the other hand, when developed countries allow to release those technologies and share profits for low and middle-income countries that have limited access to technology and capital.

A most recent study that is close to our work in terms of geographical coverage (for ASEAN with a 22-year span of time) is Azam, Khan, Zaman, and Ahmad [25], who find that trade openness has a positive relationship with energy consumption in Thailand, Malaysia, and Indonesia. Apart from trade openness, they discover that population growth increases the energy consumption in Malaysia, while it decreases energy consumption in Indonesia. Real GDP is found to have a positive relationship with energy consumption in Thailand, Malaysia, and Indonesia.

The causal effects between international trade (exports or imports) and renewable energy use in both short- and long- terms that have been found so far are weak [16]. The results from Sadorsky [20] for a sample of Middle Eastern countries show that international trade increases domestic use of energy. In addition, Shahbaz, Nasreen, Ling, and Sbia [21] conclude that a U-shaped relationship exists for high-income countries, whereas an inverted U-shaped relationship is found for middle- and low-income countries for the relationship between international trade and energy use.

In sum, none of the previous studies examine the impact of medium-and high-tech export on renewable energy use. Our study is thus aimed to contribute to filling this gap. The contribution of our study to the literature are two-fold. First, our study is the first effort to examine the effects of medium- and high-tech export on renewable energy share in total energy consumption for this group of rapidly expanding economies. To our best knowledge, only Shahbaz, Nasreen, Ling, and Sbia [21] discover non-linear relationships between trade openness and energy consumption for two country groups, high-income countries and middle- and low-income countries. Second, from a methodological perspective, we use panel data and employ a fixed-effects regression model with the Driscoll–Kraay nonparametric covariance matrix estimator to account for unobservable time-invariant

factors and sectoral and temporal dependence. These concerns have not been successfully addressed in many previous studies (Azam et al. [25]).

3. Data and Methods

3.1. Data Source

We extract the data needed for our study from the World Bank World Development Indicators (WDI). This database has been established for years in many countries. For the ASEAN countries, the data are available from 1994 to 2015. This allows us to establish a balanced panel dataset for these nine ASEAN countries. As explained above, we exclude Brunei from the sample since it is an outlier in terms of renewable energy use. Our variables of interest are medium- and high-tech export, and renewable energy use of these countries over time as we would like to investigate the association between these two important variables. From the literature review presented in the previous section, it is clear that a direct relationship between them can be established.

We use the share of renewable energy in the total energy consumption of each country in each year (% of total final energy consumption) to represent renewable energy use. It is the explained variable. For explanatory variables, the share of medium- and high-tech export in total manufactured export (% of manufactured export) is our key variable of interest. In addition, we control for inflation, employment, population growth, and GDP per capita. Inflation is measured as the change in the consumer price index (%, year 2010 is the base year, 2010 = 100), employment is measured as the share of employment in the industrial sectors in total employment of an economy in each year (% of total employment) while population growth is also reported annually for each country. The GDP per capita is measured in purchasing power parity (PPP, constant 2017). These variables are coded as follows: REU for renewable energy share (%), MHTE for medium- and high-tech export share in total manufactured export (%), INF for inflation (%), POPG for population growth (%), EMP for the employment share in industry (%), and GDPPC for GDP per capita (PPP, constant 2017). All these variables are annual for these ASEAN-9 countries from 1994 to 2015. A more detailed definition of these variables is in Appendix A. The description of the data for each country is presented in Table 1 whereas Table 2 summarizes the data for the whole block. These tables show the *REU* mean value is 42.673 (%), the mean value of *MHTE* is 37.168 (%), while the mean value of INF is 77.034 (%). On average, POPG is 1.575%, EMP is 17.915 (%), and GDPPC is 13,417 (USD PPP 2007). The correlation coefficients between these variables are in Appendix B which show that NHTE, INF, POPG, EMP, and GDPPC all have a negative association with *REU* at a 1% level of significance.

Table 1. Descriptive statistics of variables of each ASEAN country (1994–2015).

Country	REU	(%)	MHT	E (%)	INF	(%)	РОРС	G (%)	EMP	(%)	GDPPC	C (USD PPP, 2007)
	Mean	SD	Mean	SD	Mean	SD	Mean	SD	Mean	SD	Mean	SD
Cambodia	75.59	6.58	3.34	2.93	76.00	25.42	2.01	0.63	12.99	5.50	2118	789
Indonesia	42.65	4.63	29.22	3.70	69.59	36.22	1.38	0.08	19.03	1.29	7150	1501
Laos	75.94	9.94	5.45	4.34	68.51	40.83	1.72	0.27	6.43	2.43	3858	1329
Malaysia	5.79	1.63	67.45	6.13	88.99	13.80	2.00	0.42	30.19	2.17	18,262	3316
Myanmar	80.64	6.29	1.66	2.78	55.45	43.77	0.94	0.23	14.36	2.28	2080	1175
Philippines	32.64	3.78	73.81	10.44	79.34	23.13	1.94	0.27	15.51	0.39	5230	966
Singapore	0.51	0.08	68.87	16.55	92.89	11.308	2.33	1.44	25.03	4.45	65,782	14,060
Thailand	21.85	1.27	58.68	3.99	87.08	15.89	0.74	0.28	20.50	1.65	12,286	2386
Vietnam	48.45	11.12	26.04	10.84	75.46	37.01	1.13	0.26	17.20	4.26	3986	1337

Source: Authors' estimation based on WDI database; SD: Standard deviations.

** * 11	N.	SD					
Variable	Mean —	overall	between	within			
REU (%)	42.67	29.26	30.29	6.07			
MHTE (%)	37.17	29.70	30.26	8.02			
INF (%)	77.03	31.26	11.72	29.23			
POPG (%)	1.58	0.77	0.55	0.57			
EMP (%)	17.92	7.24	6.93	3.07			
GDPPC (USD PPP, 2007)	13,419	1983	20,340	4877			

Table 2. Descriptive statistics of variables of nine ASEAN countries (1994–2015).

Source: Authors' estimation based on WDI database; SD: Standard deviations; No. observations: 198; No. of countries: 9; No. of years: 22.

3.2. Methods

We employ an econometric analysis to examine the effect of the explanatory variable (medium- and high-tech export) on the explained variable (renewable energy share). Following previous studies, we control for GDP per capita, inflation, employment, and population growth as these factors have been found to influence energy use. They are the factors affecting energy demand. GPD per capita, inflation, and employment are key drivers of changes in purchasing power parity; and for population, Samargandi [35] argues that population growth positively influences energy usage and energy intensity, which might be harmful to the environment. The employment share of the industry might have either a positive or a negative on renewable energy use [73–75]. For the effects of these factors on energy use, see [7,16,20–22,35] for GDP, [73,74] for inflation, [75] for employment, and [35,76] for population growth. A factor affecting renewable energy use is its price. The declining price of renewable energy driven by technological progress could be important in increasing the renewable energy share in total energy consumption. Unfortunately, we cannot collect sufficient data. Oil price could be another significant factor. However, as some studies find the least influence of oil price on renewable energy production [7,8,11,20], or insignificant [77], especially in the case of oil-exporting countries like these ASEAN-9 countries, and/or in the case that these countries have subsidized oil prices to avoid any adverse effect of oil price fluctuations on the economy [11]. In addition, incorporating oil prices in our analysis could lead to an endogenous issue that must be addressed. Therefore, we excluded prices factors in our analysis.

Econometrically, the causal relationship could be examined using either a pooled ordinary least square (OLS) technique or the panel data method, including either a fixedeffects model (FEM) or a random-effects model (REM) [78]. To choose either the OLS technique or the panel data method, we conducted an F-test for the joint significance of differing group means. Results of the F-test presented in Appendix C (F (8, 184) = 76.97 with *p*-value 0.0000) indicate the null hypothesis that the pooled OLS model is appropriate is rejected. Thus, panel data analysis was chosen. We advanced further to choose either FEM or REM by performing a Hausman test. Results of this test in Appendix C indicate the FEM is a more suitable specification. From a theoretical point of view, FEM has the advantage of controlling for time-invariant unobservable factors. An alternative test to choose either FEM or REM was the overidentifying restriction test [79,80] which also indicates that the FEM model is a more suitable specification (results of this test in the last two rows in Appendix C). In addition, FEM is also recommended to estimate the parameters for a small cross-sectional sample [81] which is our case as we have only 9 countries. Our specification is also in line with previous studies on factors affecting renewable energy use such as Bamati and Raoofi [7] for 25 developed and developing countries, Alam and Murad [16] for 25 OECD countries, Azam, Khan, Zaman, and Ahmad [25] for three ASEAN countries (namely Indonesia, Malaysia, and Thailand); Kahia, Aïssa, and Lanouar [66] for 7 MENA Net Oil Importing Countries; Marques and Fuinhas [68] for 24 European countries; Sadorsky [20] for eight Middle Eastern countries; Bashir, Sheng, Doğan, Sarwar,
and Shahzad [31] for 29 OECD countries; Beser and Soyyigit [2] for G20 countries (except Russia), Chen, Du, Huang, and Huang [5] for 34 industrial sectors in China, and Waheed et al. [82] for 6 Gulf Cooperation and Council countries.

Therefore, our econometric regression model is specified as follows (Equation (1)):

$$REU_{it} = \alpha_0 + \beta_1 MHTE_{it} + \beta_2 INF_{it} + \beta_3 POPG_{it} + \beta_4 EMP_{it} + \beta_5 PGDPC_{it} + v_i + e_{it}$$
(1)

where *REU* is the share of renewable energy in total energy consumption; *MHTE* is the share of medium- and high-tech export in total export; *INF* is the inflation rate; *POPG* is the population growth rate; *EMP* is the employment share of industry (%), and *GDPPC* is GDP per capita (PPP, constant 2017) as defined in Section 3.1 (see Tables 1 and 2). Subscripts *i* and *t* denote country and time, respectively; α is the fixed country effect while v_i is the country-specific effect, and e_{it} is the error term.

Several tests were undertaken to ensure the validity of our regression model. First, to control for possible multicollinearity between explanatory variables, the variance inflation factor (VIF) values were checked [83,84] and the results documented in Appendix D do not signal that problem. Second, as our sample is small in terms of both the number of countries (9 countries) and the number of time periods (22 years), we undertook the following tests: (i) Modified Wald test for group-wise heteroskedasticity, (ii) the Breusch-Pagan LM test for cross-sectional dependence [85,86]; (iii) the slope homogeneity test introduced by Swamy [87], and (iv) the stationarity test for each variable.

Cross-sectional heterogeneity should be controlled for when conducting a panel data empirical analysis [88]. Swamy [87] proposes the homoskedasticity assumption test for the slope homogeneity assumption. Results of this test presented in Appendix E show that we can reject the null hypothesis of the slope homogeneity for our sample. In addition, results of the Breusch–Pagan LM test of independence (also in Appendix E) indicate that the null hypothesis of no cross-sectional independence is rejected at the 1% level of significance, indicating strong cross-sectional dependence.

Once there is cross-sectional dependence across countries in the panel, it is needed to perform the cross-sectionally augmented Dickey-Fuller (CADF) procedure from Pesaran [89]. Results presented in Appendix F show that we were not able to reject the null unit root hypothesis for the *GDPPC* series; but when taking the first difference, the null hypothesis of the unit root is rejected for variables POPG and INF. However, this is not sufficient for us to conclude that there is a long-run equilibrium relationship among the concerned variables, namely *REU*, *MHTE*, *INF*, *EMP*, *POPG*, and *GDPPC*.

Given the presence of cross-sectional dependence and/or heteroscedasticity, we adopted our FEM with Driscoll–Kraay standard errors. According to Hoechle [90], the Driscoll–Kraay standard errors are heteroskedasticity-, and autocorrelation-, consistent and robust to general forms of cross-sectional and temporal dependence.

4. Findings and Discussion

4.1. Findings

Table 3 presents the results of our estimation, including both a normal fixed effects specification and the fixed effects specification with the Driscoll–Kraay standard errors. The R squared value of 0.718 from these two specifications indicates that shows our model can predict about 72% of the variation in the share of renewable energy in total energy consumption.

Regarding the effect of medium- and high-tech export, Table 3 shows that the mediumand high-tech export has a significant U-shaped relationship with the share of renewable energy use of these ASEAN-9 countries. This U-shaped relationship implies that at the beginning stage of economic development, a higher level of medium- and high-tech export would lead to a lower share of non-renewable energy in total energy consumption. However, once the economy has reached a certain level of medium- and high-tech export, then the higher the level of medium- and high-tech export, the higher the share of renewable energy in the total energy consumption of that economy. In our case, the threshold value for the turning point of the U-shaped relationship is 64.47%. Within these ASEAN-9 countries, Malaysia and Singapore have passed this turning point since 1994, and the Philippines since 1995.

With regards to the controlled variables, inflation, employment in the industry sector, and GDP per capita are significantly and negatively associated with the share of renewable energy in total energy consumption, meanwhile, population growth is significantly and positively associated with the share of renewable energy.

Variable	Normal Fixed Effects Model	Fixed Effects Model with xxx Driscoll-Kraay Standard Errors
MHTE	-0.420 *** (0.105)	-0.420 ** (0.163)
MHTE, squared	0.0036 *** (0.0011)	0.0036 ** (0.0015)
INF	-0.0895 *** (0.0133)	-0.0895 *** (0.0130)
POPG	0.989 * (0.518)	0.989 * (0.476)
EMP	-0.719 *** (0.121)	-0.719 *** (0.199)
GDPPC	$-7.29 imes 10^{-5}$ (7.32 $ imes 10^{-5}$)	$-7.29 imes 10^{-5}$ (0.0001)
Constant	69.28 *** (2.460)	69.28 *** (3.043)
No. of observations	198	198
No. of time periods	22	22
No. of countries	9	9
F test that MHTE and MHTE squared are	F(2, 183) = 8.46	F(2, 8) = 3.32
jointly equal to zero	Prob > F = 0.0003	Prob > F = 0.0890
Overall R squared	0.718	0.718

Table 3. Impact of medium- and high-tech export on renewable energy share.

Source: Authors' estimation, standard errors in parentheses, ***, **, and * denote statistical significance at the 1%, 5%, and 10% levels, respectively.

4.2. Discussion

With respect to the influences of medium- and high-tech export on renewable energy use, our results indicate that medium- and high-tech export has a U-shaped association with renewable energy use in the ASEAN-9 countries. A couple of previous studies have not found any short- and long-term effects of trade on renewable energy use [8,91]. Our finding is a contribution to a new strand of literature on the non-linear effects of trade on resource use. This strand includes, for example, Shahbaz, Nasreen, Ling, and Sbia [21] who find the pattern of a U-shaped relationship exists in high-income countries, and an inverted U-shaped relationship in the middle- and low-income countries for the relationship between international trade and energy use.

Most previous studies so far have found only a positive effect of high-tech export. It is generally argued that under high-tech export orientation, there is peer-to-peer lending which enables them to adopt such technology that uses lower energy; in addition, different technologies and resources can be shared. Thus, energy consumption is also shared and reduced and it helps save energy costs, including renewable energy [2,7]. Bamati and Raoofi [7] provide an analysis of the influence of high-tech export on renewable energy production by levels of development and find that high-tech export increases renewable energy production for 10 developed countries, but for 15 developing countries the effect is insignificant. A stronger effect of high-tech export on developed countries rather than developing countries is also confirmed by Beser and Soyyigit [2] with a sample of the G20 countries (except Russia).

Our result can be explained by the fact that, in terms of medium- and high-tech export, most of the ASEAN-9 countries do not seem to have high shares of high-tech export or at least technologies that could affect promoting renewable energy use. This means, at a lower level of development, the effect is negative but later on, it turns out to be positive.

Renewable energy use is upon the spending and expectations toward the behaviors of consumers that are affected by different aspects of inflation [92–94]. While depicting the

concern of medium- and high-tech export toward renewable energy use, the controlled factors have a specific influence on renewable energy use in the ASEAN-9 countries. This is widely stated by the variables like GDP per capita and employment. Inflation is also counted as a major factor that influences renewable energy use. Inflation has a dominant impact on economic performance due to the implications of targeting the environment by high inflation [73]. Our results indicate that inflation has a negative relationship with renewable energy use. These results are supported by Kosai, Yuasa, and Yamasue [74]. Although prices of various products are associated with inflation and wide interpretation enhances the role of renewable energy use in the ASEAN-9 countries, a hike in prices of various goods and services induces a significant impact on renewable energy use. Therefore, inflation with the relevance of medium- and high-tech export inserts a negative role toward the renewable energy use of many ASEAN-9 countries [95]. It could be due to the devaluation of the currency that renders renewable energy use with a hike in prices. Inflation could help in uplifting the economies of various countries but could lead to some adverse effects too.

Our regression model shows that population growth has a positive effect on renewable energy use. In fact, the introduction of technology and innovation in renewable energy use has increased the demand of the population for renewable energy. While enumerating the ecological problems with energy-related factors, population growth is the main driver of environmental degradation [96]. Therefore, population growth might result in a negative contribution toward renewable energy use. Our results contribute, to some extent, to this disputable discussion. Recall that Azam, Khan, Zaman, and Ahmad [25] when examining the impact of various factors on energy consumption in three ASEAN countries (Indonesia, Malaysia, and Thailand) in the period 1980–2012, find that population growth increases the energy consumption in Malaysia, while it decreases energy consumption in Indonesia.

The advantage of renewable energy use is indisputable. It is upon the governments of these ASEAN-9 countries to induce needed measures to promote energy use. Potential impacts of employment on renewable energy use in various industries are evident with local renewable resources [97]. Many ASEAN-9 countries have contributed to a significant rise in the employment rate due to its influence on renewable energy use. For the regional development policy of renewable energy use, the employment regimes and challenges insert an important role [98]. The improper sharing of the economy has been eliminated by the positive enclosure of employment elements that widely induce technological innovation in ASEAN-9 countries.

The empirical results also show that GDP per capita has a negative but insignificant effect on renewable energy use. The result is in line with Marques and Fuinhas [68] who investigate drivers promoting renewable energy in 24 European countries and find that the per capita income (in natural logarithm) is not statistically significant in explaining the use of renewables. A negative effect is found by Waheed, Sarwar, and Mighri [82], and Samargandi [35]. Samargandi [35], for example, examines the impacts of trade openness, technological innovation, and energy price on energy intensity in OPEC countries, in which GDPPC is used as a controlling variable.

5. Conclusions

Our study investigates the influences of medium- and high-tech export on the renewable energy use of nine ASEAN countries in the period 1994–2015. We control for inflation, population growth, employment, and GDP per capita. Our findings suggest that mediumand high-tech export reduces the share of renewable energy in total energy consumption during an earlier stage of economic development but then increases the share of renewable energy consumption during the later stage of economic development. This seems to be a characterized feature observed in these ASEAN-9 countries, contributing to the complexity of trade-renewable energy nexus in the literature. Our study also elaborates that under high inflation, individuals and firms cannot afford costly high technology, effective techniques, and skilled human resources, which consume a smaller amount of renewable energy in production. Growth in the population provides human resources and at the same time promotes renewable energy use. High employment opportunities indicate high economic growth, which can help to save energy.

Our study has some limitations despite its theoretical and empirical importance. First, due to the availability of the data provided by the World Bank in the World Development indicator, we are not able to control for other factors such as oil prices and/or price of renewable energy that might have significant effects on renewable energy use. Therefore, our results should be interpreted with care. Second, our study is at the macro level and thus not on the behavior of individual energy users (i.e., firms and businesses). The changes in the behavior of energy users should be examined as well to provide a better understanding of energy transition. Third, our sample is small with only nine ASEAN countries in a short-term period. Expanding the study to cover more countries in a longer time period would provide a more comprehensive picture of the effect of medium- and high-tech export on renewable energy use. Last, we are unable to undertake a measurement uncertainty analysis [99]. These issues should be themes for future studies.

Author Contributions: Conceptualization, T.T.N., and Q.T.N.; methodology, T.T.N.; data curation, Q.T.N.; writing—original draft preparation, T.T.N., Q.T.N., C.K.D.; writing—review and editing, T.T.N., Q.T.N., C.K.D.; project administration, T.T.N.; funding acquisition, Q.T.N. All authors have read and agreed to the published version of the manuscript.

Funding: This research is funded by the University of Economics Ho Chi Minh City, Vietnam.

Data Availability Statement: The data presented in this study are available on request from the corresponding author.

Conflicts of Interest: The authors declare no conflict of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript, or in the decision to publish the results.

Variable	Definition & Measurement
REU	Share (%) of renewable energy consumption in total final energy consumption in
	a year of a country
INF	Change (%) in consumer price index with year 2010 as the base year (100%). It
	reflects changes in the cost to the average consumer of acquiring a basket of
	goods and services that may be fixed or changed at specified intervals. It is
	annual for each country.
EMP	Share (%) of employment in industry in total employment of an economy.
	Employment is defined as persons of working age who are engaged in any
	activity to produce goods or provide services for pay or profit, whether at work
	during the reference period or not at work due to temporary absence from a job,
	or to working-time arrangement. The industry sector consists of mining and
	quarrying, manufacturing, construction, and public utilities (electricity, gas, and
	water), in accordance with divisions 2–5 (ISIC 2) or categories C-F (ISIC 3) or
DODG	categories B-F (ISIC 4).
POPG	Annual growth (%) of the population of a country
GDPPC	GDP per capita measured in purchasing power parity (PPP) 2017. PPP GDP is
	gross domestic product converted to international dollars using purchasing
	power parity rates. An international dollar has the same purchasing power as a
	U.S. dollar has in the United States. GDP at purchaser's prices is the sum of
	gross value added by all resident producers in the country plus any product
	taxes and minus any subsidies not included in the value of the products. It is
	calculated without making deductions for depreciation of fabricated assets or for
	depiction and degradation of natural resources. Change $(0')$ of modium, and high task summation tasks around a sum out
	Share (//) of medium- and nign-tech export in total manufactured export
Source: WDI.	

Appendix A. Definition of Variables Used in the Analysis

Variable	REU	MHTE	INF	POPG	EMP	GDPPC
REU	1.0000					
MHTE	-0.9015 ***	1.0000				
INF	-0.4541 ***	0.3562 ***	1.0000			
POPG	-0.1662 ***	0.2578 ***	-0.1246 *	1.0000		
EMP	-0.8293 ***	0.6782 ***	0.4239 ***	0.07722	1.0000	
GDPPC	-0.6737 ***	0.5109 ***	0.3077 ***	0.2964 ***	0.4741 ***	1.0000
	10		1 1	1 0 4 1	1	

Appendix B. Correlation Matrix of the Variables Used in the Analysis

***, **, and *: Significant at the 1%, 5% and 10% level, respectively. Source: Authors' estimation.

Appendix C. F test, Hausman Test, and Test of Overidentifying Restrictions

F Test	Coef.
F test that all u_i = 0: F(8, 184)	76.97
<i>p</i> -value	0.0000
Hausman Test	Coef.
Chi-square test value	40.83
<i>p</i> -value	0.0000
Test of overidentifying restrictions: fixed vs. random effects	
Sargan–Hansen statistic Chi-sq(5)	51.628
<i>p</i> -value	0.0000
Source: Authors' estimation.	

Appendix D. Variance Inflation Factor (VIF) Value

Variable	VIF	1/VIF
MHTE	2.15	0.464597
EMP	2.08	0.479953
GDPPC	1.53	0.651664
INF	1.33	0.753238
POPG	1.23	0.813535
Mean VIF	1.67	

Source: Authors' estimation.

Appendix E. Cross-Sectional Dependence, Heteroskedasticity, and Slope Homogeneity Tests

Breusch-Pagan LM Test of Independence	Coef.
Chi-square test value: χ^2 (36)	134.156
<i>p</i> -value	0.0000
Modified Wald test for groupwise heteroskedasticity	
χ^2 (10)	509.54
<i>p</i> -value	0.0000
Swamy slope homogeneity test	
Chi-square test value: χ^2 (48)	41658.21
<i>p</i> -value	0.0000

Note: The null hypothesis of the cross-sectional dependence test is no cross-sectional dependence. The null hypothesis of the slope homogeneity test is slope homogeneity. The Cross-sectional dependence and slope homogeneity tests are conducted by using respectively 'xttest2' [100] and 'xtrchh2' [101] commands in Stata. Source: Authors' estimation.

17 1 1	Lev	Level (CIPS)		1st Difference (CADF)	
Variable —	Intercept	Intercept and Trend	Intercept	Intercept and Trend	
REU	-2.055	-1.883	-0.913	1.004	
MHTE	-0.865	-1.4615	1.977	1.2513	
INF	-1.683	-2.435	0.671	-1.499 *	
POPG	-3.124	-2.536	-2.927 ***	5.047	
EMP	-1.507	-1.670	1.081	1.339	
GDPPC	-1.437 ***	-1.784 ***	0.839	1.452	

Appendix F. Panel Unit Root Test

Note: *** and * denote statistical significance at the 1% and 10% levels, respectively. The null hypothesis is nonstationarity. The panel unit root test is conducted by using the 'xtcips' command in Stata [102]. Source: Authors' estimation.

References

- 1. Nguyen, T.T.; Tenhunen, J. Review of integrated ecological—Economic analyses for bioenergy plants under climate change at local scale. *Int. J. Clim. Chang. Strat. Manag.* **2013**, *5*, 324–343. [CrossRef]
- 2. Beser, N.O.; Soyyigit, S. The effects of high technology export and per capita income on carbon emission: An investigation on G20 countries. *Bus. Econ. Horiz.* 2019, *15*, 542–559.
- 3. Seyoum, B. The role of factor conditions in high-technology exports: An empirical examination. *J. High. Technol. Manag. Res.* 2004, 15, 145–162. [CrossRef]
- 4. OECD (Organisation for Economic Co-operation and Development). Technology Intensity Definition. Classification of manufacturing industries into categories based on R&D intensities. In OECD Directories for Science, Technology and Industry Economic Analysis and Statistics Division; OECD: Paris, France, 2011; Volume 9.
- 5. Chen, S.; Du, X.; Huang, J.; Huang, C. The impact of foreign and indigenous innovations on the energy intensity of China's industries. *Sustainability* **2019**, *11*, 1107. [CrossRef]
- 6. MacKenzie, J.J. Technology growth curves: A new approach to reducing global CO₂ emissions. *Energy Policy* **2003**, *31*, 1183–1187. [CrossRef]
- Bamati, N.; Raoofi, A. Development level and the impact of technological factor on renewable energy production. *Renew. Energy* 2020, 151, 946–955. [CrossRef]
- Sadorsky, P. Renewable energy consumption, CO₂ emissions and oil prices in the G7 countries. *Energy Econ.* 2009, 31, 456–462.
 [CrossRef]
- 9. Sadorsky, P. Renewable energy consumption and income in emerging economies. Energy Policy 2009, 37, 4021–4028. [CrossRef]
- 10. Payne, J.E. The causal dynamics between US renewable energy consumption, output, emissions, and oil prices. *Energy Sources Part B* **2012**, *7*, 323–330. [CrossRef]
- 11. Salim, R.A.; Rafiq, S. Why do some emerging economies proactively accelerate the adoption of renewable energy? *Energy Econ.* **2012**, *34*, 1051–1057. [CrossRef]
- 12. Lean, H.H.; Smyth, R. Will policies to promote renewable electricity generation be effective? Evidence from panel stationarity and unit root tests for 115 countries. *Renew. Sustain. Energy Rev.* 2013, 22, 371–379. [CrossRef]
- 13. Smyth, R. Are fluctuations in energy variables permanent or transitory? A survey of the literature on the integration properties of energy consumption and production. *Appl. Energy* **2013**, *104*, 371–378. [CrossRef]
- 14. Omri, A.; Nguyen, D.K. On the determinants of renewable energy consumption: International evidence. *Energy* **2014**, *72*, 554–560. [CrossRef]
- 15. Omri, A.; Daly, S.; Nguyen, D.K. A robust analysis of the relationship between renewable energy consumption and its main drivers. *Appl. Econ.* 2015, 47, 2913–2923. [CrossRef]
- 16. Alam, M.M.; Murad, M.W. The impacts of economic growth, trade openness and technological progress on renewable energy use in organization for economic co-operation and development countries. *Renew. Energy* **2020**, *145*, 382–390. [CrossRef]
- 17. Bakker, M.H.; Duncan, J.A. Future bottlenecks in international river basins: Where transboundary institutions, population growth and hydrological variability intersect. *Water Int.* **2017**, *42*, 400–424. [CrossRef]
- Hasanov, F.J.; Liddle, B.; Mikayilov, J. The impact of international trade on CO₂ emissions in oil exporting countries: Territory vs. consumption emissions accounting. *Energy Econ.* 2018, 74, 343–350. [CrossRef]
- 19. Mikayilov, J.I.; Galeotti, M.; Hasanov, F.J. The impact of economic growth on CO₂ emissions in Azerbaijan. *J. Clean. Prod.* **2018**, 197, 1558–1572. [CrossRef]
- 20. Sadorsky, P. Trade and energy consumption in the Middle East. Energy Econ. 2011, 33, 739–749. [CrossRef]
- 21. Shahbaz, M.; Nasreen, S.; Ling, C.H.; Sbia, R. Causality between trade openness and energy consumption: What causes what in high, middle and low income countries. *Energy Policy* **2014**, *70*, 126–143. [CrossRef]
- 22. Sohag, K.; Begum, R.; Abdullah, S.M.S.; Jaafar, M. Dynamics of energy use, technological innovation, economic growth and trade openness in Malaysia. *Energy* 2015, *90*, 1497–1507. [CrossRef]

- 23. Corcelli, F.; Fiorentino, G.; Vehmas, J.; Ulgiati, S. Energy efficiency and environmental assessment of papermaking from chemical pulp—A Finland case study. *J. Clean. Prod.* **2018**, *198*, 96–111. [CrossRef]
- 24. East-West Center. ASEAN Matters for America; East-West Center: Washington, DC, USA, 2019.
- 25. Azam, M.; Khan, A.Q.; Zaman, K.; Ahmad, M. Factors determining energy consumption: Evidence from Indonesia, Malaysia and Thailand. *Renew. Sustain. Energy Rev.* **2015**, *42*, 1123–1131. [CrossRef]
- Mohsin, M.; Kamran, H.W.; Nawaz, M.A.; Hussain, M.S.; Dahri, A.S. Assessing the impact of transition from non-renewable to renewable energy consumption on economic growth-environmental nexus from developing Asian countries. *J. Environ. Manag.* 2021, 284, 1–8. [CrossRef]
- 27. Androniceanu, A.-M.; Căplescu, R.; Tvaronavičienė, M.; Dobrin, C. The interdependencies between economic growth, energy consumption and pollution in Europe. *Energies* **2021**, *14*, 2577. [CrossRef]
- Grabara, J.; Tleppayev, A.; Dabylova, M.; Mihardjo, L.W.; Dacko-Pikiewicz, Z. Empirical research on the relationship amongst renewable energy consumption, economic growth and foreign direct investment in Kazakhstan and Uzbekistan. *Energies* 2021, 14, 332. [CrossRef]
- 29. Hsu, C.C.; Chien, F.; Ngo, Q.T.; Nguyen, T.D.; Vu, H.M.; Hugo, C. Nexus between energy security and economic development: Empirical evidence from BRICS economies. *Singap. Econ. Rev.* **2021**, *165*, 381–390. [CrossRef]
- 30. Nguyen, T.T.; Nguyen, T.-T.; Hoang, V.-N.; Wilson, C.; Managi, S. Energy transition, poverty and inequality in Vietnam. *Energy Policy* **2019**, *132*, 536–548. [CrossRef]
- 31. Bashir, M.A.; Sheng, B.; Doğan, B.; Sarwar, S.; Shahzad, U. Export product diversification and energy efficiency: Empirical evidence from OECD countries. *Struct. Chang. Econ. Dyn.* **2020**, *55*, 232–243. [CrossRef]
- 32. Ngo, Q.-T.; Nguyen, C.T. Do export transitions differently affect firm productivity? Evidence across Vietnamese manufacturing sectors. *Post Communist Econ.* **2020**, *32*, 1011–1037. [CrossRef]
- 33. Shahbaz, M.; Gozgor, G.; Hammoudeh, S. Human capital and export diversification as new determinants of energy demand in the United States. *Energy Econ.* **2019**, *78*, 335–349. [CrossRef]
- 34. Gozgor, G.; Can, M. Effects of the product diversification of exports on income at different stages of economic development. *Eurasian Bus. Rev.* **2016**, *6*, 215–235. [CrossRef]
- 35. Samargandi, N. Energy intensity and its determinants in OPEC countries. *Energy* 2019, 186, 115803. [CrossRef]
- 36. Lv, Y.; Chen, W.; Cheng, J. Direct and indirect effects of urbanization on energy intensity in Chinese cities: A regional heterogeneity analysis. *Sustainability* **2019**, *11*, 3167. [CrossRef]
- 37. Huo, T.; Ren, H.; Cai, W. Estimating urban residential building-related energy consumption and energy intensity in China based on improved building stock turnover model. *Sci. Total Environ.* **2019**, 650, 427–437. [CrossRef] [PubMed]
- Sun, H.; Edziah, B.K.; Sun, C.; Kporsu, A.K. Institutional quality, green innovation and energy efficiency. *Energy Policy* 2019, 135, 111002. [CrossRef]
- 39. Yan, H. Provincial energy intensity in China: The role of urbanization. Energy Policy 2015, 86, 635–650. [CrossRef]
- 40. Adom, P.K. Asymmetric impacts of the determinants of energy intensity in Nigeria. Energy Econ. 2015, 49, 570–580. [CrossRef]
- 41. Adom, P.K.; Minlah, M.K.; Adams, S. Impact of renewable energy (hydro) on electricity prices in Ghana: A tale of the short- and long-run. *Energy Strat. Rev.* 2018, 20, 163–178. [CrossRef]
- 42. Elliott, R.; Sun, P.; Chen, S. Energy intensity and foreign direct investment: A Chinese city-level study. *Energy Econ.* 2013, 40, 484–494. [CrossRef]
- 43. Yu, H. The influential factors of China's regional energy intensity and its spatial linkages: 1988–2007. *Energy Policy* 2012, 45, 583–593. [CrossRef]
- 44. Rafiq, S.; Salim, R.; Nielsen, I. Urbanization, openness, emissions, and energy intensity: A study of increasingly urbanized emerging economies. *Energy Econ.* **2016**, *56*, 20–28. [CrossRef]
- 45. Adom, P.K.; Kwakwa, P.A. Effects of changing trade structure and technical characteristics of the manufacturing sector on energy intensity in Ghana. *Renew. Sustain. Energy Rev.* 2014, *35*, 475–483. [CrossRef]
- 46. Adom, P.K. Determinants of energy intensity in South Africa: Testing for structural effects in parameters. *Energy* **2015**, *89*, 334–346. [CrossRef]
- 47. Cole, M.A. Does trade liberalization increase national energy use? Econ. Lett. 2006, 92, 108–112. [CrossRef]
- 48. Barnsley, A.B.; Brown, A. Enabling Renewable Energy and Energy Efficiency Technologies: Opportunities in Eastern Europe, Caucasus, Central Asia, Southern and Eastern Mediterranean; International Energy Agency (IEA): Paris, France, 2015.
- 49. Fisher-Vanden, K.; Jefferson, G.H.; Liu, H.; Tao, Q. What is driving China's decline in energy intensity? *Resour. Energy Econ.* 2004, 26, 77–97. [CrossRef]
- 50. Zhou, N.; Levine, M.D.; Price, L. Overview of current energy-efficiency policies in China. *Energy Policy* **2010**, *38*, 6439–6452. [CrossRef]
- 51. Huang, J.; Hao, Y.; Lei, H. Indigenous versus foreign innovation and energy intensity in China. *Renew. Sustain. Energy Rev.* 2018, *81*, 1721–1729. [CrossRef]
- 52. Borghesi, S.; Crespi, F.; D'Amato, A.; Mazzanti, M.; Silvestri, F. Carbon abatement, sector heterogeneity and policy responses: Evidence on induced eco innovations in the EU. *Environ. Sci. Policy* **2015**, *54*, 377–388. [CrossRef]
- 53. Mamun, A.; Sohag, K.; Shahbaz, M.; Hammoudeh, S. Financial markets, innovations and cleaner energy production in OECD countries. *Energy Econ.* **2018**, *72*, 236–254. [CrossRef]

- 54. Rathore, N.S.; Panwar, N. Renewable Energy Sources for Sustainable Development; New India Publishing: New Delhi, India, 2007.
- 55. Strantzali, E.; Aravossis, K. Decision making in renewable energy investments: A review. *Renew. Sustain. Energy Rev.* 2016, 55, 885–898. [CrossRef]
- 56. Liu, S.; Xia, X.; Tao, F.; Chen, X. Assessing urban carbon emission efficiency in China: Based on the global data envelopment analysis. *Energy Procedia* 2018, 152, 762–767. [CrossRef]
- 57. Wang, Y.; Zhao, T.; Wang, J.; Guo, F.; Kan, X.; Yuan, R. Spatial analysis on carbon emission abatement capacity at provincial level in China from 1997 to 2014: An empirical study based on SDM model. *Atmos. Pollut. Res.* **2019**, *10*, 97–104. [CrossRef]
- 58. Zeng, L.; Lu, H.; Liu, Y.; Zhou, Y.; Hu, H. Analysis of regional differences and influencing factors on China's carbon emission efficiency in 2005–2015. *Energies* **2019**, *12*, 3081. [CrossRef]
- 59. Mathiesen, B.V.; Lund, H.; Karlsson, K. 100% Renewable energy systems, climate mitigation and economic growth. *Appl. Energy* **2011**, *88*, 488–501. [CrossRef]
- 60. Menyah, K.; Wolde-Rufael, Y. CO₂ emissions, nuclear energy, renewable energy and economic growth in the US. *Energy Policy* **2010**, *38*, 2911–2915. [CrossRef]
- 61. Apergis, N.; Payne, J.E. Renewable energy consumption and economic growth: Evidence from a panel of OECD countries. *Energy Policy* **2010**, *38*, 656–660. [CrossRef]
- 62. Fang, Z.; Chang, Y. Energy, human capital and economic growth in Asia Pacific countries—Evidence from a panel cointegration and causality analysis. *Energy Econ.* 2016, *56*, 177–184. [CrossRef]
- 63. Oyedepo, S.O. Energy and sustainable development in Nigeria: The way forward. Energy Sustain. Soc. 2012, 2, 1–17. [CrossRef]
- 64. Ahmad, S.; Tahar, R.M. Selection of renewable energy sources for sustainable development of electricity generation system using analytic hierarchy process: A case of Malaysia. *Renew. Energy* **2014**, *63*, 458–466. [CrossRef]
- 65. Lund, H. Renewable energy strategies for sustainable development. *Energy* 2007, 32, 912–919. [CrossRef]
- Kahia, M.; Ben Aïssa, M.S.; Lanouar, C. Renewable and non-renewable energy use-economic growth nexus: The case of MENA Net Oil Importing Countries. *Renew. Sustain. Energy Rev.* 2017, 71, 127–140. [CrossRef]
- 67. Le, H.P.; Sarkodie, S.A. Dynamic linkage between renewable and conventional energy use, environmental quality and economic growth: Evidence from emerging market and developing economies. *Energy Rep.* **2020**, *6*, 965–973. [CrossRef]
- 68. Marques, A.C.; Fuinhas, J.A. Drivers promoting renewable energy: A dynamic panel approach. *Renew. Sustain. Energy Rev.* 2011, 15, 1601–1608. [CrossRef]
- 69. Ahmed, S.; Alam, K.; Sohag, K.; Gow, J.; Rashid, A.; Akter, M. Renewable and non-renewable energy use and its relationship with economic growth in Myanmar. *Environ. Sci. Pollut. Res.* **2019**, *26*, 22812–22825. [CrossRef] [PubMed]
- 70. Sbia, R.; Shahbaz, M.; Hamdi, H. A contribution of foreign direct investment, clean energy, trade openness, carbon emissions and economic growth to energy demand in UAE. *Econ. Model.* **2014**, *36*, 191–197. [CrossRef]
- Wan, J.; Baylis, K.; Mulder, P. Trade-facilitated technology spillovers in energy productivity convergence processes across EU countries. *Energy Econ.* 2015, 48, 253–264. [CrossRef]
- Khan, M.T.I.; Yaseen, M.R.; Ali, Q. Dynamic relationship between financial development, energy consumption, trade and greenhouse gas: Comparison of upper middle income countries from Asia, Europe, Africa and America. J. Clean. Prod. 2017, 161, 567–580. [CrossRef]
- 73. Tweneboah, G.; Alagidede, P. Dollarization, inflation targeting, and inflationary dynamics in Ghana. J. Afr. Bus. 2019, 20, 358–375. [CrossRef]
- 74. Kosai, S.; Yuasa, M.; Yamasue, E. Chronological transition of relationship between intracity lifecycle transport energy efficiency and population density. *Energies* **2020**, *13*, 2094. [CrossRef]
- 75. Wang, W.; Yang, H.; Zhang, Y.; Xu, J. IoT-enabled real-time energy efficiency optimisation method for energy-intensive manufacturing enterprises. *Int. J. Comput. Integr. Manuf.* **2017**, *31*, 362–379. [CrossRef]
- 76. Rahman, M.; Vu, X.-B. Are energy consumption, population density and exports causing environmental damage in China? autoregressive distributed lag and vector error correction model approaches. *Sustainability* **2021**, *13*, 3749. [CrossRef]
- 77. Marques, A.C.; Fuinhas, J.A.; Manso, J.P. Motivations driving renewable energy in European countries: A panel data approach. *Energy Policy* **2010**, *38*, 6877–6885. [CrossRef]
- 78. Baltagi, B.H. Econometric Analysis of Panel Data; Springer Science and Business Media LLC: Berlin, Germany, 2021.
- 79. Arellano, M. On the testing of correlated effects with panel data. J. Econ. 1993, 59, 87–97. [CrossRef]
- 80. Wooldridge, J.M. Econometric Analysis of Cross Section and Panel Data; MIT Press: Cambridge, UK, 2010.
- 81. Seddigi, H.; Lawler, K.A. Econometrics: A Practical Approach; Routledge: New York, NY, USA, 2000.
- 82. Waheed, R.; Sarwar, S.; Mighri, Z. Role of high technology exports for energy efficiency: Empirical evidence in the context of Gulf Cooperation Council countries. *Energy Environ.* **2020**, 0958305x20954196. [CrossRef]
- 83. Kennedy, P. A Guide to Econometrics; John Wiley and Sons: Hoboken, NJ, USA, 2008.
- 84. Goldberger, A.S. A Course in Econometrics; Harvard University Press: Cambridge, MA, USA, 1991.
- 85. Breusch, T.S.; Pagan, A.R. The lagrange multiplier test and its applications to model specification in econometrics. *Rev. Econ. Stud.* **1980**, 47, 239–253. [CrossRef]
- 86. Greene, W.H. Econometric Analysis, 4th ed.; Prentice Hall: Upper Saddle River, NJ, USA, 2000; pp. 201–215.
- 87. Swamy, P.A.V.B. Efficient inference in a random coefficient regression model. Econometrica 1970, 38, 311–323. [CrossRef]
- 88. Grossman, G.M.; Krueger, A.B. Economic growth and the environment. Q. J. Econ. 1995, 110, 353–377. [CrossRef]

- 89. Pesaran, M.H. A simple panel unit root test in the presence of cross-section dependence. *J. Appl. Econ.* 2007, 22, 265–312. [CrossRef]
- 90. Hoechle, D. Robust standard errors for panel regressions with cross-sectional dependence. *Stata J. Promot. Commun. Stat. Stata* 2007, *7*, 281–312. [CrossRef]
- 91. Aïssa, M.S.B.; Jebli, M.B.; Youssef, S.B. Output, renewable energy consumption and trade in Africa. *Energy Policy* **2014**, *66*, 11–18. [CrossRef]
- 92. Yanikkaya, H. Trade openness and economic growth: A cross-country empirical investigation. *J. Dev. Econ.* 2003, 72, 57–89. [CrossRef]
- 93. Al-Mulali, U.; Ozturk, I.; Lean, H.H. The influence of economic growth, urbanization, trade openness, financial development, and renewable energy on pollution in Europe. *Nat. Hazards* **2015**, *79*, 621–644. [CrossRef]
- 94. Premik, F.; Stanisławska, E. The Impact of Inflation expectations on Polish consumers' spending and saving. *East. Eur. Econ.* **2017**, 55, 3–28. [CrossRef]
- 95. Avina, J.M. City building energy efficiency programs: Hindering real energy efficiency? ... One practitioner's real world marketplace observations. *Energy Eng.* 2017, 114, 22–36. [CrossRef]
- 96. Carmi, N.; Tal, A. The perceived relationship between population growth and current ecological problems using repertory grid technique. *Hum. Ecol. Risk Assess. Int. J.* **2018**, *25*, 1773–1788. [CrossRef]
- 97. Füllemann, Y.; Moreau, V.; Vielle, M.; Vuille, F. Hire fast, fire slow: The employment benefits of energy transitions. *Econ. Syst. Res.* **2019**, *32*, 202–220. [CrossRef]
- Allan, G.; McGregor, P.; Swales, K. Greening regional development: Employment in low-carbon and renewable energy activities. *Reg. Stud.* 2017, 51, 1270–1280. [CrossRef]
- 99. Halkos, G.; Gkampoura, E.-C. Coping with energy poverty: Measurements, drivers, impacts, and solutions. *Energies* **2021**, *14*, 2807. [CrossRef]
- 100. Baum, C.F. Residual diagnostics for cross-section time series regression models. *Stata J. Promot. Commun. Stat. Stata* 2001, 1, 101–104. [CrossRef]
- 101. Poi, B.P. From the help desk: Swamy's random-coefficients model. Stata J. 2003, 3, 302–308. [CrossRef]
- 102. Burdisso, T.; Sangiácomo, M. Panel time series: Review of the methodological evolution. *Stata J. Promot. Commun. Stat. Stata* 2016, 16, 424–442. [CrossRef]





Jamal Mamkhezri^{1,*}, Leonard A. Malczynski² and Janie M. Chermak³

- ¹ Department of Economics, Applied Statistics & International Business, New Mexico State University, Las Cruces, NM 88003, USA
- ² Anderson School of Management, University of New Mexico, Albuquerque, NM 87131, USA; lajmalcz@gmail.com
- ³ Department of Economics, University of New Mexico, Albuquerque, NM 87131, USA; jchermak@unm.edu
- Correspondence: Jamalm@nmsu.edu

Abstract: State-mandated renewable portfolio standards affect substantial portions of the total U.S. electricity supply. Renewable portfolio standards are environmentally motivated policies, yet they have the potential to greatly impact economy. There is not an agreement in the literature on the impact of renewable portfolio standards policies on regional economies, especially on job creation. By integrating various methodologies including econometrics, geographic information system, and input-output analysis into a unique system dynamics model, this paper estimates the economic and environmental impacts of various renewable portfolio standards scenarios in the state of New Mexico, located in Southwestern U.S. The state is endowed with traditional fossil fuel resources and substantial renewable energy potential. In this work we estimated and compared the economic and environmental tradeoffs at the county level under three renewable portfolio standards: New Mexico's original standard of 20% renewables, the recently adopted 100% renewables standard, and a reduced renewable standard of 10%. The final one would be a return to a more traditional generation profile. We found that while the 20% standard has the highest market-based economic impact on the state as a whole, it is not significantly different from other scenarios. However, when environmental impacts are included, the 100% standard yields the highest value. In addition, while the state level economic impacts across the three scenarios are not significantly different, the county-level impacts are substantial. This is especially important for a state like New Mexico, which has a high reliance on energy for economic development. A higher renewable portfolio standard appears to be an economic tool to stimulate targeted areas' economic growth. These results have policy implications.

Keywords: renewable portfolio standards; employment; economic output; water use; greenhouse gases; emissions; social benefits

1. Introduction

Electric utilities in the United States (U.S.) are integrating more renewable energy (RE) sources in their energy mix. In May 2020, 24.3% of electricity generation in the U.S. came from renewable sources (Energy Information Administration, Form EIA-860M). This is partly a result of policies and regulations aimed at mitigating greenhouse-gas (GHG) emissions through programs such as the Regional Greenhouse Gas Initiative in the northeastern part of the U.S., and through the renewable portfolio standard (RPS) at the state level. While the primary objective of these regulations is to address global warming, there can be potential impacts on the economy at a microlevel (i.e., state and county levels). For rural western states, this has become increasingly important, as they strive to diversify their economies.

Debates are ongoing in the literature as to whether RPS policies have a positive (i.e., job creation, GHG and air pollution reduction), negative, or no impact on an economy and the environment (e.g., [1–5]). The main reason for the divergent findings is the inclusion or

Citation: Mamkhezri, J.; Malczynski, L.A.; Chermak, J.M. Assessing the Economic and Environmental Impacts of Alternative Renewable Portfolio Standards: Winners and Losers. *Energies* 2021, *14*, 3319. https://doi.org/10.3390/en14113319

Academic Editor: George Halkos

Received: 13 May 2021 Accepted: 1 June 2021 Published: 5 June 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). exclusion of market failure due to environmental amenities in the analyses. For example, NYSERDA [1] assessed New York's RPS impact and found a gain of 24,000 job-years from 2002 to 2037. Divounguy et al. [2] investigated Ohio's 12.5% by 2025 RPS and found that it would result in a loss of more than 134,000 jobs. Upton and Snyder [3] evaluated states with an RPS versus those without, and they found that an RPS standard has no significant impact on increasing RE or reducing GHGs. Zhou and Solomon [4] found that more stringent RPSs result in lowering RE capacity additions, while Carley et al. [5] found the opposite. Further, most of the existing literature focused on either an aggregate scope (e.g., nation-wide) or state-specific assessments and has not considered impacts at lower-level jurisdictions (e.g., county level). Lastly, much of the literature overlooked the fundamental dynamics within the energy sector. The objective of this paper is to contribute to this line of research and assess the economic and environmental impact of renewable energy and the tradeoffs on a regional economy.

In particular, we are interested in answering the question of what are the economic and environmental impacts of varying RPSs on regional economies. This is a rather complex question, and answering it is aided by the use of system simulation [6–11]. Thus, in this work we develop, validate, and utilize a system dynamics (SD) based simulation model that integrates results from various methodologies such as input–output analysis, econometrics, and Geographic Information System (GIS). Combining these methodologies in an innovative approach to analyze the SD model is one key contribution of this paper. We execute our analysis in our case study of New Mexico, a southwestern state in the U.S. with an RPS and high potential for both fossil fuel (traditional) and RE sources. We hypothesize that RPS levels have substantial environmental and economic impacts on regional economies. This paper is an attempt to quantify those impacts.

Our findings suggest a net increase of jobs in rural counties that are most suitable for future RE installations. Depending on the scenario, our model estimated increasing 137–156 thousand cumulative, full-time equivalent jobs, 19 to 24 billion USD (2017\$) cumulative gross economic output, and 12,987 to 13,219 and 974 to 1122 billion liters of cumulative water withdrawal and consumption respectively from 2017 to 2050. These scenarios also resulted in increasing millions of avian mortalities, as well as millions of tonnes of GHG emissions and thousands of tonnes of air pollutants, each of which leads to billions of dollars in climatic and air-quality costs (social costs). Lastly, we found that higher RPS standards lead to greater benefits to the state when externalities and social benefits/costs are taken into account.

2. Background

The burning of fossil fuel (i.e., coal, natural gas, and oil) is the main source of GHG emissions in the U.S., and this contributes to climate change. Combusting fossil fuels for electricity generation not only emits air pollution but also requires an immense amount of water. There is extensive literature that demonstrates the correlation between air pollution and premature mortality/morbidity [12–18]. Maupin et al. [19] showed that roughly 40% of all of the U.S. freshwater withdrawal was used for thermoelectric power plants in 2010. Policymakers, as a result, are seeking to promote policies that lead to integrating more environmentally friendly generation sources with less externalities.

Electricity generation is moving towards integrating a higher level of RE and a lower level of fossil fuels in the U.S. due to regulatory mandated laws such as the RPS as well as cost competitiveness. Thirty states and the District of Columbia currently have an RPS in place. RPSs mandate that electric utilities source a portion of their generation from RE within a certain timeframe. Although the main goal of an RPS is environmentally oriented—that is, to mitigate GHG emissions and/or save water—these policies have the potential to impact economies. Previous research on the impact of RPSs shows that the policies are capable of yielding positive economic impacts if positive externalities (zero or close to zero water usage, zero emission, etc.) are taken into account [20,21]. Barbose et al. [20] demonstrated that meeting requirements mandated by RPSs led to supporting 200 thousand jobs and a reduction of 59 million tonnes of CO_2 in the U.S. in 2013. Wiser et al. [21] quantified positive externalities of RE and estimated that existing RPS policies lead to improving air quality and reducing climatic damages (258 billion USD), which not only compensates the increase in electric system costs (23 to 194 billion USD) but also exceeds those costs over the period of 2015–2050.

There are a handful of peer-reviewed papers and national laboratory reports that look at the feasibility of providing global energy through RE (e.g., [22–25]). For example, Jacobson and colleagues [22] estimated a portfolio mix that enables the United States to sustain its entire energy needs—including electricity, transportation, heating/cooling, and industry—using renewable energy by 2050. Similarly, Cole et al. [25] assessed different scenarios of achieving various levels of RE in only the power sector by 2050. Further, previous economic impact studies of constructing and operating RE projects suggested that the economic impacts to states are considerable [17,26,27]. Similarly, studies on environmental impact of RE showed significant climate and air quality benefits [28–31]. For instance, Millstein et al. [29] found that solar and wind development resulted in benefits of 30-113 billion USD (2015) and 5-107 billion USD from air quality and climate, respectively, while avoiding 3000–12,700 premature mortalities in 2007–2015. Most of these studies produced state-level or nation-wide job/environmental impact estimates, which in turn meant less understanding of lower-level dynamics such as job/environmental impacts across counties. These studies also did not consider the underlying dynamics within the energy sector.

To address the aforementioned gaps in the literature, we combine various methodologies to develop an SD model. SDs are a derivative of the work developed by Forrester [32], in which he introduced a novel approach to integrate multiloop feedback systems. So long as relationships among variables are known, this approach makes modeling complex systems possible [33]. The SD model of this paper integrates results from input–output analysis, econometrics, and GIS to form a unique framework that provides both the public and policymakers improved information with which to make decisions. The model is developed at a monthly time-step from January 2004 through January 2050. Multiple programs are used to analyze the complex electricity problems common to most utilities. Specifically, Jobs and Economic Development Impact (JEDI) coupled with Impact Analysis for Planning (IMPLAN) are used to calculate job multipliers by energy type and by county; Stata is used to estimate electricity generation by county, as well as the optimal location for siting additional power plants; lastly, results from previous models are all embodied in Powersim Studio, which is used to analyze various energy mix scenarios.

The objective of the SD model is to estimate electricity generation and consumption by different fuel sources and various sectors respectively. We provide a roadmap to assess the explicit and implicit impacts of various energy mix scenarios at the state and county level and at different points in time. Explicit impacts may include potential jobs and economic gross output associated with current and potential future electricity generation, and implicit impacts may include positive health effects and social benefits of reducing ambient emissions. We apply this roadmap to our case study of New Mexico.

2.1. Study Area: New Mexico

New Mexico has considerable potential for both fossil fuel and RE resources. It holds about 3%, 4%, and 5% of the United States' total estimated recoverable coal reserves, proved crude oil, and natural gas respectively and it possesses the second-largest uranium reserves in the nation. Most of the state's natural gas and crude oil are located in the San Juan and Permian Basins in the northwestern and southeastern part of the state, respectively, while coal reserves are mainly located in the San Juan and Raton Basins in the northern part of the state. The vast areas of New Mexico with available geophysiological landmass that receives high wind and sunlight levels are optimal for increasing RE usage. New Mexico ranks third in both solar and wind potential in the U.S. [34].

New Mexico's economy is ranked 46th in the nation. The energy industry, especially oil and natural gas extraction, is a main contributor to New Mexico's economy. The state receives approximately 2 billion and 300 million USD per year in direct (e.g., severance, property taxes, royalty, and rental income) and indirect (sales and income taxes) revenues, respectively, from oil and gas production. Depending on the state of the economy, based on recent state finance facts, revenues from oil and gas can contribute about 40% to New Mexico's general fund tax revenue. Thus, fluctuating oil and gas prices affect New Mexico's economy immensely.

On one hand, the energy industry is responsible for emitting GHG and ambient pollution as well as increased water usage in New Mexico. GHG contributes to climate change, while air pollution causes premature mortality and morbidity, and freshwater has historically been insufficient in New Mexico. On the other hand, RE is becoming more and more cost-competitive compared to fossil fuel technologies. Thus, it makes logical and economic sense for policymakers to promote policies such as an RPS in order to integrate more RE into New Mexico's energy mix.

At the time of analysis, New Mexico's RPS required all large electric utilities to generate 20% of their in-state electricity sales from RE resources by 2020. Although it did not pass, a bill (Senate Bill 312) was introduced to increase New Mexico's RPS previous level to 25% by 2020, 35% by 2025, 50% by 2030, 65% by 2035, and 80% by 2040 in the 53rd legislative session in 2017. A modified version of this bill was reintroduced in January 2019 (House Bill 15) and was passed in the 54th legislative session in March 2019 (Senate Bill 489). In addition to the requirements set by Senate Bill 312, Senate Bill 489 sets a 100% RPS by 2045 that is sourced from zero carbon resources. This makes New Mexico the third state in the U.S. after California and Hawaii and before Washington, New York, Maine, and Virginia to mandate a 100% RPS. Thus, New Mexico's current RPS policy requires 20% in-state electricity sales from RE resources by 2020, 40% by 2025, 50% by 2030, 80% by 2040, and 100% by 2045.

Currently, there are three large electric utilities in New Mexico: the Public Service Company of New Mexico, El Paso Electric, and Xcel Energy, with the first serving the largest customer pool in the state. Further, as New Mexico has considerable potential in both wind and solar energy, the Public Regulation Commission set diversity targets (carve-outs) for different types of RE to create a diversified portfolio. Based on this portfolio, the utilities are required to source at least 30%, 20%, and 3% of their in-state electricity sales from wind, utility-scale photovoltaic solar (PV), and residential photovoltaic solar (RPV), respectively, by 2020 (see Table 1). RPS requires the New Mexico's rural electric distribution cooperatives to generate 10% of their in-state electricity sale from renewable sources. We did not consider a rural cooperatives constraint in our analysis.

Table 1. Carve-outs regulated by renewable portfolio standard.

Source	Minimum Amount
Wind	30%
Utility-scale solar (PV)	20%
Residential/distributed solar (RPV)	3%

2.2. Scenario Construction

Our analysis investigated the number of jobs and their locations by energy source, as well as environmental impact based on thirty-four prices, three technological-costs, and three RPS scenarios. Each of these scenarios are described briefly below.

We adopted 34 price scenarios—i.e., electricity price by sector, Henry Hub natural gas price, and electric sector fuel cost (coal and natural gas)—developed by the Energy Information Administration's (EIA) Annual Energy Outlook 2018 (AEO2018), along with three technology cost scenarios developed by the National Renewable Energy Laboratory (NREL) [25]. A list of AEO2018 scenarios are summarized in the supplementary document (Table S1). The cost scenarios includes low, mid, and high (constant) cost and performance estimates for wind, PV, RPV, natural gas (both baseload (combined-cycle; NGb) and peaker (single-cycle; NGp)), and coal from 2016 to 2050. Low-cost wind and solar scenarios utilize low-cost estimates for land-based wind, along with PV and RPV technologies, while high-cost scenarios use constant costs at or near the 2018 cost estimates. The mid-case scenario assumes prospective advances in the RE arena technology. The low- and highcost scenarios for fossil fuel beyond 2016 relies on two case estimates from AEO2018, i.e., the high oil and gas resource and technology case and the low oil and gas resource and technology case, respectively. The mid-case scenario serves as a reference case for fossil fuel technology costs adopted from AEO (2018). Overall, the SD model is capable of assessing 918 ($34 \times 3 \times 3 \times 3$) different scenarios. For the purpose of brevity, we focus on the three most plausible future scenarios: the new RPS, the previous RPS, and a future where integrating RE in the electric grid is discouraged. Under the first scenario, i.e., 100% RPS, we assume a future with scarce natural resources with costly fossil fuel and cheap RE technologies that make 100% RPS by 2050 possible. The second scenario, i.e., 10% RPS, is the opposite of the first scenario, in that we assume abundant natural resources with cheap fossil fuel and expensive RE technologies, hence a decreased RPS (10% by 2050). Lastly, we implement a status quo scenario, i.e., 20% RPS, that assumes reference case AEO prices with mid-case technology cost of fossil fuel and constant RE technology cost, along with business-as-usual RPS (RPS 20% by 2020 and on). Below we summarize each scenario.

- I. 100% RPS: AEO price = low oil and gas resource and technology; RE cost = low; fossil fuel cost = high; RPS = Senate Bill RPS (100%)
- II. 10% RPS: AEO price = high oil and gas resource and technology; RE cost = high; fossil fuel cost = low; RPS = decrease RPS (10%)
- III. 20% RPS: AEO prices = reference case; RE cost = high; fossil fuel cost = mid; RPS = status quo RPS (20%)

3. Materials and Methods

Our model consists of five submodels: (1) demand; (2) supply; (3) gap between supply and demand (hereafter "gap"); (4) jobs; and (5) environmental impact, with more than 1200 variables. The first submodel consists of two modules that together estimate electricity demand beyond 2016. The second submodel includes six modules that altogether project megawatt-hour (MWh) generation. The gap and the jobs submodels each contain seven modules. Finally, the environmental impact submodel contains only one module. A detailed description of the model is provided in the Supplementary Materials (Section B) and a related work [35]. Here, we briefly describe the overarching dynamics of the model.

Our model is based on a series of stocks and flows. Stocks can change from period to period; the changes are governed by "flows". The flows, based on either natural sciencebased rules, human choice, or policies, or a combination thereof, are the set of rules that dictate the change in the stocks. Figure 1 provides a schematic of the model. Arrows provide the connections between stocks and flows. In all cases, the arrows represent the direction of interaction. Associated with each stock, flow, and connecting arrow is a set of quantifiable relationships and rules that allow us to model the system and assess the impact and tradeoffs between sectors within a time period as well as over time.

The basic structure of the modeling components is the physical market for electricity, which (in the figure) is at the intersection of supply and demand and is governed by an exogenous price path. As mentioned in the scenario definitions section, we implemented 34 price scenarios developed by EIA's 2018 AEO report. Thus, "exogenous" here does not mean a fixed value over time but rather is a given independent variable that fluctuates by month and over time. Supply depends on capacity and capacity utilization, which is aggregated from individual generation sources of capacity, utilization, and net exports into/out of state, while demand depends on in-state (domestic) consumption. In-state demand is the aggregation of individual sectoral demands, which can be impacted by market conditions (price) and population impacts.



Figure 1. Modeling schematic.

The electricity market outcome at each time step maps into economic activity, estimated in dollars, which is one part of our macroeconomic module. The level of economic activity impacts the job outcome through changes in the demand for workers. It should be noted that population can also be impacted by the job impact as it could result in net outor in-migration.

The linkage between the electricity market and the environment (potential external impacts) is represented through a pollution component, where emissions during a time step add to the concentration level of the pollutant. We depict direct impacts of pollution through impacts on economic activity and through population (e.g., health impacts). It should be noted that there are a number of potential indirect links through, for example, consumer groups. In addition to pollution, our basic model includes water resources and human and avian mortalities. Further, RPS policies are included. Depending on the policy, the generation capacity, supply, demand, market prices, emissions, economic activity, or jobs could be impacted. Finally, all of these modules and methodologies are gathered in a unique SD model. Figure 2 summarizes the causal loop diagram utilized in developing the SD model.

In order to read the causal loop diagram depicted in Figure 2, we begin by imagining the variable at the base of the arrow increasing in value; the sign at the arrowhead indicates the increase (+) or decrease (-) in the variable at the arrowhead, all other variables unchanged. Lastly, parallel lines crossing an arrow indicate delay in impact from the variable at the base of the arrow to the variable in the head of the arrow. The causal loop diagram presents the logic behind our SD model. The following is an explanation of one path in the diagram.



The required generation to achieve a certain level of RPS increases as in-state electricity demand increases, which increases the need for additional RE capacity to meet the corresponding RPS level. The higher the need for additional RE capacity, the higher the new capacity of RE. As the new capacity of RE rises, the total RE capacity rises, and the capacity that is decommissioned in the future increases with a delay. A higher level of RE capacity that is to be decommissioned decreases the total RE capacity, creating an enforcing loop (see Figure 2). On one hand, the higher the RE capacity, the higher the RE generation, hence the higher the need for peaker natural gas, storage, and transmission lines. On the other hand, if we assume that a higher level of RE generation replaces fossil fuel generation, then a higher level of RE generation results in lower GHG and air pollution, thereby lowering population mortality and morbidity (social cost). A higher level of RE generation can also decrease the gap caused by a discrepancy between supply and demand for electricity and/or RPS requirement. The same logic holds true for the remaining components of the causal loop diagram.

Data

Data were obtained from numerous sources including, the U.S. Energy Information Administration (various survey forms, AEO2018, and Layer Information for Interactive State Maps shapefiles), Emissions and Generation Resource Integrated Database (eGRID) of the U.S. Environmental Protection Agency (EPA), the National Renewable Energy Laboratories (JEDI, Annual Technology Baseline, wind data, and solar data), the New Mexico Public Regulation Commission, the United States Geological Survey, the U.S. Bureau of Economic Analysis, the United States Census Bureau, and the Western Electricity Coordinating Council, as well as the energy literature. Except for RPV data, we obtained generation data from EIA-923 and EIA-861 (annual and monthly data). The data includes historical nameplate capacity of the existing power plants, generation, power plants' locations (county and latitude/longitude), operating and planned retirement year times, and capacity factors. The data for the existing RPV capacity were obtained from the New Mexico Public Regulation Commission. Further, IMPLAN 2016 data were used to calculate jobs and output multipliers for each energy source. Lastly, economic benefit/cost of air pollution and GHG reduction multipliers came from the energy literature. Table S2 of the Supplementary Materials summarizes the key data sources.

4. Results

In this section, we present our results. We first review electricity generation under the three modeled scenarios. Next, we discuss state-level and county-level economic and environmental impacts. Economic impact results are presented for full-time equivalent employment and gross economic output. Environmental impacts, on the other hand, are reported in terms of GHG emissions, air pollution, water usage, and human and avian mortality associated with each of our three modeled scenarios. These impacts are experienced once the plants are in the O&M phase. Thus, environmental impact results are reported for the operations period solely and on a state- and county-level basis. Finally, we compare results across scenarios to expose whether results are statistically significantly different. If they are, we then identify state and county levels that experience job gains (winners) and job losses (losers).

4.1. Generation

Figure 3 shows the total electricity generation under the three modeled scenarios, and Figure 4 presents the generation mix through 2050. Based on the 20% RPS scenario, as with the other two scenarios, RE and fossil fuel generations encompassed respectively 17% and 83% of total generation in 2017. In 2030, generation shares are 15% and 85% for RE and fossil fuel, respectively. Compared to the 20% RPS scenario, RE generations are 9% higher in the generation mix under the 100% RPS scenario (24%) and 5% lower under the 10% RPS scenario (12%). All scenarios estimated a dip in electricity generation from 2036 until

the end of 2037. This is due to the decommissioning of the existing coal-fired power plants in that period. The dip in the overall electricity generation is expected to be compensated by importing nuclear energy from Arizona. The figures depict generation from both within the state and not imported into the state.



Figure 3. Total annual electricity generation under the three modeled scenarios.



Figure 4. Annual electricity generation (in thousand GWh or TWh) by all six energy sources: (**a**) 10% RPS; (**b**) 100% RPS; (**c**) 10% RPS scenarios.

As presented in Figure 4, for the scenarios we estimated the amount and type of energy source to replace coal generation. By 2050, RE generation increases to 52%, while fossil fuel generation drops to 48% under the 20% RPS scenario. The 100% RPS scenario and the 10% RPS scenario result in a 11% higher and a 48% lower RE generation, respectively, when compared with the 20% RPS scenario. As mentioned, RPS requires utility companies to generate a portion of their in-state sales from RE. Thus, it is possible to have fossil fuel

generation even under the 100% RPS scenario. The takeaway here is that different energy scenarios lead to different energy mixes, which therefore means different environmental and economic impacts.

4.2. Economic Impacts

Our model is capable of estimating employment and gross economic output by three categories: direct (onsite), indirect, and induced. Total impact is the sum of direct, indirect, and induced impacts. Since direct, indirect, and induced impacts are a fixed fraction of total impact, we only discuss total impacts here. In what follows, we first discuss employment impact at the state and county level. We then compare results across scenarios and identify whether there are winners or losers. Next, we summarize total economic output results in a similar approach. Further discussion of the results, especially more granular level results (e.g., different types of energy sources during different phases), can be found in the Supplementary Materials (Section C).

Figure 5 summarizes the cumulative total employment impact by the 20% RPS scenario and the other two modeled scenarios. We estimated a total employment impact on New Mexico in construction and O&M to be as follows: 151,857 (42,517 RE and 109,340 fossil fuel), 151,284 (112,593 RE and 38,691 fossil fuel), and 155,520 (26,271 RE and 129,248 fossil fuel) full-time equivalent jobs according to the 20% RPS, 100% RPS, and 10% RPS scenarios, respectively, from January 2017 to January 2050. Thus, compared to the 20% RPS scenario, the 100% RPS one (RE intensive scenarios) results in 573 fewer cumulative (construction and O&M) full-time equivalent jobs, while the 10% RPS one (most fossil fuel intensive scenario) yields 3663 more cumulative jobs. Note that these results are based on the assumption that all labor is provided locally. This assumption, which is on a 0–100% scale, can also be changed in the original SD model. What is important here is that this assumption does not impact the dynamics within modules and only results in lower direct economic impact (labor and economic output) across scenarios.



Figure 5. Temporal cumulative jobs (construction and O&M) by modeled scenarios from January 2017 to January 2050.

As demonstrated in Figure 5, all scenarios estimate a boost in energy employment after 2037. This is because existing coal-fired power plants are expected to retire in 2037, meaning there should be no new installation. Depending on the scenario, coal generation is expected to be replaced by either renewables or natural gas after 2037, and thus jobs related to coal are also likely to be replaced by renewable or natural gas jobs. Although the 100% RPS scenario yields fewer cumulative total jobs than the 20% RPS case, its impact fluctuates and is more diverse throughout the timespan of the study. Figure 3 depicts the employment distribution by the three modeled scenarios from 2017 through January 2050. Any spikes in the estimated employment numbers can be due to whether RPS and RE carve-

out requirements are met. We performed nonparametric tests such as the Kolmogorov– Smirnov tests to compare the equality of distributions of total employment across scenarios. The test results suggest that the null hypothesis of equality of distributions across the three scenarios cannot be rejected. In other words, at the state level, the employment impact of these three scenarios are not statistically significantly different. Thus, we found that the state of New Mexico is not a winner or a loser in terms of job gains or losses at the state level under all of the assessed scenarios. Temporal and cumulative employment impacts during construction and O&M phases are provided in the Supplementary Materials (Section C).

Now, we turn our attention to county-level employment results. Table 2 summarizes the annual average employment for all three scenarios by county. This table demonstrates an important result of the current study: some counties will be winners, and others will be losers. Figure 6 puts these results in perspective; it shows employment gain and loss per 10,000 labor force for the 100% RPS case versus the reference case of 20% RPS. Lastly, the Kolmogorov–Smirnov test results at the county level support the statistically significantly different employment distributions as well.

County	20% RPS	100% RPS	10% RPS
Bernalillo	301	83	320
Catron	22	93	3
Chaves	19	121	8
Cibola	23	118	4
Colfax	26	128	6
Curry	211	275	188
De Baca	23	94	3
Dona Ana	397	225	393
Eddy	285	121	289
Grant	284	119	291
Guadalupe	44	132	24
Harding	23	115	3
Hidalgo	294	120	300
Lea	536	443	498
Lincoln	7	45	3
Los Alamos	41	31	35
Luna	442	334	419
Mc Kinley	303	67	325
Mora	23	126	3
Otero	16	114	6
Quay	79	160	59
Rio Arriba	23	112	3
Roosevelt	137	210	117
Sandoval	39	58	34
San Juan	764	569	762
San Miguel	23	110	4
Santa Fe	12	35	7
Sierra	22	108	3
Socorro	22	108	3
Taos	23	97	3
Torrance	155	235	136
Union	81	163	61
Valencia	306	84	322

Table 2. Annual average employment by county and modeled scenarios *.

* Average values are from 2017 to 2050.

Economic output follows the employment results closely: when there is employment impact, there is economic output impact as well. Construction and O&M employees, depending on type of energy source, earn an average annual salary (with benefit) of 35,000 to 58,000 USD (2017\$) and 56,000 to 76,000 USD (2017\$) per year, respectively [36]. Under the 20% RPS scenario, these employments result in a cumulative (sum of construction and

O&M) total economic output of 24 billion USD (2017\$) (18% RE and 49% O&M) per year from 2017 to 2050. The 100% RPS and 10% RPS scenarios respectively lead to roughly 4 (20 USD: 94% RE and 54% O&M) and 2 (22 USD: 4% RE and 45% O&M) billion USD (2017\$) per year less than the 20% RPS case. In other words, the 20% RPS scenario yields a cumulative economic output that is 20% and 9% higher than the 100% RPS and 10% RPS scenarios, respectively. Figure 7 summarizes these results.



Job Gain/Loss Per 10k Labor Force - 100% RPS Vs. Ref. Case



Figure 6. Employment gain and loss per 10,000 labor force under the 100% RPS case compared to the 20% RPS case. Note: Positive values indicate job gains; negative values are job losses.



Figure 7. Total annual economic output by energy source and modeled scenarios from 2017 to 2050.

Similar to the state-level employment, economic impact distributions under the three assessed scenarios are not statistically significantly different. However, at the county level, rural counties can benefit under the RE intensive scenario, and counties with fossil fuel infrastructure in place can benefit from the fossil fuel intensive scenarios (namely the more populous counties).

4.3. Environmental Impacts

Based on all of the three modeled scenarios, coal-fired power plants are assumed to fully retire after 2037. This is mainly due to the fact that the existing coal-fired power plants are aging (>40 years), and fuel contracts with coal mines are ending; more importantly, it is highly likely that coal will no longer be cost-competitive. Given these situations, we expect that there would be no new coal-fired power plants constructed in the future (see Figure 4). Note that these power plants are the most water-intense and polluting technologies in our set of energy sources (see Table S6). Eliminating coal from New Mexico's energy mix would result in fewer negative externalities (GHG, ambient pollutions, and water usage) from fossil fuel overall. Different technology costs along with RPS requirements drive the energy source that would eventually replace coal. The more RE replaces coal, the fewer negative externalities and the higher the social benefit from the replacement.

In what follows, we first discuss cumulative water withdrawal and consumption results at the state and county level. We then compare results across scenarios and identify whether there is water saved at the state and county levels. We take a similar approach in explaining emissions. Finally, we discuss the social benefit/cost of different scenarios.

4.3.1. Water Usage

Figure 8 depicts the temporal water withdrawal and consumption from 2017 to 2050. The 20% RPS scenario suggests a cumulative 13,178 and 1096 billion liters of water withdrawal and consumption throughout the study timeline. Compared to the 20% RPS scenario, the 100% RPS scenario uses less water for withdrawal and for consumption by 190 and 122 billion liters, respectively. The 10% RPS scenario, with the highest level of fossil fuels in the energy mix, uses 41 and 26 billion liters of water more than the 20% RPS scenario for water withdrawal and consumption, respectively.



Figure 8. Water withdrawal and consumption over time by the electric sector under the three modeled scenarios. (**a**) Water withdrawal; (**b**) Water consumption.

Considering an average price of 0.00689 USD/liter for water consumption by each energy source [37], the 20% RPS scenario results in a total cost of 527 million USD (\$2017) in water consumption for electricity generation. Compared to the 20% RPS scenario, the 100% RPS scenario results in saving 58 million USD for water savings, while the 10% RPS scenario increases costs by 13 million USD, as it is more water intense.

To compare water consumption distributions across scenarios, we performed Kolmogorov–Smirnov tests. Test results provided us with evidence to reject the null hypothesis of equality of distributions between the 100% RPS and 20% RPS scenarios even at the state level. On the whole, we did not find similar results when comparing the 10% RPS scenario against the 20% RPS one.

Table 3 summarizes the annual average million liters of water consumption by county and the three scenarios; Figure 9 translates this information to per capita (county) water consumption saved or lost. While the majority of counties see no changes, the majority of impacts are the savings. Our nonparametric test results further support the alternative hypothesis of unique water consumption distributions across scenarios at the county level.

County	20% RPS	100% RPS	10% RPS
Bernalillo	46	16	53
Catron	0	0	0
Chaves	0.08	0.68	0.04
Cibola	0	0	0
Colfax	0.23	0.23	0.23
Curry	0	0	0
De Baca	0	0	0
Dona Ana	119	89	126
Eddy	31	1	38
Grant	34	4	41
Guadalupe	0	0	0
Harding	0	0	0
Hidalgo	39	8	45
Lea	217	187	223
Lincoln	0	0	0
Los Alamos	13	7	14
Luna	140	109	146
Mc Kinley	203	173	209
Mora	0	0	0
Otero	0.08	0.68	0.04
Quay	0	0	0
Rio Arriba	0	0	0
Roosevelt	0.08	0.68	0.04
Sandoval	0	0	0
San Juan	1873	1843	1879
San Miguel	0	0	0
Santa Fe	0.08	0.68	0
Sierra	0	0	0
Socorro	0	0	0
Taos	0	0	0
Torrance	0	0	0
Union	0	0	0
Valencia	50	20	56

Table 3. Annual average water consumption by county and modeled scenarios *.

* Average values are in million liters and from 2017 to 2050; "0" means no change.

4.3.2. Air Pollution and Greenhouse-Gas Emissions

Figures 10 and 11 depict the cumulative impact of air pollution and GHG emissions, along with consecutive social benefit to the state from 2017 to 2050. Cumulatively, the RE intensive scenario emits roughly 91 million tonnes GHG less than the 20% RPS scenario throughout the study timeline, leading to more than 6.8 billion USD (2010\$) in cumulative climate benefit. The fossil fuel intensive scenario, on the other hand, emits 3% (19 million tonnes) higher GHG than the 20% RPS one, which causes more than 1400 million USD (2010\$) social cost compared to the 20% RPS one. Each one million tonnes of GHG emissions is equivalent to GHG emissions by approximately 2250 million miles driven by an average passenger vehicle. Table 4 summarizes the county level results only for GHG. Based on

the Kolmogorov–Smirnov test results, we can reject the null hypotheses of equality of GHG emission distributions when comparing both the 100% RPS and 10% RPS scenarios against the 20% RPS scenario. In other words, the 100% RPS scenario results in statistically significantly lower GHG than the 20% RPS scenario, while the opposite holds true for the 10% RPS scenario. We found similar results at both state and county levels.



Figure 9. Per capita water consumption saved/lost under the 100% RPS scenario compared to the 20% RPS scenario. Note: Negative value indicates water saved; 1 gallon is \approx 3.79 liters.



Figure 10. State-level cumulative tonnes of GHG and air emission under the three modeled scenarios from 2017 to 2050.



Figure 11. Social impact of air pollution and GHG emission reduction for the 100% RPS and 10% RPS scenarios compared to the 20% RPS scenario from 2017 to 2050.

County	20% RPS	100% RPS	10% RPS
Bernalillo	38	18	43
Catron	0	0	0
Chaves	0.09	0.78	0.02
Cibola	0	0	0
Colfax	0.24	0.24	0.24
Curry	0	0	0
De Baca	0	0	0
Dona Ana	90	69	95
Eddy	23	1	27
Grant	26	4	31
Guadalupe	0	0	0
Harding	0	0	0
Hidalgo	32	10	36
Lea	160	139	165
Lincoln	0	0	0
Los Alamos	9	5	10
Luna	101	79	105
Mc Kinley	93	71	97
Mora	0	0	0
Otero	0.09	0.78	0.02
Quay	0	0	0
Rio Arriba	0	0	0
Roosevelt	0	0	0
Sandoval	0.09	0.78	0.02
San Juan	785	764	790
San Miguel	0	0	0
Santa Fe	0.09	0.78	0.02
Sierra	0	0	0
Socorro	0	0	0
Taos	0	0	0
Torrance	0	0	0
Union	0	0	0
Valencia	45	23	49

Table 4. Annual average thousand tonnes of GHG emissions by county and modeled scenarios.

Note: Average values are in thousand tonnes and from 2017 to 2050; "0" means no change.

Since coal is the only energy source that emits mercury and since it stays unchanged throughout our study period, mercury is therefore assumed to be the same amount in all three scenarios, i.e., 3 tonnes. The 100% RPS scenario results in a roughly 500 tonne reduction in SO₂ emissions (approximately 3 million USD (2010\$) in social benefit) compared to the 20% RPS scenario, while the 10% RPS scenario results in an increase of more than

100 tonnes of SO₂ (1 million USD (2010\$) in social cost) cumulatively from 2017 to 2050. NO_x emissions in the RE intensive scenario are reduced by 6649 tonnes and a 7 million USD (2010\$) increase in social benefits compared to the 20% RPS scenario, while the fossil fuel intensive scenario yields 1990 tonnes more NO_x and 2 million USD (2010\$) more in social costs. Lastly, PM emission in the 100% RPS scenario is reduced by 5612 tonnes, resulting in a 123 million USD (2010\$) increase in social benefits compared to the reference case scenario, while the 10% RPS scenario yields 1215 tonnes more and 27 million USD (2010\$) in social costs.

Table 5 summarizes the cumulative avoided air pollution, social benefit, and the premature mortality and morbidity impact of air pollution under the 100% RPS and the 10% RPS scenarios from 2017 to 2050 compared to the 20% RPS scenario. The 20% RPS scenario is estimated to have 408 to 924 adult fatalities caused by a combination of SO₂, NO_x, and PM pollutants. The 100% RPS scenario has the potential to avoid 23 to 52 premature mortality incidences, while the 10% RPS scenario increases 5 to 11 fatalities due to exposure to ambient pollution, when compared to the reference scenario. While the majority (>90%) of social benefits for each scenario comes from avoiding premature mortality [12], we also estimated a number of additional morbidity benefits, from avoiding nonfatal heart attacks, hospital visits for asthma, or other cardiopulmonary conditions, to fewer lost work or school days. For example, the 100% RPS scenario is estimated to result in avoiding 19 visits to the emergency department or hospital for cardiopulmonary conditions as well as approximately 3000 fewer lost work or school days from 2017 to 2050.

Quitcome	100% RPS			10% RPS			20% RPS		
	SO ₂	NO _x	PM	SO ₂	NO _x	PM	SO ₂	NO _x	PM
Emission Reductions (Thousand Tonnes)									
	0.45	6.7	5.7	-0.1	$^{-2}$	-1.2	254	1094	29
Social Benefits (2010 million USD)									
	3.4	6.6	122.8	-0.7	-2.0	-26.6	1928	1078	640
Premature Mortality Incidences									
Krewski et al. [13] ^a	0	1	22	0	0	-5	207	95	107
Lepeule et al. [14] ^a	1	2	50	0	0	-11	464	219	241
Morbidity Incidences									
Emergency department visits for asthma	0	0	7	0	0	-1	81	37	36
Acute bronchitis	1	2	36	0	0	-8	372	251	181
Lower respiratory symptoms	9	20	464	-2	-6	-100	4699	3178	2312
Upper respiratory symptoms	13	29	679	-3	-9	-146	6735	4540	3334
Minor restricted-activity days	319	643	16,682	-69	-193	-3586	169,722	99,043	82,791
Lost workdays	54	110	2782	-12	-33	-600	28,485	16,673	13,975
Asthma exacerbation	5	1892	630	-2	-522	-156	5900	226,256	3938
Hospital admissions, respiratory	0	0	5	0	0	-1	48	20	24
Hospital admissions, cardiovascular	0	0	6	0	0	-1	61	26	30
Nonfatal Heart Attacks Incidences (Age > 18)									
Peters et al. (2001) ^b	0	1	22	0	0	-5	202	85	105
Pooled estimate of 4 studies	0	0	2	0	0	-1	22	9	11

Table 5. Accumulated emissions, social benefits, and mortality and morbidity incidence reductions compared to the reference case scenario (20% RPS) using SO₂, NOx, and PM reductions as a result of RE installation from 2017–2050.

Note: Positive value means reduction, whereas negative value indicates addition. ^{a,b} Multipliers from these studies are used to calculate the mortality and morbidity incidences.

Fossil fuel and RE power plants are contributors to avian mortality; fossil fuel plants induce fatality through plant operation, acid rain, mercury, and climate change, while bird fatality associated with wind and PV power plants is mainly due to bird colliding with turbine blades and panels respectively [15,38,39]. Figure 12 summarizes avian mortality caused by different energy sources (i.e., coal, NG, wind, and PV) under different scenarios. The 20% RPS scenario leads to 5.131 million avian fatalities, of which fossil fuel

is responsible for approximately 99% of the overall number of deaths. Compared to the 20% RPS scenario, the 100% RPS scenario has the potential to save 441 thousand deaths, while the 10% RPS scenario leads to 106 thousand more avian fatalities. Lastly, the RE intensive scenario leads to more than 4.69 million bird deaths, with fossil fuel sources being responsible for 97% of the overall number.



Figure 12. Avian mortality caused by coal, NG, wind, and PV power plants under the three modeled scenarios from 2017 to 2050.

Dissanayake and Ando [40] conducted a choice experiment survey in Illinois and found that their respondents are willing to pay between 1.11 and 1.13 USD for each extra bird per year, and between 7.72 and 10.22 USD for each endangered species annually. Since we were unable to discern different types of birds (generic versus endangered species) in our analysis, we utilized the mean value of the upper-level estimates as to how much each bird death is worth. We estimated that the 100% RPS scenario is capable of saving 3 million USD in bird mortality, while the 10% RPS scenario costs the state 1 million USD more in avian mortality, when compared with the 20% RPS scenario. We performed Kolmogorov–Smirnov tests and *t*-tests on human and avian mortality, and also on air pollutants (except mercury), and we found similar results to those of GHG and water consumption.

4.4. Summary of Cumulative Results

Our analysis seeks to investigate the economic and environmental impacts of the status quo scenario, along with two future scenarios. Without considering environmental impacts such as water usage, air pollution, GHG, and avian mortality, our results suggest that the reference case and the fossil fuel intensive scenarios lead to higher economic output and total employment impacts than the RE intensive scenario, though not statistically significant. Once the environmental impacts are included, these results no longer hold. Compared to the 20% RPS scenario, cumulatively, the 100% RPS scenario results in 3095 million USD (2017\$) higher benefit and the 10% RPS scenario in 3325 million USD (2017\$) more cost to the state. This makes the 100% RPS the best scenario, 20% RPS the second best, and 10% RPS the worst-case scenario, when both environmental and economic impacts are taken into account. Thus, the higher the RPS level, the higher the overall benefit to the state. Table 6 summarizes the state cumulative results in relation to the 20% RPS scenario. At the county level, compared to the 20% RPS case, we found that RE suitable counties are net gainers (in terms of both economic and environmental impacts), while fossil fuel counties suffer economically and benefit environmentally under the 100% RPS scenario. The opposite holds true when comparing the 10% RPS scenario against the 20% RPS scenario.

Outcome	100% RPS, in Million USD	10% RPS, in Million USD
Economic Output	-3962 (-120)	-1881 (-57)
Water benefit	59 (2)	-13(-0.4)
CO ₂	6865 (208)	-1402(-42)
SO ₂	3 (0.1)	-1(-0.03)
NO _x	7 (0.2)	-2(-0.1)
PM2.5	123 (4)	-27 (-1)
Bird mortality	+3 (0.1)	-1(-0.03)
Total monetary value	3095 (94)	-3325 (-101)
Employment value ^a	-180 (-5)	+328 (10)

Table 6. Summary of cumulative results in relation to the 20% RPS scenario from 2017–2050.

Numbers in parentheses are annual values. ^a Employment monetary values are based on salary of 46,500 USD for construction and 66,000 USD for O&M jobs [36], calculated based on the employment count (see Section 4.2) of -573 (-17 annual) jobs and 3663 (111 annual) jobs for the 100% RPS and 10% RPS scenarios, respectively.

5. Conclusions and Policy Implications

Legislators across the globe are supporting policies that move toward electricity generation from renewable resources. To this end, some jurisdictions in the U.S. have enacted regulations, such as the RPS. These provide a mechanism that can result in not only GHG emission reduction but also water preservation. This is especially prudent in geographic locations with limited water resources. Moreover, RPS can support jobs, although the primary policy target of an RPS is not focused squarely on job creation.

This study provided a roadmap of how to quantify the economic and environmental impacts of three scenarios, in which not only the RPS level varies but also the energy sector dynamics, technological cost, and price of energy. Specifically, we modelled New Mexico's newly enacted RPS policy, where it increases from the status quo of 20% by 2020 to 100% by 2050. We also studied a scenario where RPS decreases to 10% by 2050. In so doing, we combined results from input–output (JEDI and IMPLAN) analyses, econometrics (Stata), and GIS (ArcGIS), and we created a unique SD model that enabled us to assess regional economic and environmental impacts of different scenarios. Our contribution to the current body of literature is twofold: not only did we assess different RPS scenarios by considering the underlying dynamics within the energy sector, but we also assessed these impacts at a lower granular level (i.e., county level).

Under the status quo scenario, the estimates in our model accounted for 152 thousand cumulative full-time equivalent jobs, 24 billion USD in economic output, 3648 million USD in air quality cost, 36 billion USD in climatic cost, 527 million USD worth of water use, 5 million avian mortality, and 409-924 premature mortality. Compared with this status quo scenario, our analysis suggests that the RE intensive scenario (100% RPS) leads to less cumulative employment and economic output, but much higher social benefits compared to the 20% RPS scenario, i.e., 500–15,000 fewer cumulative jobs, 3–4 billion USD less in cumulative economic output, 132 million USD less in air quality cost, 7 billion USD less in climatic cost, 58 million USD less in value of water use, 441–485 thousands less in avian mortality, and 23-53 less in premature mortality. The 10% RPS scenario leads to approximately 4000 more jobs, 2 billion USD less in cumulative economic output, 29 million USD more in air quality cost, 1 billion USD more in climatic cost, 13 million USD more in value of water use, 100 thousand more in avian mortalities, and 5–11 more in premature mortality than the 20% RPS scenario. Considering the environmental impacts, our analysis finds that the Senate Bill RPS scenario (100% RPS) is the best scenario, followed by the status quo scenario, and the 10% RPS scenario is the worst case.

Higher levels of RPS policy aligns with support from New Mexicans. In separate work by the co-authors [41,42], we estimated that a sample of New Mexicans are willing to pay 5.4 USD per year on top of their annual electricity bill for each 1% increase in the current level of RPS (20%). To achieve a 100% RPS by 2050, we extrapolate that, all else equal, New Mexicans are willing to pay 58, 180, 373, 581, 803, and 1144 million USD (2017\$) in 2020, 2025, 2030, 2035, 2040, and 2050, respectively. Note that the wide range of willingness to pay is due to the way the bill requirements of achieving 80% RPS by 2040 were designed. Under this bill, electric utility companies were required to increase current RPS level to 25% by 2020, 35% by 2025, 50% by 2030, 65% by 2035, and 80% by 2040. The higher the percentage, the higher residents are willing to pay.

Although scenarios with a lower level of RPS might result in supporting a higher number in employment (in the fossil fuel sector), these scenarios lead to much higher social cost of GHG and ambient pollution (i.e., premature mortality and morbidity) and water usage. This suggests that coming up with an overarching policy that benefits both the environment and economy is not an easy task. Policymakers seeking to promote energy policies may need to consider not only the economic benefit associated with energy development but also social welfare. In other words, RPS policies are more desirable when internalizing external costs and hence correcting for market failure [21,43,44].

Further, the most decisive conclusion that can be drawn from job comparison across different scenarios is that the higher the RE development level, the more disperse and rural the employment impact. On the contrary, the higher the level of fossil fuel deployment, the less diverse and rural the job impact. San Juan County among all is expected to experience a net negative (loss) in O&M jobs, i.e., a loss of 780 jobs from coal-fired power plant retirements after 2037 and depending on the scenario, a gain of an annual average of 84 (100% RPS) to 601 (10% RPS) jobs. Concurrently, the state is estimated to experience nearly 686 billion USD (100% RPS) in social benefits, particularly from the coal power plants retirement. The disparity in job and economic output distribution across counties and energy sources suggests that counties with varying energy potential and population density may experience variation in impacts. In other words, some counties are likely to be net gainers while others may suffer.

The results of this study are broadly consistent with that in the literature [20,26,29,45–49]. We do recognize that the majority of these studies had explicit research questions only on wind energy. For example, some studies sought to measure the actual economic impact of a particular wind installation at county level (e.g., [48]), while others estimated a wind vision for the U.S. (e.g., [26,49]) or the environmental and economic impact of RPS policies nationwide for solely one year (i.e., [20]). Similar to Barbose et al. [20], Millstein et al. [29], and Wiser et al. [21], our model suggests that RPS policies have the potential to yield billions of dollars in climatic and air-quality benefits as well as economic benefits. Similar to preceding studies, we found that increasing RPS does not result in stimulating the economy of a state [3,4], but it does impact the environment positively [29,45,50]. Our contribution to the literature is that we demonstrated that increasing RPS does stimulate the economy of the state at the more granular levels (especially rural counties).

The tools and theories integrated for the analysis in this research are broadly transferable across a wide range of topics and/or regions. For example, a similar approach can be taken to evaluate RPS policies in each one of the other 28 states with such regulations. Our model can be modified and used for states with existing 100% RPS policies (Hawaii, California, Washington, Maine, New York, and Virginia), and those with promises for 100% clean electricity (Colorado, Connecticut, Massachusetts, Illinois, Oregon, New Jersey, Nevada, Wisconsin, and Puerto Rico). Additionally, our state-of-the-art modeling and set of methods are applicable to other topics, such as the impact of decarbonization through a battery of smart grid (e.g., smart meter), transportation (e.g., electric vehicle), and energy-efficient buildings; 100% RE for all sectors (i.e., electricity, heating/cooling, transportation, and industry); oil and natural gas extraction; and the agriculture sector on regional economies. Another expansion of this analysis could include developing nations, as well as other developed countries with similar regulatory mandates. One potential limitation of this work is that our model does not calculate electricity rates for each scenario and takes rates as independent. More expensive scenarios could potentially result in higher electricity rates, which can impact economic activity. This is also important as it has the potential to impact customers' perspective and willingness to pay towards higher level of RE diffusion. Another caveat is that we assume that employment impacts are fully

provided (100%) by local residents, which is not typically the case in real-world settings; although the model is capable of varying this assumption, we chose not to include this here for the purpose of brevity. Future research should account for data uncertainty and present results as confidence intervals rather than precise values. This can be done by using Monte-Carlo simulations. This study's results provided improved information for state policymakers seeking to alter RPS policies and can also be extrapolated to states with similar energy policies.

Supplementary Materials: The following will be available online at www.jamalmamkhezri.weebly. com/research.html.

Author Contributions: J.M.: Conceptualization, data curation, formal analysis, funding acquisition, investigation, methodology, resources, software, validation, writing—original draft preparation; L.A.M.: methodology, software, validation, formal analysis, writing—review and editing; J.M.C.: conceptualization, supervision, resources, funding acquisition, writing—review and editing. All authors have read and agreed to the published version of the manuscript.

Funding: This research was partially funded by the U.S. National Science Foundation awards #IIA-1301346 and OIA-1757207. Jamal Mamkhezri acknowledges support received from the University of New Mexico Center for Regional Studies.

Institutional Review Board Statement: Not Applicable.

Informed Consent Statement: Not Applicable.

Data Availability Statement: Publicly available datasets were used in this study. Readers are directed to Table S2 for an overview of the key variables' data sources.

Acknowledgments: We thank the late Ronald Cummings, Jennifer Thacher, and Robert Berrens, as well as participants at the 2019 Southern Economic Association (SEA) conference, the 2020 Association of Environmental and Resource Economists at Western Economic Association International conference, the 2020 Energy Policy Institute workshop, the 2020 System Dynamics Society Economics special interest group, and at the 37th and 38th International Conference of the System Dynamics Society for on an earlier version of this paper. We also thank the four anonymous reviewers for their insightful comments. Any errors are, of course, our own.

Conflicts of Interest: The authors declare no conflict of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript, or in the decision to publish the results.

Abbreviations

Abbreviation	Definition
AEO	Annual Energy Outlook
eGRID	Emissions and Generation Resource Integrated Database
EIA	Energy Information Administration
EPA	Environmental Protection Agency
GHG	Greenhouse-gas
IMPLAN	Impact Analysis for Planning
JEDI	Jobs and Economic Development Impact
MWh	Megawatt-hour
NGp	Peaker Natural Gas
NGb	Baseload Natural Gas
NREL	National Renewable Energy Laboratory
O&M	Operating and maintenance
PV	Utility-scale photovoltaic solar
RE	Renewable energy
RPS	Renewable portfolio standards
RPV	Residential photovoltaic solar
SD	System dynamics

References

- 1. NYSERDA. NYSERDA Renewable Portfolio Standard Main Tier 2013 Program Review, Volume 2—Main Tier Current Portfolio Analysis; NYSERDA: Albany, NY, USA, 2013.
- Divounguy, O.; Rea, S.H.; Nichols, J.; Spitzwieser, L. The Impact of Renewables Portfolio Standards on the Ohio Economy. Available online: https://www.buckeyeinstitute.org/research/detail/the-impact-of-renewables-portfolio-standards-on-theohio-economy (accessed on 22 July 2018).
- 3. Upton, G.B.; Snyder, B.F. Funding Renewable Energy: An Analysis of Renewable Portfolio Standards. *Energy Econ.* 2017, 66, 205–216. [CrossRef]
- 4. Zhou, S.; Solomon, B.D. Do Renewable Portfolio Standards in the United States Stunt Renewable Electricity Development beyond Mandatory Targets? *Energy Policy* **2020**, *140*, 111377. [CrossRef]
- Carley, S.; Davies, L.L.; Spence, D.B.; Zirogiannis, N. Empirical Evaluation of the Stringency and Design of Renewable Portfolio Standards. *Nat. Energy* 2018, *3*, 754–763. [CrossRef]
- 6. Ford, A.; Bull, M. Using System Dynamics for Conservation Policy Analysis in the Pacific Northwest. *Syst. Dyn. Rev.* **1989**, *5*, 1–16. [CrossRef]
- 7. Olaya, Y.; Dyner, I. Modelling for Policy Assessment in the Natural Gas Industry. J. Oper. Res. Soc. 2005, 56, 1122–1131. [CrossRef]
- 8. Qudrat-Ullah, H. Understanding the Dynamics of Electricity Generation Capacity in Canada: A System Dynamics Approach. *Energy* **2013**, *59*, 285–294. [CrossRef]
- 9. Qudrat-Ullah, H.; Seong, B.S. How to Do Structural Validity of a System Dynamics Type Simulation Model: The Case of an Energy Policy Model. *Energy Policy* **2010**, *38*, 2216–2224. [CrossRef]
- 10. Tidwell, V.C.; Kobos, P.H.; Malczynski, L.; Klise, G.; Hart, W.E.; Castillo, C. *Decision Support for Integrated Water-Energy Planning*; Sandia National Lab.(SNL-NM): Albuquerque, NM, USA, 2009; p. 79.
- 11. Ying, Z.; Xin-gang, Z.; Zhen, W. Demand Side Incentive under Renewable Portfolio Standards: A System Dynamics Analysis. *Energy Policy* **2020**, *144*, 111652. [CrossRef]
- 12. US EPA. Regulatory Impact Analyses for Air Pollution Regulations. Available online: https://www.epa.gov/economic-and-costanalysis-air-pollution-regulations/regulatory-impact-analyses-air-pollution (accessed on 20 December 2018).
- 13. Krewski, D.; Jerrett, M.; Burnett, R.T.; Ma, R.; Hughes, E.; Shi, Y.; Turner, M.C.; Pope III, C.A.; Thurston, G.; Calle, E.E. *Extended Follow-up and Spatial Analysis of the American Cancer Society Study Linking Particulate Air Pollution and Mortality*; Health Effects Institute: Boston, MA, USA, 2009.
- 14. Lepeule, J.; Laden, F.; Dockery, D.; Schwartz, J. Chronic Exposure to Fine Particles and Mortality: An Extended Follow-up of the Harvard Six Cities Study from 1974 to 2009. *Environ. Health Perspect.* **2012**, *120*, 965. [CrossRef]
- 15. Sovacool, B.K. Contextualizing Avian Mortality: A Preliminary Appraisal of Bird and Bat Fatalities from Wind, Fossil-Fuel, and Nuclear Electricity. *Energy Policy* **2009**, *37*, 2241–2248. [CrossRef]
- 16. Woodruff, T.J.; Grillo, J.; Schoendorf, K.C. The Relationship between Selected Causes of Postneonatal Infant Mortality and Particulate Air Pollution in the United States. *Environ. Health Perspect.* **1997**, *105*, 608. [CrossRef]
- 17. Steinberg, D.; Porro, G.; Goldberg, M. Preliminary Analysis of the Jobs and Economic Impacts of Renewable Energy Projects Supported by the.. Section.. 1603 Treasury Grant Program; National Renewable Energy Lab.(NREL): Golden, CO, USA, 2012.
- 18. Mamkhezri, J.; Bohara, A.K.; Camargo, A.I. Air Pollution and Daily Mortality in Mexico City Metropolitan Area. *Atmósfera* **2020**, 33, 249–267. [CrossRef]
- 19. Maupin, M.A.; Kenny, J.F.; Hutson, S.S.; Lovelace, J.K.; Barber, N.L.; Linsey, K.S. Estimated Use of Water in the United States in 2010; US Geological Survey. 2014. Available online: http://pubs.er.usgs.gov/publication/cir1405 (accessed on 5 October 2020).
- 20. Barbose, G.; Wiser, R.; Heeter, J.; Mai, T.; Bird, L.; Bolinger, M.; Carpenter, A.; Heath, G.; Keyser, D.; Macknick, J.; et al. A Retrospective Analysis of Benefits and Impacts of U.S. Renewable Portfolio Standards. *Energy Policy* **2016**, *96*, 645–660. [CrossRef]
- 21. Wiser, R.; Mai, T.; Millstein, D.; Barbose, G.; Bird, L.; Heeter, J.; Keyser, D.; Krishnan, V.; Macknick, J. Assessing the Costs and Benefits of US Renewable Portfolio Standards. *Environ. Res. Lett.* **2017**, *12*, 094023. [CrossRef]
- 22. Jacobson, M.Z.; Delucchi, M.A.; Bazouin, G.; Bauer, Z.A.; Heavey, C.C.; Fisher, E.; Morris, S.B.; Piekutowski, D.J.; Vencill, T.A.; Yeskoo, T.W. 100% Clean and Renewable Wind, Water, and Sunlight (WWS) All-Sector Energy Roadmaps for the 50 United States. *Energy Environ. Sci.* 2015, *8*, 2093–2117. [CrossRef]
- 23. Heard, B.P.; Brook, B.W.; Wigley, T.M.; Bradshaw, C.J. Burden of Proof: A Comprehensive Review of the Feasibility of 100% Renewable-Electricity Systems. *Renew. Sustain. Energy Rev.* 2017, *76*, 1122–1133. [CrossRef]
- 24. Shaner, M.R.; Davis, S.J.; Lewis, N.S.; Caldeira, K. Geophysical Constraints on the Reliability of Solar and Wind Power in the United States. *Energy Environ. Sci.* 2018, *11*, 914–925. [CrossRef]
- 25. Cole, W.; Frazier, W.; Donohoo-Vallett, P.; Mai, T.; Das, P. 2018 Standard Scenarios Report: A U.S. Electricity Sector Outlook; National Renewable Energy Lab. (NREL): Golden, CO, USA, 2018.
- 26. Wiser, R.; Lantz, E.; Mai, T.; Zayas, J.; DeMeo, E.; Eugeni, E.; Lin-Powers, J.; Tusing, R. Wind Vision: A New Era for Wind Power in the United States. *Electr. J.* 2015, *28*, 120–132. [CrossRef]
- 27. Godby, R.; Taylor, D.; Coupal, R. An Assessment of Wyoming's Competitiveness to Attract New Wind Development and the Potential Impacts Such Development May Bring the State; University of Wyoming Center for Energy Economics & Public Policy: Laramie, WY, USA, 2016.

- 28. Considine, T.J.; Manderson, E.J.M. The Cost of Solar-Centric Renewable Portfolio Standards and Reducing Coal Power Generation Using Arizona as a Case Study. *Energy Econ.* **2015**, *49*, 402–419. [CrossRef]
- 29. Millstein, D.; Wiser, R.; Bolinger, M.; Barbose, G. The Climate and Air-Quality Benefits of Wind and Solar Power in the United States. *Nat. Energy* **2017**, *2*, 17134. [CrossRef]
- 30. Novan, K. Valuing the Wind: Renewable Energy Policies and Air Pollution Avoided. *Am. Econ. J. Econ. Policy* 2015, *7*, 291–326. [CrossRef]
- 31. Fell, H.; Kaffine, D.T.; Novan, K. Emissions, Transmission, and the Environmental Value of Renewable Energy. *Am. Econ. J. Econ. Policy* **2021**, *13*, 241–272. [CrossRef]
- 32. Forrester, J.W. Counterintuitive Behavior of Social Systems. Technol. Forecast. Soc. Chang. 1971, 3, 1–22. [CrossRef]
- Sterman, J. System Dynamics: Systems Thinking and Modeling for a Complex World; Massachusetts Institute of Technology. Engineering Systems Division. 2002. Available online: https://dspace.mit.edu/handle/1721.1/102741 (accessed on 5 October 2020).
- 34. NREL WINDExchange: U.S. Installed and Potential Wind Power Capacity and Generation. Available online: https://windexchange.energy.gov/maps-data/321 (accessed on 16 December 2018).
- 35. Mamkhezri, J. Market and Non-Market Valuation of Renewable Energy. 2019. Available online: https://digitalrepository.unm. edu/econ_etds/106/ (accessed on 6 February 2021).
- 36. Mamkhezri, J.; Thacher, J.; Chermak, J. Socioeconomics and Environmental Impacts of Solar and Wind Projects Tied to Renewable Portfolio Standards. In Proceedings of the Riding the Energy Cycles, 35th USAEE/IAEE North American Conference, Houston, TX, USA, 12–15 November 2017; Available online: http://www.iaee.org/proceedings/article/14863 (accessed on 6 February 2021).
- 37. Cohen, M. Avoided Water Cost of Electricity Generation for Solar PV and Wind Technologies in Southern California. Master's Theses and Project Reports. 2014. Available online: https://digitalcommons.calpoly.edu/cgi/viewcontent.cgi?referer=https: //scholar.google.com/&httpsredir=1&article=2403&context=theses (accessed on 6 February 2021).
- McCubbin, D.; Sovacool, B.K. Quantifying the Health and Environmental Benefits of Wind Power to Natural Gas. *Energy Policy* 2013, 53, 429–441. [CrossRef]
- 39. Walston, L.J.; Rollins, K.E.; LaGory, K.E.; Smith, K.P.; Meyers, S.A. A Preliminary Assessment of Avian Mortality at Utility-Scale Solar Energy Facilities in the United States. *Renew. Energy* **2016**, *92*, 405–414. [CrossRef]
- 40. Dissanayake, S.T.M.; Ando, A.W. Valuing Grassland Restoration: Proximity to Substitutes and Trade-Offs among Conservation Attributes. *Land Econ.* **2014**, *90*, 237–259. [CrossRef]
- 41. Mamkhezri, J.; Thacher, J.A.; Chermak, J.M. Consumer Preferences for Solar Energy: A Choice Experiment Study. *Energy J.* 2020, 41. [CrossRef]
- 42. Mamkhezri, J.; Thacher, J.A.; Chermak, J.M.; Berrens, R.P. Does the Solemn Oath Lower WTP Responses in a Discrete Choice Experiment Application to Solar Energy? *J. Environ. Econ. Policy* **2020**, *9*, 447–473. [CrossRef]
- 43. Borenstein, S. The Private and Public Economics of Renewable Electricity Generation. J. Econ. Perspect. 2012, 26, 67–92. [CrossRef]
- 44. Change, I.P. (Ed.) *Climate Change 2014: Mitigation of Climate Change*; Cambridge University Press: Cambridge, UK, 2015; ISBN 978-1-107-05821-7.
- 45. Hollingsworth, A.; Rudik, I. External Impacts of Local Energy Policy: The Case of Renewable Portfolio Standards. J. Assoc. Environ. Resour. Econ. 2019, 6, 187–213. [CrossRef]
- Lantz, E.; Tegen, S.; Paper, I. NREL Is Operated by Midwest Research Institute

 Battelle Contract No. DE-AC36-99-GO10337 NOTICE. 2008. Available online: http://www.osti.gov/servlets/purl/1219190/ (accessed on 6 February 2021).
- 47. Llera Sastresa, E.; Usón, A.A.; Bribián, I.Z.; Scarpellini, S. Local Impact of Renewables on Employment: Assessment Methodology and Case Study. *Renew. Sustain. Energy Rev.* 2010, 14, 679–690. [CrossRef]
- 48. Slattery, M.C.; Lantz, E.; Johnson, B.L. State and Local Economic Impacts from Wind Energy Projects: Texas Case Study. *Energy Policy* **2011**, *39*, 7930–7940. [CrossRef]
- 49. Wiser, R.; Bolinger, M.; Heath, G.; Keyser, D.; Lantz, E.; Macknick, J.; Mai, T.; Millstein, D. Long-Term Implications of Sustained Wind Power Growth in the United States: Potential Benefits and Secondary Impacts. *Appl. Energy* **2016**, *179*, 146–158. [CrossRef]
- 50. Palmer, K.; Burtraw, D. Cost-Effectiveness of Renewable Electricity Policies. Energy Econ. 2005, 27, 873–894. [CrossRef]





Article Comprehensive Benefit Analysis of Port Shore Power Based on Carbon Trading

Yang He and Yun Zhu *

School of Electrical Engineering, Guangxi University, Nanning 530004, China * Correspondence: zhuyun@gxu.edu.cn

Abstract: The concept of "oil to electricity" is crucial for expanding the share of electricity in final energy consumption as well as for encouraging energy efficiency and emission reduction. Initially, a multidimensional strategy analysis is conducted for the government, ports, and ships concerned. From an economics perspective, a mathematical model of electricity substitution benefit analysis based on multiagent cooperative game theory under cap and trade and carbon tax policies is constructed, and the effect of carbon emissions caused by ships on the environment and society is converted into economic value. How several variables, such as transformation costs, ship electricity consumption, subsidy rates, carbon tax prices, and the ratio of shore power usage time to berthing time, affect the functioning of shore power is analyzed. The best electricity price under various circumstances is determined while considering the benefits of the three parties to maximize social welfare. The reduction in carbon dioxide and pollutant emissions is calculated. Meanwhile, the environmental advantages of the "replacement of oil with electricity" procedure are estimated. An example supports the claim that the suggested modeling approach can successfully resolve the economic benefits of each participant for the period that fosters the growth of electricity replacement projects and offers a sound scientific foundation for the formation of pertinent legislation.

Keywords: carbon trading; cooperative game; pollutant emission; shore power

1. Introduction

Seaborne trade stalled in 2020 amid the COVID-19 epidemic and the anticipated downturn in global economic growth. Surveys conducted by UNCDAT have revealed that the world container throughput declined by 1.2% to 815.6 million 20-foot TEUs [1]. Governments advocated for citizens to stay inside to prevent contact, which tremendously accelerated the rise of international e-commerce. The distribution of vaccines has slowed the epidemic's growth and deaths, enabling the recovery of international trade. The beginning of the economic recovery was heralded in 2021, with seaborne trade predicted to increase by 4.3% [2]. The maritime sector's quick ascent has resulted in significant emissions of pollutants such as CO_2 , SO_2 , and NO_x . As of 2012, shipping was responsible for 972 million tons of greenhouse gas release, or 2.5% of all releases worldwide [3]. It not merely raises the global temperature but also leads to respiratory illnesses in those who live close to ports and coastlines [4]. There is a consensus among experts that 60% to 90% of the diffusion in ports stems from ships, which also account for 70% of marine diffusion [5].

Several nations and international organizations are pursuing numerous explorations and research to lessen the issue of pollution discharge from ships. Based on the "IMO 2020" guideline, ships operating outside specified emission-control areas can diminish sulfur oxide outflow by 8.5 million tons while exploiting low-sulfur oil with a sulfur content of 0.50% m/m [6]. More than 570,000 residents will die prematurely if the SO_x limit reduction is postponed from 2020 to 2025 [7]. Even though low-sulfur oil modestly reduces NO_x discharge by 10% [8], the maritime industry still contributes to 250,000 fatalities [9]. The usage of shore power minimizes pollution emissions by 94–97% when berthing at the

Citation: He, Y; Zhu, Y. Comprehensive Benefit Analysis of Port Shore Power Based on Carbon Trading. *Energies* **2023**, *16*, 2755. https://doi.org/ 10.3390/en16062755

Academic Editor: George Halkos

Received: 7 February 2023 Revised: 12 March 2023 Accepted: 13 March 2023 Published: 15 March 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). port [10]. It satisfies fundamental needs such as lighting, cooling, and communication, notably improving air quality [11], decreasing carbon emissions by 800,000 tons, and elevating environmental benefits [12,13].

The present literature on shore power is confined and centers especially on technology [14,15], economy [16,17], management [18,19], policy promotion [20], etc. One study by Qi [8] observed the trend in obstacles to the upgrading of shore power in China, focusing on the economic evaluation for different stakeholders. Zhao [21] considered the effects of port size, fines, and subsidies on the evolutionary game to analyze the financial relationship between the government and the port. A mathematical model was constructed by Wu [22] to investigate how government subsidy schemes might help shorten the outflow from ship berths. Song [23] set up four parties, the government, the port, the ship, and the power grid, then pondered the cost-effectiveness of each in the shore power system to calculate the optimal shore power price. Through quantitative evaluation, Tseng [24] demonstrated that environmental policies levying pollution taxes can immensely suppress pollutant discharge.

The global community has agreed on limiting carbon emissions since the "Kyoto Protocol" took effect. On the one hand, developed nations such as the European Union, Japan, and Australia were the first to adopt cap-and-trade and carbon tax policies, which victoriously decreased CO_2 emissions [25]. On the other hand, China pledged to achieve carbon neutrality by 2060 at the 75th UN General Assembly. However, there are few surveys on the carbon trade mechanism of shore power, and the majority of studies concentrate on economic factors. Murray [26] discovered the carbon price legislation lowered British Columbia's emissions by 5–15% through modeling. The EU Emissions Trading Scheme's implementation has resulted in a 0.5–2 million ton CO_2 reduction [27]. Song [28] developed a stochastic model to explore the effects of various carbon tax rates on the growth of logistical capacity. A dual-objective optimization model was developed by Liu [29] to discuss the liner's best performance under the carbon tax policy. Chen [30] described a social optimal welfare model to assess how carbon taxes affect production, consumption, and redistribution.

This study uses multiagent game behavior as its research object in the "oil-to-electricity" conversion process. Economically speaking, the government, the port, and the ship are strategically examined, and the cost of the port includes the value of the carbon. The whole social welfare is maximized by the favorable shore electricity price. The environmental advantages of "oil-to-electricity" are counted simultaneously.

2. Electricity Substitution Multiagent Game Model

2.1. Multiagent Game Analses

Figure 1 depicts the evolution of port shore power based on a cooperative game. Due to the contradictory objectives being sought by the government, port, and ships while replacing electricity, a game of interests has developed.

Ships consuming fuel oil pollute the environment, harm locals' health, and make it more difficult to achieve carbon neutrality, all of which run counter to the government's stated environmental objectives. From an economic and market standpoint, shipping corporations feel there are few ports with the ability to supply shore power, whereas ports think there are not many ships with the capability to use shore power. Nobody wants to take the initiative and make themselves passive. Now is the time for the government to take a two-pronged strategy and develop policies that would encourage the implementation of shore power projects through incentives and sanctions.

The government subsidizes port renovation and applies carbon emission controls to increase port operating costs. The government encourages the development of the energy structure by funding ship retrofits while imposing environmental protection tariffs on pollutants released through the use of fuel oil. By giving priority to berthing for ships using shore power, ports can entice ship retrofits. Ships can decide whether to employ shore power depending on their financial circumstances.



Figure 1. Multiagent game of oil-to-electricity conversion.

Consequently, the three parties' game is impacted by the subsidy rate, carbon mechanism, environmental protection tax, and shore electricity price. The ideal shore power pricing can be attained in the endless game, allowing the electric energy replacement effort to continue.

This article proposes a simple method to calculate the carbon emissions of ports, which only considers the carbon emissions of ships. Other equipment will not be considered, such as harbor railway, quayside container crane, locomotive, and other special machinery. We calculate the carbon emissions of ships using fuel oil and shore power and add the carbon emissions to the economic value, including the port cost. Then, the tripartite economic game models under the two carbon mechanisms are established, respectively, and the impact of the power supply service price, the carbon price, and the proportion of the time using shore power to the docking time on the social welfare, government benefits, port benefits, and ship benefits is discussed.

2.2. Assumptions

To simplify the problem and facilitate modeling and subsequent analysis and discussion, the following assumptions are proposed for the research content:

- The government stipulates carbon emission caps or implements carbon tax policies for ports;
- The created carbon emissions belong to the port once the ship has berthed there.

2.3. Methodology

In order to address climate change, the Ministry of Ecology and Environment of China has formulated "Implementation Plan for Setting and Allocating the Total Amount of Carbon Emission Trading Quotas for 2019–2020 (Power Generation Sector)", which includes enterprises or other economic organizations that emit 26,000 ton of carbon dioxide equivalent or more in any one year from 2013 to 2019. As carbon emissions trading management has only just started, the central government is currently only regulating the power sector. However, in places such as Guangdong Province and Shanghai, it has already started to cover industries such as steel, chemicals, cement, paper, and aviation. The allocation methods are mainly the historical intensity method and the historical emissions method. The former is applicable to industrial enterprises with a high correlation between product output and carbon emissions, and well measured. The latter is suitable for industrial enterprises where the boundary has changed significantly in recent years and it is difficult to apply the industry baseline method or the historical intensity method.
In the historical intensity method, the annual base quota for enterprises is equal to the historical intensity base multiplied by the annual business volume. The historical intensity base is the weighted average of the enterprise's annual business volume of carbon emissions in the previous three years. The annual business volume is the business volume data of the enterprise for the current year verified by a third party verification agency and validated and confirmed by the relevant departments. In the historical emissions approach, an enterprise's annual base allowance is equal to the historical emissions base. In this article, port companies use the historical intensity method and carbon emissions from ports only consider carbon emissions from ships, not from other equipment such as shoreside cranes and locomotives. Thus, carbon emissions are only relevant to the activities of the ship, and the oil or electricity consumed per unit of power has a relevant carbon emission factor to calculate carbon emissions.

2.4. Parameter Descriptions

The following is a description of the parameters that appear in the mathematical model of the three-way game.

$$\begin{cases}
P^{k,i} = W_{sp}^{k,i} / T_{sp}^{k,i} \\
W_{sp}^{k} = \sum_{i=1}^{N_{i}} W_{sp}^{k,i} \\
W_{oil}^{k} = \sum_{i=1}^{N_{i}} P^{k,i} \left(T^{k,i} - T_{sp}^{k,i} \right) \\
W^{k} = W_{sp}^{k} + W_{oil}^{k}
\end{cases}$$
(1)

where *k* denotes the year; *i* denotes the type of ships; $P^{k,i}$ is the power of the ship's auxiliary engine, kWh; W_{sp}^k is the power consumed by the ship using shore power, kWh; W_{oil}^k is the power consumed by the ship using fuel oil, kWh; W^k is the total power consumption of the ship, kW h; $T^{k,i}$ is the berthing time of the ship, h; $T_{sp}^{k,i}$ is the time when the ship uses shore power, h.

$$C_{dj}^{k} = C_{grid}^{k} + C_{serve}^{k}$$

$$C_{e}^{k} = W_{sp}^{k} C_{grid}^{k}$$

$$C_{sp}^{k} = W_{sp}^{k} C_{dj}^{k}$$

$$T_{e}^{k} = C_{e}^{k} / (1 + 16\%) \times 16\%$$

$$(3)$$

$$T_s^k = W_{sp}^k C_{serve}^k / (1 + 6\%) \times 6\%$$

where C_{grid}^k is the electricity basic price, CMY/kWh; C_{serve}^k is the electricity service price, CMY/kWh; C_{dj}^k is the electricity actual price, CMY/kWh; C_e^k is the cost of purchasing electricity for the port, CMY; C_{sp}^k is the ship paying the port for electricity, CMY; T_e^k is to pay value-added tax to the government as the power grid provides electricity to the port, CMY; T_s^k is the port that makes a profit from providing shore power service to the ship and pays value-added tax to the government, CMY.

$$C_{oil,sp}^{k} = 10^{-6} W_{sp}^{k} E_1 C_{per-oil}$$

$$C_{oil,oil}^{k} = 10^{-6} W_{oil}^{k} E_1 C_{per-oil}$$
(4)

where $C_{oil,sp}^k$ represents the fuel cost savings by the ship using shore power, CMY; $C_{oil,oil}^k$ represents the cost of fuel oil used by the ship, CMY; E_1 is the fuel consumption per unit

of electricity emitted by the auxiliary engine, g/kWh; $C_{per-oil}$ is the price of marine fuel oil, CMY/Mt.

$$\begin{cases} T_{ep,sp}^{k} = 10^{-3} C_{dl} W_{sp}^{k} \sum_{n=1}^{N_{n}} F_{n} / E_{dl}^{n} \\ T_{ep,oil}^{k} = 10^{-3} C_{dl} W_{oil}^{k} \sum_{n=1}^{N_{n}} F_{n} / E_{dl}^{n} \end{cases}$$
(5)

where *n* denotes the type of pollutant; $T_{ep,sp}^k$ is the environmental protection tax saved by ships using shore power, CMY; $T_{ep,ol}^k$ is the environmental protection tax paid by ships using fuel oil, CMY; C_{dl} is the pollution factor pollutant discharge fee standard per unit of pollution equivalent, CMY/equivalent; F_n is the emission factor of pollutants discharged from the fuel oil of the ship's auxiliary engine, g/kWh; E_{dl}^n is the pollution equivalent value of pollutants, kg.

$$\begin{cases}
V_{sp}^{k} = 10^{-6} W_{sp}^{k} F_{e} \\
V_{oil}^{k} = 10^{-6} W_{oil}^{k} F_{c} \\
V_{actual}^{k} = V_{sp}^{k} + V_{oil}^{k}
\end{cases}$$
(6)

$$\begin{cases} F_{cef}^{k} = 10^{6} V_{actual}^{k} / W^{k} \\ F_{wef}^{k} = 10^{6} \frac{\sum (V_{actual}^{k-3} + V_{actual}^{k-2} + V_{actual}^{k-1})}{\sum (W^{k-3} + W^{k-2} + W^{k-1})} \end{cases}$$
(7)

where V_{sp}^k and V_{oil}^k are the CO₂ emissions of the ship using shore power and fuel oil, respectively, Mt; V_{actual}^k is the total CO₂ emissions of the ship, Mt; F_e is the annual average power supply emission factor of the regional power grid, g/kWh; F_c is the emission factor of carbon dioxide pollutants emitted by marine auxiliary engine fuel oil, g/kWh; F_{cef}^k is the comprehensive CO₂ emission factor of the ship, g/kWh; F_{wef}^k is the ship's weighted CO₂ emission factor, g/kWh. Equation (6) gives the calculation of the carbon emissions from the use of shore power and the use of oil, respectively, as well as the total emissions for the year. In Equation (7), the ship CO₂ weighted emission factor is taken as the weighted average of the carbon emissions per unit of business of the port in the previous three years, which is used as the base of the historical carbon emission intensity.

$$\begin{cases} V_{cap}^{k} = 10^{-6} \eta F_{wef}^{k} W^{k} \\ T_{c1}^{k} = \left(V_{actual}^{k} - V_{cap}^{k} \right) \times P_{c1}^{k} \end{cases}$$

$$\tag{8}$$

where V_{quato}^k is the port's carbon emission cap, Mt; η is the annual decline coefficient, taking 1; T_{c1}^k is the total amount of carbon trading, CMY; P_{c1}^k is the carbon price in the carbon trading market, CMY/Mt CO₂. Since only the power sector is currently subject to government regulation and other sectors are not yet subject to excessive restrictions, the annual decline coefficient in Equation (8) is set to 1. This gives the port room to strengthen its efforts to reduce carbon emissions. Carbon emission allowances for the year were calculated, as well as the fees paid in excess of the allowances.

$$T_{c2}^{k} = V_{actual}^{k} \times P_{c2}^{k} \tag{9}$$

where T_{c2}^k is the carbon emission tax paid by the port, CMY; P_{c2}^k is the carbon tax price, CMY/Mt CO₂.

$$T_u^k = W_{sp}^k / S\omega \cos\varphi \tag{10}$$

where T_u^k is the annual utilization hours, h; *S* is the total installed capacity of the shore power system, kVA; ω means that there is a certain margin between the actual use of the shore power capacity of the terminal and the planned construction capacity of shore power; cos φ is the comprehensive power factor of the shore power frequency conversion equipment and the ship's load. We experimentally set $\omega = 0.8$ and $\cos \varphi = 0.7$. ω is needed to make sure that the shore power equipment is working properly under the load. Part of the shore power equipment must be suspended during equipment repairs and maintenance in order to prolong the equipment's useful life. $\cos \varphi$ is related to the nature of the load: different devices have different power factors, so, here, the integrated power is used instead. When W_{sp}^k and *S* are fixed, obviously, the larger ω and $\cos \varphi$, the smaller T_u^k .

3. Game Research under Two Carbon Mechanism

3.1. Game Research on the Use of Shore Power under Cap and Trade

Figure 2 explains the three-way game model in the case of cap and trade. The government defines carbon emission rights as a commodity and establishes carbon emission caps for ports. By incorporating the value of carbon emissions into port costs, ports can directly buy or sell allowances in the carbon trading market. Ships can voluntarily choose to use shore electricity while they are docked in port and the amount of time they utilize it has been rising every year.



Figure 2. Tripartite benefit analysis under cap and trade.

1. Government Benefit Analysis Model

$$\begin{cases} B_g = T_e + T_s + T_{ep,sp} + T_{ep,oil} \\ C_g = C_1 \alpha_1 + C_2 \alpha_2 + T_{ep,oil} \\ F_g = B_g - C_g \end{cases}$$
(11)

where B_g is the government income; C_g is the government cost; F_g is the government profit; α_1 and α_2 are the subsidy rates for shore power equipment transformation given by the government to the port and the ship, respectively; C_1 and C_2 are the transformation costs of the port and the ship, respectively, CMY.

2. Port Benefit Analysis Model

$$\begin{cases}
B_{p} = C_{sp} + C_{1}\alpha_{1} \\
C_{p} = C_{1} + C_{e} + T_{s} + T_{c1} \\
F_{p} = B_{p} - C_{p}
\end{cases}$$
(12)

where B_p is the port income; C_p is the port cost; F_p is the port profit.

3. Ship Benefit Analysis Model

$$\begin{cases} B_{s} = C_{2}\alpha_{2} + C_{oil,sp} + T_{ep,sp} \\ C_{s} = C_{sp} + C_{2} + C_{oil,oil} + T_{ep,oil} \\ F_{s} = B_{s} - C_{s} \end{cases}$$
(13)

where B_s is the ship income; C_s is the ship cost; F_s is the ship profit.

$$SW_1 = F_p + F_s + T_e + T_s - T_{c1}$$
(14)

where SW_1 refers to social welfare, including consumer surplus, producer surplus, government tax, and environmental benefits [31]. $T_{ep,sp}$ represents the environmental benefit, which not only reduces the environmental protection fee levied on the ship but also is included in the government income. Note that social welfare is only counted once and does not accumulate repeatedly.

The use of shore power is abandoned after the ship converts to shore power owing to the high price. In the worst case, all ships consume fuel for power supply. In this way:

$$\begin{cases}
B_{s,oil} = C_2 \alpha_2 \\
C_{s,oil} = C_{oil,oil} + C_{oil,sp} + T_{sp,oil} + T_{ep,sp} + C_2 \\
F_{s,oil} = B_{s,oil} - C_{s,oil}
\end{cases}$$
(15)

where $B_{s,oil}$ is the ship income; $C_{s,oil}$ is the ship cost; $F_{s,oil}$ is the ship profit.

When some ships use shore power, the benefits of the entire ship are improved:

$$F_{s,save} = F_s - F_{s,oil} \tag{16}$$

When the ships use all fuel oil, the costs come from the fuel costs and the environmental protection taxes paid. The income comes from freight and is unincluded in the scope of the three-party game, so the ship's benefit is negative, which is similar to social welfare.

3.2. Game Researches on the Use of Shore Power under the Carbon Tax Policy

Figure 3 illustrates the three-way game model in the case of carbon tax policy. The government sets the carbon tax rate, and ports must offer the government a carbon tax for every metric ton of carbon dioxide they emit.



Figure 3. Tripartite benefit analysis under carbon tax policy.

1. Government Benefit Analysis Model

$$\begin{cases} B_g = T_e + T_s + T_{c2} + T_{ep,sp} + T_{ep,oil} \\ C_g = C_1 \alpha_1 + C_2 \alpha_2 + T_{ep,oil} \\ F_g = B_g - C_g \end{cases}$$
(17)

where B_g is the government income; C_g is the government cost; F_g is the government profit.

2. Port Benefit Analysis Model

 $\begin{cases} B_{p} = C_{sp} + C_{1}\alpha_{1} \\ C_{p} = C_{1} + C_{e} + T_{s} + T_{c2} \\ F_{p} = B_{p} - C_{p} \end{cases}$ (18)

where B_p is the port income; C_p is the port cost; F_p is the port profit.

3. Ship Benefit Analysis Model

$$\begin{cases} B_{s} = C_{2}\alpha_{2} + C_{oil,sp} + T_{ep,sp} \\ C_{s} = C_{sp} + C_{2} + C_{oil,oil} + T_{ep,oil} \\ F_{s} = B_{s} - C_{s} \end{cases}$$
(19)

where B_s is the ship income; C_s is the ship cost; F_s is the ship profit.

Social welfare is given by:

$$SW_2 = F_p + F_s + T_e + T_s \tag{20}$$

If all types of ships are powered by fuel oil, the benefit analysis of the ship is as follows:

$$\begin{cases}
B_{s,oil} = C_2 \alpha_2 \\
C_{s,oil} = C_{oil,oil} + C_{oil,sp} + T_{ep,oil} + T_{ep,sp} + C_2 \\
F_{s,oil} = B_{s,oil} - C_{s,oil}
\end{cases}$$
(21)

where $B_{s,oil}$ is the ship income; $C_{s,oil}$ is the ship cost; $F_{s,oil}$ is the ship profit.

Some ships are converted from oil to electricity, and the overall benefit of the ship is improved:

$$F_{s,save} = F_s - F_{s,oil} \tag{22}$$

4. Data Selection

4.1. Government Data Acquisition

According to the "Interim Measures for the Management of Subsidy Funds for Ports, Ship Shore Power Facilities and Marine Low Sulfur Oil Subsidy Funds in Shenzhen", the subsidy will be provided for the reconstruction of port shore power facilities, which will not exceed 30% of the project construction costs [32].

4.2. Port Data Acquisition

According to the survey, a port has built 14 sets of shore power systems, covering a total of 23 berths. The installed capacity of the shore power system has reached 11,600 kVA. Including the equipment purchase fee and construction installation fee, the investment and renovation costs of the power equipment are CMY55,144,134. Taking the service life as 30 years, the interest rate of the annualized cost is 8%, which is equivalent to the annual renovation cost:

$$C_1 = \frac{8\% \times (1+8\%)^{30}}{(1+8\%)^{30} - 1} \times 55,144,134 = 4,898,312 \text{ CMY}$$

4.3. Ship Data Acquisition

The investment and transformation cost of onboard electrical equipment, referring to the report of the European Commission Environment Directorate (ECDGE) [33], converted into unit power is 1530 CMY/kW.

The berthing time of the ship in the port, the length of the use of shore power in a certain year, and the electricity consumed by the use of shore power are shown in Figure 4.



Figure 4. Ship information.

4.4. Electricity Price Acquisition

According to the "Notice on Clarifying the Electricity Price and Service Price of Ship's Shore-based Power Supply Facilities" of the price bureau of Jiangsu province, C_{grid}^k takes the electricity price of large industrial electricity at 0.6601 CMY/kWh. The maximum price of shore power used by ships in Taizhou is 1.20 CMY/kWh [34].

In order to standardize the accounting of carbon dioxide emissions implied by electricity consumption by regions, industries, enterprises, and other units, and to ensure comparability of results, the government organized a study to determine the average carbon dioxide emission factor for regional power grids in China. It refers to the carbon emissions generated by one unit of electricity used in the grid, and is obtained by dividing the total emissions of the entire grid by the total electricity generation. As the port study is in the southern region, the average CO₂ emission factor for the southern regional grid was used. F_e is the annual average power supply emission factor of the regional power grid, taking 527.1 g/kWh.

4.5. Pollutant Data Acquisition

According to the literature [33], E_1 is taken as 213 g/kWh. According to the "2020 Implementation Plan for the Global Sulfur Restriction Order for Marine Fuel Oil" issued by the China Maritime Safety Administration, those entering the country's inland river ships' air pollutant emission control areas should use fuel oil with a sulfur content of no more than 0.10% [35]. $C_{per-oil}$ was taken as 3800 CMY/Mt.

According to the "Decision of the Standing Committee of the Jiangsu Provincial People's Congress on the Applicable Tax Amount of Environmental Protection Tax for Air Pollutants and Water Pollutants", the tax rate in Nanjing is CMY8.4 per pollution equivalent of air pollutants [36].

The annual emission factors and pollution equivalent values of various pollutants caused by marine auxiliary engine fuel are shown in Table 1 [37,38].

Marine Fuel Emission Pollutants	F_n (g/kWh)	E_{dl}^n (kg)
SO ₂	0.46	0.95
NO _x	11.80	0.95
VOC	0.53	0.05
СО	1.68	16.7
general dust PM10	0.30	4
CO ₂	698	5000

Table 1. Annual emission factors and pollution equivalent values of pollutants.

4.6. Carbon Price Data Acquisition

The carbon price data from 2020 to 2050 comes from the "2020 China Carbon Price Survey" [39]. Figure 5 reveals that there has been a steady increase in the carbon price in China since 2020.



Figure 5. The expected price of the national carbon emissions trading market in 2021–2050.

5. The Impact of Various Factors on the Benefits of Each Party

5.1. Comparison of the Impact of Subsidy Rates

Based on the maximization of social welfare, we explore the impact of subsidy rates under two carbon mechanisms on the optimal price and the benefits to all parties. Restrictions:

$$\begin{cases} 0 \le a_1 \le 30\% & 0 \le a_2 \le 30\% & C_{serve} \ge 0 & 0 \le C_{dj} \le 1.2 \\ F_g \ge 0 & F_p \ge 0 & F_{s,save} \ge 0 \end{cases}$$
(23)

Objective function:

$$max.SW$$
 (24)

Tables 2 and 3 state the impact of the subsidy rate on the economic and environmental benefits of each party in the three-way game model under the two carbon regimes, respectively.

Carbon Mechanism	Cap and Trade	Carbon Tax
Subsidy rate a1 (%)	30	30
Subsidy rate a2 (%)	30	30
Electricity service price (CMY/kWh)	0.1151	0.2855
Electricity actual price (CMY/kWh)	0.7752	0.9456
Electricity consumption (GWh)	30.484	30.484
Annual utilization hours (h)	4692.8	4692.8
Social welfare (million CMY)	-104.937	-109.959
Government benefit (million CMY)	7.570	12.647
Port benefit (million CMY)	4.657×10^{-16}	$9.313 imes10^{-16}$
Ship benefit (million CMY)	-108.031	-113.228
Savings from "oil-to-electricity" ships (million CMY)	37.917	32.720

Table 2. Comparison of the impact of subsidy rates on economic benefits to all parties under the two carbon regimes.

Table 3. Comparison of the impact of subsidy rates on environmental benefits to all parties under the two carbon regimes.

Carbon Mechanism	Cap and Trade	Carbon Tax
Carbon cap (Mt)	98,051.842	
Carbon price (CMY/Mt)	50	50
Actual CO ₂ emissions (Mt)	95,649.611	95,649.611
Tax involved in carbon mechanism (million CMY)	-0.120	4.782
Environmental tax savings (million CMY)	6.100	6.100
CO ₂ reduction (Mt)	5209.746	5209.746
SO ₂ reduction (Mt)	14.023	14.023
NO _x reduction (Mt)	359.713	359.713
VOC reduction (Mt)	16.157	16.157
CO reduction (Mt)	51.213	51.213
General dust PM10 reduction (Mt)	9.145	9.145

As can be seen from Figure 6a, whether under the cap-and-trade or the carbon tax policy, social welfare decreases when the electricity service price rises with roughly a linear negative correlation between the two. The cost of power supply services is rising, which does not promote social welfare, and there is clear resistance to the use of shore power. In addition, the lowest value of social welfare under cap and trade is much higher than the maximum value under a carbon tax policy. As a result, cap and trade offers a significant benefit in this case and merits consideration. From Figure 6b–e, we can observe that the subsidy rate a1 affects social welfare and ship benefits rise as the subsidy rate a1 rises. The effects of a1 and a2 on government benefits are identical. Government benefits rise with an increase in a2, while they decline with an increase in a1. A comparison of the two results reveals that the cap-and-trade group reported far more social welfare and ship benefits than the other one. On the contrary, the government benefit and port benefit under cap and trade are in every case short of what they are under the other system.



Figure 6. The relationship between subsidies and the benefits to all parties. (**a**) The relationship between electricity service price and social welfare; (**b**) The relationship between subsidy and social welfare; (**c**) The relationship between subsidy and government benefit; (**d**) The relationship between subsidy and port benefit; (**e**) The relationship between subsidy and ship benefit.

5.2. Carbon Price Impact Comparisons

We evaluated how the price of carbon affects the best price and the gains for all parties under two carbon mechanisms based on the maximization of social welfare.

Restrictions:

$$\begin{cases} 20 \le P_c \le 100 \quad C_{serve} \ge 0 \quad 0 \le C_{dj} \le 1.2 \\ F_g \ge 0 \quad F_p \ge 0 \quad F_{s,save} \ge 0 \end{cases}$$
(25)

Objective function:

max.SW

(26)

Tables 4 and 5 indicate the impact of the carbon price on the economic and environmental benefits of each party in the three-way game model under the two carbon regimes, respectively.

Table 4. Comparison of the impact of carbon price on economic benefits to all parties under the two carbon regimes.

Carbon Mechanism	Cap and Trade	Carbon Tax
Subsidy rate a1 (%)	30	30
Subsidy rate a2 (%)	30	30
Electricity service price (CMY/kWh)	0.1109	0.1857
Electricity actual price (CMY/kWh)	0.7710	0.8458
Electricity consumption (GWh)	30.484	30.484
Annual utilization hours (h)	4692.8	4692.8
Social welfare (million CMY)	-104.696	-107.090
Government benefit (million CMY)	7.563	9.605
Port benefit (million CMY)	0	0
Ship benefit (million CMY)	-107.904	-110.186
Savings from "oil-to-electricity" ships (million CMY)	38.045	35.762

Table 5. Comparison of the impact of carbon price on environmental benefits to all parties under the two carbon regimes.

Carbon Mechanism	Cap and Trade	Carbon Tax
Carbon cap (Mt)	98,051.842	
Carbon price (CMY/Mt)	100	20
Actual CO ₂ emissions (Mt)	95,649.611	95,649.611
Tax involved in carbon mechanism (million CMY)	-0.240	1.913
Environmental tax savings (million CMY)	6.100	6.100
CO_2 reduction (Mt)	5209.746	5209.746
SO ₂ reduction (Mt)	14.023	14.023
NO _x reduction (Mt)	359.713	359.713
VOC reduction (Mt)	16.157	16.157
CO reduction (Mt)	51.213	51.213
General dust PM10 reduction (Mt)	9.145	9.145

Figure 7 compares the outcomes gained from the analysis of the two carbon mechanisms. Under both the cap-and-trade and carbon tax policies, social welfare decreases as electricity service prices rise. The social welfare under cap and trade is always greater than that under the carbon tax policy. Figure 7b–e shows that the price of carbon has a significant impact on social welfare, government benefits, port benefits, and ship benefits under the carbon tax policy. The four are not significantly impacted by the carbon price under cap and trade. According to the cap-and-trade model, while government benefits decline as the price of carbon rises, social welfare and ship benefits increase. The carbon tax policy states that when carbon prices rise, the government gains more advantages whereas social welfare and ship benefits decline. Social welfare and ship benefits under cap and trade tend to be invariably greater than those under the carbon tax approach. In comparison to a carbon tax, the government's benefit under cap and trade is always less. The port efficiency fluctuates positively around 0.





5.3. Comparison of Time-Proportional Effects of Using Shore Power

This study investigates the impact of the ratio of time spent using shore power to total docking time on the best pricing and the gains for all parties under two carbon mechanisms, based on the maximization of total social welfare.

$$T_{u}^{k} = \lambda \sum_{i=1}^{N_{i}} T^{k,i} P^{k,i} / S\omega \cos \varphi$$
⁽²⁷⁾

where λ is the ratio of using shore power to the berthing time. As the total port call time is constant, T_u^k and λ are proportional. T_u^k decreases, so λ decreases, which affects all aspects of electricity prices, social welfare, government benefits, port benefits, ship benefits, etc. Restrictions:

 $C_{serve} \ge 0 \quad 0 \le C_{dj} \le 1.2 \quad T_u \le 8760$ (28)

Objective function:

Tables 6 and 7 explicate the impact of the time-proportional on the economic and environmental benefits of each party in the three-way game model under the two carbon regimes, respectively.

Table 6. Comparison of the impact of time-proportional on economic benefits to all parties under the two carbon regimes.

Carbon Mechanism	Cap and Trade	Carbon Tax
Subsidy rate a1 (%)	30	30
Subsidy rate a2 (%)	30	30
Electricity service price (CMY/kWh)	0.5399	0.5399
Electricity actual price (CMY/kWh)	1.2	1.2
Proportion of time using shore power (%)	39	39
Electricity consumption (GWh)	56.354	56.354
Annual utilization hours (h)	8675.2	8675.2
Social welfare (million CMY)	-66.986	-72.230
Government benefit (million CMY)	16.625	21.187
Port benefit (million CMY)	25.616	20.713
Ship benefit (million CMY)	-99.796	-99.796
Savings from "oil-to-electricity" ships (million CMY)	46.153	46.153

Table 7. Comparison of the impact of time-proportional on environmental benefits to all parties under the two carbon regimes.

Carbon Mechanism	Cap and Trade	Carbon Tax
Carbon cap (Mt)	98,051.842	
Carbon price (CMY/Mt)	50	50
Actual CO ₂ emissions (Mt)	91,228.445	91,228.445
Tax involved in carbon mechanism (million CMY)	-0.341	4.561
Environmental tax savings (million CMY)	11.276	11.276
CO ₂ reduction (Mt)	9630.913	9630.913
SO ₂ reduction (Mt)	25.923	25.923
NO _x reduction (Mt)	664.978	664.978
VOC reduction (Mt)	29.868	29.868
CO reduction (Mt)	94.675	94.675
General dust PM10 reduction (Mt)	16.906	16.906

What stands out in Figure 8a is the price of electricity supply service changes regularly between 0 and 0.5399 CMY/kWh, and social welfare is also changing. The service price must be 0.5399/kWh under cap and trade in order to reach its maximum value, and it must be 0 under the carbon tax policy in order to reach its minimum value. In Figure 8b–d, the similarity between the two carbon mechanisms is highlighted above. The overall level of social welfare rises as the ratio of time spent utilizing shore power to docking time

grows. Under the two carbon mechanisms, the change curves for overall social welfare, government benefits, and port benefits are comparable and can be attained by moving up and down. The curve of ship benefit with the time proportion of using shore power is exactly the same in Figure 8e. With the share of time spent utilizing shore power increasing, the overall level of social welfare rises monotonically. However, government benefits, port benefits, and ship benefits show an upward trend with the increase in the proportion of time when shore power is used.



Figure 8. The relationship between time proportion and the benefits to all parties. (**a**) The relationship between electricity service price and social welfare; (**b**) The relationship between time proportion and social welfare; (**c**) The relationship between time proportion and government benefit; (**d**) The relationship between time proportion and port benefit; (**e**) The relationship between time proportion and ship benefit.

6. Conclusions

The two systems of cap and trade and carbon tax are clear and easy to understand, but too many similar parameters appear in the modeling process, leading to easy confusion. In both cap and trade and carbon tax, the specific components of tripartite benefits and social welfare are different and are simply represented by the same parameters, and the specific values change as the electricity service price, carbon tax, and the ratio of using shore power to the berthing time change. Also note that the government is happy to see and promote the use of shore power, which is in line with the plan to reduce carbon emissions; the port is to take the responsibility of a state-owned enterprise, respond to the national policy, take responsibility for emission reduction and establish a good image among the public. Ships have the least public pressure and social responsibility, and they are oriented by economic interests, so they are more unstable, and they need government guidance and support because they have to face technical and economic difficulties in the "oil-to-electricity" conversion. In the model, we should focus on understanding the cost saving of some ships after "oil-to-electricity" conversion, which is the key to decide whether ships should insist on using shore power. There is a difference between the CO₂ emitted from using oil to meet the power and the CO₂ emitted from switching to shore power to replace this power, and the focus is on calculating the difference and the accompanying economic and environmental benefits.

Depending on the port's yearly business volume and historical carbon emission intensity base, the government calculates the annual carbon cap for the port. The comprehensive CO_2 emission factor of ships in the first three years is greater than the current year's CO_2 emission factor. Therefore, the carbon cap is frequently higher than the actual CO_2 emission, according to calculations of the port's real energy usage. Currently, compared to other businesses, the government's criteria for reducing carbon emissions in the maritime sector are not stringent enough.

In summary, these results indicate that the annual hours of shore power facility use, the amount of environmental protection tax saved by utilizing shore power, and the reduction in pollutants and carbon dioxide are all equivalent as long as the ships' energy consumption is constant. The social welfare, governmental benefits, port benefits, and ship benefits appear to vary depending on the subsidy rate, carbon price, and percentage of time using shore power.

It is reasonable for the port to set the shore electricity price within the range of 0.6601 to 1.2 CMY/kWh, taking into consideration both its own transformation costs and government subsidies. As long as ships use electricity instead of fuel, the economic benefits will be significantly improved, no matter what changes in various carbon mechanisms and influencing factors. The primary source of improvement may partly be related to reduced fuel costs. The government's benefits are always greater than zero, mostly due to the environmental benefits of decreasing pollutant emissions. Low economic advantages are expected for the port, as a result of high costs of self-renovation and government limits on the actual electricity price.

The major limitation of this study is that ships are assumed to be equivalent to a virtual ship although they have distinct types and numbers when in berth. It is a macro analysis for multiagent games; hence, it cannot accurately reflect the benefit to an individual ship. The three-party game is dynamic and played repeatedly, and ships are allowed to choose whether to utilize shore power or not, which makes it challenging to calculate the best electricity price. By examining the status of the tripartite game in electricity substitution, this study offers a particular reference value for the cycle planning and benefit distribution of electric energy replacement projects in the future.

Author Contributions: Conceptualization, Y.H. and Y.Z.; methodology, Y.H.; software, Y.H.; validation, Y.H.; formal analysis, Y.H.; investigation, Y.H.; resources, Y.Z.; data curation, Y.H.; writing—original draft preparation, Y.H.; writing—review and editing, Y.Z.; visualization, Y.H.; supervision, Y.Z.; project

administration, Y.Z.; funding acquisition, Y.Z. All authors have read and agreed to the published version of the manuscript.

Funding: Guangxi Special Fund for Innovation-Driven Development (AA19254034).

Data Availability Statement: Publicly available datasets were analyzed in this study. These data can be found in the References section.

Conflicts of Interest: The authors declare no conflict of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript; or in the decision to publish the results.

References

- 1. Handbook of Statistics 2021. In Proceedings of the United Nations Conference on Trade and Development. Available online: https://unctad.org/system/files/official-document/tdstat46_en.pdf (accessed on 6 February 2023).
- Review of Maritime Transport 2021. In Proceedings of the United Nations Conference on Trade and Development. Available online: https://unctad.org/publication/review-maritime-transport-2021 (accessed on 6 February 2023).
- 3. Sheng, D.; Li, Z.-C.; Fu, X.; Gillen, D. Modeling the effects of unilateral and uniform emission regulations under shipping company and port competition. *Transp. Res. E Logist. Transp. Rev.* 2017, 101, 99–114. [CrossRef]
- 4. Rolán, A.; Manteca, P.; Oktar, R.; Siano, P. Integration of cold ironing and renewable sources in the barcelona smart port. *IEEE Trans. Ind. Appl.* **2019**, *55*, 7198–7206. [CrossRef]
- 5. Ballini, F.; Bozzo, R. Air pollution from ships in ports: The socio-economic benefit of cold-ironing technology. *Res. Transp. Bus. Manag.* **2015**, *17*, 92–98. [CrossRef]
- 6. IMO 2020—Cutting Sulphur Oxide Emissions. Available online: https://www.imo.org/en/MediaCentre/HotTopics/Pages/ Sulphur-2020.aspx (accessed on 1 August 2022).
- Corbett, J.; Winebrake, J.; Carr, E.; Jalkanen, J.; Johansson, L.; Prank, M.; Sofiev, M.; Winebrake, S.; Karppinen, A. Health impacts associated with delay of MARPOL global sulphur standards. In *Air Pollution and Energy Efficiency*; Annex II MEPC; International Maritime Organization: London, UK, 2016; Volume 70.
- 8. Qi, J.; Wang, S.; Peng, C. Shore power management for maritime transportation: Status and perspectives. *Marit. Transp. Res.* 2020, 1, 100004. [CrossRef]
- 9. Sofiev, M.; Winebrake, J.J.; Johansson, L.; Carr, E.W.; Prank, M.; Soares, J.; Vira, J.; Kouznetsov, R.; Jalkanen, J.-P.; Corbett, J.J. Cleaner fuels for ships provide public health benefits with climate tradeoffs. *Nat. Commun.* **2018**, *9*, 406. [CrossRef] [PubMed]
- 10. Corbett, J.J.; Winebrake, J.J.; Green, E.H.; Kasibhatla, P.; Eyring, V.; Lauer, A. Mortality from ship emissions: A global assessment. *Environ. Sci. Technol.* 2007, 41, 8512–8518. [CrossRef]
- 11. Chen, J.; Zheng, T.; Garg, A.; Xu, L.; Li, S.; Fei, Y. Alternative maritime power application as a green port strategy: Barriers in China. *J. Clean. Prod.* **2019**, *213*, 825–837. [CrossRef]
- 12. Radwan, M.E.; Chen, J.; Wan, Z.; Zheng, T.; Hua, C.; Huang, X. Critical barriers to the introduction of shore power supply for green port development: Case of Djibouti container terminals. *Clean Technol. Environ. Policy* **2019**, *21*, 1293–1306. [CrossRef]
- Winkel, R.; Weddige, U.; Johnsen, D.; Hoen, V.; Papaefthimiou, S. Shore side electricity in Europe: Potential and environmental benefits. *Energy Policy* 2016, *88*, 584–593. [CrossRef]
- 14. Pettit, S.; Wells, P.; Haider, J.; Abouarghoub, W. Revisiting history: Can shipping achieve a second socio-technical transition for carbon emissions reduction? *Transp. Res. D Transp. Environ.* **2018**, *58*, 292–307. [CrossRef]
- 15. Vaishnav, P.; Fischbeck, P.S.; Morgan, M.G.; Corbett, J.J. Shore power for vessels calling at US ports: Benefits and costs. *Environ. Sci. Technol.* **2016**, *50*, 1102–1110. [CrossRef] [PubMed]
- 16. Sciberras, E.A.; Zahawi, B.; Atkinson, D.J. Electrical characteristics of cold ironing energy supply for berthed ships. *Transp. Res. D Transp. Environ.* **2015**, *39*, 31–43. [CrossRef]
- 17. Peterson, K.L.; Islam, M.; Cayanan, C. State of shore power standards for ships. In Proceedings of the 2007 IEEE Petroleum and Chemical Industry Technical Conference, Calgary, AB, Canada, 17–19 September 2007; pp. 1–6.
- 18. Lam, J.S.L.; Notteboom, T. The greening of ports: A comparison of port management tools used by leading ports in Asia and Europe. *Transp. Rev.* 2014, 34, 169–189. [CrossRef]
- 19. Zis, T.; North, R.J.; Angeloudis, P.; Ochieng, W.Y.; Harrison Bell, M.G. Evaluation of cold ironing and speed reduction policies to reduce ship emissions near and at ports. *Marit. Econ. Logist.* **2014**, *16*, 371–398. [CrossRef]
- Styhre, L.; Winnes, H.; Black, J.; Lee, J.; Le-Griffin, H. Greenhouse gas emissions from ships in ports–Case studies in four continents. *Transp. Res. D Transp. Environ.* 2017, 54, 212–224. [CrossRef]
- Zhao, X.; Liu, L.; Di, Z.; Xu, L. Subsidy or punishment: An analysis of evolutionary game on implementing shore-side electricity. *Reg. Stud. Mar. Sci.* 2021, 48, 102010. [CrossRef]
- 22. Wu, L.; Wang, S. The shore power deployment problem for maritime transportation. *Transp. Res. E Logist. Transp. Rev.* 2020, 135, 101883. [CrossRef]

- Song, T.; Li, Y.; Hu, X. Cost-effective optimization analysis of shore-to-ship power system construction and operation. In Proceedings of the 2017 IEEE Conference on Energy Internet and Energy System Integration (EI2), Beijing, China, 26–28 November 2017; pp. 1–6.
- 24. Tseng, P.-H.; Pilcher, N. Evaluating the key factors of green port policies in Taiwan through quantitative and qualitative approaches. *Transp. Policy* **2019**, *82*, 127–137. [CrossRef]
- 25. Drake, D.F.; Kleindorfer, P.R.; Van Wassenhove, L.N. Technology choice and capacity portfolios under emissions regulation. *Prod. Oper. Manag.* **2016**, *25*, 1006–1025. [CrossRef]
- 26. Murray, B.; Rivers, N. British Columbia's revenue-neutral carbon tax: A review of the latest "grand experiment" in environmental policy. *Energy Policy* **2015**, *86*, 674–683. [CrossRef]
- 27. Ellerman, A.D.; Buchner, B. Over-Allocation or Abatement? A Preliminary Analysis of the EU Emissions Trading Scheme Based on the 2005 Emissions Data; MIT Joint Program on the Science and Policy of Global Change: Cambridge, MA, USA, 2006.
- Song, S.; Govindan, K.; Xu, L.; Du, P.; Qiao, X. Capacity and production planning with carbon emission constraints. *Transp. Res. E Logist. Transp. Rev.* 2017, 97, 132–150. [CrossRef]
- 29. Liu, Y.; Xin, X.; Yang, Z.; Chen, K.; Li, C. Liner shipping network-transaction mechanism joint design model considering carbon tax and liner alliance. *Ocean Coast. Manag.* 2021, 212, 105817. [CrossRef]
- 30. Chen, Z.-y.; Nie, P.-y. Effects of carbon tax on social welfare: A case study of China. Appl. Energy 2016, 183, 1607–1615. [CrossRef]
- 31. Wang, C. Monopoly with corporate social responsibility, product differentiation, and environmental R&D: Implications for economic, environmental, and social sustainability. *J. Clean. Prod.* **2021**, *287*, 125433.
- Interim Measures of Shenzhen Municipality on the Administration of Subsidies for Ports, Ship Shore Power Facilities and Low Sulfur Oil for Ships. Available online: http://jtys.sz.gov.cn/jtzx/wycx/slcx/khzc/content/post_4302973.html (accessed on 1 August 2022).
- Ritchie, A.; de Jonge, E.; Hugi, C.; Cooper, D. European Commission Directorate General Environment, Service Contract on Ship Emissions: Assignment, Abatement, and Market-based Instruments. In *Task 2c–SO2 Abatement*; Entec UK Limited: Northwich, Cheshire, UK, 2005.
- Notice on Defining the Price of Ship Shore Based Power Supply Service. Available online: http://zwgk.taizhou.gov.cn/art/2016 /12/26/art_46293_1282722.html (accessed on 1 August 2022).
- Implementation Plan of Global Marine Fuel Sulfur Limit Order in 2020. Available online: https://www.msa.gov.cn/page/article. do?articleId=7917B172-1CB6-421E-881C-25E5D00001B3 (accessed on 1 August 2022).
- Decision of the Standing Committee of the People's Congress of Jiangsu Province on the Applicable Amount of Environmental Protection Tax for Air Pollutants and Water Pollutants. Available online: http://czt.jiangsu.gov.cn/art/2017/12/5/art_7805_693 9598.html (accessed on 1 August 2022).
- Environmental Protection Tax Law of the People's Republic of China. Available online: https://www.mee.gov.cn/ywgz/fgbz/fl/ 201811/t20181114_673632.shtml (accessed on 1 August 2022).
- 38. Hu, X.; Wang, B.; Huang, J.; Ruan, W. Cost-benefit optimization analysis of all parties involved in shore power construction and operation and comparison of coastal power characteristics in rivers and lakes. *Electr. Power Autom. Equip.* **2018**, *9*.
- 39. Slater, H.; De Boer, D.; Qian, G.; Shu, W. 2021 China Carbon Pricing Survey; ICF: Beijing, China, 2021.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.



Article



Combustion, Ecological, and Energetic Indicators for Mixtures of Hydrotreated Vegetable Oil (HVO) with Duck Fat Applied as Fuel in a Compression Ignition Engine

Oleksandra Shepel¹, Jonas Matijošius¹, Alfredas Rimkus¹, Olga Orynycz^{2,*}, Karol Tucki³ and Antoni Świć⁴

- ¹ Department of Automobile Engineering, Faculty of Transport Engineering,
- Vilnius Gediminas Technical University, J. Basanavičiaus Street 28, LT-03224 Vilnius, Lithuania
 ² Department of Production Management, Faculty of Engineering Management,
- ³ Department of Production Finance Finance Figure Figure
 - Warsaw University of Life Sciences, Nowoursynowska Street 164, 02-787 Warsaw, Poland
- ⁴ Department of Production Computerisation and Robotisation, Faculty of Mechanical Engineering, Lublin University of Technology, Nadbystrzycka Street 36, 20-618 Lublin, Poland
- * Correspondence: o.orynycz@pb.edu.pl; Tel.: +48-746-9840

Abstract: The aim of the present study was to investigate the effects of the application of hydrotreated vegetable oil (HVO) mixed with pure duck fat (F100) as fuel, replacing the conventional fossil diesel fuel (D100). The tests were performed using a four-stroke direct injection CI engine diesel engine. Six fuel samples were used: D100, HVO100, F100, as well as three HVO–fat mixtures F25, F50, and F75. To further study the main characteristics of fuel combustion, the AVL BOOST software (Burn program) was applied. The results of experimental studies showed that with the addition of pure fat to HVO, the ignition delay phase increased with an increase in the amount of heat released during the premix combustion phase and the pressure and temperature rise in the cylinder increased; however, the mentioned parameters were not higher as compared to diesel fuel. It was found that as the concentration of fat in the HVO–fat mixtures increases, the viscosity and density increases, while LHV was decreased, which thereby increases brake specific fuel consumption and slightly decreases brake thermal efficiency in comparison to diesel fuel. A decrease of CO₂, HC, NO_x emissions, and smoke was established for all HVO–fat mixtures as compared to diesel fuel at all loads; however; under low loads, CO emissions increased.

Keywords: combustion; fuel; emissions; engine

1. Introduction

The lack of components and raw materials mined in Ukraine and Russia, as well as international sanctions, are only part of the war-related crises that the automotive industry has experienced.

The rapid decline in hydrocarbon reserves and the constant rise in prices for them require large-scale development of renewable energy sources [1–3]. Moreover, an important reason stimulating the transition to alternative energy sources are the problems of global climate change, which will reduce the impact on the environment of harmful factors, improve the ecology of our planet, as well as implement the recommendations written in the "Paris Convention on Climate Change", which entered into action on 4 November 2016 [4,5].

The current situation calls for regulation of the biofuels sector in 2023 with regard to the possibilities and obligations for the use of biocomponents in fuels. The aim is to contain further increases in fuel prices, to stabilize the situation involving the national fuel and biofuel markets, and to increase the state's fuel safety.

Citation: Shepel, O.; Matijošius, J.; Rimkus, A.; Orynycz, O.; Tucki, K.; Świć, A. Combustion, Ecological, and Energetic Indicators for Mixtures of Hydrotreated Vegetable Oil (HVO) with Duck Fat Applied as Fuel in a Compression Ignition Engine. *Energies* **2022**, *15*, 7892. https:// doi.org/10.3390/en15217892

Academic Editor: George Halkos

Received: 30 September 2022 Accepted: 21 October 2022 Published: 24 October 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). The EU biofuel policy aims to promote and encourage the development of biofuels, such as biodiesel, bioethanol, and biomass [6,7]. The Energy Union strategy, entitled "Clean energy for all Europeans" published in 2016, highlighted the further measures to reduce CO_2 emissions by up to 40% by 2030 and have net zero by 2050 [8,9].

As a result of environmental policies, many automotive alliances and partnerships have emerged to work together on large-scale biofuel projects. For example, FCA teamed up with Tesla and Honda, Mazda with Toyota, and Ford with Volvo and Polestar [10]. Since 1990 emissions of CO_2 in Europe have reduced by about 24% [11].

The European Union is the leader in the use of diesel biocomponents, the region accounts for 41% of the world demand for diesel biocomponents, which is 15.9 million tons, or about 7.4% of the volume of diesel fuel consumption in the EU. The vast majority of biocomponents used in the EU—about 85.5% (13.6 million tons)—are FAME (fatty acid methyl esters), the remaining 14.5% (2.3 million tons) are HVO. Of this volume, 11.6 million tons of FAME are produced directly in the EU countries and 2.7 Mt of HVO [12]. The HVO system enables the industrial application of the hydrogenation of rapeseed oil, used cooking oil (UCO), or a mixture of both. The final product may be used as an additive to diesel or jet fuel.

Experts predict that this figure will be 70% by 2030. Thus, the environmental and economic indicators for numerous countries, for instance Poland, Slovakia, Malta, Bulgaria, and Estonia, will be even more vulnerable to the impact of the transport industry [13].

The recently adopted European Climate Law raises the EU's 2030 vehicle emissions target from 40% to a minimum of 55% and introduces a legislative commitment to carbon neutrality by 2050, which, in turn, should help reduce emissions of CO_2 in the volume of about 420 million tons per year [13].

Today, most experts [14–19] agree that an important factor for the creation and use of innovative fuels for diesel engines is the availability of extensive raw materials for the production of alternative motor fuels. The energy features of the presented sources of raw materials, similar to mineral fuels, make it possible to use the latter as motor fuels [20].

The balance of the combination of rational prices for raw materials and measures to regulate social and environmental risks is of particular importance [21].

Hydrogenated vegetable oil (HVO) was introduced in 2005 when it was derived exclusively from palm oil [22]. Free of aromatics, oxygen, and sulfur, hydrogenated vegetable oil has a high cetane number, resulting in reduced NO_x emissions, improved stability in storage, and low temperature properties, making it suitable for almost all diesel engines [23,24]. The main limiting factor in the industrial producing of biodiesel is the cost of vegetable oil. The purchase of oilseeds, transportation, storage, and extraction of oil are the main items of expenditure related with the production of biodiesel. The production of fuel from plants takes up agricultural land, while more pesticides, herbicides, and fertilizers are used for higher yields, making it impossible to continue growing any other plants suitable for food on this area [25].

At the same time, intensive animal husbandry and subsequent processing of raw meat leads to the accumulation of a significant amount of fat-containing raw materials and waste [17,26–28]. This resource can be used to further solve energy problems for the production of biofuel.

Biodiesel can be exploited as pure (B100) or mixed with diesel fuel at any combination in most diesel engines. Generally, the use of such fuel does not require modification of the vehicle's engine [29–31].

Analysis of recent studies and publications suggests that numerous research on alternative fuels for diesel engines focuses more heavily on blending ratios with diesel [32–35], but there is little research evaluating the use of clean duck fat fuels as oxygenated fuels in combination with HVO.

Thus, in the opinion of the authors of this work, it is also important to study the potential of the presented samples of mixtures for further assessment of the main criteria of fuel quality during operating in a diesel engine.

According to most research [18,32,35–39], when biodiesel is burned, the greenhouse effect does not increase; it decreases the content of hydrocarbons, soot, and carbon monoxide exhaust gases. Biodiesel does not contain carcinogenic substances, such as polycyclic aromatic hydrocarbons and especially benzopyrene, with comparison to fossil fuel [40].

Some of the important indicators of engine efficiency are the parameters of the fuel used, in particular: density and kinematic viscosity [18,41,42]. In this paper, [43] noted that biodiesel extracted from duck fat has favorable properties of density, kinematic viscosity, and also, lower heating value compared to diesel fuel.

Animal fat as a fuel component can be widely used due to its cheapness (because it is obtained as a by-product of meat processing) and availability (every country has a meat processing industry). Pure lard is filtered before use, its preparation is carried out (heating to a uniform consistency and repeated filtering) [44]. When choosing the proportions of fat mixtures with traditional fuels, their physicochemical properties must be taken into account [45]. Depending on the type of fat, it is necessary to select measures to prevent it from solidifying in fuel mixtures (chemical stabilizers are used, additional heating, and constant mixing) [46].

The combustion parameters of chicken fat are different compared to traditional diesel fuel, such as a lower rate of heat release, which is determined by prolonged reactions at low temperatures [47]. To improve them, additional hydrogen can be added to fuel containing chicken fat—a better energetic (increased BTE) and ecological effect (decrease in sharpness, CO, and UHC emissions) is achieved [48].

The peculiarity of the high viscosity of the biodiesels is that it tends to negatively affect the loss of engine power. Due a high viscosity, large droplets and a short jet are formed; therefore, it takes more time for the fuel to evaporate, the ignition delay phase increases, incomplete combustion occurs, carbon deposits form, and fuel consumption increases [49]. Poor sprinkling, in turn, leads to clogging of the nozzle and fuel pump, which directly affect the increase in toxic emissions, such as CO, CO₂, and SO_x [50,51]. Besides, the straight using of pure vegetable oil causes the formation of injector sediments, a result of which gives rise to higher exhaust gases [52].

As known, density is one of the key characteristics of petroleum products for diesel engines. The density is determined by the parameters of the fuel itself. The higher the fractional composition, the more difficult the processes of evaporation and atomization of fuel in the injectors become [53].

Hoekman et al. [54] indicated that due to the oxygen content of biofuels, it has a lower content of energy (MJ/kg) than diesel fuel.

It should be noted that biodiesels have a higher cetane number than diesel fuel, which indicates a good ignition rate of the fuel [26,55].

A large number of studies have been carried out to study the consequence on the performance of a diesel engine of biodiesel based on various animal fats [38,56–58].

Sen et al. [38] used chicken fat for the making of biodiesel. In a pilot study, it was noted that the use of biodiesel blends led to a reduce in emissions CO, CO_2 , HC, and smokiness, but slightly increased the torque values and indicators NO_x . Raman et al. [49] also found that CO emissions from biodiesel blends are lower than those of a diesel engine, but CO values are higher at low loads. One of the possible reasons is the presence of a rich fuel mixture at higher loads.

In this research [56], the authors investigated duck fat oil used in a single cylinder Kirloskar TV-1 diesel engine. The results have shown that emissions of CO and HC were increased. Opposite results were for CO_2 and NO_x , which was reduced compared to diesel fuel.

Goga et al. [59] used a fuel mixture in their experiment (10% rice bran oil and 90% diesel) and it was found that hydrocarbon emissions are decreased when biodiesel was used as opposed to diesel fuel, which may indicate a shorter ignition delay phase due to a higher cetane number of the biodiesel mixture. Furthermore, it should be said that a shorter ignition delay phase contributes to a more complete combustion of the fuel; consequently,

there is less hydrocarbon emission. Other authors [33] also obtained similar results of a decrease in the emission of HC.

Results obtained from experiment with edible sunflower oil and non-edible Karanj oil indicated longer ignition, which consecutively caused an increase in pressure in a cylinder and higher CO and NO_x emissions, while in contrast, demonstrated lower BSFC in comparison to diesel fuel [52].

Additionally, diesel fuel is composed of alkines, cycloalkine, and aromatic hydrocarbons, which, in turn, also increase the formation of smoke.

In this research [23], the effect of pure chicken fat and various mixtures of fat and diesel in the ratio of 70/30, 50/50, and 30/70 was studied. An increase was noted in emissions of CO and CO₂ for mixtures with pure fat. Reduced emissions of NO_x during low engine loads for mixtures due to larger droplets of fuel, which caused a decrease in temperature, was also observed.

The authors [60] point out that high NO_x emissions from biodiesel mixtures may result from the high oxygen content of biodiesel. Barrios et al. have the same opinion [61].

Several publications [36,62] indicate that NO_x emissions are influenced by fuel density. NO_x emissions are also dependent on engine load and rpm, injection timing, and ignition delay [63,64].

Many studies have found that the use of biofuels in a diesel engine improves environmental performance, but at the same time increases BSFC [33,35,36,49,60,62,65,66].

Emiroğlu et al. [42] studied turkey fat as the main raw material for biodiesel in blends. It was found that the mixtures had at all loads with an engine speed from 1600 rpm to 2400 rpm, higher specific fuel consumption (BSFC) values and, at the same time, lower brake thermal efficiency (BTE) values compared to diesel fuel. Rao et al. [67] have similar conclusions. They pointed out that as the percentage of chicken fat biodiesel increased, exhaust temperatures, CO emissions, and BTE declined, while BSFC and NO_x increased. In [68], biodiesel based on chicken fat (B) was blended with diesel fuel (D) in specific blending percent: B20D80, B30D70, B40D60, and B50D50. It was indicated that the lower the engine power, the higher the fat content in the mixture, which is associated with a low calorific value.

Selvan, V.A.M. has made a major contribution to the development of knowledge about the use of fats for energy purposes [69–73]. His experimental studies have shown that chicken fat and egg shell are suitable as catalysts for the production of biodiesel. In addition, he was able to demonstrate that the physicochemical properties of the biodiesel produced from chicken oil comply with the ASTM D 6751 standard. In his scientific studies, Selvan, V.A.M. has shown that skin fat is an excellent source of energy.

Mikulski et al. [33] conducted experimental work on a four-stroke Common Rail diesel engine investigating pork fat methyl esters. It was noted that increasing the methyl ester in the blend increased the BSFC. This was due to the low calorific value of the tested mixtures, and also indicated a shorter ignition delay phase of the fuel. At the same time, an increase in BTE values was observed with an increase in the amount of biodiesel, on average, by 1.6%, 4.8%, and 7.8% for B25, B50, and B75, respectively. The same results are consistent with the solution indicated by Abed et al. [74] and Jayaprabakar et al. [75]. Consequently, the higher fuel consumption of biodiesel fuel contributes to improved fuel combustion due to oxygen enrichment, which also affects the performance of higher exhaust gas temperatures. Analysis of some publications recommends the use of biodiesel blends containing no more than 20% fuel based on renewable sources to minimize losses in engine performance [16,58].

The European Union's climate policy aims at climate neutrality. One way to achieve this is to reduce emissions of harmful gases (including greenhouse gases) from transport as much as possible. A large proportion of vehicle manufacturers selling their products in Europe have declared that they will not sell combustion vehicles between 2030 and 2040.

Electromobility is being developed and promoted in many countries of the European Union. However, there is no way to remove all of the combustion vehicles (approx. 2.5 to 3 billion). This applies to both passenger and truck transport. In the case of heavy vehicles,

the problem is even greater. Currently, diesel tractors are responsible for international traffic. They travel thousands of kilometers to transport between states. In this case, electromobility is not yet equipped for these major challenges. On the one hand, there are vehicles that do not yet have sufficient range, and on the other, there is the infrastructure.

Hydrotreated vegetable oil (HVO) could be a solution. This is fuel derived from waste from the food industry, i.e., in reality from residues of vegetables, fruit, and fat products (even animal origin).

Preliminary studies have shown that its benefits include reduced carbon dioxide emissions (between 50% and 90%, depending on the purity) and the absence of sulfur compounds. A 2011 study by VTT of Finland found that older cars can emit up to 30% less carbon dioxide. HVO100 is a pure hydrogenated vegetable oil without the addition of fossil fuels. HVO can also be mixed with conventional diesel in different proportions, e.g., HVO30, HVO50, etc. For new vehicles, HVO reduces CO_2 emissions by about 90%.

In the case of HVO, emissions cannot be avoided during production. This process still requires oil extraction and processing.

Leading truck manufacturers support the spread of the HVO. Compliance with the standards for its entire fleet has been announced by the DAF and has been declared for several years by Scania, MAN, Volvo, Mercedes, Renault, and Iveco. In particular, owners of Euro 5 and Euro 6 compliant lorries will be able to use the new biodiesel, i.e., practically the entire Polish fleet serving international transport as well as most local vehicles. After verification, the HVO mixtures can also refuel Euro 3 and 4 vehicles.

Volkswagen announced that from July 2021 diesel vehicles can be operated with pure HVO. In addition, the group estimates that the share of this fuel will reach up to 30 percent of the energy mix needed for transportation within a decade.

The reduction of pollutant concentrations during the combustion process and the ability of the HVO to act as a substitute fuel for most compression ignition engines makes it worthwhile to develop. Hydro-refined vegetable oil is not emission-free and consumes a fairly large amount of energy, but it is produced from waste that would have to be disposed of anyway.

The main aim of this research is to evaluate the energy and ecological benefits obtained with blends containing HVO and pure fat, as opposed to diesel fuel.

2. Methods and Materials

The study of engine performance indicators using HVO and fat fuel mixtures was carried out by means of experimental and numerical analysis. In the course of the experimental analysis, energy and ecological indicators were determined and the pressure in the cylinder was measured. Analysis of the combustion process was performed with the help of the BURN subroutine of the AVL BOOST program. Summarizing conclusions are presented based on the indicators of experimental and numerical analysis (Figure 1). The algorithms for controlling combustion engines require a considerable amount of time and cost. Engine manufacturers and research centers are increasingly using advanced tools to simulate engine operation. These tests allow a significant reduction in the analysis time and a reduction in the costs of engine design and development. AVL BOOST is a multi-level computing system with the possibility of real-time operation to simulate variable engine conditions. The calculation program simulates engine operation over time using current and constant zero-dimensional and quasi-dimensional components of the model.



Figure 1. Nomenclature of research.

2.1. Fuel Preparation

The main properties of fuel samples for a CI engine were examined in accordance with fuel standards in the European Union. In the EU, there are two standards for diesel fuel: standard EN 590 for mineral diesel fuel, to which it is allowed to add up to 5% fatty acid methyl esters, and the standard EN 14214—fatty acid methyl esters applying for diesel engines.

For HVO, we applied the recommendation of Neste Renewable Diesel, in that hydrotreated vegetable oil contains paraffinic hydrocarbons, and cannot be equivalent to the requirements of EN 14214, which was proposed solely for fatty acid methyl esters, that is, FAME. Nonetheless, HVO is close to standard EN 590, not including density.

Biodiesel concentrations studied, include mixing, was carried out in the following proportions. Blends of hydrotreated vegetable oil (HVO100) and pure duck fat (F100) mixed in the ratio of F25, F50, and F75 by volume. The comparison of physical and chemical properties of various fuels are given in Table 1.

Fuels	Density (kg/m³) at 15°	Viscosity (mm ² /s) at 40°	Sulfur Content (mg/kg)	Water Content (mg/kg)	Total Con- tamination (mg/kg)	Cetane Number	Hydrogen %	Carbon %	Oxygen%	C/H%	LHV (MJ/kg)
	Allowe	d value in aco	cordance wit	th quality sta	ndard EN 590						
	820-845	2-4.5	<10	<200	<24	>51					
D100	823.00	3.5	7.25	85	20	45	0.130	0.870	0.000	6.69	42.70
	Allowed	value in acco	ordance with	n quality stan	dard EN 14214						
	860-900	3.5–5	≤ 10	\leq 500	≤ 24	≥51					
F25	800	4.7	4.52	690	43.27	72.04	0.146	0.827	0.027	5.64	42.40
F50	831	9.8	4.87	770	-	67.19	0.141	0.804	0.055	5.70	40.70
F75	867	18.8	5.21	925	-	62.34	0.136	0.782	0.082	5.77	39.00
F100	908	34.8	5.31	1450	-	57.49	0.130	0.760	0.110	5.85	37.30
Allowed	value in acc	ordance with	the booklet for HV	information O	on Neste Renew	able Diesel					
	770–790	2–4	≤ 5	≤200	≤10	>70					
HVO100	776.00	2.9	4.16	20	5.52	76.89	0.152	0.848	0.000	5.58	43.70

Table 1. Physicochemical properties of the analyzed fuels.

Physicochemical properties of biodiesel made from animal fats or vegetable oils, in particular, viscosity, density, heat of combustion, cetane number, etc., differ from those for diesel fuel. It can be noted that the fuel mixtures presented are within the normal range. To ensure proper viscosity, duck fat fuel was heated to 40–50 $^{\circ}$ C.

Thus, the above analysis of the physicochemical properties of hydrotreated vegetable oil and its mixtures with pure duck fat indicates the possibility of using most of them to power diesel engines, despite the weighted fractional composition of fat; hence, with increased viscosity. However, these differences in the properties of pure fat and mixtures based on them from the properties of diesel fuel can lead to a deterioration in the quality of fuel atomization and mixture formation [68]. Therefore, it is preferable to use low fat hydrotreated vegetable oil in diesel engines; the less the viscosity of the fuel, the easier the fuel supply and its atomization (Table 1).

2.2. Test Bench, Measuring Instruments, and Data Processing

Specifications of the engine used in the experiment are given in Table 2. During the experimental part, the engine was taken at fixed speed n = 2000 rpm, the engine brake torque (MB) was presented in 30 Nm, 60 Nm, 90 Nm, and 120 Nm, which meant the brake mean effective pressure (BMEP) was 0.2 MPa, 0.4 MPa, 0.6 MPa and 0.8 MPa.

Specification	Parameter
Engine	1.9 Turbodiesel Direct Injection
Number of cylinders	4
Compression ratio	19.5
Stroke	95.5 mm
Bore	79.5 mm
Maximum power output	66 kW at 4000 rpm
Maximum torque	182 Nm at 2000–2500 rpm

Table 2. Specifications of the engine used in the experiment.

The tests were carried out at the stand used in the direct injection of the CI engine equipped with the electronic control unit. For measuring the composition of exhaust gases CO, HC, NO_x, smoke, and CO₂, the instrument AVL DiCom 4000 was applied, with precision of the result of 0.01% for CO, and for HC and NO_x, respectively, 1 ppm, and smoke 0.01 m^{-1} , and 0.1% for CO₂. The consumption of the tested fuel samples was carried out by weighing them on an electronic balance, CK-5000, with precision of 1.0 g. Therefore, an air meter was used to measure the air flow BOSCH HFM 5, with an accuracy of 2%. Pressure sensor Delta OHM HD 2304.0 measured the pressure of the turbocharger, with an accuracy of 0.0002 MPa. The temperature was measured using a thermoelectric converter, with an accuracy of 1.5 C (Figure 2).

In order to ensure uniformity of results and to avoid random errors, each test point was repeated 5 times. Such repeatability showed that during the test the recording of the results was done only when the smooth operation of the engine was established.

During the experimental tests, the CO concentration was measured, the accuracy of the measurement was 0.01%. At low engine load (BMEP = 0.2 MPa) and the engine running on HVO fuel, the CO concentration was 0.01%, with F100 fuel the CO concentration was close to 0.03%. When the engine was running at a load of BMEP = 0.8 MPa, the CO concentration of all fuels was the same—0.01%. The pollutant concentration was recalculated into a specific emission g/kWh and the obtained results correlate with the experimental data.



Figure 2. Schematic internal combustion engine testing equipment.

The piston position at the top dead center (TDC) was determined by an optical crankshaft position sensor, A58M-F, with signal repeatability of 0.176 CAD. To convert the signals from the pressure and crankshaft position sensors, the device AVL DiTEST DPM 800 was used. A quartz piezoelectric sensor was used to measure the gas pressure in the cylinder. AVL GH13P had a sensitivity of 15.84 ± 0.09 pC/bar. LabView Real software recorded the engine pressure (100 cycles). Registration of the start of fuel injection was noted by the equipment VAG-COM. The fuel injection timing control equipment controlled the fuel injection process.

During the tests, each point was repeated 5 times after the engine had stabilized. Standard error statistical evaluation was used:

$$u\left(\bar{x}\right) = u(x)/\sqrt{n} \tag{1}$$

where, the number of repetitions, in the case, was equal to 5.

Further, the errors were evaluated according to the sources [76,77] in the calculation and expanded uncertainly $U_{0.99}$.

The error values are shown in Table 3.

Parameter	Standard Uncertainly u	Expanded Uncertainly U _{0.99}
ROHR, J/CAD	0.003657	0.025
Temperature rise K/deg	0.002987	0.017
Pressure rise in cylinder, MPa/deg	0.005987	0.036
Pressure in cylinder, MPa	0.005745	0.034
CO_2 , g/kWh	0.000301	0.007
CO, g/kWh	0.000258	0.006
HC, g/kWh	0.006987	0.041
Smoke, m^{-1}	0.000249	0.005
NO _x , g/kWh	0.007459	0.052
BSFC, g/kWh	0.005221	0.032
BŤE	0.003698	0.025

Table 3. The error values.

2.3. Analysis of Experimental Results with the Use of AVL Boost Software

The fuel combustion processes were further studied by means of the software AVL Boost. AVL Boost is a software that includes of a pre-processing program for the starting data and description of the engine that will be represented as a model. Thereafter, the system applications form the mathematical equations and algorithmic program with a illustrative user interface, and inspect and calculate the processes that will be needed in the analysis and modeling. The software AVL BOOST's subprogram BURN uses the experimental data: cylinder pressure, as well as fuel and air consumption, properties of tested fuel samples, etc. By means of the subprogram BURN, the start of combustion (SOC), combustion duration (CD), and shape parameter (m) was determined. Furthermore, the rate of heat release (ROHR), temperature, and pressure rise in the cylinder were observed.

3. Results and Discussion

3.1. Indicators of Combustion

The combustion process is affected by both the structure and size of fuel droplets, the difference in the molecular structure of fuel hydrocarbons, the types of hydrocarbon compounds, and the types of chemical intermolecular bonds [78]. These characteristics of the fuel supplied for combustion have a significant effect on the qualitative and quantitative characteristics of the combustion process, and on the oxidation reactions of hydrocarbon compounds in the combustion zone [53].

The start of combustion (SOC) and ignition delay (ID) for various fuels at engine load BMEP = 0.8 MPa are shown in Figure 3. From the analysis of the experimental data, it can be found that SOI = 7 CAD bTDC for all fuels. The ignition delay phase for different fuels increases in the following order: HVO100, F25, F50, F75, F100, and D100. The shorter ignition delay phase of biofuel mixtures compared to diesel is explained by its higher cetane number [79]. Moreover, Sivalakshmi et al. [80] explained that low molecular weight gaseous compounds degraded from biodiesel during injection into an engine cylinder at high temperatures can ignite earlier; thereby reducing ignition delay phase and accelerating the onset of biofuel combustion.



Figure 3. Dependence of the rate of heat release in the cylinder using different fuels.

At high engine load (BMEP = 0.8 MPa), the amount of fuel consumed increases in the order of HVO100, D100, F25, F50, F75, and F100 due to the lower calorific value of the mixtures compared to diesel (Table 1). An increase in the mass of injected fuel occurs, which leads to an increase temperature rise in the combustion chamber (Figure 4). Adding more fat to the HVO increases the mass of fuel injected, which leads to a delay in ignition, which is associated with a large consumption of heat for the evaporation of fuel droplets.



Figure 4. Dependence of the temperature rise in the cylinder using different fuels.

Furthermore, in having a shorter ignition delay, less fuel for the samples of fuel is burned in the premix mode and more during the mixing-controlled combustion phase. A decrease in the ignition delay phase for HVO, in comparison to mineral fuel, will lead to a reduced part of fuel that burns during the flash period (premixed combustion phase), and accordingly, the proportion of fuel burned in the time period of diffusion combustion will increase (mixing-controlled combustion phase).

One of the important factors causing such differences in the combustion process is the viscosity of the fuels. Table 1 shows that the viscosity of HVO is 20% less than the diesel fuel. The addition of duck fat to the fuel mixture causes a significant increase in the fuel viscosity and the ignition delay. Thus, the correlation between the viscosity of the fuel and the ignition delay phase is visible.

As can be seen from the data presented in Figure 3, for HVO the first ROHR peak corresponding to the premixed combustion phase is about 20–25% lower than for diesel fuel, and this peak is reached 1 degree earlier. This regularity can be explained by a reduced ignition delay phase, and consequently, by a smaller amount of fuel that enters the diesel cylinder during this period of time. The addition of duck fat to the fuel causes an increase in the ignition delay phase and an increase in the intensity of the combustion process during the premixed combustion phase.

The regularity described above causes a reduction in the proportion of fuel that burns out over the ignition delay phase, and consequently, an increase in the proportion of fuel that burns during the mixing-controlled combustion phase with an increase in the concentration of animal fat in the fuel mixture. This is clearly seen from the data presented in Figure 3—the maximum ROHR level in the "mixing-controlled combustion phase" (Crank angle 11–12 CAD) for pure diesel fuel is the smallest of all the presented samples. With an increase in the concentration of fat (duck fat) in the fuel mixture, an increase in the level of the maximum ROHR in the "mixing-controlled combustion phase" is observed. For F100 fuel, the maximum ROHR level during the mixing-controlled combustion phase is the highest, which confirms the described tendency.

As the percentage of fat in the mixtures increased, the ignition delay phase increased; thereby increasing the peak rate of heat release. A longer ignition delay phase was observed with the F100 mixture than with HVO and other mixtures.

It was found that the ignition delay phase for pure HVO and mixtures with pure fat is lower in contrast to diesel fuel. A possible explanation is the higher cetane number for pure HVO [81]. The higher the amount of cetanes, the shorter the ignition time. The amount of cetanes increases with the length of the unbranched carbon chain. Therefore, the lower the content of "harmful" aromatic hydrocarbons in the fuel, the higher the cetane number will be [82].

During the combustion phase, including a premix at 2 CAD, the rate of heat release for HVO is ~21% less than for D100. It should also be noted that the ignition delay phase also depends on the viscosity and density of the test mixture samples. Since HVO has a lower viscosity than diesel, this, in turn, contributes to better mixing in the premix phase. Furthermore, due to the fact that the HVO has a chain with paraffinic hydrocarbon which decomposes and evaporates faster, this contributes to a more intensive mixing with the ambient air in comparison with diesel fuel. The fat increases the viscosity of the mixture with HVO, and increases in the ignition delay phase and ROHR in the premix combustion phase approaches that of diesel. The variance between the heat release rate in 1 CAD for F25 is ~24%, F50 is ~26%, F75 is ~14%, and F100 is ~10% compared to fossil fuel.

Examining the mixing-controlled combustion phase the maximum rate of heat release for D100 (at 10–12 CAD) was ~1.2% less than for HVO100 (at 11 CAD). Furthermore, comparing mixtures with pure fat, we observed that for F100 the maximum heat release is ~4.5% higher compared to fossil fuel; for F25, F50, and F75 the results were ~2.5%, 0.7%, and 0.8%.

In Figure 4, the maximum temperature rise at the premixed combustion phase observed for diesel fuel was 1–2 CAD—34 K/deg. When the temperature rise for D and for HVO was compared, it was found that diesel fuel had a higher rate at ~26% as compared to HVO. The HVO mixtures with pure fat and the premixed combustion phase temperature rise show a related trend as compared to fossil fuel. For F25, F50, F75, and F100, they were smaller than that for diesel: ~28%, ~27%, ~16%, and ~12%, respectively. This effect may be due to the higher viscosity, later start of combustion, longer fuel injection, higher injection pressure, and velocity along with lower heating value. The intense combustion in the premix phase that influences the formation of NO_x should also be noted. Thus, the rate of formation of nitrogen oxide for mineral fuel will be higher compared to other mixtures.

In Figure 5, it was found when testing diesel fuel the pressure rise at 2 CAD (premixed combustion phase) was higher ~28% compared with HVO. Similar results were obtained for other blends. For F25, F50, F75, and F100 they were accordingly, ~20%, ~34%, ~26%, and ~14% less compared to diesel fuel. The pressure rise correlates with ROHR and the temperature rise in the cylinder. During the mixing-controlled combustion phase, the minimum pressure rise fixed using pure fat F100. This was due to the decreased fuel injection rate due to the high viscosity.



Figure 5. Dependence of the pressure rise in the cylinder using different fuels.

Peak pressure varies little, but pressure at the end of compression and start of combustion is higher with added fat (Figure 6). This means that the burning of fat is longer. The longer combustion duration was determined due to the higher consumption (lower LHV), the longer injection duration, which was further increased by the higher fuel viscosity. Exhaust gas flow energy became higher and this increased the turbocharger pressure. However, longer combustion duration of fat reduced the BTE.



Figure 6. Dependence of the pressure in the cylinder using different fuels.

Figure 6 shows the pressure in the cylinder when BMEP = 0.8 MPa. We do not see any significant pressure differences because, for all fuel mixtures, the start of the fuel injection is the same (SOI = 7 BTDC), there is no very significant difference in fuel properties, and the engine load is the same. However, after performing the analysis of the combustion process (using these pressures), we see more pronounced differences in the various combustion indicators (Figures 3–5) when the studied fuel mixtures are used.

Higher maximum cylinder pressures were observed when the engine was running on diesel. This resulted in the longest ignition delay phase, maximum ROHR, temperature rise, and pressure rise in the premixed combustion phase. The maximum cylinder pressure was slightly reduced with the use of HVO and fat mixtures.

3.2. Ecological Indicators

3.2.1. Emissions of Carbon Dioxide (CO₂)

Specific CO₂ emissions, that are shown Figure 7, decrease for all samples of fuel with growth in the load. The BMEP = 0.8 MPa for HVO emissions of carbon dioxide were ~4.6% lower compared with fossil fuel. Furthermore, a decrease in CO₂ for F25 and F50 ~ 3.2% and 1.7% appropriately, was established. For F75 and F100, the CO₂ emissions were on average ~0.9% and ~2.1% higher compared with fossil fuel. The higher rates of CO₂ in the mixtures are because of the higher carbon and oxygen content of the examined fuels in comparison with mineral fuel and due to higher fuel consumption.



Figure 7. Dependence of carbon dioxide emissions on the load using different fuels.

Additionally, in the time of testing, it was found that blends that have a smaller ratio of C/H contribute the most to CO₂ reduction (Table 1). HVO, in turn, has a smaller ratio of C/H (5.7%), which allows for the reduction of CO₂ for this fuel sample compared to the mixtures and fossil fuel. With decreasing emissions of CO₂ less fuel consumption was noted. Perhaps the lack of air in the mixture of F25 and F50 slows down the combustion process, and thus reduces the production of CO₂ compared to D100, F75, and F100. The rate of the combustion process has little effect on the level of CO₂ formation. Several factors prevail: specific fuel consumption and specific carbon content in the fuel.

The level of specific CO_2 emissions of diesel exhaust gases for different fuels (Figure 7) at the same load is directly proportional to the specific fuel consumption and is directly proportional to the percentage of carbon in the fuel. It can be seen from Table 1 that with an increase in the concentration of fat (duck fat) in the fuel mixture, the percentage of carbon in the fuel decreases, but at the same time, the lower specific heat of combustion of the fuel also decreases, which causes an increase in specific fuel consumption. That is, the influence of the above factors on the level of CO_2 emissions with exhaust gases is opposite. Consequently, the final effect of fuel on the level of CO_2 emissions from the exhaust gases is determined by which of the two factors will dominate over the other.

So, for F75 and F100 fuels, an increase in the specific fuel consumption by an average of 12–18%, respectively, is the dominant factor over a 7–9% decrease in the specific carbon content in the fuel. As a consequence, there is an increase in specific CO_2 emissions for F75 and F100 fuels in comparison to diesel fuel throughout the entire range of engine operating loads. For HVO, F25, and F50 fuels, a decrease of 7%, 5%, and 3%, respectively, of the carbon content in the fuel is the dominant factor over the change in the average specific fuel consumption. As a consequence, for these fuels, a decrease in the specific CO_2 emission is observed over the entire load range.

3.2.2. Emissions of Carbon Monoxide (CO)

Since the fuel is split into CO at the time of the combustion process and then oxidized to carbon dioxide, the amount of CO tends to reduce with the growing temperature. The presence of hydrogen-containing substances, such as hydrogen, accelerates this process [83]. Moreover, pure fat blends are oxygenated fuel and the extra oxygen molecule helps the fuel burn better, which helps lower CO emissions.

With an increase in the concentration of fat (duck fat) in the mixture, the viscosity of the fuel increases, and also the specific net calorific value decreases (Table 1). A decrease in the specific net calorific value causes an increase in the cycle fuel supply. Both of these factors cause an increase in the maximum fuel injection pressure, and that can decrease the average diameter of fuel droplets in the cylinder and improve the distribution of fuel to the periphery of the combustion chamber. In turn, a decrease in the average diameter of fuel droplets has a positive effect on the completeness of fuel combustion, and an improvement in the distribution of fuel droplets to the periphery of the combustion chamber contributes to the elimination of zones with low local oxygen deficiency, which also reduces the formation of CO. However, with a significant increase in fuel viscosity, this effect of reducing CO emissions may not be achieved and pollutant emissions may increase.

Thus, at low load (BMEP = 0.2 MPa), the CO emission of F100 is ~160% higher compared to fossil diesel fuel (Figure 8). This indicates incomplete burning of fuel that consists of pure fat. A low cycle rate results in a low pressure, which, in turn, causes the formation of large droplets of high density and viscosity fuel, which burn much worse. The same trend was noticed for all HVO and fat blends. By increasing the load to BMEP = 0.4 MPa, the maximum difference of emissions for F100 and diesel fuel was ~63%. By increasing the load to BMEP = 0.6 MPa and BMEP = 0.8 MPa, the CO emissions of all fuel mixtures became similar.



Figure 8. Dependence of carbon monoxide emissions on the load using different fuels.

3.2.3. Emissions of Hydrocarbons (HC)

As can be seen from the Figure 9, at all loads the mixtures have lower HC values, unlike mineral fuel, and for HVO this indicator is the smallest. The high-rise cetane number of HVO, and thus of blends with HVO, reduces hydrocarbon exhaust gases in comparison to diesel fuel [84]. That is due to the low content of aromatic compounds in the fuel mixtures. Furthermore, it should be noted that the fuel mixture with duck fat contains some oxygen in the structure, so it improves the fuel combustion process, and HC emissions will be reduced when using these blends with a percentage of pure fat.



Figure 9. Dependence of hydrocarbon emissions on the load using different fuels.

For example, F100 for BMEP = 0.2 MPa has ~4% less HC emissions than D, while HVO has ~ 47% less, opposite with diesel fuel. At higher loads, a similar trend was determined for all fuel samples. On average, the HC values for the combinations F25, F50, F75, and F100 were lower, opposite to fossil diesel fuel, respectively, ~28%, ~23%, ~19%, and ~7%. HC emissions from HVO fuel at higher engine loads are ~45% lower compared to diesel. This can be explained by the above-discussed influence of fuel on the quality of the atomization and combustion processes. In case the fuel has a lower cetane number, it takes longer to start, which causes higher HC emissions [85].

3.2.4. Smoke

Smoke is generated by partial combustion of the fuel. The HVO smoke levels that we see in the graph (Figure 10) are reduced by an average of ~18% compared to diesel. This can be explained by the fact that the C/H ratio in HVO in its composition was 17% less (Table 1), and also does not have such components as sulfur, aromatic hydrocarbons, and other mineral impurities in its chemical composition, which form the formation of soot [86]. As discussed earlier, testing mixtures with duck fat (Table 1) contains some oxygen in its molecule, which improves combustion. Blends of HVO and pure fat have lower values in comparison with diesel fuel. The average values for all tested loads were: for F25~51%, F50~54%, F75~56%, and F100~59%. The decrease in smoke emission for the mixtures can be explained by the high mass oxygen content and the lower C/H ratio (Table 1). The presence of excess oxygen in mixtures with pure fat leads to better combustion and results in less smoke generation under all engine load conditions.



Figure 10. Dependence of smoke emissions on the load using different fuels.

3.2.5. Emissions of Nitrogen Oxide (NO_x)

Figure 11 shows increasing NO_x emissions with increasing loads for all testing fuels. This was due to the higher combustion temperature. D100 has the highest NO_x emissions. Furthermore, the variation between D100 and HVO100 to various loads on average is ~18%, and between D100 and F100, ~5%. At a higher load (BMEP = 0.8 MPa), the NO_x emissions were reduced for HVO~17.6%, F25~9.1%, F50~7.2%, F75~5.4%, and F100~3%, than for D100. This means that NO_x emissions from conventional diesel are highest in all cases. Nitrogen oxide emissions are lowest with HVO, which has the highest CN count of all samples. The ambiguous impact on nitrogen oxide emissions may depend on the cumulative effects of ignition delay, fuel injection quantity, and injection quantity distribution between the pilot and main injection [62]. Ignition quality is often influenced by cetane number; therefore, a high CN value indicates a short ignition delay, which means less fuel energy (ROHR) in the premix stage, and therefore, lower NO_x emissions.



-D -HVO -F25 -F50 -F75 -F100

Figure 11. Dependence of nitrogen oxide emissions on the load using different fuels.

It can be assumed that the emissions grow for HVO–fat mixtures is related to the presence of oxygen molecules in the fuel. The increase of NO_x emissions can also be explained by an increase in the iodine value. The amount of iodine is related to the cetane number, and the density and compressibility of fuel samples [87]. Thus, the experimental pure fat mixtures improve the oxidation of the fuel during combustion, leading to a higher local temperature, and therefore, a rise in nitrogen oxide emissions.

3.3. Energy Indicators

As presented in Figure 12, the brake specific fuel consumption (BSFC, g/kWh) for all duck fat blends at high loads was higher compared with pure diesel. However, the BSFC for HVO was lower by ~2.4% in comparison with pure diesel. It was noted that with an increase in the percent of pure fat in the samples, the fuel consumption rose for F100 ~ 17.7% in comparison to diesel fuel. When comparing fossil diesel fuel and other HVO mixtures with pure fat, an increase in BSFC for F25 ~ 1.6%, F50 ~ 6.8%, and F75 ~ 11.8% was noted. On condition that the calorific value of fat has a low value, then to maintain a constant speed at a certain load, the engine needs, accordingly, more fuel, and hence, we have an increased consumption of fuel [88]. One of the reasons that may influence the increase in BSFC with a percentage of pure fat is associated with a higher density compared to fossil fuels [89].



Figure 12. Dependence of BSFC, g/kWh on the load using different fuels.

Figure 13 presents that, for all samples of fuel, BTE was increased with increasing load, due to a rise in the ratio of indicated power and internal mechanical losses of the engine. With an increase in the load, the quality of the processes of mixture formation and fuel combustion improves as the combustion temperature rises [90], which also determines the above-described dependence of the BTE on the engine load.



Figure 13. Dependence of BTE on the load using different fuels.

Figure 13 shows the highest BTE for HVO, followed by mineral diesel fuels. The contrast between these samples of fuels was ~0.14%, which is directly related to LHV and the combustion process. A total of 1 kg of HVO fuel has ~2.4% more energy and HVO cyclic fuel mass is ~2.4% less compared to diesel. HVO fuels have shorter injection duration and shorter combustion duration. Lower cooling and exhaust heat losses increase the BTE of HVO in comparison to diesel fuel.

There is a clear trend in Figure 13 that as the pure fat in the blends increases, the BTE tends to decrease. At loads of BMEP = 0.8 MPa for F100 it was found that a reduction of BTE was on average ~2.7%, compared to diesel fuel. Brake thermal efficiency depends on how efficient the combustion is. The amount of oxygen in the fuel with the duck fat additive increases, this reduces the LHV and requires longer fuel injection duration. The BTE tends to decrease, since poor atomization is important due to the high viscosity of fat. With longer combustion duration, more energy of fuel is transferred to the cooling system and to the exhaust, which reduces the BTE. Higher amounts of oxygen, which accelerates combustion, have a smaller effect here.

4. Conclusions

On the basis of the studies carried out, it can be concluded that:

- The HVO compared to diesel fuel has ~1 CAD shorter ignition delay, ~20% lower ROHR during the premixed combustion phase, and slightly higher ROHR during the mixing-controlled combustion phase. The addition of pure fat to the mixtures increased the ignition delay phase compared to HVO, causing a shorter period of the premixed combustion phase and increasing ROHR at same time, but did not reach the level of diesel;
- 2. CO₂ emissions at all engine loads were reduced for HVO ~ 7% for F25 and F50 mixtures approximately by 3–5%, except F75 and F100, which were ~1.2% and ~2.8% higher than diesel fuel. This is mainly due to the specific fuel consumption and C/H ratio in fuels, as well as the efficiency of the mixing and combustion processes;
- 3. CO emission at a low load for F100 increased by 160–60% compared to diesel fuel. The reason for this is a low fuel pressure, which, in turn, causes the formation of large droplets of high density and viscosity fuel, which burn much worse. Increasing the engine load significantly reduces the CO emissions of pure fat. HC emissions for HVO100, F25, F50, F75, and F100 were lower, opposite to fossil diesel fuel, respectively, ~45%, ~28%, ~23%, ~19%, and ~7%. This is explained by the simpler molecular
structure of HVO and better injection and combustion properties. The fat changes the quality of the injection and the combustion deteriorates, especially at low loads;

- 4. Smoke values, on average, decreased by 18% for HVO and 51%, 54%, 56%, and 59% for F25, F50, F75, and F100 compared to diesel fuel. The decrease in smoke emission for the mixtures can be explained by the high mass oxygen content in duck fat and the lower C/H ratio in HVO. Due to the fat additive, the worst fuel injection did not increase the smoke;
- 5. The maximum difference of NO_x emissions was observed between D and HVO and amounted to ~18%. By increasing to F100 emission of NO_x , it remained 6% lower compared to diesel. It can be assumed that the maximum temperature rise during the combustion of diesel fuel is higher; therefore, the level of formation of nitrogen oxides for diesel is higher. Due to the worst fat injection, the maximum combustion temperature was lower compared to diesel;
- 6. BSFC of pure HVO fuel was ~2.4% lower compared to conventional diesel due to 2.4% higher LHV. HVO fuel mixtures with a duck fat additive can only be used after heating them to 40–50 °C. With an increase in the concentration of fat to 100% in the HVO–fat mixtures, a proportional increase in the BSFC was observed on average up to 17.5% in comparison with conventional fossil diesel fuel due to the ~13% lower calorific value of the fat and slightly (2–3%) lower BTE for fat containing mixtures, due to their higher viscosity, and accordingly, poor atomization and combustion compared to diesel fuel.

Author Contributions: Conceptualization, O.S., J.M. and A.R.; methodology, O.S., J.M., A.R. and O.O.; validation, O.O., K.T. and A.Ś.; investigation, J.M., A.R., K.T. and O.S.; writing—original draft preparation O.S., J.M. and A.R.; writing—review and editing, J.M. and O.O.; supervision, J.M., O.S., A.R., O.O. and A.Ś.; project administration, J.M. and A.R.; software J.M., A.R. and K.T. All authors have read and agreed to the published version of the manuscript.

Funding: The research was carried out under financial support obtained from the research subsidy of the Faculty of Engineering Management (WIZ) of Bialystok University of Technology. From the grant No. WZ/WIZ-INZ/4/2022 (Olga Orynycz). Part of this research was carried out during the exchange stays of O. Orynycz at Vilnius University of Technology (Lithuania 2021, 2022), and cooperation stays at the Department of Computer Science Technology and Production Robotics of the Faculty of Mechanical Engineering, Lublin University of Technology (2021, 2022). Part of the research was also carried out during the scientific stay of O. Orynycz at the Department of Transport and Logistics of Institute of Technology and Business in České Budějovice (2022).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: All data are presented in this article. Data sharing is not applicable to this article.

Acknowledgments: Vilnius Tech has many years of cooperation with this company. The used simulation package allows proper evaluation of various parameters of combustion processes and is one of the best packages of its kind on the market.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Anifantis, A.S.; Colantoni, A.; Pascuzzi, S.; Santoro, F. Photovoltaic and Hydrogen Plant Integrated with a Gas Heat Pump for Greenhouse Heating: A Mathematical Study. *Sustainability* **2018**, *10*, 378. [CrossRef]
- Sarkan, B.; Skrucany, T.; Semanova, S.; Madlenak, R.; Kuranc, A.; Sejkorova, M.; Caban, J. Vehicle coast-down method as a tool for calculating total resistance for the purposes of type-approval fuel consumption. *Sci. J. Sil. Univ. Technol.-Ser. Transp.* 2018, 98, 161–172.
- Orynycz, O. Influence of tillage technology on energy efficiency of rapeseed plantation. *Procedia Eng.* 2017, 182, 532–539. [CrossRef]

- 4. Gao, Y.; Gao, X.; Zhang, X. The 2 °C Global Temperature Target and the Evolution of the Long-Term Goal of Addressing Climate Change—From the United Nations Framework Convention on Climate Change to the Paris Agreement. *Engineering* **2017**, *3*, 272–278. [CrossRef]
- 5. Nazarko, J.; Czerewacz-Filipowicz, K.; Kuzmicz, A.K. Comparative analysis of the Eastern European countries as participants of the new silk road. *J. Bus. Econ. Manag.* 2017, *18*, 1212–1227. [CrossRef]
- 6. Stattman, S.L.; Gupta, A.; Partzsch, L.; Oosterveer, P. Toward Sustainable Biofuels in the European Union? Lessons from a Decade of Hybrid Biofuel Governance. *Sustainability* **2018**, *10*, 4111. [CrossRef]
- 7. Ponte, S.; Daugbjerg, C. Biofuel sustainability and the formation of transnational hybrid governance. *Environ. Politics* **2015**, *24*, 96–114. [CrossRef]
- 8. Fact Sheets on the European Union. Energy Policy: General Principles. Available online: https://www.europarl.europa.eu/factsheets/en/sheet/68/energy-policy-general-principles (accessed on 23 June 2022).
- Clean Energy for All Europeans Package. Available online: https://energy.ec.europa.eu/topics/energy-strategy/clean-energyall-europeans-package_en (accessed on 23 June 2022).
- Renewable Energy Policies in a Time of Transition. Available online: https://www.irena.org/-/media/Files/IRENA/Agency/ Publication/2018/Apr/IRENA_IEA_REN21_Policies_2018.pdf (accessed on 23 June 2022).
- 11. Robaina, M.; Neves, A. Complete decomposition analysis of CO2 emissions intensity in the transport sector in Europe. *Res. Transp. Econ.* **2021**, *90*, 101074. [CrossRef]
- Mizik, T.; Gyarmati, G. Economic and Sustainability of Biodiesel Production—A Systematic Literature Review. *Clean Technol.* 2021, *3*, 19–36. [CrossRef]
- 13. European Climate Law. Available online: https://ec.europa.eu/clima/eu-action/european-green-deal/european-climate-law_en (accessed on 23 June 2022).
- 14. Hazrat, M.A.; Rasul, M.G.; Khan, M.M.K.; Ashwath, N.; Rufford, T.E. Emission characteristics of waste tallow and waste cooking oil based ternary biodiesel fuels. *Energy Procedia* 2019, *160*, 842–847. [CrossRef]
- 15. Simsek, S.; Uslu, S. Determination of a diesel engine operating parameters powered with canola, safflower and waste vegetable oil based biodiesel combination using response surface methodology (RSM). *Fuel* **2020**, 270, 117496. [CrossRef]
- 16. Emiroglu, A.O.; Keskin, A.; Sen, M. Experimental investigation of the effects of turkey rendering fat biodiesel on combustion, performance and exhaust emissions of a diesel engine. *Fuel* **2018**, *216*, 266–273. [CrossRef]
- 17. Adewale, P.; Dumont, M.; Ngadi, M. Recent trends of biodiesel production from animal fat wastes and associated production techniques. *Renew. Sustain. Energy Rev.* 2015, 45, 574–588. [CrossRef]
- Verma, T.N.; Shrivastava, P.; Rajak, U.; Dwivedi, G.; Jain, S.; Zare, A.; Shukla, A.K.; Verma, P. A comprehensive review of the influence of physicochemical properties of biodiesel on combustion characteristics, engine performance and emissions. *J. Traffic Transp. Eng.* 2021, *8*, 510–533. [CrossRef]
- 19. Szmigielski, M.; Zarajczyk, J.; Węgrzyn, A.; Leszczyński, N.; Kowalczuk, J.; Andrejko, D.; Krzysiak, Z.; Samociuk, W.; Zarajczyk, K. Testing the technological line for the production of alternative fuels. *Przem. Chem.* **2018**, *97*, 1079–1082.
- Singh, D.; Sharma, D.; Soni, S.L.; Inda, C.S.; Sharma, S.; Sharma, P.K.; Jhalani, A. A comprehensive review of physicochemical properties, production process, performance and emissions characteristics of 2nd generation biodiesel feedstock: Jatropha curcas. *Fuel* 2021, 285, 119110. [CrossRef]
- Gebremariam, S.N.; Marchetti, J.M. Economics of biodiesel production: Review. *Energy Convers. Manag.* 2018, 168, 74–84. [CrossRef]
- 22. Dimitriadis, A.; Natsios, I.; Dimaratos, A.; Katsaounis, D.; Samaras, Z.; Bezergianni, S.; Lehto, K. Hydrotreated Vegetable Oil (HVO) and Effects on Emissions of a Passenger Car Diesel Engine. *Front. Mech. Eng.* **2018**, *4*, 7. [CrossRef]
- 23. Rimkus, A.; Vipartas, T.; Matijošius, J.; Stravinskas, S.; Kriaučiūnas, D. Study of Indicators of CI Engine Running on Conventional Diesel and Chicken Fat Mixtures Changing EGR. *Appl. Sci.* **2021**, *11*, 1411. [CrossRef]
- McCaffery, C.; Zhu, H.; Ahmed, C.M.S.; Canchola, A.; Chen, J.Y.; Li, C.; Johnson, K.C.; Durbin, T.D.; Lin, Y.H.; Karavalakis, G. Effects of hydrogenated vegetable oil (HVO) and HVO/biodiesel blends on the physicochemical and toxicological properties of emissions from an off-road heavy-duty diesel engine. *Fuel* 2022, 323, 124283. [CrossRef]
- Atabani, A.E.; Silitonga, A.S.; Ong, H.C.; Mahlia, T.M.I.; Masjuki, H.H.; Badruddin, I.A.; Fayaz, H. Non-edible vegetable oils: A critical evaluation of oil extraction, fatty acid compositions, biodiesel production, characteristics, engine performance and emissions production. *Renew. Sustain. Energy Rev.* 2013, 18, 211–245. [CrossRef]
- 26. Stopka, O.; Stopková, M.; Pečman, J. Application of Multi-Criteria Decision Making Methods for Evaluation of Selected Passenger Electric Cars: A Case Study. *Commun. Sci. Lett. Univ. Zilina* **2022**, *24*, A133–A141. [CrossRef]
- 27. Tucki, K.; Orynycz, O. Bioenergy and Biofuels. Sustainability 2021, 13, 9972. [CrossRef]
- Lópeza, E.M.T.; Mauricioa, W.H.B.; Oliveraa, C.A.C.; Floresa, J.W.V.; Alfaroa, E.G.B.; Gonzales, L.G.V. Comparison of Biodiesel Extracted from Pork and Duck Fat. *Chem. Eng. Trans.* 2021, *87*, 505–510. [CrossRef]
- 29. Kinnal, N.; Sujaykumar, G.; D'costa, S.W.; Girishkumar, G.S. Investigation on Performance of Diesel Engine by Using Waste Chicken Fat Biodiesel. *IOP Conf. Ser. Mater. Sci. Eng.* **2018**, *376*, 012012. [CrossRef]
- Ramalingam, S.; Rajendran, S.; Ganesan, P.; Govindasamy, M. Effect of Operating Parameters and Antioxidant Additives with Biodiesels to Improve the Performance and Reducing the Emissions in a Compression Ignition Engine—A Review. *Renew. Sustain. Energy Rev.* 2018, *81*, 775–788. [CrossRef]

- Caban, J.; Jandacka, J.; Nieoczym, A.; Holubcik, M.; Vrabel, J. Preliminary Studies of Fuel Briquettes from Wood Waste. Engineering for Rural Development. Available online: https://www.tf.llu.lv/conference/proceedings2021/Papers/TF293.pdf (accessed on 23 June 2022).
- 32. Kumar, M.S.; Jaikumar, M. A Comprehensive Study on Performance, Emission and Combustion Behavior of a Compression Ignition Engine Fuelled with WCO (Waste Cooking Oil) Emulsion as Fuel. *J. Energy Inst.* **2014**, *87*, 263–271. [CrossRef]
- Mikulski, M.; Duda, K.; Wierzbicki, S. Performance and Emissions of a CRDI Diesel Engine Fuelled with Swine Lard Methyl Esters–Diesel Mixture. *Fuel* 2016, 164, 206–219. [CrossRef]
- 34. Elkelawy, M.; Alm-Eldin Bastawissi, H.; Esmaeil, K.K.; Radwan, A.M.; Panchal, H.; Sadasivuni, K.K.; Ponnamma, D.; Walvekar, R. Experimental studies on the biodiesel production parameters optimization of sunflower and soybean oil mixture and DI engine combustion, performance, and emission analysis fueled with diesel/biodiesel blends. *Fuel* 2019, 255, 115791. [CrossRef]
- 35. Kanthasamy, S.; Karthick, A.; Ramalingam, V.; Praveen, M.V.J. Performance and emission characteristics of diesel engine fueled with ternary blends of linseed and rubber seed oil biodiesel. *Fuel* **2021**, *285*, 119255. [CrossRef]
- Ghazali, W.N.M.W.; Mamat, R.; Masjuki, H.H.; Najafi, G. Effects of biodiesel from different feedstocks on engine performance and emissions: A review. *Renew. Sustain. Energy Rev.* 2015, *51*, 585–602. [CrossRef]
- Gad, M.S.; El-Shafay, A.S.; Abu Hashish, H.M. Assessment of diesel engine performance, emissions and combustion characteristics burning biodiesel blends from jatropha seeds. *Process Saf. Environ. Prot.* 2021, 147, 518–526. [CrossRef]
- 38. Sen, M.; Emiroglu, A.O.; Keskin, A. Production of biodiesel from broiler chicken rendering fat and investigation of its effects on combustion, performance, and emissions of a diesel engine. *Energy Fuels* **2018**, *32*, 5209–5217. [CrossRef]
- 39. Ge, S.; Brindhadevi, K.; Xia, C.; Elesawy, B.H.; Elfasakhany, A.; Unpaprom, Y.; Doan, H.V. Egg shell catalyst and chicken waste biodiesel blends for improved performance, combustion and emission characteristics. *Fuel* **2021**, *306*, 121633. [CrossRef]
- 40. Karmakar, A.; Karmakar, S.; Mukherjee, S. Properties of various plants and animals feedstocks for biodiesel production. *Bioresour. Technol.* **2010**, *101*, 7201–7210. [CrossRef]
- 41. Du, E.; Cai, L.; Huang, K.; Tang, H.; Xu, X.; Tao, R. Reducing viscosity to promote biodiesel for energy security and improve combustion efficiency. *Fuel* **2018**, *211*, 194–196. [CrossRef]
- Keskin, A.; Şen, M.; Emiroğlu, A.O. Experimental studies on biodiesel production from leather industry waste fat and its effect on diesel engine characteristics. *Fuel* 2020, 276, 118000. [CrossRef]
- 43. Milojevic, S.; Savić, S.; Marić, D.; Stopka, O.; Krstić, B.; Stoja, B. Correlation between Emission and Combustion Characteristics with the Compression Ratio and Fuel Injection Timing in Tribologically Optimized Diesel Engine. *Teh. Vjesn.* **2022**, *29*, 1210–1219.
- Alptekin, E.; Canakci, M.; Ozsezen, A.N.; Turkcan, A.; Sanli, H. Using Waste Animal Fat Based Biodiesels–Bioethanol–Diesel Fuel Blends in a DI Diesel Engine. *Fuel* 2015, 157, 245–254. [CrossRef]
- 45. Banković-Ilić, I.B.; Stojković, I.J.; Stamenković, O.S.; Veljkovic, V.B.; Hung, Y.-T. Waste Animal Fats as Feedstocks for Biodiesel Production. *Renew. Sustain. Energy Rev.* **2014**, *32*, 238–254. [CrossRef]
- El-Shafay, A.S.; Ağbulut, Ü.; Attia, E.-A.; Touileb, K.L.; Gad, M.S. Waste to Energy: Production of Poultry-Based Fat Biodiesel and Experimental Assessment of Its Usability on Engine Behaviors. *Energy* 2022, 262, 125457. [CrossRef]
- Karuppan, D.; Muthu Manokar, A.; Vijayabalan, P.; Sathyamurthy, R.; Madhu, B.; Mageshbabu, D.; Bharathwaaj, R.; Jenoris Muthiya, S. Experimental Investigation on Pressure and Heat Release HCCI Engine Operated with Chicken Fat Oil/Diesel-Gasoline Blends. *Mater. Today Proc.* 2020, 33, 437–444. [CrossRef]
- Suhel, A.; Abdul Rahim, N.; Abdul Rahman, M.R.; Bin Ahmad, K.A. Engine's Behaviour on Magnetite Nanoparticles as Additive and Hydrogen Addition of Chicken Fat Methyl Ester Fuelled DICI Engine: A Dual Fuel Approach. *Int. J. Hydrog. Energy* 2021, 46, 14824–14843. [CrossRef]
- 49. Raman, A.L.; Deepanraj, B.; Rajakumar, S.; Sivasubramanian, V. Experimental investigation on performance, combustion and emission analysis of a direct injection diesel engine fuelled with rapeseed oil biodiesel. *Fuel* **2019**, *246*, 69–74. [CrossRef]
- Taghizadeh-Alisaraei, A.; Ghobadian, B.; Tavakoli-Hashjin, T.; Mohtasebi, S.S.; Rezaei-Asl, A.; Azadbakht, M. Characterization of Engine's Combustion-Vibration Using Diesel and Biodiesel Fuel Blends by Time-Frequency Methods: A Case Study. *Renew. Energy* 2016, 95, 422–432. [CrossRef]
- Cheikh, K.; Sary, A.; Khaled, L.; Abdelkrim, L.; Mohand, T. Experimental Assessment of Performance and Emissions Maps for Biodiesel Fueled Compression Ignition Engine. *Appl. Energy* 2016, 161, 320–329. [CrossRef]
- 52. Shah, P.R.; Ganesh, A. A comparative study on influence of fuel additives with edible and non-edible vegetable oil based on fuel characterization and engine characteristics of diesel engine. *Appl. Therm. Eng.* **2016**, *102*, 800–812. [CrossRef]
- 53. Stone, R. Intergovernmental Panel on Climate Change. Summary for Policymakers. In *Climate Change 2013—The Physical Science Basis: Working Group I Contribution to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change*; Cambridge University Press: Cambridge, UK, 2014; pp. 1–30. [CrossRef]
- Hoekman, S.K.; Broch, A.; Robbins, C.; Ceniceros, E.; Natarajan, M. Review of biodiesel composition, properties, and specifications. *Renew. Sustain. Energy Rev.* 2012, 16, 143–169. [CrossRef]
- 55. Asikin-Mijan, N.; Rosman, N.A.; AbdulKareem-Alsultan, G.; Mastuli, M.S.; Lee, H.V.; Nabihah-Fauzi, N.; Lokman, I.M.; Alharthi, F.A.; Alghamdi, A.A.; Aisyahi, A.A.; et al. Production of Renewable Diesel from Jatropha Curcas Oil via Pyrolytic-Deoxygenation over Various Multi-Wall Carbon Nanotube-Based Catalysts. *Process Saf. Environ. Prot.* 2020, 142, 336–349. [CrossRef]
- 56. Selvam, D.J.P.; Vadivel, K. Performance and Emission Analysis of DI Diesel Engine Fuelled with Methyl Esters of Beef Tallow and Diesel Blends. *Procedia Eng.* 2012, *38*, 342–358. [CrossRef]

- 57. Gad, M.S.; EL-Seesy, A.I.; Hashish, H.M.A.; He, Z.; Alshaer, W.G. Combustion and emissions aspects of a diesel engine working with sheep fat oil biodiesel-diesel blends. *Case Stud. Therm. Eng.* **2021**, *26*, 101162. [CrossRef]
- Krishania, N.; Rajak, U.; Chaurasiya, P.K.; Singh, T.S.; Birru, A.K.; Verma, T.N. Investigations of spirulina, waste cooking and animal fats blended biodiesel fuel on auto-ignition diesel engine performance, emission characteristics. *Fuel* 2020, 276, 118123. [CrossRef]
- 59. Goga, G.; Chauhan, B.S.; Mahla, S.K.; Cho, H.M. Performance and emission characteristics of diesel engine fueled with rice bran biodiesel and n-butanol. *Energy Rep.* **2019**, *5*, 78–83. [CrossRef]
- Jayabal, R.; Thangavelu, L.; Subramani, S. Combined Effect of Oxygenated Additives, Injection Timing and EGR on Combustion, Performance and Emission Characteristics of a CRDi Diesel Engine Powered by Sapota Biodiesel/Diesel Blends. *Fuel* 2020, 276, 118020. [CrossRef]
- 61. Barrios, C.C.; Domínguez-Sáez, A.; Martín, C.; Álvarez, P. Effects of animal fat based biodiesel on a TDI diesel engine performance, combustion characteristics and particle number and size distribution emissions. *Fuel* **2014**, *117*, 618–623. [CrossRef]
- 62. Graboski, M.S.; McCormick, R.L.; Alleman, T.L.; Herring, A.M. Impact of biodiesel source material and chemical structure on emissions of criteria pollutants from a heavy-duty engine. *Environ. Sci. Technol.* **2001**, *35*, 1742–1747. [CrossRef]
- 63. Sakthivel, R.; Ramesh, K.; Purnachandran, R.; Shameer, P.M. A review on the properties, performance and emission aspects of the third generation biodiesels. *Renew. Sustain. Energy Rev.* **2018**, *83*, 2970–2992. [CrossRef]
- 64. Rajak, U.; Nashine, P.; Verma, T.N.; Pugazhendhi, A. Performance and emission analysis of a diesel engine using hydrogen enriched n-butanol, diethyl ester and Spirulina microalgae biodiesel. *Fuel* **2020**, 271, 117645. [CrossRef]
- 65. Atabani, A.E.; Silitonga, A.S.; Badruddin, I.A.; Mahlia, T.M.I.; Masjuki, H.H.; Mekhilef, S. A comprehensive review on biodiesel as an alternative energy resource and its characteristics. *Renew. Sustain. Energy Rev.* **2012**, *16*, 4. [CrossRef]
- 66. Datta, A.; Mandal, B.K. A comprehensive review of biodiesel as an alternative fuel for compression ignition engine. *Renew. Sustain. Energy Rev.* **2016**, *57*, 799–821. [CrossRef]
- 67. Rao, K.; Ramakrishna, A.; Rao, P. Effect of fuel injection pressure on performance and emission characteristics of DI-CI engine fueled with chicken fat biodiesel. *Int. J. Heat Technol.* **2013**, *71*, 53–59.
- Mohiddin, M.N.; Saleh, A.A.; Reddy, A.N.R.; Hamdan, S. A Study on Chicken Fat as an Alternative Feedstock: Biodiesel Production, Fuel Characterisation, and Diesel Engine Performance Analysis. *Int. J. Automot. Mech. Eng.* 2018, 15, 5535–5546. [CrossRef]
- 69. Kanthasamy, P.; Selvan, V.A.M. FTIR and GCMS analysis on useful methyl ester compound from industrial waste animal fleshing oil (WAFO). *Mater. Today Proc.* 2021, *46*, 10072–10078. [CrossRef]
- 70. Kirubakaran, M.; Selvan, V.A.M. Experimental investigation on the effects of micro eggshell and nano-eggshell catalysts on biodiesel optimization from waste chicken fat. *Bioresour. Technol. Rep.* **2021**, *14*, 100658. [CrossRef]
- 71. Kirubakaran, M.; Selvan, V.A.M. Biodiesel Prod. Waste Chick. Oil Using Nanoeggshell Heterog. Catal. Isopropyl Ether Cosolvent. *Environ. Qual. Manag.* 2020, *31*, 51–64. [CrossRef]
- Kanthasamy, P.; Selvan, V.A.M.; Shanmugam, P. Investigation on the performance, emissions and combustion characteristics of CRDI engine fuelled with tallow methyl ester biodiesel blends with exhaust gas recirculation. *J. Therm. Anal. Calorim.* 2020, 141, 2325–2333. [CrossRef]
- 73. Kirubakaran, M.; Selvan, V.A.M. Eggshell as heterogeneous catalyst for synthesis of biodiesel from high free fatty acid chicken fat and its working characteristics on a CI engine. *J. Environ. Chem. Eng.* **2018**, *6*, 4490–4503. [CrossRef]
- 74. Abed, K.A.; El Morsi, A.K.; Sayed, M.M.; El Shaib, A.A.; Gad, M.S. Effect of waste cooking-oil biodiesel on performance and exhaust emissions of a diesel engine. *Egypt. J. Pet.* **2018**, *27*, 985–989. [CrossRef]
- Jayaprabakar, J.; Dawn, S.S.; Ranjan, A.; Priyadharsini, P.; George, R.J.; Sadaf, S.; Rajha, C.R. Process optimization for biodiesel production from sheep skin and its performance, emission and combustion characterization in CI engine. *Energy* 2019, 174, 54–68. [CrossRef]
- ISO 21748:2017; Guidance for the Use of Repeatability, Reproducibility and Trueness Estimates in Measurement Uncertainty Evaluation. ISO: Geneva, Switzerland, 2017. Available online: https://www.iso.org/standard/71615.html (accessed on 14 October 2022).
- 77. JCGM 100:2008; Evaluation of Measurement Data—Guide to the Expression of Uncertainty in Measurement. JCGM: Geneva, Switzerland. Available online: https://www.bipm.org/documents/20126/2071204/JCGM_100_2008_E.pdf/cb0ef43f-baa5-1 1cf-3f85-4dcd86f77bd6 (accessed on 14 October 2022).
- 78. Topare, N.S.; Jogdand, R.I.; Shinde, H.P.; More, R.S.; Khan, A.; Asiri, A.M. A short review on approach for biodiesel production: Feedstock's, properties, process parameters and environmental sustainability. *Mater. Today Proc.* 2022, 57, 1605–1612. [CrossRef]
- Syed, A.; Quadri, S.A.P.; Rao, G.A.P.; Mohd, W. Experimental investigations on DI (direct injection) diesel engine operated on dual fuel mode with hydrogen and mahua oil methyl ester (MOME) as injected fuels and effects of injection opening pressure. *Appl. Therm. Eng.* 2017, 114, 118–129. [CrossRef]
- 80. Sivalakshmi, S.; Balusamy, T. Influence of Ethanol Addition on a Diesel Engine Fuelled with Neem Oil Methyl Ester. *Int. J. Green Energy* **2012**, *9*, 218–228. [CrossRef]
- 81. Bello, E.I.; Out, F.; Osasona, A. Cetane number of three vegetable oils, their biodiesels and blends with diesel fuel. *J. Pet. Technol. Altern. Fuels* **2012**, *3*, 52–57. [CrossRef]

- 82. Knoth, G. Cetane Numbers-Heat of Combustion-Why Vegetable Oils and Their Derivatives Are Suitable as a Diesel Fuel, 1st ed.; The Biodiesel Handbook; AOCS Press: Champaign, IL, USA, 2005; pp. 4–16.
- 83. Martinka, J.; Rantuch, P.; Wachter, I. Impact of Water Content on Energy Potential and Combustion Characteristics of Methanol and Ethanol Fuels. *Energies* 2019, *12*, 3491. [CrossRef]
- 84. Singer, A.; Schröder, O.; Pabst, C.; Munack, A.; Bünger, J.; Ruck, W.; Krahl, J. Aging studies of biodiesel and HVO and their testing as neat fuel and blends for exhaust emissions in heavy-duty engines and passenger cars. *Fuel* **2015**, *153*, 595–603. [CrossRef]
- 85. Mofijur, M.; Hazrat, M.A.; Rasul, M.G.; Mahmudul, H.M. Comparative Evaluation of Edible and Non-edible Oil Methyl Ester Performance in a Vehicular Engine. *Energy Procedia* 2015, 75, 37–43. [CrossRef]
- Rimkus, A.; Žaglinskis, J.; Rapalis, P.; Skačkauskas, P. Research on the Combustion, Energy and Emission Parameters of Diesel Fuel and a Biomass-to-Liquid (BTL) Fuel Blend in a Compression-Ignition Engine. *Energy Convers. Manag.* 2015, 106, 1109–1117. [CrossRef]
- Abdulkadir, L.N.; Adisa, A.B.; Kyauta, E.E.; Raheem, M.A. Corrosion and Engine Test Analysis of Neem (*Azadirachta indica*) Oil Blends in a Single Cylinder, Four Stroke, and Air-cooled Compresion Ignition Engine. *Am. J. Mech. Eng.* 2014, 2, 151–158.
- Levine, F.; Kayea, R.V.; Wexler, R.; Sadvary, D.J.; Melick, C.; La Scala, J. Heats of Combustion of Fatty Acids and Fatty Acid Esters. J. Am. Oil Chem. Soc. 2014, 91, 235–249. [CrossRef]
- 89. Duda, K.; Wierzbicki, S.; Śmieja, M.; Mikulski, M. Comparison of performance and emissions of a CRDI diesel engine fuelled with biodiesel of different origin. *Fuel* **2018**, *212*, 202–222. [CrossRef]
- 90. Barua, P.; Hossain, N.; Chowdhury, T.; Chowdhury, H. Commercial Diesel Application Scenario and Potential of Alternative Biodiesel from Waste Chicken Skin in Bangladesh. *Environ. Technol. Innov.* **2020**, *20*, 101139. [CrossRef]

MDPI

St. Alban-Anlage 66 4052 Basel Switzerland Tel. +41 61 683 77 34 Fax +41 61 302 89 18 www.mdpi.com

Energies Editorial Office E-mail: energies@mdpi.com www.mdpi.com/journal/energies







Academic Open Access Publishing

www.mdpi.com

ISBN 978-3-0365-8333-4